# NATIONAL FORENSIC DNA STUDY REPORT

#### FINAL REPORT

Prepared in Partnership Between the:

# DIVISION OF GOVERNMENTAL STUDIES AND SERVICES WASHINGTON STATE UNIVERSITY

Nicholas P. Lovrich, PhD, Director

Michael J. Gaffney, J.D., Associate Director

Travis C. Pratt, PhD, Assistant Professor

Charles L. Johnson, MA, Doctoral Candidate

701 Johnson Tower Washington State University Pullman, Washington 99164

#### **AND**

# SMITH ALLING LANE, P.S.

Christopher H. Asplen, JD

Lisa H. Hurst

Timothy M. Schellberg, JD

1102 Broadway Plaza, Suite 403 Tacoma, Washington 98402

927 – 15<sup>th</sup> Street NW, Suite 1200 Washington, District of Columbia 20005

> 47 Marlborough Crescent London, England W4 1HG

# TABLE OF CONTENTS

Executive Summary	2
Acknowledgements	6
Authors' Note	7
	0
I. Introduction	8
II. Methodology	9
III. Casework Backlogs	11
A. Unsolved Rapes and Homicides	12
B. Unsolved Property Offenses	13
C. Other Offenses	14
W. A. C. 'd	1.5
IV. Agency Capacity	15
A. Law Enforcement B. Local and State Crime Laboratories	15 24
B. Local and State Crime Laboratories	24
V. CODIS Analysis	37
VI. Forensic DNA and Crime Prevention	47
VII. Cost and Benefit Questions	67
VIII. Conclusions and Policy Implications	69
The Application of Forensic DNA Technology in England and Wales	
Appendix	
1 Forensic DNA Assessment Forms	
a. Cover Letter	
b. Local Law Enforcement Agency Form and Instructions	
c. Local Laboratory Form and Instructions	
d. State Laboratory Form and Instructions	
2 Calculations of Local Law Enforcement Backlog	
3 Response Frequencies by Question	
a. Local Law Enforcement Response Frequencies by Question	
b. Local Law Enforcement Response Frequencies by Question (by size strata)	
c. Local Laboratory Response Frequencies by Question	
d. State Laboratory Response Frequencies by Question	
4 "Salaries" Clarification	
5 CODIS Data	
a. SDIS Hits	
b. NDIS Hits	1

### NATIONAL FORENSIC DNA STUDY REPORT

#### **EXECUTIVE SUMMARY**

This report was prepared by Smith Alling Lane in partnership with Washington State University through the support of a grant awarded by the National Institute of Justice, Office of Justice Programs, U.S. Department of Justice (Grant 2002-LT-BX-K 003). Points of view or opinions in this document are those of the authors and do not necessarily represent the official position or policies of the United States Department of Justice.

In recent years forensic DNA evidence has been used by agencies and actors in the criminal justice system more and more frequently to **both** convict the guilty and exonerate the innocent. Cases that previously may have been unsolvable have been transformed into solvable cases where viable suspects can be identified and arrested or removed from suspect lists. While this new technology enhances the opportunity for the criminal justice system to "do justice" in more cases of suspected wrongdoing, it also creates an added need for law enforcement agencies to review old cases for potential biological evidence that should be sent to a crime laboratory for testing. Many policy makers are interested in discussing solutions to this problem of an added burden on law enforcement and crime laboratories, but they need to have a more complete understanding of the major dimensions of the problem – How big is the case backlog? How did it grow so large? What is the capacity of the evidence collection, storage, analysis and retrieval system to reduce that backlog and provide timely information for the criminal justice system?

This Report relied largely on a nation-wide mail assessment of local and state forensics laboratories and local law enforcement agencies, as well as extensive interviews of prosecutors, investigators and laboratory administrators to arrive at answers to the questions posed above. While it is generally acknowledged that there is likely a significant backlog of criminal cases that might benefit from DNA analysis, no effort had been made to quantify this number. This study arrives at estimates for the numbers of unsolved criminal cases in the U.S. which might benefit from DNA analysis, assesses both law enforcement and laboratory capacities for dealing with cases involving DNA, and identifies significant issues relating to the expansion of the use of DNA forensic analysis in criminal cases.

### SUMMARY OF REPORT FINDINGS

#### **BACKLOGS AND CAPACITY**

#### The backlog of unsolved rapes and homicides in the U.S. is massive.

Through the data collected from a large, representative sampling of local law enforcement agencies in the United States, the study arrives at the following pertinent estimates:

- The number of rape and homicide cases with possible biological evidence which local law enforcement agencies have not submitted to a laboratory for analysis is over 221,000.
  - Homicide cases 52,000 (approximate)
  - o Rape cases 169,000 (approximate)
- The number of property crime cases with possible biological evidence which local law enforcement agencies have not submitted to a laboratory for analysis is over 264,000.
- The number of unanalyzed DNA cases reported by State and local crime laboratories is more than 57,000.
  - State laboratories 34,700 cases (approximate)
  - Local laboratories 22,600 cases (approximate)
- Total crime cases with possible biological evidence either still in the possession of local law enforcement, or backlogged at forensic laboratories is over one half million (542,700).

# A significant proportion of law enforcement agencies continue to misunderstand the potential benefits of DNA testing.

While these figures address the first question as to the approximate size of the case backlog, the second question remains regarding how the backlog came to grow so large. The answer to this question is quite complicated, and the phenomenon of a substantial growth in criminal case backlogs involves a variety of factors – some of which are vexing and difficult to manage. However, a series of questions posed to local law enforcement agencies and forensic laboratories reveal several interesting patterns of response which, when considered collectively, begin to provide an accurate picture of how the backlogs tend to develop and why they continue to exist. The following provides some of these responses:

#### Reasons why cases with DNA evidence have not been submitted to the laboratory:

- 50.8 percent of responding local law enforcement agencies indicated that forensic DNA was not considered a tool for crime investigations.
  - o 31.4 percent responded that no suspect had been identified
  - o 9.2 percent indicated that the prosecution had not requested testing
  - o 10.2 percent responded that a suspect had been identified, but not yet charged
- 23.6 percent of responding agencies suggested that DNA evidence from unsolved cases was not submitted for reasons relating to poor funding.

- o 9.4 percent indicate a lack of funding for DNA analysis
- o 10.4 percent indicate inability of laboratories to produce timely results
- 3.8 percent indicate crime laboratory is not processing requests for DNA testing

# Both state and local crime laboratories are overworked, understaffed, and insufficiently funded.

Processing times at crime laboratories pose significant delays in many jurisdictions. State laboratories take an average of 23.9 weeks to process an unnamed suspect rape kit, and local laboratories average 30.0 weeks for such tests. The cost for testing these rape kits was estimated at \$1,100 per case, and does not account for many overhead costs. Both state and local laboratories indicated that personnel needs were among the most significant concerns for their DNA programs. Specifically, most crime laboratories expressed the need for supplemental funding for additional DNA staff; several laboratories indicated that their priority concern was for additional funding to augment current salaries to avoid the loss of skilled personnel to other prospective employers. A strong need was also reported for reagents (chemicals used in DNA analysis) and for technical equipment used for DNA analysis.

While the need for the expansion of mitochondrial DNA testing is substantial, the need for the more common nuclear DNA testing is even more pressing. Forensic laboratories reported nearly three thousand (2,999) cases over the last twelve-month period that could have benefited from mitochondrial testing.

# The role of the federal government in funding forensic DNA analysis has been, up to this point, important but rather minimal.

Finally, most laboratories also reported that while federal funding has played an important role in assisting with backlogged DNA cases, the proportion of their overall DNA budgets funded through federal sources is minimal. Only 20.5 percent of state crime laboratories receive 50 percent or more of their funding from federal sources; that figure is only 4.5 percent for the local laboratories.

#### BENEFITS FROM DNA ANALYSIS

Armed with a better understanding of the problem, policy makers subsequently ask for a definition of the expected benefits of devoting limited resources to the solution. Although this Report does not attempt a comprehensive, econometrically-derived cost-benefit analysis, the Report includes case studies on "preventable" crimes through strong DNA programs, along with an overview of DNA database successes.

# Forensic DNA databases are important tools in solving a variety of crimes, committed by a variety of criminals.

The analysis of DNA database hits presented in the Report provides a breakdown of 4,092 DNA database hits in 38 states. The majority of database hits have linked repeat offenders to violent crimes. In fact, for several years the Virginia Division of Forensic Science has reported that more than 80% of the hits on the state's DNA database would have been missed if the database had been limited to only violent offenders. Most states, however, have only recently begun to require DNA testing for non-violent offenders for placement on their DNA database. Similarly, several laboratories will not process DNA evidence from property crimes if it is sent to the laboratory for analysis.

# There are numerous crimes that are potentially preventable through better, more efficient use of forensic DNA analysis.

A review of specific cases in nineteen states reveals over 100 serious crimes that could have been prevented through either the inclusion of all convicted felons in the database or shorter DNA analysis processing times. The early identification of repeat offenders is possible with comprehensive DNA databases, and just how that identification would be made is highlighted in each actual case.

# The cost and the offsetting benefits associated with DNA testing present several questions relating to return on investment which merit further study.

Finally, there are many additional benefits, and indeed costs, that are derived from expanded utilization of forensic DNA analysis. Although this Report discusses some of those benefits resulting from DNA database hits and preventable crimes, there are several additional areas for further study which may give policy makers a more vivid picture of how forensic DNA issues impact their communities.

#### **ACKNOWLEDGEMENTS**

This Report could not have been completed without the participation and support of a great number of criminal justice professionals spread throughout the country. We wish to thank the nation's public forensic DNA laboratories whose response rate of 100 percent has guaranteed confidence in many of the results documented in this Report. The willingness of the heavily burdened state and local forensic laboratories to devote their limited resources to the detailed assessment tool sent for their completion is a testament not only to their professional dedication, but also to their recognition of the great need for national policy attention in this arena from all levels of government – local, state and federal.

We are also particularly grateful to those professionals in the law enforcement and forensic communities who participated in the pre-administration review of the draft assessment tool developed for this study. Given the considerable diversity in law enforcement agencies throughout the country, and the substantial variation in level of understanding and rate of utilization of forensic DNA, a thorough review of issues relating to the interpretation of and possible responses to the assessment tool was a primary concern of the researchers. Our topnotch review team was extremely instrumental in examining the draft assessment tools for weaknesses and omissions, and in providing a high level of confidence in the responses the assessment instrument ultimately generated.

In addition to the criminal justice professionals specified above, we also wish to express our gratitude to each local level law enforcement official who took the considerable time required to fill out the detailed assessment form. This group of public servants includes Police Chiefs and Sheriffs across the nation, as well as many tribal law enforcement executives. Similar thanks are owed to those many local prosecutors and crime investigators who assisted with the case studies and other research on DNA forensics.

And lastly, a very special thank you is owed as well to the officials at the *National Institute of Justice* who recognized at an early stage the importance of forensic DNA to the protection of public safety. This study is a reflection of their commitment to seeking to quantify the problems related to DNA analysis backlogs and to provide law enforcement and crime laboratories with workable solutions to those problems.

### **AUTHORS' NOTE**

In order to ensure accurate and honest responses, law enforcement agencies and crime laboratories were assured anonymity in the final published document. For this reason, some data are intentionally broad or not attributed to a specific source. In those instances where a law enforcement agency or crime laboratory is named, permission was obtained to both identify the agency and utilize specific data.

#### I. INTRODUCTION

This Report summarizes the findings of an in-depth Forensic DNA Assessment project undertaken by Smith Alling Lane in partnership with the Division of Governmental Studies and Services at Washington State University. The study employed multiple data collection approaches, including a nation-wide mail assessment of local and state forensics laboratories and law enforcement agencies, as well as extensive interviews of prosecutors, investigators and laboratory administrators. The findings detailed herein are based upon data obtained through this process from significant numbers of law enforcement agencies, local forensics laboratories, state forensics laboratories, and dedicated individuals in every U.S. state.

This study brings to bear scientific quantitative methodology and rigorous qualitative analysis on a question that has increasingly occupied public debate in both law enforcement and public policy circles. Recent newsworthy developments have fueled this broad interest – ranging from the activities of the *Innocence Project* in applying DNA analysis in pursuit of the release of those wrongfully convicted, to cases in which DNA has been used to solve long-unresolved cases and obtain convictions in extended serial cases. Heretofore *not* well addressed, however, have been the related questions of **capacity** and **backlog** at local law enforcement agencies and the corresponding impact, or potential for impact, at crime laboratories. While popular wisdom has acknowledged the existence of a backlog of cases that might benefit from the application of forensic DNA analysis, no clear insight into the extent of any such backlog has previously been available. Using the data compiled from this study, it is possible to make scientifically supportable estimates of the numbers of unsolved criminal cases in the U.S. which might benefit from DNA analysis, to assess both law enforcement and laboratory capacities for dealing with cases involving DNA, and to identify significant issues relating to the expansion of the use of DNA forensic analysis in criminal cases.

Given this general background, this Report addresses a number of key issues for the continued development of comprehensive statistical data on the effectiveness of forensic DNA in both solving and preventing crime. In doing so, Section II describes the methods used in administering the assessment tool to law enforcement agencies, local crime laboratories and state crime laboratories. Section III presents the analyses results that estimate the backlog of unsolved homicide and rape cases in the U.S. Section IV then provides an overview of a number of issues associated with the capacity of these various agencies to collect and process forensic DNA evidence. The issue of unsolved case "hits" (based on CODIS data) is then addressed in Section V; Section VI considers the issue of DNA testing and crime prevention; and a discussion of cost-benefit analysis issues is in Section VII. The Report concludes in Section VIII with a review of the major findings derived from the study and the resulting implications for public policy development.

#### II. METHODOLOGY

As noted above, this study applies several distinct data collection methods. Section VI, for example, relies entirely on information collected using telephone interviews of local crime investigators and prosecutors. Sections III, IV, and V, on the other hand, report analysis of data collected through a series of self-administered assessment tools. The methodology developed for this portion of the study involves an application of Don Dillman's *Total Design Method* to a self-administered mail survey of law enforcement agencies and forensics laboratories. The ultimate goal of the study was to arrive at defensible estimates of the numbers of **unsolved cases** (homicides, rapes, and property crimes) extant in the U.S., of the **backlog** pressure that might be anticipated if DNA testing were applied to those cases, and of the impact which such testing might have upon agency (e.g., law enforcement and crime laboratory) **capacity.** To accomplish this goal a sampling protocol was constructed to allow for the assessment of a representative sample of law enforcement agencies, as well as nearly all state and local crime laboratories.

All state and local forensic DNA laboratories were thus assessed. (For the purposes of this study, the term "local" laboratory(ies) is defined to include those laboratories serving city, county or regional jurisdictions.) Responses were obtained from 50 state laboratories and 70 local laboratories across the country. This level of response provides an overwhelming confidence in the analysis conducted using the data provided by these laboratories. Because of the large number of local law enforcement agencies in this country (between 15,000, and 18,000 - depending upon the inclusion or exclusion of "special police" agencies such as the U.S. Border Patrol, airport police, etc.), the study design called for the development of a law enforcement agency sampling protocol. This protocol allowed for the selection of agencies from within twelve basic selection strata. Working from up-to-date national mailing lists, both Sheriff offices and municipal police departments were divided into strata on the basis of the number of commissioned officers employed by a law enforcement agency. All police agencies with over 100 commissioned officers (both Sheriff and municipal police agencies) were included. Random samples ranging from 500 to 1000 were taken from the remaining sampling frames thus identified (Unspecified size, Under 25 Officers, 25-50 Officers, 50-100 Officers) to which all tribal law enforcement agencies were added to arrive at a final mail sample list of approximately 3,400 law enforcement agencies.

As discussed below, a significant number of agencies, particularly the larger organizations, completed the forensic DNA assessment questionnaire. Again, this relatively high level of response provides significant confidence in the results reported here. Local law enforcement agencies from every state, and representing approximately 154,467,000 citizens of the United States, responded to this assessment. Based upon the broadly representative nature of the many responding agencies, and based upon the overall response rate, the authors have a high level of confidence that the conclusions drawn from the analysis of the responses to the mail assessments are methodologically supportable and representative of law enforcement in the nation as a whole.

Working from the list of law enforcement agencies developed through the sampling protocol, and similar current lists for state and local forensic laboratories, a three-wave mailing process was administered, with each non-responding agency ultimately receiving three separate mailings of the assessment tool and a letter encouraging completion of the form. Once the mail process was concluded, select non-responding laboratories and agencies were contacted by telephone in order to check for any bias related to non-response error in the survey process. The

number of final responses (1,692 law enforcement agencies, 70 local laboratories, 50 state laboratories) provides significant confidence for the observations and conclusions contained in this Report. Response rates for each size and type of entity are set forth in the table below. These rates of response compare favorably to the anticipated 40 percent minimum response rate set at the outset for this study, with higher response rates from the laboratories and larger law enforcement organizations increasing confidence in findings relating to those entities.

	State Forensics Laboratories	Local/ Forensics Laboratories	Large LE Agencies	LE Agencies 50-100	LE Agencies 25-50	LE Agencies < 25	LE Agencies Unspecified
Responses	50 <sup>i</sup>	70	586	287	476	170	98
Percent	100%	100%	59%	52%	47.6%	34%	19.8%

194 Tribal law enforcement agencies were mailed assessment forms. However, with a response rate of approximately 27.8 percent (54 responses) the data did not generate statistically reliable information.

Analysis of the responses obtained from this mail-out assessment collection process indicates that the final assessment dataset derived from the data collection effort is broadly representative of the local law enforcement agency community nationally. The law enforcement agency responses are (in a geographic sense) proportionately spread across the agency size strata identified above, and they represent local police agencies from each of the fifty states. In this analysis, only tribal agencies and those of unspecified size are under-represented among participating agencies. In many instances the tribal agencies did not have access to the data sought by the assessment since the federal government (Bureau of Indian Affairs) typically handles these types of cases on reservations. One issue does arise from the consideration of the level of response achieved in the study. The smaller agencies did not respond at as high a rate as did the large agencies, and those agencies of unspecified size responded at a rather low rate. This unfortunate fact influenced both our analyses and some of the conclusions drawn; in light of this limitation, particularly conservative approaches to the analysis of central tendencies and in the estimation of backlogs were employed. Researchers compensated for this reduced confidence regarding the smaller agencies when dealing with total numbers or issues specific to these classes of law enforcement agencies. In follow-up questioning with those non-responding agencies, a lack of resources was the primary reason given for failure to complete the assessment. In many jurisdictions, case management systems are either non-existant or so antiquated as to not be of much utility in developing the numbers requested in the assessment. Moreover, local jurisdictions did not have the manpower to devote scarce personnel resources to manually reviewing old case files for open cases that may contain DNA evidence.

While due care was taken to develop conservative estimates, it should be noted that the analysis confirms the anticipated utility of the assessment process and the data obtained from the effort. For instance, relatively few agencies report difficulty in responding to the questionnaires (e.g., only 10.9 percent of large agencies report being unable to formulate an educated guess in response to a question about rape cases with potential DNA evidence). In short, sufficient data of good quality have been received as a consequence of the mail assessment collection process to allow estimation of national totals with considerable confidence, especially given the cautionary note above and the conservative estimation approach applied as a result.

#### III. CASEWORK BACKLOGS

One of the single most troublesome obstacles to generating accurate figures from the local law enforcement assessment is that a significant portion of respondents (particularly from the larger jurisdictions) indicate that an educated guess on issues of unsolved rapes and homicides is not possible without a time-consuming comprehensive review of case records. Many jurisdictions are not able, or possibly were not willing, to venture a guess as to how many cases are still open—much less speculate as to whether or not there may be biological evidence associated with such criminal offenses. In order to generate the most accurate estimates possible, project researchers worked directly with these jurisdictions over the telephone and through email communications to encourage supportable "educated guesses" for these few agencies. The researchers did not suggest to the agencies methodologies for developing numbers, but rather spoke with various departments within agencies (such as homicide units, sexual assault units and property rooms) to encourage coordination and completion of the assessment. combination of assessment instrument-based evidence and these "refined guesses" for a small number of law enforcement agencies, it was possible to generate figures for consideration which provide some needed perspective on the likely scale of the current forensic DNA analysis backlog issue in this country.

It is possible to apply multiple methods to the estimation of backlogs of cases involving forensic DNA evidence. Federal Bureau of Investigation (FBI) crime statistics serve as one source of estimation, the mail assessment process serves as a second source, and conventional statistical estimation techniques serve as yet a third. Of some limited assistance in this regard are national Uniform Crime Report (UCR) figures, reported annually by the FBI. These crime figures, unfortunately, are summary at best, and hence of rather limited utility. With these caveats in mind, a rough calculation of a 10-year figure for backlogged cases (murders and rapes) can be made using an average generated from the most recent three years reported by the FBI. This estimation approach leads to the conclusion that there are at least 49,000 unsolved murder cases in the U.S. (as of January 1, 2002). Because this method of backlog estimation does not assess truly old cases (older than 10 years), and because it does not include the immediately previous calendar years' figures, this figure is in all probability a low estimate. Applying the same cautious approach to the estimation of unsolved rapes leads to a figure of at least 470,000 unsolved cases. One significant shortcoming of UCR data as they relate to the purposes of this Report is that the FBI crime data summaries do not identify cases with biological evidence which could yield DNA findings.

Using the data obtained through the mail assessment data collection process, it is possible (as was done for the Preliminary Report on this study) to extrapolate total numbers of backlogged cases as well. For purposes of validation, this was done by calculating an average figure for each type of case for each of the six strata of law enforcement agencies. That mean number was then multiplied by the total number of agencies of that size in the United States to obtain a national backlog subtotal for each stratum. Adding up these figures provides an estimated national total for each type of crime. This approach provides an estimated backlog of unsolved murder cases of more than 170,000 cases, more than 60,000 of which involve DNA evidence. The same estimation approach yields figures of more than 593,000 total unsolved rapes and more than 430,000 unsolved rape cases featuring forensic DNA evidence. The full range of calculations associated with this approach and the UCR estimations are set forth in the appendices to this Report.

However useful these figures might be for a rough estimation, both of these approaches are somewhat problematic. The UCR data are too summary in character, and they do not include DNA information. The raw data from the mail assessment study are skewed upwards by the presence of the very large agencies, and/or agencies with very strong evidence retention policies, and rely uncomfortably on data obtained from the "unspecified size" agency category – a category with low response rate and a high degree of variation in response. In order to compensate for these shortcomings in the data, other more complex statistical techniques must be employed which provide a higher level of confidence.

Given the caution related to response rates discussed above, and the extremely skewed distributions of each of the measures being assessed in this section (unsolved rapes and homicides, and unsolved property crimes), which was driven by the large estimates from the larger law enforcement agencies, the following estimates were calculated in multiple ways to ensure that they are not the artifacts of a single method of statistical estimation. In particular, the following summary estimate figures were generated by a semi-parametric method that involved a weighted decile approach (excluding the cases at the 10<sup>th</sup> and 90<sup>th</sup> percentiles in the calculation of the mean and confidence interval estimates, and adding the excluded cases back into the end totals). This conventional statistical estimation approach was taken so that the estimates from the large law enforcement agencies would be reflected fully in the totals calculated, but would not disproportionately influence the calculation of the mean estimates.

These results were then checked according to several standard statistical diagnostic procedures. These commonly used procedures in the management of estimation problems included performing log transformations on each variable of interest in order to correct for skewness (separate analyses for both log base 10 and "e" were conducted), and calculating separate estimates within each strata of agency size (under the assumption that pattern of skewness may vary according to the size of the agency, which it certainly did) with and without log transformations. In the end, none of the results from these several diagnostic statistical procedures produced backlog estimates that differed significantly (at p<.05) from those reported in this Report. The researchers, therefore, quite confident that the estimates arrived at here are not contingent upon any particular analytic strategy.<sup>ii</sup>

#### A. Unsolved Rapes and Homicides

Table 3.1 contains the estimated total number of unsolved homicides and unsolved rapes extant in the U.S. at the time of the mail assessment data collection effort, along with estimates of the number of cases which likely feature possible forensic DNA evidence. Agencies were asked to count unsolved cases dating back to 1982 (twenty years). With regard to unsolved rape and unsolved homicide cases, there are currently an estimated 96,141 unsolved homicide cases and an estimated 304,178 unsolved rape cases reported by local law enforcement agencies in the U.S. in 2002-2003. Upon combining these two estimates, researchers arrive at an extrapolated total of 400,319 unsolved homicide and rape cases nationally. If the upper limits of the confidence intervals are used as an estimate (what will be referred to as "adjusted totals"), there may actually be as many as 432,179 unsolved homicides and rape cases nationally.

**Table 3.1.** Estimated national totals for unsolved homicides, rapes, and property offenses and estimates of cases with possible DNA evidence.<sup>iii</sup>

Offense Type	Total	95% Confidence Interval
Unsolved Homicides	96,141	91,281 to 101,001
Unsolved Homicides with DNA	48,324	44,904 to <b>51,774</b>
Unsolved Rapes	304,178	277,178 to 331,178
Unsolved Rapes with DNA	154,649	140,069 to <b>169,229</b>
Total Property Crimes with DNA	253,931	243,491 to <b>264,371</b>

Note: A similar table reports crime laboratory backlog estimates as well later in the text

Local law enforcement agencies were also asked how many murder and rape cases contained possible biological evidence that had not been sent to a laboratory for testing. These findings, also displayed in Table 3.1, show that a substantial portion of the adjusted totals for rape (169,229) and homicide (51,774) cases have not been sent to a forensic laboratory for testing—indeed, using the adjusted totals, an extrapolated total of 221,003 cases may contain biological evidence that has not been sent to a forensic laboratory for DNA testing.

### **B.** Unsolved Property Offenses

The majority of law enforcement agencies report that forensic DNA evidence is *not* routinely collected from property crime scenes, and similarly a preponderant majority of local and state crime laboratories report that law enforcement does not perform "routine" collection of DNA from property crimes. The common assumption (made by nearly half of the law enforcement agencies) is that their local or state crime laboratory does *not* accept DNA evidence from property crimes for unnamed suspects. However, 75 percent of reporting local laboratories and 88 percent of state laboratories report that they *do* accept and process unnamed suspect property crime evidence. These mail assessment findings point to a shortcoming in the understanding by law enforcement of the policies of forensic laboratories that serve them. These findings also signal an inconsistency between local law enforcement's understanding of what forensic analysis is being done versus what the actual services of their crime laboratories are in practice. They could also signal an inconsistency between the reported versus actual practices of those forensic laboratories.

Despite this apparent "information gap" between local law enforcement agencies' perceptions and crime laboratory policies, the information collected in the law enforcement assessment process still allows an estimated total number of property crimes that may contain biological evidence across the nation. Accordingly, using the adjusted totals in Table 3.1, **there** 

is an extrapolated total of as many as 264,371 property offenses with possible biological evidence in the U.S.

Thus, upon adding the adjusted numbers of homicides, rapes, property crimes and "other" cases with possible DNA evidence from law enforcement agencies and state and local crime laboratories, there are currently as many as 542,723 unsolved homicide, rape, and property offense cases for possible DNA analysis known to law enforcement agencies and crime laboratories in the U.S.

# C. Other Offenses

Other types of offenses for which DNA evidence could exist are not reflected in Table 3.1 because the assessment tool did not seek this information from local law enforcement agencies. Such offenses could include crimes such as assault, battery and kidnapping.

In designing the assessment tool, the researchers placed a considerable amount of emphasis on developing questions that would generate accurate responses without being overly burdensome on the responding agency. The initial emphasis of this question set was on counting the potential number of unsolved homicides and rapes with possible DNA evidence – those crimes typically considered the most heinous by society, and therefore the most important to solve. The researchers feel the general public come to expect DNA analysis for these types of cases.

The question regarding the number of unsolved property crimes was intended to provide a picture of where the United States could be headed in terms of future DNA analysis demands. While also asking for the number of "other" types of crimes with potential DNA evidence might have provided a broader picture of the number of backlogged DNA cases in the country, the researchers were concerned that requesting additional figures would only result in higher numbers of local agencies determining that they did not have the staff resources or automated systems required to complete the assessment.

Therefore, in reading the figures in the preceding sections A and B, it is important to bear in mind that these figures potentially exclude a large number of other types of crimes for which DNA evidence may be instrumental in solving. Because many crime laboratories do routinely track the types of cases for which DNA analysis is needed, a category for "other" case backlog was included in their assessment instrument. For comparison purposes, consider that of the over 57,300 backlogged DNA cases reported by state and local crime laboratories, at least 5,200 of these cases were classified as "other" – accounting for nearly 10 percent of all backlogged cases. (See Chapter IV, *Crime Laboratory Backlogs* for a further discussion of DNA backlogs at crime laboratories).

#### IV. AGENCY CAPACITY

This section addresses a number of issues concerning the capacity of the various agencies to collect and process forensic DNA evidence (see the appendix for the distributions of values for all variables in the law enforcement, local laboratory, and state laboratory assessments). In particular, this section is separated into the capacity issues facing law enforcement agencies versus local and state crime laboratories. The selected law enforcement issues to be considered include: (1) where law enforcement agencies typically send cases for DNA analysis; (2) biological evidence storage issues; (3) the reasons why DNA evidence for either unsolved homicides or rapes is not sent to a crime laboratory for testing; and, (4) cold case squad reviews.

With regard to the local and state crime laboratories, the selected capacity issues highlighted in this section include: (1) those associated with the evidence typically compiled in backlogged cases; (2) the cases that are currently within the statute of limitations; (3) the expected backlog of cases estimated by the crime laboratories; (4) the average time for analysis and the output capacity of the laboratories; (5) the major barriers associated with processing DNA evidence, with particular attention to property offenses; (6) the potential need for mitochondrial DNA testing; and, (7) cost/funding issues.

#### A. Law Enforcement

# Case Processing

With regard to general case processing, the findings set forth in Table 4.1 suggest that 80.1 percent of law enforcement agencies report that the primary location for sending DNA evidence for processing is state crime laboratories. Conversely, only 11.7 percent of law enforcement agencies reported that local and regional crime laboratories are the primary places to which they send forensic DNA evidence for testing. Another 4.2 percent reported that their evidence is sent elsewhere ("other"). However, in reviewing the description that these agencies included with their description of "other", it appears that this category in fact meets the definition of a "local" laboratory as used in this study; thus bringing the number of agencies using local laboratories up to 15.9 percent

**Table 4.1.** Primary locations where law enforcement agencies send DNA evidence for testing.

Location to which DNA Evidence is Sent	Law Enforcement Agencies Responding (%)
State Agency Laboratory	80.1
FBI Laboratory	1.1
Private Laboratory/Commercial Laboratory	2.9
Local Agency Laboratory	11.7
Other (regional, medical examiner, county)	4.2

At first glance it may appear that state crime laboratories are performing the majority of casework DNA analysis. Even so, many of the larger U.S. cities and counties, where the majority of reported homicides and rapes take place, are served by local crime laboratories. Thus, although a greater number of law enforcement agencies report that they send their cases to state laboratories for DNA analysis, a relatively significant proportion of the overall DNA casework in the U.S. is in fact conducted in local crime laboratories. In point of fact, an estimated minimum of 80 million U.S. residents are being served by such local crime laboratories. With an estimated U.S. population of slightly more than 280 million according to the 2000 U.S Census Bureau data, local laboratory DNA testing accounts for nearly 30 percent of all DNA testing being done in the country. Moreover, of the 25 U.S. cities with the highest crime rate per capita (as reported in the 2001 UCR data), more than half (14) are being served by local crime laboratories.

Additional analyses (see appendix) indicate that the larger jurisdictions also have greater needs for DNA testing. Specifically, in addition to having a larger caseload of rapes and homicides in general, the large law enforcement agencies reported a slightly higher estimate of the proportion of rape cases that are likely to contain DNA evidence, where 53 percent of large agencies estimated that between 75-100 percent of rape cases are likely to contain DNA evidence relative to 47 percent of all other law enforcement agencies. With regard to homicides, roughly 58 percent of both large and all other law enforcement agencies estimated that between 75-100 percent of all homicides are likely to contain DNA evidence. It should be noted that these estimates reflect the *expectations* of law enforcement, and are not necessarily indicative of the percentage of cases that test positive for DNA at the crime laboratory.

These types of "local considerations" must be taken into full consideration in reviewing the findings reported here. Therefore, although it is undoubtedly true that the majority of forensic DNA analysis is performed by state laboratories, it is also the case that the local laboratories primarily serve major metropolitan populations that have high crime rates, and as a consequence may generate a higher level of demand for DNA analysis. Many of these large local jurisdictions may also benefit from from better evidence collection training programs, developed with the assistance of the local crime laboratory.

Also of note, a relatively low number of law enforcement agencies utilize DNA analysis by either the FBI crime laboratory or private forensic laboratories. The FBI crime laboratory serves the agencies of the federal government, including the Bureau of Indian Affairs, and the District of Columbia almost exclusively. Though there is capacity at private laboratories for additional DNA analysis, many jurisdictions do not have the financial resources to pay the fees associated with such forensic testing.

## Storage

Of those cases where biological crime scene evidence is not sent to a crime laboratory for testing, or evidence is returned to the local agency if current technology cannot produce results, a central concern for many law enforcement agencies is the availability of appropriate storage space for unanalyzed evidence. Pressures on evidence storage space can result in degradable biological evidence being maintained under improper conditions—or worse yet, being discarded or not collected at all for a lack of space to store it safely. Lack of appropriate storage space can lead to valuable DNA evidence becoming degraded and requiring a more expensive and

potentially less exact DNA analysis process to be employed (see "Mitochondrial DNA Testing" section for further discussion).

In addition to the need to retain evidence from unsolved crimes in the event that new advances in forensic technology (such as DNA testing) may identify a suspect in the future, law enforcement agencies are also facing increasing statutory requirements to preserve evidence pertinent to cases considered "solved." Cases where post-conviction DNA testing has resulted in extraordinary exoneration have led a number of state legislatures to impose requirements for the indefinite storage of evidence used in serious crime convictions. While such systematic storage activity is important to the enhancement of the criminal justice system's capacity to "do justice" for its citizens, such requirements for evidence storage frequently take the form of an unfunded mandate passed down to local jurisdictions from their respective state governments.

It is also worth noting that over a fifth (22 percent) of law enforcement agencies reported that some of their unanalyzed evidence is stored at the crime laboratory rather than in agency evidence repositories. Among police agencies reporting that biological evidence is stored at the crime laboratory, nearly 40 percent were the largest jurisdictions. Again, considering that larger jurisdictions typically have higher crime rates and a high demand for forensic DNA analysis, this finding suggests that many crime laboratories may also be facing similar strains on evidence storage capacity. Moreover, many laboratories are also required to store evidence after analysis is completed, and therefore face many of the same unfunded mandates for evidence storage for "solved" cases as well as for "unsolved" cases. As a consequence many laboratories find they are responsible for not only storing their own cases – both unanalyzed backlog cases and analyzed evidence – but also those of the law enforcement jurisdictions they serve.

In the area of lost forensic evidence, there have been widely publicized reports of large metropolitan law enforcement agencies discarding potential DNA evidence (e.g., rape kits) in an effort to create additional storage space for new evidence. Regardless of whether these reported actions were the result of honest mistakes or the consequence of faulty agency decision-making processes, the fact remains that this critical evidence is forever lost to future crime investigations.

Table 4.2 summarizes these storage issues for local law enforcement agencies. As can be seen quite clearly in this table, most law enforcement agencies (79.0 percent) indicate that unanalyzed evidence is typically held in a centralized storage area. Six-in-ten (61.0 percent) of these agencies indicated that they currently have insufficient storage capacity for evidence retention needs relating to DNA evidence. It is also important to note that three-in-four (75.7 percent) of the large law enforcement agencies indicated that gaining additional space for the effective preservation of evidence was either of "critical" or "highly critical" importance (with 70.3 percent of all responding agencies responding similarly).

**Table 4.2.** Storage locations for unanalyzed evidence and long-term storage needs.

Storage Issue for Local Law Enforcement Agencies	Law Enforcement Agencies Responding (%)
Where unanalyzed evidence is stored	
Centralized storage area	79.0
Decentralized storage areas/various district locations	3.1
Prosecutor's facility	2.0
Crime laboratory facility	22.2
Other	5.6
Does agency have sufficient space for long-term evidence storage?	
Yes	39.0
No	61.0
Is the need for more storage space "critical" or "highly critical"?	70.3

### Failure to Send DNA for Testing in Rape and Homicide Cases

Part of what may be driving these backlog numbers and storage capacity issues at local law enforcement agencies are the specific reasons behind why forensic DNA evidence from unsolved homicides and unsolved rapes have not been sent to a crime laboratory for testing. Table 4.3 contains the percentages for common explanations provided to law enforcement agencies to choose from in the mail assessment instrument.

For those agencies indicating that rape or homicide evidence has not been sent to a crime laboratory for testing, Table 4.3 indicates that the primary reason is that a suspect has not yet been identified (31.4 percent of all responding agencies). Clearly, these "no suspect" cases are exactly the types of crime scene evidence that need to be submitted in order for the DNA database to be effective.

This finding is a strong indication that forensic DNA testing is not considered an investigatory tool by a significant portion of law enforcement agencies which have chosen not to send biological evidence to a crime laboratory for testing. Two other categories of response to this question are of a similar nature, and they warrant consideration as part of this discussion. The responses "A suspect has been identified but not yet charged," and "Analysis not requested by prosecutors" again show the bias towards using DNA analysis as a tool for the prosecution but not necessarily as an aid to identifying a suspect. When all three categories are combined, an estimated 50.8 percent of respondents indicated that forensic DNA was not considered a tool for

law enforcement criminal investigations. Rather, DNA evidence is considered a tool for the prosecution – evidence to secure a conviction after traditional police investigations have already identified the suspected criminal.

This revelation is particularly important for the corresponding impact it has upon DNA databases – crime scenes in which there are no suspects are precisely the types of cases that need to be submitted in order for the DNA database to be effective. The purpose of the DNA database is to link known offenders to crimes with no known suspects, and to link unsolved crimes together, thereby providing detectives either with suspects or with new investigatory leads. The fact that law enforcement agencies are purposely not submitting these cases indicates that there is limited understanding as to the nature and purpose of DNA databases. In fact, the study found that in many instances law enforcement officials were unaware of the fact that a DNA database has been established. For illustration, consider the following comments:

One large law enforcement agency commented simply:

"National DNA Database Needed."

Three other law enforcement agencies recorded these comments:

"A database similar to automated fingerprinting should be the goal. Perhaps regional database established by the DOJ for the nation."

"This department would be interested in seeing a DNA database started. It could be accomplished by making the giving of samples mandatory after conviction of a crime."

"Need national database with rules for collecting samples, like with fingerprints."

It should be noted that these comments came from law enforcement agencies scattered throughout the U.S., and from agencies of varying sizes (NOT exclusively small agencies). This limited knowledge of the DNA database is troublesome, but it should not be construed as an accurate reflection of law enforcement's desire for such a tool. In fact, when asked if the law enforcement agency filling out the assessment would be interested in using forensic DNA databases more frequently if there was a reasonable expectation that an unnamed suspect could be quickly identified, an overwhelming 96.9 percent responded positively. Many of the agencies responding negatively indicated that they needed no access to the database since the state laboratory took care of database searches. However, a handful of agencies also explained that they believed DNA to be too expensive for their jurisdiction or they had too few crimes to justify DNA testing. Also reflective of law enforcement's interest in becoming more knowledgeable about forensic DNA evidence in crime fighting is the fact that two National Institute of Justice resources - What Every Law Enforcement Officer Should Know About DNA Evidence -Beginning Level Module and Advanced Level Module - are frequently requested by law enforcement professionals. In fact, numerous agencies responding to the assessment specifically requested additional information about DNA and expressed an interest in the availability of pertinent training on the use of DNA databases. (These agencies were all put in contact with the appropriate personnel at the National Institute of Justice for further assistance.)

Another theme in agency responses regarding failure to submit biological evidence for testing relates to cost and resource (fiscal considerations). Individually, few agencies reported

that evidence is not sent to crime laboratories because of a lack of funding for DNA analysis (9.4 percent), the inability of crime laboratories to produce timely results (10.4 percent), or because crime laboratories are not processing requests for DNA (3.8 percent). However, each of these reasons may be considered in the overall context of *fiscal limitations*. Timely results require the existence of adequate capacity on the part of crime laboratories to handle demand, and such capacity is limited by resources. Crime laboratories that may not be processing requests for DNA testing do so primarily as a fallback means of *caseload management*, which is limited by capacity, which is limited in turn by resources. So, taken together, an estimated 23.6 percent of agencies do not submit DNA cases for reasons relating to poor funding. This grouping represents the second most frequently indicated reason for not submitting evidence for DNA testing.

In fact, the issue of *lengthy delays* in DNA analysis time was identified as a major concern by a significant number of respondents who chose to include comments with their assessment instrument responses. Police agencies throughout the country often face long delays in requested DNA analysis, which in turn limits the usefulness of DNA as an investigative tool for the police. Delays meanwhile are typically caused by crime laboratory resources that are inadequate to meet the demand for testing. The following comments regarding laboratory delays also reinforce the results discussed previously regarding the fact that law enforcement tends to view DNA as a tool for prosecutors and not for detectives:

"[The] state laboratory has over a four-year wait for processing DNA from property crimes. Therefore we do not submit it unless the crime is of major significance."

"DNA cases sent to laboratory with no suspect having been <u>arrested</u> are 1-2 year wait, even if a suspect is named and in [a] DNA database."

[The] State Police Laboratory system is inadequate for producing timely results – average waiting period is  $1\frac{1}{2}$  - 2 years."

"DNA evidence is beginning to have a significant impact on our investigations...However, the process of DNA evidence is expensive and creates a financial burden for the local police and the state laboratories. If funding was not an issue, DNA would prove to be on [sic] of the most valuable tools in solving cases."

These comments represent a sampling of the over 100 responding agencies which felt strongly enough about this issue that they were compelled to add written comments on their assessment response. Clearly, the delay in processing time and associated fiscal considerations are issues that law enforcement agencies have identified as obstacles to their use of DNA evidence. There also appears to be a disconnect between local agencies who complain of one to two year delays on DNA testing (and one agency reporting four year), and laboratories who report average processing times of 23-30 weeks (see further discussion in the section on "Processing Time and Output Capacity").

These issues bring to bear a larger question relating to evidence collection. Specifically, if law enforcement officers do not see DNA as a primary part of their investigation, and those agencies which would like to use DNA for investigations are limited in doing so due to evidence analysis turnaround times that are not constructive to ongoing crime investigations, then what

effect do these factors have on the likelihood that investigators will identify, collect and submit DNA evidence? On the basis of discussions with law enforcement officials, that this situation of systematic disincentives occurs more often than would be preferred. Consider the following true story related to researchers in one of these background discussions.

The owners of a car parked on a residential street heard a collision one night and ran outside to find their car had been hit by another vehicle. The local police department was called to the scene for an accident report. The owners of the damaged vehicle had noticed a section of the car where paint from the offending vehicle had been transferred during the collision, and asked the officers if they intended to take paint chip samples for forensic analysis. The officers agreed to collect a sample, but informed the owners that the sample probably would not be sent to the state laboratory for testing because it would be "too expensive." This was news to one of the owner's, who happened to be a state crime laboratory employee. Not only was he certain that local jurisdictions were not asked to pay for such testing, but the trace evidence unit also verified that they had the time and capacity to analyze these types of cases. After some follow-up with the investigating agency, the evidence sample was eventually forwarded to the laboratory for analysis.

While this example is not specific to DNA, it illustrates a line of reasoning which is worth contemplation: how much potentially valuable DNA evidence is simply not collected by law enforcement officials who have little hope that forensic analysis will be conducted in a timely manner? And when collected, how often is such evidence actually sent to the laboratory for testing? This scenario is quite likely for non-violent crimes such as property offenses where a large majority of law enforcement agencies report that their respective crime laboratories do not conduct such testing. In fact, some comments from law enforcement agencies on this issue seem to indicate that this may be a bigger problem than anticipated. For example, the comments from three responding agencies clearly suggest the scale of the problem in question:

"I have just recently been trained in collection of DNA. Before my training, I am unaware of any cases where DNA was collected."

"I am not familiar enough with DNA collecting to know how it effects [sic] my agency."

"We provided all property detectives with DNA evidence training 2 months ago with the request to make testing routine. This is in progress. However, it is a cultural change that is difficult."

This type of observation makes accurate extrapolation extremely difficult, but by looking at the converse of this scenario, the potential value of a fully functional forensic DNA crime laboratory setting may be understood. Virginia's forensic DNA program is among the most mature in the nation and the state database has been averaging one cold hit per day for the last two years. Virginia's processing time for DNA evidence, while not ideal, is by far more efficient than the majority of other forensic crime laboratories in the country. Additionally, and perhaps most importantly, Virginia's crime laboratory does not limit (within reason) the type of case or the type of evidence that can be submitted. This means that the Virginia crime laboratory is just as willing to conduct DNA analysis in a murder case as it is for a breaking and entering case under investigation. These factors – short processing time, database successes, and liberal case submission policy – have resulted in a steady rate of growth every year in the number of cases

submitted for biological testing. In fact the crime laboratory estimates that the amount of evidence submitted by law enforcement for DNA analysis grows by 30 percent every year. Does this mean that there were far *fewer* available DNA cases in previous years? Not likely, but rather the fact that crime investigators feel more encouraged to submit DNA evidence for analysis in Virginia likely accounts for that growth rate. This encouragement, coupled with the positive reinforcement of frequent DNA database matches, has resulted in a cadre of crime investigators across the state who tend to view the processing of crime scene DNA evidence as an effective means of reducing their caseload.

**Table 4.3.** Reasons why law enforcement agencies do not send DNA evidence from unsolved homicide and rape cases to a crime laboratory for testing.

Explanation for Not Sending DNA Evidence from Unsolved Crimes to Crime Laboratories for Analysis	% of Law Enforcement Agencies Responding
A suspect has not yet been identified	31.4
A suspect has been identified but not yet charged	10.2
Guilty plea is expected	13.9
Uncertain how DNA analysis would be useful in case	6.1
Lack of funding for DNA analysis	9.4
Inability of laboratories to produce timely results	10.4
Laboratory is not processing requests for DNA testing	3.8
Uncertain where to send the case for DNA analysis	0.9
Analysis not requested by prosecutors	9.2
Other	9.7

#### Cold Case Review

The final backlog issue to explore is cold case review squads and policies. Cold case squads have been used in various jurisdictions throughout the country to reopen old cases for new investigatory leads. Many such cases have been found to contain DNA evidence which has been tested to solve crimes that are decades old. This type of review recently led to the identification of a suspect in the "Green River Killer" serial murders. The Green River Killer was thought to be responsible for 49 deaths over three years in the Seattle, Washington area. Biological evidence was collected from many of the murders, which began in 1982, and was kept in storage. After a review of the evidence in 2001, some pieces were selected and sent to the state crime laboratory for possible DNA analysis. The state crime laboratory was able to extract a DNA sample, and the analysis produced a match to a man who had always been considered a

prime suspect by investigators. The suspect was initially charged with seven counts of murder, and eventually pleaded guilty to 48 counts of murder. The example of the Green River case is a valuable one because it demonstrates the utility of DNA to old investigations in which primary suspects are known but could not be linked directly to the crime.

Law enforcement agencies were asked if they have a "cold case squad" or other established policy for reviewing unsolved violent crimes for fresh investigatory leads, such as new DNA testing. It is important to note that many agencies, particularly the smaller ones, do not have the organizational resources to form actual "cold case" squads. For this reason, the assessment instrument question on this topic included language that allowed for "other policy for reviewing unsolved violent crimes." Of those agencies with a cold case squad (or a related periodic unsolved case review policy), the assessment also inquired as to the number of cases potentially eligible for review, how many cases have been reviewed recently, and how many cases have been sent for DNA testing.

Table 4.4 contains the results posed by these questions. As indicated, only slightly more than one-in-five (22.8 percent) law enforcement agencies report that they have a cold case squad or similar review policy, and possibly due to resource issues mentioned above the percentage for large agencies is somewhat higher at 38.6 percent. Among the group of agencies without a current cold case squad or related policy, 49.4 percent of the large agencies (and 37.9 percent of all agencies) report that they plan to review old, unsolved cases for new investigatory opportunities such as DNA testing. This leaves just under half (48.6 percent) of the agencies which neither have cold case review policies in place nor any plans for such a review. One large agency indicated that the task of responding to the DNA assessment inspired a discussion on cold cases in the jurisdiction that could potentially benefit from DNA testing. The agency had not previously considered reviewing cold cases for DNA, but was now planning a systematic agency review. Although it is encouraging to learn that this data collection effort has possibly spurred the review of solvable cases that may otherwise have remained in storage, it is also somewhat telling that the agency had not considered such a review beforehand.

Table 4.4 also shows a significant variation in volumes of cold cases that have been reviewed (or are considered eligible for review) by those agencies reporting the existence of cold case squads or periodic unsolved cases review policies. This variation is evidenced by the fact that the mean and median values are substantially different from one another—indicating significant positive skewness—in the distributions of these variables. In other words, there is a high degree of variation in the responses where many agencies report relatively small numbers and a few agencies report extremely large numbers on these items. Thus, a median number of 0.0 arose due to this severe "upward skewness." A median number of 0 indicates that a significant number of agencies with cold case review squads have not sent any of their cases to a crime laboratory for DNA testing; only a handful of agencies have been responsible for the majority of DNA testing coming out of these specialized units. This failure to submit DNA evidence could be due to a variety of factors, ranging from a determination of non-probative value, to a tendancy of law enforcement to rely more heavily on "traditional" investigative techniques -- particularly when confronted with long backlogs at public crime laboratories and testing fees at private laboratories. While this assessment did not specifically ask cold case squads to quantify underlying reasons for why DNA evidence has not been submitted, clues to possible answers may be found in the previous section's general discussion on treatment of DNA evidence by law enforcement agencies ("Failure to Send DNA for Testing in Rape and Homicide Cases").

**Table 4.4.** Cold case squad review.

Assessment Item	ent Item Statistics	
Does agency have a "cold case squad"?	Yes = 22.8%	No = 77.2%
Of those jurisdictions that have a cold case squad	Median	Mean
Cases eligible for review	5.0	71.3
Cases reviewed	4.0	22.3
Reviewed cases sent for DNA testing	$0.0^{\mathrm{vi}}$	6.3

#### **B.** Local and State Crime Laboratories

### Crime Laboratory Backlogs

State and local crime laboratories reported an estimated 57,349 backlogged cases awaiting DNA analysis, with 34,723 cases at state laboratories and 22,626 cases at the local and regional laboratories. It may be important to note that these backlogged cases appear to be only those crimes that have been identified by the crime laboratory as requiring DNA analysis. The assessment instructions defined a "backlogged case" as an unreported forensic case that has been in the possession of the crime laboratory for more than 30 days. However, after looking at data reported by one state crime laboratory it appears that law enforcement officials in most states have been counting only those cases for which it had already been determined that DNA analysis was needed. The state in question reported 14,695 rape cases, 12,023 homicide cases, 51,616 property crime cases, and 168,414 "other" cases in the "backlogged" crime cases category. While this state's crime laboratory will certainly not be performing DNA analysis on all 210,748 backlogged cases, it is a useful number in illustrating the great demand potentially facing crime laboratories in general, and in explaining the long turnaround times over which some police agencies express considerable concern. (This state crime laboratory also provided the number of backlogged cases that had been identified for DNA testing, which was used in calculating the national total.)

Table 4.5 displays the mail assessment-generated information regarding backlogged rape cases featuring rape kits and known DNA evidence. As indicated, both the local and state crime laboratories report that the majority of their backlogged rape cases (between 75-100 percent) contain rape kits only (additional analyses indicate that both local and state laboratories are also generally consistent in that the majority of these rape kits are likely to contain DNA evidence). Furthermore, a slightly larger percentage of local crime laboratories reported that over half of their backlogged rape cases feature rape kits as well as other forms of DNA evidence. Finally, both state and local laboratories note that backlogged rape cases with other DNA evidence exclusively are quite rare. These responses potentially point to a bigger issue of heavy reliance on rape kits for evidence. As the public acceptance and understanding of forensic DNA continues to grow, there is also a good chance that criminals may become increasingly savvy and begin leaving fewer obvious DNA clues behind at the scene of their crimes. Moreover, not all

rape kits will necessarily yield a testable DNA sample. For these reasons, it is important that investigators do not begin to rely too heavily on the collection of a rape kit in sexual assaults for all of their physical evidence. Full processing of the crime scene may capture DNA evidence that may prove invaluable if the rape kit does not produce results.

One backlog issue *not* discussed in this report regards those cases that have previously been analyzed by crime laboratories, but which require additional analysis before they can be entered into the national DNA index system. These cases fall into one of two categories: cases analyzed using the RFLP method of testing, and cases for which only partial STR analysis was completed. For several years before the current method of DNA testing (STR analysis) was introduced and accepted by the scientific community, forensic DNA analysis was completed using a method called RFLP. Unfortunately, results from cases analyzed using RFLP are not transferrable into an STR result. This means that any testing that yielded RFLP profiles (either offender profiles, or unknown suspect profiles) must be reanalyzed before they can be compared to the database population of STR profiles. These particular cases are not typically counted in a laboratory's "backlog" of cases that need to be analyzed, since, indeed, these cases have already been analyzed. The number of RFLP cases requiring STR analysis for entry into the DNA index system is unknown.

The other type of possible uncounted backlogged DNA cases are those that have only been partially profiled. In order to be eligible for entry into the National DNA Index System, a DNA profile must be analyzed at 13 points, called *loci*. For a number of years many laboratories reduced their costs by analyzing fewer than 13 loci (usually 9 or 10 loci). While these results are typically adequate for evidentiary purposes, and are sufficient for entry into the Local or State level of DNA Index Systems, they are *not* permitted to be loaded to the national system for searching between states. Today there are only a handful of laboratories that are not using all 13 loci in their analysis of National DNA Index System eligible cases. However, the number of cases that were analyzed using partial profiles in the past has not been counted and could be quite sizeable in some jurisdictions. Moreover, since there are a number of laboratories that continue to use partial profiles, this number is likely increasing.

**Table 4.5.** Backlogged rape cases, rape kits, and DNA evidence from local and state crime laboratories.

Backlog with biological evidence (%)	Jurisdictions with Backlogged Cases with Rape Kits Only (%)		Backlogged Rape Kits	ions with   Cases with   and Other   dence (%)	Backlogged Other DNA	ions with Cases with A Evidence (%)
	Local	State	Local	State	Local	State
0-10%	10.0	3.0	13.7	2.9	56.9	64.7
11-25%	30.0	18.2	13.7	14.7	31.4	26.5
25-50%	14.0	21.2	19.6	41.2	5.9	5.9
50-75%	12.0	24.2	35.3	20.6	3.9	2.9
75-100%	34.0	33.3	17.6	20.6	2.0	0.0

## Statutes of Limitations and Anticipated Backlogs

Crime laboratories reported at least 1,637 backlogged rape cases that were expected to exceed the statute of limitations for prosecution by June 30, 2003. A significant number of these cases were reported to be held by the local laboratories. Furthermore, this number was generally expected to increase in the coming six-month period rather than decrease, thereby furthering the possibility that additional cases could have expiring statutes of limitation before the backlog is eliminated.

It should be noted that local jurisdictions were also asked to estimate the number of cases for which the statute of limitations may be a factor, but a very high number of agencies either did not respond to the question or indicated that an educated guess was not possible. This inability to track important information relating to cases points to a considerable deficiency in case management systems. Without such systems, the burden placed on law enforcement to review cases and respond to opportunities provided by advancements in forensic and other crime fighting technologies is overwhelming.

**Table 4.6.** Status of cases with regard to statutes of limitations (SOL) and anticipated backlogs.

Laboratory Type	Cases within SOL	Cases to Exceed SOL by 6/30/03	Anticipated Backlog
Local Laboratories	Mean = $232.5$	Mean = $49.1$	Mean = $229.6$
	Total = $7,440$	Total = $1,522$	Total = $10,789$
State Laboratories	Mean = $372.1$	Mean = 5.5	Mean = $377.6$
	Total = $7069$	Total = 115	Total = $20,371$
Totals	14,509	1,637	31,160

Criminal justice professionals and policy makers also have been considering statute of limitations issues in conjunction with changes to forensic DNA policies. The issue is twofold. First, there are known instances where DNA backlogs – either in casework samples or offender samples – have resulted in DNA database matches that occurred after the statute of limitations for prosecution had lapsed for the likely guilty party. One state reported an estimated 150 such positive matches being made after the statutory period of exposure to prosecution had expired for repeat offenders in that state.

Secondly, an abbreviated statute of limitations period negates the long-term effectiveness of the DNA database in that the crime investigators making use of the database will not be given a full opportunity to succeed in matching across crimes and/or persons. The DNA database is an effective tool to catch recidivists. In many instances, a known offender entered into the database will be identified at a later date when he or she commits a subsequent offense that leads to the collection of DNA forensic evidence. Even with a year-long backlog, DNA testing should still be completed in sufficient time for the prosecutors to bring forward legal charges. However, the reverse scenario is also equally important in making DNA matches. For example, a crime occurs, and at some later date when the offender commits a crime that requires a DNA sample

for the database, a match is made retrospectively to the earlier unsolved crime(s). These "backwards" matches are those limited by short statute of limitations periods. Consider a state where the statute of limitations is six years where an unknown rapist victimizes a citizen in 2003 and eight years pass before the offender commits a crime that qualifies for the DNA database. In this situation, the 2003 rape can never be prosecuted. In such cases the crime victim may never receive closure, and the public safety is seriously compromised.

To be certain, the statute of limitations represents a very important element of "quality control" in the criminal justice system. The limitation on prosecutorial exposure protects individuals from the natural tendency for most evidence to become increasingly less reliable over time. Eyewitness testimony becomes less certain; a person near the scene of a crime may have difficulties rebuilding the reason for his/her proximity over the passage of several years. The reliability of DNA testing, however, is quite different; it is largely undamaged by the passage of time. DNA forensic evidence has been used quite frequently to convict criminals decades after the commission of their crime. It should also be noted that while a fingerprint may be found at a crime scene for a variety of legitimate and/or illegitimate reasons, a semen sample has very few legitimate reasons for being part of the evidence found in a rape examination of an unknown suspect assault.

Many states have responded to this statute of limitations problem by enacting new legislation intended to extend or remove the statute of limitations for specific violent crimes. One option that is quite popular among U.S. states is to remove the statute of limitations temporarily in sexual assault cases where DNA evidence is available. Once a match is made on the database, however, prosecutors have a set number of years to bring charges against the suspect in question. In this way, matches made on the DNA database can still be prosecuted, but prosecutors cannot indefinitely postpone a trial. Wisconsin, Delaware, and the federal government have enacted statutes that specifically permit "John Doe warrants" for specific crimes. However, Wisconsin subsequently completely removed the statute of limitations, negating the need for such warrants.

### Processing Time and Output Capacity

State and local crime laboratories also differ substantially in terms of the average time for processing DNA evidence and the average annual output capacity. Laboratories were asked for the approximate length of time required for the analysis of a typical (non-priority) unnamed suspect case rape kit, assuming a vaginal swab with one perpetrator, one victim, and that the time runs from the date the rape kit is received by the laboratory until analytical results are reported. This question was worded very specifically to control for a number of factors that can affect processing times. For example, cases that have more than one perpetrator, multiple pieces of evidence, or where analysis must control for known consensual partners of the victim, can lead to longer analysis periods. Delays between the collection of evidence and actual submission of the evidence to the crime laboratory can also lengthen the overall turnaround time. Conversely, cases that become a priority, either for public safety reasons or because of publicity, can be sped through the system at a much quicker rate.

The findings reported in Table 4.7 show that state crime laboratories take an average of 23.9 weeks, and local laboratories average 30.0 weeks for DNA processing. In addition, state crime laboratories process an average of 1,284.5 samples per year as opposed to the local

laboratories which average an output of 771.4 samples per year (with an extremely wide degree of variation between lows and highs). A potential explanation for this difference may lie in the fact that state crime laboratories tend to be slightly more process oriented because DNA evidence is generally submitted to the laboratory from remotely located agencies. In contrast, local laboratories are generally embedded in the law enforcement agencies they serve, and hence may have a more significant role in determining which evidence is of the most probative value.

A thorough understanding of the scope of capacity problems must also consider the wide range in existing capacity at crime laboratories. Consider the neighboring states of Virginia and North Carolina in this regard. Virginia has a population slightly smaller than North Carolina (by approximately 1 million, according to 2000 U.S. Census Bureau data), and Virginia's crime rate of 3,178 crimes per 100,000 is lower than that of North Carolina's at 4,938 (according to FBI UCR data). However, Virginia's leaders recognized the importance of DNA to public safety in the late 1980's and the state has steadily built its DNA program to rank among the finest in the To date, Virginia has approximately thirty-eight DNA analysts and processes approximately 8,172 DNA samples per year (with approximately 8 samples per case). In contrast, North Carolina has been slow to build its DNA program – its crime laboratories (a state forensic laboratory and a local laboratory serving Charlotte-Mecklenberg) employed a combined total of six DNA analysts (four at the state facility and two at the local laboratory) at the time of this Report. Moreover, as a matter of policy, North Carolina's state laboratory will not accept evidence for DNA analysis for any case in which there is no known suspect (with obvious exceptions for high profile serial cases) and has a processing time of three months. The local crime laboratory for Charlotte-Mecklenberg, which has a six-month processing time, accepts no suspect cases for violent crimes, but places limits on acceptance of property crime cases with no suspects. The two crime laboratories in North Carolina estimate their combined backlog of DNA cases at 720 (320 for Charlotte-Mecklenberg, 400 for the state). Virginia, in contrast, accepts nearly all cases (within reason), has a processing time of four months, and an estimated backlog of just over 2,000 cases.

Intuitively, it is reasonable to assume that higher output capacity should result in shorter processing times and smaller backlogs. However, when considering the Virginia-North Carolina comparison, the opposite was found to be the case. Virginia's capacity nearly doubles that of North Carolina, yet Virginia also has a larger backlog and generally a longer processing time. It seems unlikely that North Carolina simply has fewer crimes with biological evidence. In fact, given the higher crime rate in North Carolina, it is reasonable to assume just the opposite – that the state should have a larger number of crimes for DNA testing. This scenario begs the question of what is happening to biological criminal evidence in states such as North Carolina. There is good reason to believe that many of these cases are sitting in storage, either in crime laboratory evidence rooms or in local law enforcement evidence storage areas. It is also possible that biological evidence simply will not be collected for crimes if there is little real hope that it will be analyzed.

Another important comparison to make at this point involves the average DNA processing time considered acceptable in England. England's nationally established goal for the processing time of forensic DNA evidence is 24 days, and the Government has deemed as unacceptable the current processing time of 33 days for sex assault cases (See the attached report, *The Application of DNA Technology in England and Wales*). By comparison, in the U.S., the current average processing time for unnamed suspect rape kits is between 24 and 30 weeks (on the extremes, one local laboratory reported a 208 week backlog, and a state laboratory

reported a 78 week backlog). Similar to England, many laboratories throughout the U.S. also have a stated goal of reaching a 30-day processing time. While this is a laudable goal, the problem is that most crime laboratories have no actual means of reaching this goal. Take for example the Oregon crime laboratory. In the late 1990's, crime laboratory administrators had announced a processing time goal of 30 days, although many on the DNA staff privately believed that target to be unreachable. However, due to a steady stream of funding and some changes to operational procedures, in 2000 the DNA section had reduced the DNA evidence processing time to sixty days or less. The next target of thirty days suddenly seemed to be an achievable goal. Unfortunately, at about the same time the DNA section was faced with a hiring freeze while demand for DNA testing continued to grow. Then, a budget crisis in 2002 saw the State Police budget decimated and the DNA staff was reduced by 65 percent (a staff of twelve DNA personnel were reduced to only four). By mid-2003, the DNA section was largely reconstituted, but the damage had been done. Oregon currently has a processing time of at least 10 weeks for a non-priority rape kit, and is still not operating at pre-2002 levels with regards to trained staff and output capacity. The goal of 30 days for processing of DNA evidence again seems far-fetched without a significant infusion of additional funding.

**Table 4.7.** Average analysis time and output capacity for state and local crime laboratories.

Laboratory Type	Avg. Analysis Time (in Weeks)	Avg. Annual Output Capacity
Local Laboratories	30.0	771.4
State Laboratories	23.9	1,284.5

Note: Average analysis time is based on a non-priority, no suspect rape kit. Average annual capacity is measured in samples per year.

# Barriers to Processing DNA Evidence of Property Crimes

State and local crime laboratories were also asked about the barriers associated with processing DNA evidence from property crimes (in particular, unnamed suspect property crimes). Table 4.8 displays the results of these items. The most frequently indicated barriers to the processing of evidence in such cases have to do with *personnel shortages* and the *backlog of cases* with higher priority. The more secondary concerns expressed by crime laboratories include reagents and equipment needs and the inadequacy of operational space.

**Table 4.8.** Reported barriers associated with processing DNA evidence from unnamed suspect property crimes for local and state crime laboratories.

Barriers to Processing DNA Evidence	Local Laboratories (%)	State Laboratories (%)
Lack of Storage Space	5.2	9.3
More Personnel Needed	34.5	30.2
Backlogs/Other Priority Cases	39.7	30.2
Reagents and Equipment Needed	12.1	18.6
Lack of Operational Space	10.3	16.3
Prosecution Must Request	8.6	4.7
Investigator Must Request	17.2	4.7
Other	6.9	4.7

On a related note, an item was included in the local laboratory mail assessment instrument concerning the various obstacles facing local crime laboratories in getting DNA evidence from either unsolved rapes or unsolved homicides in queue at the laboratory. As indicated, and consistent with the previous set of analyses, the findings reported in Table 4.9 indicate that funding and backlog considerations both play an important role in getting DNA evidence from unsolved homicide and rape cases into the queue. This question was not posed to state laboratories because it essentially asks the respondent to opine on why detectives in their jurisdictions have not submitted cases. While local laboratories could reasonably be expected to have good insight on this issue given their closer proximity to the investigative unit, state laboratories, which sometimes serve hundreds of jurisdictions, cannot be expected to have the same detailed understanding.

Even so, several important differences do emerge here. In particular, local laboratories noted the importance of at least three additional reasons why DNA evidence from these types of cases may not be processed: a suspect has not yet been identified (noted as important by 37.0 percent of local laboratory respondents), a guilty plea is anticipated (30.4 percent), or the DNA analysis may not be requested by the prosecution (34.8 percent). Other important factors included the situation in which the subject has yet to be charged (26.1 percent) and cases wherein the agency may be uncertain as to how the DNA analysis my help close the case (28.3 percent).

In short, while both local and state crime laboratories indicate that staff personnel and backlog issues are the primary barriers to processing DNA evidence from property crimes, the results from the local crime laboratory assessment indicate a much more complex picture with regard to the processing of DNA evidence from unsolved homicide and unsolved rape cases.

**Table 4.9.** Reported barriers associated with processing DNA evidence from unsolved homicides and/or unsolved rapes for local crime laboratories.

Barriers to the Processing of DNA Evidence	Local Crime Laboratories (%)
A suspect has not yet been identified	37.0
Suspect identified but not yet charged	26.1
Guilty plea is anticipated	30.4
Agency uncertain how DNA may help case	28.3
Lack of funding for DNA analysis	30.4
Backlog at laboratory prevents timely results	45.7
Laboratory is not processing requests for DNA testing	2.2
Agencies uncertain as to where to send case for analysis	15.2
Analysis not requested by prosecution	34.8
Other	30.4

# Mitochondrial DNA Testing

Another DNA testing option that warrants some discussion in this study is the analysis of mitochondrial DNA (or mtDNA). Testing of nuclear DNA is by far the most common analysis conducted in the United States and in other countries. This testing results in the measurement of the *short tandem repeats* of nuclear cells in DNA, and is therefore commonly referred to as STR testing. However, only two copies of nuclear DNA are present in human cells (once copy from the biological mother and one copy from the biological father), which means that a deterioration of the cell sample, or a sample that has only a few cells, may not be testable through STR analysis. On the other hand, mitochondrian, which contain mtDNA, are present in hundreds to thousands of copies per cell, and therefore not only withstand degradation better but can also provide more testable molecules per sample. This type of testing is particularly useful for evidence such as bone fragments and hair samples where nuclear DNA may not be present.

However, mtDNA testing is rather limited in criminal forensic identification applications. MtDNA is passed down through maternal lines, with all descendants in a given maternal line sharing the same mtDNA sequence. So, while mtDNA may conclusively prove that a particular person is related through maternal lineage to a specified woman, it does not serve the purpose of uniquely identifying the individual in question. Indeed, persons who would appear to be unrelated may in fact share a common maternal relative at some point in their lineage, and this would lead to the same mtDNA results. As a result, mtDNA testing can be used in criminal cases to positively include or exclude a person as a possible suspect, but does not provide the

discriminatory power that the forensic community has come to expect from nuclear DNA analysis. MtDNA testing is also more labor intensive, and therefore considerably more expensive than nuclear DNA testing.

Both state and local laboratories were asked about the degree to which their unsolved cases may benefit from mitochondrial DNA testing. Specifically, the laboratories were asked to estimate: (1) the total number of forensic cases (for the calendar year 2002) analyzed in the laboratory that failed to yield testable amounts of DNA for STR analysis, but could potentially benefit from mitochondrial DNA testing; and, (2) the number of hair analysis cases received by the laboratory that would benefit from additional mitochondrial testing.

Table 4.10 contains the results of these items. Both local and state laboratories reported that, although not as substantial as that for nuclear (STR) DNA analysis, a number of cases could potentially benefit from mtDNA testing. To be sure, local laboratories note that a total of 956 cases could benefit from such testing, and state laboratories indicated an additional 2,043 cases.

This question was not posed to local law enforcement agencies given that a need for mtDNA testing is primarily a scientific determination for which general law enforcement personnel do not have the appropriate expertise to ascertain. However, it is probable that, of the estimated 542,743 backlogged cases with biological evidence held by local law enforcement agencies and crime laboratories, some number of these cases may require mtDNA testing.

This may be especially true for those jurisdictions (70.3 percent) that reported that the need for additional evidence storage space was either "critical" or "highly critical." Many of the evidence storage rooms across the country do not necessarily maintain the climate control conditions that are ideal to DNA evidence preservation. Although DNA can typically withstand fluctuating temperatures and endure over very long periods of time, it is not indestructible. Moreover, many of the evidence policies and protocols at local law enforcement agencies may not be conducive to long-term storage of biological materials. Although this study turned up no such admissions, stories of biological evidence being "stored" in squad car trunks or under stairwells at the stationhouse are commonplace in the forensic science community.

In addition to the evidence contamination and chain of custody problems that the inappropriate storage of any crime scene evidence could present, haphazard storage is also far from an ideal manner of storage for the preservation of biological evidence. In cases where DNA evidence has deteriorated to a point where the more common STR testing cannot be completed, mtDNA testing may be appropriate. Currently, there are only a few crime laboratories around the country with the ability to conduct mtDNA testing. With demand being present but not overwhelming, the question has been raised as to whether it may prove more cost-effective to maintain regional mtDNA testing laboratories rather than attempting to establish this capacity at each state and local forensic laboratory.

Another factor possibly affecting demand for mtDNA testing is that fewer than half of all states have obtained court decisions permitting mtDNA test results as evidence at trial in criminal cases. It is possible that the need for mtDNA testing could expand as more states rule on its admissibility as evidence. It is also worth noting that mtDNA testing results are not comparable to nuclear DNA testing results, and therefore cannot be loaded into criminal DNA database indexes (see CODIS discussion in section V). However, an index for mtDNA analysis has been included as part of the CODIS *Missing and Unidentified Persons Database*. MtDNA testing is being used by a growing number of state laboratories which intend to participate in this database program. According to statistics maintained by the FBI's National Crime Information

Center (NCIC), there are nearly 5,000 reported unidentified persons in the United States at this time.

**Table 4.10.** Forensic and hair analysis cases that may benefit from mitochondrial DNA testing.

Case Type	Local Laboratories Total (avg.)	State Laboratories Total (avg.)
Forensic Cases	336 (8.6)	897 (28.9)
Hair Analysis Cases	620 (17.2)	1,146 (47.8)
Total	956	2,043

## Cost/Funding Issues

Many of the major obstacles associated with DNA analysis concern the costs/funding issues involved. Accordingly, the local and state crime laboratories were assessed on a number of these issues. As a first step in this process, local and state crime laboratory administrators were asked to indicate from what sources the bulk of their funding comes. Table 4.11 shows that the majority of funding for the local crime laboratories derives from local sources. Accordingly, state crime laboratories receive most of their funding from state sources. Conversely, most state laboratories (91.6 percent) receive little to no funding from local sources, and 88.9 percent of local laboratories report a similarly low level of assistance from the state.

What this table also reveals is that federal funding is not a significant portion of the overall DNA budgets of state and local crime laboratories. Indeed, only 20.5 percent of state laboratories receive at least half of their funding from federal sources—a figure that drops to a mere 4.5 percent for local laboratories. Moreover, nearly half of all local laboratories reported that 10 percent or less of their DNA budget was attributable to federal sources. On the other hand, only 23 percent of state laboratories reported this lowest level of funding (see appendix). When considering that many of the local laboratories handle extremely high volumes of forensic DNA cases, this difference in reliance on federal money is startling.

However, there may be two plausible explanations for this difference. First, local crime laboratories are not permitted to apply directly for federal funding grants for DNA analysis. Instead, local laboratories must apply as a consortium through the state laboratory. Although this application process works well for a number of state and local laboratories, some local laboratories privately complain that they do not get a fair amount of this grant money. This situation also potentially leaves local laboratories at the mercy of a state's level of interest in

applying for federal funding. Additionally, the process of coordinating a consortium application – particularly in those states where local laboratories are numerous – can be difficult.

A second reason why these data show a higher degree of reliance at the state level on federal funding is because the question did not allow for a distinction between funding received for casework versus funding received for offender samples. Offender DNA samples are the sole responsibility of the state, and therefore local crime laboratories are not eligible to apply for federal grants for offender DNA analysis.

**Table 4.11.** Sources of funding for state and local crime laboratory budgets for DNA analysis.

Funding Source	Less than 50% of Funding	50% or Greater of Funding
Local Crime Laboratories		
Local	19.6	80.4
State	88.9	11.1
Federal	95.5	4.5
State Crime Laboratories		
Local	91.6	8.4
State	27.8	72.2
Federal	79.5	20.5

Local and state crime laboratories were also asked to estimate the average cost of processing an unnamed suspect rape kit (assuming a vaginal swab with one perpetrator and one victim). Table 4.12 indicates that both state and local laboratories arrived at the same approximate cost of \$1,100. In arriving at this estimate, the majority of local and state laboratories included the costs of reagents and salaries, but very few included costs associated with overhead and equipment. It is important to remember that these estimated costs are for what is considered a very simple rape kit case, as defined above. There are a variety of other factors that can significantly increase costs associated with DNA analysis, including the size, quantity and condition of the evidence, the number of perpetrators involved, and contaminants. To repeat, the \$1,100 per case estimate is only for the most rudimentary of rape cases, and it does *not* include many overhead considerations. As forensic cases become more complex, the cost of analysis will quickly rise.

**Table 4.12.** Average cost of rape kit analysis and frequency that factors associated with costs were included in the estimate.

	Local Laboratories	State Laboratories
Average cost of rape kit analysis	\$1,153.95	\$1,041.27
Factors associated with costs		
Reagents	75.9%	67.4%
Equipment	34.5%	27.9%
Salaries	65.5%	48.8%
Overhead	39.7%	11.6%
Other	15.5%	9.3%

Table 4.13 presents the responses from laboratories regarding how they would choose to spend supplemental funding for their DNA programs. Essentially, the question asks laboratories where there biggest needs lie. The sum totals for all laboratory responses indicates that reagents and analytical equipment were the areas most frequently identified by laboratories for such funding. However, this table shows only how often a particular response was chosen, and no limit was placed on the number of response chosen.

**Table 4.13.** Projected use of supplemental funding for DNA analysis for local and state crime laboratories (frequency of selection).

Funding Priority	Local Laboratories (%)	State Laboratories (%)
Salaries	63.8	59.5
Overtime	46.6	48.8
Training	58.6	62.8
Equipment	60.3	67.4
Robotics	31.0	58.1
Reagents	67.2	<b>76.7</b>
Construction/Lease	29.3	48.8
Offender Collection	10.3	41.9
Suspect Case Backlog	51.7	41.9
No Suspect Case Backlog	55.2	51.2
Outsourcing	29.3	34.9
Other	6.9	27.9

Table 4.14 sets forth study results in this area in the form of a priority listing of spending priorities for local crime laboratories and state crime laboratories. Of the responses listed in Table 4.13, laboratories were asked to rank the top three most significant priorities, in order of importance. A review of these data clearly shows that salaries were, by far, considered the single

most significant need. Crime laboratories were also subsequently asked if by "salaries" did they mean additional funding to augment current analyst salaries or funding to pay for hiring new personnel? Although a handful of laboratories indicated that current staff needed augmented salaries, the majority of laboratories indicated that the need was for new hires. This issue of salaries is significant because federal grants for DNA analysis may not be used in this manner. Instead, DNA laboratories are solely dependent on state and local funding for salary needs. Two other personnel issues – training and overtime – also ranked among the top needs reported.

Additionally, the fact that local laboratories ranked funding for no suspect casework as a mid-level priority, but state laboratories ranked it last is a point for consideration. Current federal grants for DNA analysis fall into two categories. One grant can be used for DNA analysis of backlogged offender samples, and the other grant is for DNA analysis of backlogged cases with no suspect. As discussed previously, local laboratories privately complain that they do not receive a fair pass-through of the federal funding that is awarded to the state. This complaint would seem to be reinforced by the high priority that local laboratories place on the need for additional money for no suspect casework. This significantly higher need could also be reflective of the fact that many local laboratories serve large metropolitan jurisdictions with high crime rates, as discussed previously in this Report.

Finally, "construction/lease" was selected as a priority by a small number of laboratories, but for those laboratories that indicated this need, it was typically identified as the most significant priority. Of the twenty-five laboratories rating construction/lease as a priority, 14 rated it as the top priority, and eight rated it as the middle priority. Therefore, funding for construction/lease, though not a priority for the majority of laboratories, becomes a significant concern once the need develops.

**Table 4.14.** Rank ordering of spending priorities for local and state crime laboratories.

Local Crime Laboratories	State Crime Laboratories			
1. Salaries	1. Salaries			
2. Reagents	2. Equipment			
3. Equipment	3. Construction/Lease			
4. Training	4. Reagents			
5. Construction/Lease	5. Outsourcing			
6. Overtime	6. Training			
7. No Suspect Case Backlog	7. Robotics			
8. Suspect Case Backlog	8. Overtime			
9. Robotics	9. Offender Collection			
10. Other <sup>vii</sup>	Other <sup>viii</sup> *			
11. Outsourcing	10. Suspect Case Backlog			
12. Offender Collection	No Suspect Case Backlog*			

<sup>\*</sup> Ranking of "Offender Collection" and "Other" categories resulted in a statistical tie.

<sup>\*\*</sup> Ranking of "Suspect Case Backlog" and "No Suspect Case Backlog" categories resulted in a statistical tie.

#### V. CODIS ANALYSIS

In 1982, California became the first state to begin requiring blood samples from certain sex offenders. Although DNA was not specifically mentioned in statute, and indeed the basis for modern forensic DNA analysis was not first introduced until 1985, California's identification of the utility of collecting blood samples from a specified group of known offenders was ground-breaking at the time. In 1988, Colorado became the first state to specify through statute the requirement for DNA samples from certain offenders, and in 1990, Virginia became the first state to enact a law requiring DNA from all convicted felons. In the following year (1991), the FBI established guidelines for states on sex offender DNA database laws and began developing the concept of a computer software program that would operate similar to the national criminal fingerprint database (IAFIS). In 1994, this DNA database program was officially created in statute by Congress through the *DNA Identification Act (P.L. 103-322)* and was formally named the Combined DNA Index System, or CODIS.

Today, every State has a DNA database statute that allows collection of DNA from specified offenders. All 50 States require DNA from sex offenders and murderers, and 46 States require DNA from all violent felony convictions (including assault and battery and robbery). Over the past several years, a growing number of states have been expanding their databases to include non-violent felony convictions; 45 States require DNA from burglary convictions, 36 States require DNA from certain drug convictions, and 31 States require DNA from all felony convictions. (These figures are current through July 2003). (See appendix for chart of state statutes.)

An easily understandable description of CODIS was included in a July 2002 Special Report from the National Institute of Justice (U.S. Department of Justice) titled, "Using DNA to Solve Cold Cases" (NCJ 194197). The Special Report's description is as follows:

CODIS is a computer software program that operates local, State, and national databases of DNA profiles from convicted offenders, unsolved crime scene evidence, and missing persons. Every State in the Nation has a DNA database that allows for the collection of DNA profiles from offenders convicted of particular crimes. CODIS software enables State, local, and national law enforcement crime laboratories to compare DNA profiles electronically, thereby linking serial crimes to each other and identifying suspects by matching DNA profiles from crime scenes with profiles from convicted offenders. The success of CODIS is demonstrated by the thousands of matches that have linked serial cases to each other and cases that have been solved by matching crime scene evidence to known convicted offenders...

...CODIS uses two indexes to general investigative leads in crimes for which biological evidence is recovered from a crime scene. The convicted offender index contains DNA profiles of individuals convicted of certain crimes ranging from certain misdemeanors to sexual assault and murder. Each State has different "qualifying offenses" for which persons convicted of them must submit a biological sample for inclusion in the database. The forensic index contains DNA profiles obtained from crime scene evidence, such as semen, saliva, or blood. CODIS uses computer software to automatically search across these indexes for a potential match.

A match made between profiles in the forensic index can link crime scenes to each other, possibly identifying serial offenders. Based on these "forensic hits," police in multiple jurisdictions or States can coordinate their respective investigations and share leads they have developed independent of one another. Matches made between the forensic and convicted offender indexes can provide investigators with the identity of a suspect(s). It is important to note that if an "offender hit" is obtained, that information typically is used as probable cause to obtain a new DNA sample from that suspect so that the match can be confirmed by the crime laboratory before an arrest is made.

## LDIS, SDIS, and NDIS

CODIS is implemented as a distributed database with three hierarchical levels (or tiers)—local, State, and national. All three levels contain forensic and convicted offender indexes and a population file (used to generate statistics). The hierarchical design provides State and local laboratories with the flexibility to configure CODIS to meet their specific legislative and technical needs.

# A description of the three CODIS tiers follows

- Local. Typically, the Local DNA Index System (LDIS) installed at crime laboratories is operated by police departments or sheriffs' offices. DNA profiles originated at the local level can be transmitted to the State and national levels.
- State. Each State has a designated laboratory that operates the State DNA Index System (SDIS). SDIS allows local laboratories within that State to compare DNA profiles. SDIS also is the communication path between the local and national tiers. SDIS is typically operated by the agency responsible for implementing and monitoring compliance with the State's convicted offender statute.
- National. The National DNA Index System (NDIS) is the highest level of the CODIS hierarchy and enables qualified State laboratories that are actively participating in CODIS to compare DNA profiles. NDIS is maintained by the FBI under the authority of the DNA Identification Act of 1994.

There are three states (Louisiana, Texas, and Virginia) which, in addition to requiring DNA from all convicted felons, also require DNA from certain offenders prior to conviction (upon arrest or indictment). However, the current national DNA database law only permits those samples that are taken from <u>convicted</u> offenders to be uploaded to NDIS. The wording of this law also does not permit samples from juveniles to be uploaded to NDIS since most juveniles are adjudicated delinquent rather than convicted of crimes. However, individual states may maintain these samples at the SDIS tier, as state laws permit.

## Offender DNA Profiles

According to the FBI CODIS program office, there were 1,321,854 offender DNA profiles in NDIS as of April 2003. Additionally, 41 State CODIS laboratories reported 463,209 offender samples that were expected to be part of the laboratory backlog by June 30, 2003. As a comparison, the English DNA database contains over two million DNA profiles. Given that the United Kingdom's population is approximately 52 million, the two million offender samples on file represent a significantly larger percentage of the entire population than what is represented in CODIS (see the attached report, *The Application of Forensic DNA Technology in England and Wales*). The U.S. DNA database system would need to reach almost ten million profiles to achieve a similar percentage of the total population covered. The difference between these proportions go a long way towards explaining the greater success of England's forensic DNA program compared to that of the United States.

The mail assessment instrument asked those state laboratories with statutes that did not yet require DNA from all convicted felons to provide an estimate of the additional samples that the enactment of such a law would add in the first year, along with the estimated number of annual samples expected in future years. Those laboratories estimated that an expanded DNA database statute would add a combined total of 2,281,000 offender profiles in the first year, with an annual rate of 504,484 additional profiles in future years. Anticipated offender profiles for the first year are more than in future years because laboratories were asked to assume that the expanded requirement would be made retroactive to include offenders who were still incarcerated. A potential pool of backlogged offender samples that these figures do not capture are the uncollected offender sample in states where a requirement for an all felons DNA database has been enacted in statute, but not yet implemented (usually as a result of funding problems).

In the majority of states, offender DNA samples are taken at some point prior to release (72 percent) rather than at the point of sentencing or intake (52 percent). These numbers do not total 100 percent due to the fact that many laboratories responded that samples are collected at both points, presumably depending on the particular circumstances of each offender's conviction. For the purpose of solving crimes, it is most desirable to capture DNA samples from the specified offenders at the earliest point possible. Collecting DNA from offenders shortly after sentencing not only allows sufficient time for the sample to make its way through the backlog, but also, if a DNA database match is made to a previously unsolved crime, investigators and prosecutors will have ample opportunity to determine whether additional charges should be pursued while the offender is still in custody.

Another factor determining the number of offender profiles on the DNA database and related backlogs is whether the state requires DNA from specified offenders who are not sentenced to state prison terms. There are a large number of felony convictions in which the offender is either sentenced to a county jail or to community probation. In 2002, Washington State expanded its offender DNA database to include all felony convictions. The State estimated that limiting the database to only those felony offenders sentenced to state prison would mean an additional 5,600 DNA profiles annually. However, including those felony offenders sentenced to jail would mean an additional 15,000 offenders annually, and felons sentenced to serving community probation would add 840 offenders per year. In addition to the large number of felony convictions that these sentences represent, these are also offenders who are likely to have shorter sentences and therefore earlier releases back into the community.

Moreover, the size of the DNA database in the U.S. is further limited by whether the state elects to require DNA samples from persons who had been convicted of qualifying crimes prior to the time of the database expansion statute and are still serving sentences. Many states choose to omit these offenders, primarily due to the additional analysis and logistical problems associated with collections. Again using Washington State as an example, including those offenders who were incarcerated at the time of enactment would have added 6,100 state prison inmates and 1,800 county jail inmates. Making the statute retroactive to include those persons currently on probation and parole would mean adding an additional 10,300 DNA profiles. While this group of offenders included retroactively represent a significant cost increase, it is also worth pointing out that retroactive provisions represent a one-time cost, with no recurring collection issues. And as with the jail and community probation sentences, the groups of felons already serving time for qualifying offenses are the ones most likely to be released into the community soonest.

The type of biological sample required from offenders is another factor affecting both collection and analysis costs. Over half of the state laboratories (52.4 percent) indicated that a blood sample is collected from offenders, with 28.6 percent of states collecting buccal (cheek) swabs. Nearly one-in-five (19 percent) states are taking both types of samples, as conditions warrant. The move towards taking buccal swabs instead of blood draws represents a recent switch. Buccal swabs are less problematic, particularly for those local jurisdictions that are responsible for taking DNA samples from jailed offenders and offenders on probation and parole. Blood collections require a trained phlebotomist and are therefore considerably more expensive for local agencies. Buccal swabs, in contrast, can be accomplished easily with a minimal amount of training. Additionally, many people consider a buccal swab to be less invasive than blood draws. Fortunately, at the same time that the demand for buccal swabs was increasing, the kits for buccal swab collection were becoming increasingly reliable. evidenced by the crime laboratory assessment responses, many states now use buccal swabs exclusively for offender DNA testing and have independently indicated that they receive adequate samples for analysis. However, for crime laboratories making the switch from the analysis of offender blood samples to offender buccal swabs, it is not an overnight change and requires some planning and changes to laboratory operations.

## Characteristics of CODIS Hits

Over the past few years as DNA databases have become increasingly successful, the need for information on the types matches made (or "hits") has become apparent. Virginia's CODIS office has a strong tracking program for identifying the types of crimes solved and the types of offenders linked to such crimes. As other states have considered following the Virginia model of an all felons DNA database, the Virginia information has been a significant tool in driving the debate. However, Virginia's data have brought to light the lack of any comprehensive CODIS hit data on a national scale. As part of the state and local laboratory assessment form, respondents were asked to supply this information. The figures contained in this Report cover 38 states that reported DNA database hits, plus 70 local laboratory hits that were determined to be cases that had not been reported to the state. The local crime laboratory hits reported are matches made at the LDIS tier, but should not be viewed to represent the only hits made at the LDIS tier since many local jurisdictions regularly report such hits to their state CODIS office.

In reporting forensic hits and offender hits, crime laboratories were requested to follow the "CODIS Hit Counting Guidelines" as prepared by the FBI. However, it is possible that there are inconsistencies among states in the manner in which hits were reported. Another issue to bear in mind when reviewing Table 5.1 is that not every "hit" results in an "investigation aided." For example, a DNA profile from an unknown suspect rape case could hit upon the victim's boyfriend, assuming that the boyfriend has a prior offense which resulted in his inclusion on the database. If the victim is certain that her boyfriend was not the rapist, then the "hit" is not valuable to the investigation. Most often consensual partners are controlled for in DNA analysis cases, but such "hits" are not unheard of.

Table 5.1 displays the characteristics of the SDIS hits (hits made within a state) and NDIS hits (hits made between states). As indicated, most of the hits are from state crime laboratories. Furthermore, most (approximately three-in-four) are SDIS offender hits. With respect to the qualifying offense distribution, offenders were more likely to be entered into the CODIS database for a non-violent qualifying offense as a violent offense. With regard to types of crimes linked to offenders, however, violent crimes predominate; approximately six-in-ten of the crimes solved with DNA hits were violent crimes in the SDIS subset of cases. For the NDIS hits, violent crimes predominate on both the qualifying offense and crimes cleared sides of the equation. A majority of the inter-state hits entailed offenders who had violent qualifying offenses, and nearly three-in-four crimes linked either to offenders or to other cases through DNA evidence were violent crimes.

As is clear from the findings set forth in Table 5.1, the majority of types of crime solved through DNA database hits in the United States have been sexual assaults and other violent crimes. Considering that many crime laboratories will not process non-violent cases (such as property crime cases), or will only do so under a narrow set of circumstances, it is not surprising that the majority of unsolved crimes matched to offenders through the DNA database have been violent crimes. However, a comparison to hits on England's database, which has had an aggressive policy of encouraging submissions from property crimes, shows over 31,000 residential burglaries and over 29,000 commercial burglaries have been matched through hits on the database. In fact, only 14 percent of all DNA cases submitted to England's crime laboratory agency in 2001-2002 were violent crimes. With the FBI reporting that the estimated dollar loss attributed to property crime (excluding arson) in the U.S. was \$16.6 billion in 2001, it is reasonable to assume that the untapped potential for using CODIS to solve non-violent crimes is immense. (See attached report, *The Application of Forensic DNA Technology in England and Wales*, for further details.)

These figures indicate that in the United States only between 16.8 percent and 25.6 percent of the qualifying offenses for offenders linked to crimes are specified as violent offenses. It is quite possible, however, that a large number of cases comprising the "unknown qualifying offense" category—which contains the majority of responses—are violent offenses. Such a high proportion of violent offenses would be expected since most state DNA databases began as sex offender databases, and slowly expanded to include other violent crimes. For a majority of states, it has only been within the last three years that the database has grown to include non-violent felonies. Consequently, for the non-violent offenders on the database a majority of the hits recorded at this time are for crimes that were previously committed. It will likely take several years before these offenders matriculate through the penal system and new offenses are committed to which they can be linked. This delay in hit levels as the database grows was

experienced by Virginia. Although Virginia has had an all felons DNA database since 1990, it has only been since 2001 that the state has been reporting approximately one hit per day.

In addition to having all convicted felons on the database, another reason that Virginia's program has been so successful is due to its policy of processing a wide array of DNA cases. DNA databases will be limited in success if either the crime file index or offender file index is limited. The FBI reports a total of 54,895 forensic case profiles on the database as of April 2003. This number pales in comparison to the more than 200,000 similar profiles included on England's database – again contributing to the relatively higher level of success of their database compared to that of the United States.

**Table 5.1.** Descriptive statistics for CODIS hit characteristics.

Characteristics of CODIS Hits	SDIS Hits % In State Matches (n = 4,092)	NDIS Hits % Between State Matches (n = 289)	
Type of Crime Laboratory*			
Local	8.5	8.5	
State	91.5	91.5	
Type of Hit*			
Offender	72.3	57.7	
Forensic	27.7	42.3	
Offense Characteristics			
Non-Violent Qualifying Offense	23.9	15.9	
Violent Qualifying Offense	16.8	25.6	
Qualifying Offense Unknown	59.3	58.5	
Non-Violent Crime Solved	16.4	12 1	
Violent Crime Solved	39.2	46.8	
Type of Crime Solved Unknown	44.4	41.1	

<sup>\*</sup> Note: these estimates are based on the 2,706 SDIS cases for which states provided case-level information.

In order to portray these findings in a more concrete and detailed format, Tables 5.2 and 5.3 set forth in raw numbers the case combinations between qualifying offenses and the unsolved crimes linked for all SDIS and NDIS cases where such information was available.

Many of the responding laboratories were able to supply both the matched offender's qualifying offense (i.e., the reason the offender was included on the database), and the type of crime linked to the offender. However, a crime laboratory's ability to provide this information depends upon several factors. First, many crime laboratories do not have ready access to offender criminal history information, and therefore may not be able to identify the qualifying offense. Also, several laboratories have not instituted programs that allow them to track database hits. Although such tracking may seem like a simple thing, it requires additional resources from laboratories that are already facing extreme resource shortages. For this reason, there are a

considerable number of offenders in the "unknown" category of Table 5.2 and 5.3. Additionally, a few agencies were able to provide the total number of hits recorded, but could supply neither the offenders' qualifying offense nor the type of crime solved. By default, these hits are recorded in the "unknown" category as well.

**Table 5.2.** SDIS hit distribution by qualifying offense and type of crime solved.

Qualifying Offense	SDIS TIER OFFENDER HITS (Within State Matches)  Forensic Case Type								
	Murder	24	18	0	3	2	3	4	54
Sex Offenses	30	220	2	5	11	6	53	327	
Assault	18	61	6	3	9	11	2	110	
Robbery	27	75	4	25	53	12	2	198	
Burglary	18	99	4	31	168	16	79	415	
Property Offenses*	18	21	3	22	43	15	0	122	
Drug Offenses	43	94	2	16	47	26	0	228	
Supervision Violation	0	4	0	0	3	0	1	8	
Misc.	21	87	4	16	55	18	2	203	
Unknown	61	485	3	33	137	36	1672	2427	
Total	260	1164	28	154	528	143	1815	N=409	

<sup>\*</sup> Note: property offenses include theft, forgery, and breaking and entering. Miscellaneous offenses include: violation of oath as a public official; obstruction as a law officer (7); criminal attempt (3); entering vehicle; driving habitual violator (2); possession of firearm during felony; miscellaneous judicial proceeding; impersonating an officer; cruelty to children; criminal mischief; public lewdness.

The findings set forth in Table 5.2 indicate the presence of many repeat offenders – that is, criminals who tend to specialize in one type of crime and re-offend frequently. For example, the most frequent qualifying offense for burglars was burglary, and for rape and sex offenses was rape and sex offenses. It is also clear, moreover, that property offenses were qualifying offenses for many violent crimes – more than 100 among the cases displayed here. A somewhat different pattern is evident for drug offenses as qualifying crimes. Such offenses are matched to a wide

range of types of crimes solved through DNA evidence. These data begin to show that the non-violent felons, in fact, commit a wider array of crimes than the violent felons. In fact, Virginia estimates that limiting its DNA database to violent offenders would have resulted in missing 82 percent of its hits to offenders.

It also bears pointing out that the "unknown" category for qualifying offenses is quite large. Also, the qualifying offense as reported by states is merely a record of the crime that resulted in the offender's inclusion on the database. As illustrated in the New York data discussed later in this Report, many of these offenders have extensive criminal histories – and many carry convictions for other "lower-level" felonies that were not crimes included on the database at the time of the conviction. Therefore, the relatively larger number of violent offenders matched to crimes on the database is possibly inflated both because most databases began as violent offender only databases, and also because the qualifying database offense is not necessarily the only conviction attributable to the offender.

The likelihood of this scenario is illustrated in the New York study that found that for more than half of the offenders studied, the qualifying offense that resulted in the inclusion of the offenders profile on the database was committed an average of four and a half years after the first criminal conviction<sup>xi</sup>. Moreover, these data are supported in some of the hit data submitted by states. Several states were able to include criminal histories for their offender hits. These criminal histories were, more frequently than not, extremely lengthy with multiple arrests and convictions. A hit in one state linking an unsolved sexual assault to a felon whose qualifying offense was robbery, shows a criminal history that covers two states as well as some federal felonies. The record includes charges on: two rapes, a kidnapping, four robberies, three assaults, two burglaries, three thefts, two counterfeiting crimes, driving while intoxicated, resisting an officer, trespassing, defrauding an innkeeper, and three parole/probation violations. These twenty-one charges (excluding the parole/probation violations) resulted in nine convictions – only the last of which required a DNA sample for the database.

Table 5.2 also reports data which may be of interest for further criminological reasearch. Of the 228 drug offenders linked through the database to another unsolved crime, almost 70 percent of these unsolved crimes were violent offenses. Overall, drug offenders accounted for more than 21 percent of the hits made to unsolved murders and 14 percent of hits made to unsolved sexual assaults. When looking more closely at the sexual assault crimes solved through the DNA database, approximately one-third of these crimes were linked to sex offenders. More surprisingly, nearly 18 percent of the unsolved sex assaults were linked to burglary or property crime offenders; this figure combined with the 14 percent attributed to drug offenders may reveal patterns of criminal activity for study and use in police investigation practices. (For the purposes of these calculations, the "unknown" number of forensic cases was subtracted from the denominator.)

The findings set forth in Table 5.3 address the same question of type of qualifying offense matched to crimes for NDIS hits. This table includes NDIS tier offender hit information for 31 states. Significant efforts were made to ensure that hit counting was not duplicated, but it is possible that some of the hits were counted twice, particularly for those states that were unable to give a breakdown of qualifying offense of type of forensic case matched. A significant number of database hits were made between neighboring states, with Illinois reporting fourteen hits with Wisconsin, and six hits with Missouri. However, hits were also reported between states with little geographical proximity – such as Alaska and Arkansas, Washington and Florida, and

Maine and Texas. In addition, there were five hits reported to the District of Columbia, one hit involving the military, and one international hit with Canada. (See appendix).

Moreover, at least 73 forensic case matches have been made between 27 states, the District of Columbia and the FBI. These cases included 56 sex offenses and 21 homicides and 15 burglaries. The total number of forensic hits is substantially lower than the number of crimes involved because one forensic hit will necessarily link at least two crimes together. Therefore, although only 73 forensic hits were made, 166 unsolved crimes were involved.

**Table 5.3.** NDIS hit distribution by qualifying offense and type of crime solved

	NDIS TIER OFFENDER HITS (Between State Matches)								
Qualifying Offense	Forensic Case Type								
	Murder	Sex Offenses	Assault/ Battery	Robbery	Burglary/ B&E	Misc.*	Unknown	Total	
Murder	0	4	0	0	1	0	0	5	
Sex Offenses	8	27	0	0	3	0	8	46	
Kidnapping / Restraint	1	3	0	0	0	0	0	4	
Assault	2	8	1	0	0	0	1	12	
Robbery	1	1	0	1	3	0	1	7	
Burglary	4	2	0	1	8	2	1	18	
Theft & Receiving	4	4	1	0	1	0	0	10	
Misc.	1	10	0	0	4	1	2	18	
Unknown	7	38	3	2	11	1	107	169	
Total	28	97	5	4	31	4	120	289	

<sup>\*</sup>Miscellaneous forensic crimes include vehicle theft, criminal trespass, DUI/vehicle accident. Miscellaneous Qualifying Offenses include theft, drug charges, forgery, tampering with evidence, weapons violations, passing bad checks, and juvenile adjudications.

It is clear from the findings set forth in Table 5.3 that most cases entailing complete data for qualifying offenses and crimes solved among the NDIS hits relate to rapes and sex offenses. The vast majority of rape and sex offense crimes cleared entailed rape and sex crime qualifying offenses. However, again there is a large number of "unknown" offenders and forensic cases, which could potentially cause the distribution. This finding, coupled with the same strong

pattern of repeat offending witnessed in Table 5.2, highlights the importance of collecting and processing rape kits and other biological evidence that might render DNA evidence for DNA databases. The following section will add further to our understanding how DNA evidence, if collected and processed with proper dispatch, could in fact prevent re-offending and spare many citizens from avoidable crime victimization.

#### VI. FORENSIC DNA AND CRIME PREVENTION

Solving a crime -- and solving it quickly -- has a direct effect on preventing additional crimes by the same perpetrator. An offender who is not apprehended in a timely manner remains free to commit more crimes. The most recent federal studies on recidivism tracked 272,111 released prisoners from 15 states<sup>xii</sup>. The study found that 29.9 percent of the released prisoners were rearrested for a new offense within six months of release; and after three years this percentage grew to 67.5 percent. The arrests in question were almost exclusively either for new felonies or serious misdemeanors. The same study found that the offenders accounted for 4.1 million arrest charges prior to their most recent incarceration, and were responsible for another 744,000 arrest charges within three years of release.

It is precisely such repeat offenders that DNA databases are established to identify to enable timely intervention and the prevention of new crimes by habitual offenders. A match on the DNA database necessarily means that the same offender has committed at least one other criminal offense. Therefore, DNA databases serve the purpose not only of bringing criminals to justice and achieving closure for the victims of crime, but they also offer the prospect of preventing the same perpetrator from victimizing additional citizens. Given the relatively high rates of recidivism in the U.S., the potential for DNA databases to help solve and prevent crime is substantial. DNA analysis cannot stop the first crime from occurring, but it can provide investigators with the tools they need to identity a suspect and remove the threat to public safety before the same perpetrator can re-offend.

However, the effectiveness of DNA databases is restricted by a number of factors. First, to have an effective DNA database, biological evidence must be routinely collected from crime scenes, and that evidence must be sent to crime laboratories for forensic analysis. This set of activities on the part of law enforcement personnel assumes that crime scene investigators have the proper training to identify and collect the DNA specimens, and that the samples are submitted to the crime laboratory in a timely manner.

Secondly, crime laboratories must have a reasonable processing time for DNA analysis of crime scene samples, along with a policy to accept and process most reasonable requests for DNA testing – including crimes with no suspects and property crimes. In England, the current processing time of 33 days has been deemed unacceptable by the Government, which has set a processing time goal of 24 days. As discussed previously in this study, the average turn-around time in the U.S. is approximately 30 weeks, or for purposes of comparison to England's numbers, 210 days.

Additionally, the crime laboratory must accept requests for DNA testing for a variety of crimes. As discussed previously, states with progressive DNA polices have had great success in solving an increasing number of crimes through their DNA database. Limiting the numbers and types of acceptable cases for DNA testing will necessarily limit the number and types of crimes that can be solved through the use of forensic DNA.

Lastly, the DNA database must have a strong pool of offenders for comparison. As described in Section V, the DNA database is a two-index system – a crime scene sample index, and an offender index. The effectiveness of either index is necessarily restricted by any limitations on the other index. As previously discussed, those states with statutes requiring DNA from all convicted felons have shown a high success rate in matches on their DNA database. A 2002 Research Note by the New York Division of Criminal Justice Services further supports this argument<sup>xiii</sup>. The study reviewed the first 102 DNA database matches occurring after a 1999

expansion of the database to include considerably more convicted felons. For 55 of the 102 offenders on the database who were matched to unsolved crimes, the review found that the first conviction was not for an offense that would be included on the DNA database under the current statute. Instead, the conviction that resulted in the offender's inclusion on the database occurred about 4½ years after the first criminal conviction – during which time these particular offenders accumulated an average of 10.2 felony and 5.6 misdemeanor arrests. These types of numbers on repeated offending by known felony offenders are supported by the recidivism study discussed previously. These observations on repeat offenders highlight a clear public safety need – that of identifying these repeat offenders as quickly as possible and monitoring their activities carefully.

Quantifying the number of crimes that could be prevented through a more systematic use of forensic DNA testing presents a difficult problem. As in all matters of documenting the "prevention" of crime, it is difficult to count crimes that did not happen. As a way to deal with this fundamental problem, the following case studies were developed to provide examples of crimes that potentially could have been prevented had a strong forensic DNA program been in existence to call upon. The research on these cases specifically focused on those states that did not require DNA from all convicted felons at the time that the research for this Report began. Such states were chosen for the greater likelihood of uncovering preventable crimes due to a limited offender database. Moreover, many of these states have not expanded their offender databases largely due to resource problems which also affect criminal evidence analysis.

The case studies were developed through the assistance of investigators and prosecutors around the country, and reflect the following assumptions:

- o State statutes requiring DNA from all convicted felons were feasible by 1990. This assumption is based on the fact that Virginia enacted such a law in 1990.
- o In addition to all felony offenses, state databases could include misdemeanor convictions that arose out of felony charges. Several states currently have such requirements due to the number of felony offenders who plea to misdemeanor crimes.
- o A 30-day turn-around time is both desirable and possible, given sufficient resources. In fact, laboratories will often expedite testing and can finish analysis in a week or less if the crime(s) warrant priority treatment.

When reviewing the cases presented below, the reader should bear in mind the following points:

- o Backlogs at crime laboratories are primarily the result of growing demand and limited resources. "Preventable crimes" that may in part be due to backlogs at crime laboratories, should be considered as a byproduct of resource limitations. In fact, crime laboratory analysts often work overtime and weekends to complete analysis on high profile cases.
- o Law enforcement and prosecutors should not be expected to be able to identify offenders of unsolved crimes when they have not been given access to the necessary tools, such as DNA. These case studies should not be read to point blame at criminal justice professionals who did not immediately detect the offender's involvement in additional crimes. Indeed, many of these cases would still be unsolved if not for other extraordinary detective work.

o States with multiple case studies should not be considered to have a particular deficit in DNA testing. For several of the case studies, it was the fact that DNA testing <u>is</u> being conducted that allows law enforcement to determine that serial offenders were at work.

#### **ALASKA**

#### FOUR PREVENTABLE RAPES

Between January of 2000 and January of 2001, an unidentified perpetrator raped five women in the area of Anchorage. All five rapes were all linked to the same unknown offender through DNA analysis.

Police identified a suspect after the January 2001 rape, and that individual has since been convicted of all five attacks.

The suspect in question had a long criminal history of felony charges which resulted in misdemeanor convictions. Felony charges levied against him included burglary with sexual intent, and his misdemeanor convictions ranged from criminal trespass to forgery. If this individual had been required to supply a DNA sample from a 1995 felony burglary charge that was reduced to a criminal trespass conviction, he could have been identified after the first rape in January 2000, *thereby preventing the following four rapes*.

#### ONE PREVENTABLE RAPE

In 2000 in Anchorage, a woman was raped by a stranger and a DNA sample from the crime was entered into the state DNA database. No DNA hit was made, and the crime remained unsolved.

In 2001, another woman was raped in the remote village of Kotzebue on the Seward Peninsula – an area primarily accessible only by airplane. In this case, authorities identified the suspected perpetrator and he was convicted of sexual assault. Upon his conviction, a DNA sample was collected for the DNA database, and was matched in June of 2002 to the 2000 Anchorage rape.

A review of the offender's criminal history shows at least nine prior convictions for a variety of crimes ranging from criminal trespass to felony burglary. Several of the convictions also carried sexual assault charges which were dismissed. In 2000, Alaska expanded the DNA database to include burglars, but did not make the provision retroactive to include prior convictions for those still serving probation or on parole. If the state had required DNA from this group of offenders, the DNA database would have matched the offender to the 2000 rape, *thereby preventing the rape in 2001*.

#### ONE PREVENTABLE PROPERTY CRIME

In 1993, a murder was committed in Kodiak, but with no suspect the case remained unsolved. In recent years, the case was reopened and a cigarette butt collected from the crime scene was tested for DNA. A DNA profile was extracted, and it matched up with a felon on the database for a 1996 felony armed robbery.

However, before this murder identification was made, the offender remained criminally active in the community and was arrested and convicted of another theft in 2002.

With proactive policies to review cold cases for possible DNA evidence, the offender's DNA sample could have been matched to the 1993 murder prior to the 2002 theft – *thereby preventing at least one property crime*.

#### **ARKANSAS**

#### FOUR PREVENTABLE RAPES

Between 1995 and 1997, seven rapes were committed against women and young girls across four counties in Arkansas. In at least four of the cases, the victims were assaulted after being stopped by a man posing as a police officer, whom newspapers subsequently dubbed the "Blue Light Rapist."

In 1997, a suspect was identified as the suspected rapist through the assistance of one of his acquaintances who was working with law enforcement. Subsequent DNA testing linked him to several of the rapes, with other evidence and victim descriptions tying in the remaining cases.

The suspect had been convicted in 1996 on charges of theft by receiving a stolen rifle. Had the state required a DNA sample for this felony conviction, *up to five of the subsequent rapes could have been prevented*.

#### **CALIFORNIA**

#### SIX PREVENTABLE RAPES

In the Alameda County area, a series of 12 sexual assaults occurred in 1997 over a two-month period, all of which were attributed to the same unidentified suspect. The offender wore a ski mask in all the attacks, often robbed his victims, and in many instances attacked two women on the same day. This rapist struck quickly and frequently – his first five detected attacks occurred over a period of 10 days. Due to the extremely violent nature of the offenses and other similarities linking the attacks, it is reasonable to expect that DNA analysis would be expedited in these cases, as is often done when a serial offender is at large in a community.

A latent fingerprint found at the scene of the 12<sup>th</sup> attack was linked to a former high school track and field coach who had an extensive criminal record. This individual was formally charged and convicted in 6 of the cases, and is currently serving a 59-year to life imprisonment sentence.

The perpetrator's criminal record included 21 prior arrests, most of which were felony charges ranging from murder to interfering with a police officer. Although most of the offenses occurred prior to the introduction of DNA databases, the man's last brush with justice was a 1993 parole violation for a 1989 felony burglary conviction; the offender was discovered to be in possession of weapons and spent another 16 months in prison.

DNA evidence from the first rape in 1997 could have been matched to the offender by the time of the sixth attack (occurring 11 days after the first attack) if the state had required DNA from all convicted felons – *thereby preventing six rapes*.

#### THREE PREVENTABLE RAPES

In 1998, a woman was raped in Ventura County. A DNA profile from the criminal evidence was uploaded to the DNA database, but no match was made. Two additional rapes that occurred four months apart in 2002 were subsequently matched to the 1998 case, and new testing of a 1993 case was also eventually linked to the same unknown offender.

The offender became a suspect in these crimes after being identified in another case. He provided a DNA sample that linked him to the four rapes. He is currently awaiting trial.

The suspected rapist had a prior conviction of felony drug possession in 1990. If the state statute had required DNA from all convicted felons, including felony drug possession, and if the 1993 rape case had been submitted for DNA testing in a timely manner, *at least three rapes could have been prevented*.

## ONE PREVENTABLE ASSAULT

On an April evening 1997, an intruder entered a residence in the Berkeley area and sexually assaulted a woman in her bed while threatening her with a knife. Before burglarizing the house and fleeing the scene, the intruder bound the victim with duct tape and stockings and viciously beat her.

In a second incident in 1997, an intruder entered a residence in Berkeley and began to burglarize the house. A male resident awoke and a struggle ensued during which the resident was stabbed in the face and back. In a sweep of the neighborhood, police stopped a suspect who was eventually determined to be responsible for both attacks.

The offender had a prior record that included charges on 13 felonies. However, only six of these charges resulted in convictions and all six convictions were reduced to misdemeanors. Had California required DNA from criminals whose misdemeanor conviction arises from the same set of circumstances as a felony charge, *at least one potentially deadly assault could have been prevented*.

#### **HAWAII**

# TWO PREVENTABLE SEXUAL ASSAULTS, TWO PREVENTABLE ASSAULTS

Over a period of two months in 1994, two women were sexually assaulted at knifepoint in parking garages in Honolulu. These two crimes occurred two weeks apart, so it is reasonable to assume that the case could have been prioritized and expedited DNA testing could have been conducted. More than a month later, a third attack was attempted, but the offender broke off the attack when a man nearby heard the victim's screams. The offender then chased after the man, but both the male and female were able to evade further contact with the offender.

Relying on latent fingerprints and victim identifications, police eventually arrested a suspect. During the course of the investigation that suspect was also linked to two unsolved home invasion sexual assaults. He pled guilty to 21 counts pursuant to a plea agreement, and ultimately was sentenced to 40 years in prison.

The criminal in question was a registered sex offender in Hawaii, but his sentence terms were completed prior to the establishment of a DNA database in Hawaii in 1992. As it turns out, the 1994 attacks coincided with dates when he was required to check in with his parole officer in the downtown area. Had Hawaii extended its DNA database to include all registered sex offenders, he could have been identified prior to the third attack.

Moreover, this individual's criminal record contains a kidnapping conviction for which he was paroled in 1994. If the state had required a DNA sample from persons incarcerated for kidnapping at the time of the DNA database establishment, *at least two sexual assaults and two related attempted attacks could have been prevented.* Hawaii is the only state that still does not require DNA from kidnapping convictions.

### **IDAHO**

### ONE PREVENTABLE DEATH

In September of 2002, a woman was kidnapped, raped and murdered. DNA collected from the crime scene was entered into CODIS, but no offender was identified. In March of 2003 another woman was raped and murdered, and the DNA database linked that crime to the 2002 unsolved murder case. The man who was a suspect in the 2003 murder case was arrested and confessed to the crime. Further DNA testing reconfirmed that the man had in fact committed both murders.

The murder suspect had been convicted on felony burglary charges in 1991. If the state had required DNA for this conviction, *one death could have been prevented*.

## **INDIANA**

## ONE PREVENTABLE DEATH, SEVERAL PREVENTABLE PROPERTY CRIMES

In May of 2000, an elderly woman was brutally raped while on her way to vote in an election. Ten days later a suspect, who had not been identified as the woman's rapist, was charged and subsequently convicted of felonious receiving of stolen property and trespass. Upon his release, he was remanded to another jurisdiction where he was wanted on felony theft charges. The suspect was released from this county correctional facility in May of 2001, and the next day an elderly woman was found raped and murdered. Tracking dogs lead police to an area where the suspect's mother lived, and within a few days the suspect was arrested for the murder. Subsequent DNA testing linked both the 2001 murder and the 2000 rape to the suspect in question.

In 1998, the perpetrator of these crimes was convicted on felony charges of receiving stolen property, and he was again convicted on a similar charge while still under confinement in prison. From 1999 through 2000, he was charged or arrested on four separate occasions for felony theft by receiving, one of which resulted in a misdemeanor conviction and the last one brought the felony conviction in May of 2001. If the offender in question had been required to give a DNA sample upon the 1998 felony conviction, the 2000 rape could have been solved, and the rape and murder of an elderly woman, and several property crimes could have been prevented.

### TWO PREVENTABLE SEXUAL ASSAULTS

In October of 2001, a woman was raped by an unknown assailant who broke into her home; a rape kit was collected for analysis in this case. In July of 2002, another woman was raped and burglarized, and another rape kit was collected for forensic analysis. Then once more in August of 2002 a third woman was burglarized and sexually battered. All three violent assaults took place in the Indianapolis area.

Use of the Indiana state DNA database eventually led to the matching of the rape that occurred in October of 2001 to the rape of July 2002. In addition, a match was also made to a DNA sample obtained from a suspect. Upon questioning by detectives, the suspect in question incriminated himself in a third sexual assault which had taken place in August of 2002.

The perpetrator in this set of cases received a 20-year sentence for his second rape conviction in December 1986, and was released from prison on parole in October 1995. However, the Indiana DNA database did not become effective until July 1996, and upon going into effect it did not require DNA samples from qualifying offenders who were still serving community sentences of probation or parole. Had the Indiana statute on DNA evidence collection included a provision to collect DNA from persons still serving community sentences, *at least two sexual assaults could have been prevented*.

#### **KENTUCKY**

#### SEVEN PREVENTABLE RAPES

In October 1998 a female attendant at a gas station was raped. In this case, DNA was collected but was not tested. Approximately ten days later, another female gas station attendant was raped, and over the next three years an additional seven rapes were committed by the same repeat offender.

Investigators identified a suspect in the last rape in August of 2002 after he used the victim's ATM card. In 2003, DNA testing finally linked the man to the first two rapes in 1998, and other evidence linked the remaining cases to one another.

The suspect was a known sex offender in Kentucky who had committed twelve documented rapes as a juvenile. His first rape committed as an adult occurred within a month of his release from juvenile detention facilities. Unfortunately, the State of Kentucky did not require DNA from juvenile offenders adjudicated delinquent of sex crimes at the time of the suspect's juvenile sentences. If the state had required DNA from juveniles adjudicated delinquent of felonies, *at least seven rapes could have been prevented.* Moreover, in 2002, the requirement was not made retroactive to encompass persons on probation or parole. Had it not been for good police work, this rapist could still be a threat to public safety today. when amending the state's DNA evidence law governing the DNA database to include juveniles

#### FIVE PREVENTABLE RAPES

From June of 1997 through October of 2000 six rapes and several attempted rapes took place in the Louisville area. In addition to DNA evidence that linked these rapes to one another, the rapist frequently used a flashlight to blind his victims -- thus earning the nickname of the "Flashlight Rapist."

In November of 2000 a suspect was arrested and subsequently convicted on numerous burglary charges in Louisville. A sex crime detective began to suspect the man as the Flashlight Rapist and obtained a warrant for a DNA sample. The DNA analysis positively identified the man as the Flashlight Rapist, and he later pleaded guilty to the crimes.

The flashlight rapist had been previously convicted on felonious burglary charges in 1991 in Florida, which did not yet require DNA from all convicted felons. Had DNA been required, at least 5 rapes could have been prevented. Moreover, although Kentucky altered its law to require DNA for burglary convictions in 2002, the provision was not made retroactive to include probationers or parolees and the implementation date of the law was delayed. Thus the 2000 burglary conviction in Kentucky may also have "slipped through the cracks" and allowed the crimes to remain unsolved if not for good detective work.

#### LOUISIANA

#### FOUR PREVENTABLE DEATHS

Between September of 2001 and March of 2003 five women were raped and murdered in the Baton Rouge area. All five crimes were linked to the same unidentified perpetrator through DNA testing. In May of 2003, a volunteered DNA sample collected from a mass sampling of potential suspects resulted in the identification of a suspect. Subsequent to this match, at least one additional murder has been tied through DNA to the suspect, and over 20 additional cases are being reviewed for possible connections to the suspect in question.

The suspect had a 1993 conviction on felony burglary charges arising from an attack on a 74 year-old man. If his DNA sample had been collected and entered into the database, *at least four deaths could have been prevented*. The suspect also had prior convictions for beating up his girlfriend in a bar and trying to run over a Sheriff's deputy at a roadblock.

#### ONE PREVENTABLE RAPE AND BURGLARY

In 1997 and unknown suspect committed attempted aggravated rape on a child in her home. A DNA sample was collected as evidence, but was not matched to an offender.

In December 1998, a woman was raped and her home burglarized by an unknown assailant. A suspect who was in close proximity to the crime scene and matching the victim's description was arrested. Subsequent DNA testing linked him to both the 1997 and 1998 crimes. The same perpetrator is suspected in the commission of two other similar crimes for which no DNA evidence was available. He has been sentenced to 50 years of imprisonment.

The offender's criminal history included a 1990 conviction on unauthorized entry of a place of business, and a 1996 felony theft conviction. If a DNA sample had been required for

either of these crimes, the database could have matched the offender to the 1997, *thereby* preventing the 1998 rape and burglary.

#### **MASSACHUSETTS**

### THREE PREVENTABLE RAPE/MURDERS

In late 1997, a woman was found raped and murdered in her home in Springfield. In early 1998, another three women were found raped and murdered – one in her home, two others in alley ways. All four murders were linked through DNA, and due to the unique positioning of the bodies which became the offender's "signature."

Following the fourth murder, a voluntary DNA sample was collected from a person who had become a suspect in the case. Within a few weeks, that suspect was tied through a DNA match to the crimes.

The suspect's criminal record included two prior felony convictions in 1996 – one for larceny and the other for breaking and entering, for which he was sentenced to community supervision. If Massachusetts had required a DNA sample for either of the 1996 non-violent felony convictions, a DNA match could have been obtained after the first rape/murder, thereby preventing the subsequent three rape/murders.

### **MISSOURI**

#### TEN PREVENTABLE RAPES

Between 1988 and 1997, an unidentified masked man was beating up and raping women in areas of Missouri and across the Mississippi River in Illinois. Because many of the 29 or more attacks happened in south St. Louis, the media dubbed the attacker the "South Side Rapist." DNA linked at least 13 of the cases together, but the police were unable to identify a perpetrator.

In October of 1998 St. Louis City Police were called when a man was seen breaking into a house. A van registered to a person known to the police was reported leaving the scene of the break-in. This person's appearance matched a physical description of the South Side Rapist and police asked for a DNA sample. The suspect voluntarily agreed to have his mouth swabbed. Several weeks later the DNA results were returned and positively identified the suspect as the South Side Rapist, who by this time had disappeared. He was arrested several months later in Albuquerque, New Mexico after a worried mother found his hotel number in her daughter's belongings. The suspect ultimately confessed to raping at least 100 women since his late teens.

Sometime in 1991 the suspect was convicted of felony burglary for which he served a sentence in the state prison. Prosecutors indicate that there were a minimum of ten rapes occurring after this conviction that are attributable to the suspect. If the state had required a

DNA sample from this perpetrator after the burglary conviction, the suspect could have been linked to serial rapes occurring prior to his conviction, *thereby preventing at least 10* additional rapes.

#### ELEVEN PREVENTABLE RAPES AND MURDERS

From April of 2001 through May of 2002, police began finding the bodies of women who had been raped and murdered in the St. Louis/western Illinois area. Investigators recovered semen from the victim's bodies that was sent for crime laboratory analysis. The DNA profiles from the semen recovered from the victims all matched.

A person known to the police became a suspect in the crimes after he wrote an anonymous letter to a local newspaper. An internet-generated map enclosed with the letter led to some computer forensics that eventually identified the specific computer that downloaded the map. The suspect was ultimately arrested in June of 2002 when police found videotapes of himself killing and torturing his victims. On the videotape, the suspect states that he had just committed "murder number seventeen". Police have conclusively linked 12 victims to this offender thus far, and believe the number could be as high as 20. This individual committed suicide shortly after his arrest.

In March of 1988, this offender committed a series of five robberies and was sentenced to fifteen years for robbery and armed criminal action. In June 1994, he was paroled after serving five years and three months of his sentence. Unfortunately, Missouri does not require DNA from convicted robbers, and this offender was released without submitting a DNA sample for the state database. With a DNA sample in the database, this offender could have been identified as the killer of the first victim *long before a minimum of eleven additional women lost their lives*.

#### **NEBRASKA**

### TWO PREVENTABLE SEXUAL ASSAULTS

In 1995, an unknown perpetrator raped a woman in Ohio. DNA was collected during the investigation, but was not submitted for testing.

In 1999, a woman was sexually assaulted by an unknown assailant at a hotel in Nebraska. DNA evidence was collected from this crime scene. Later the same day in Colorado, a woman was sexually assaulted by a man who had answered an advertisement in the newspaper for furniture that she was selling. DNA was collected from this crime scene as well.

Using an electronic bulletin board that serves a nine-state area in the Midwest, a Nebraska police detective posted a digitized photo of the suspect in the Nebraska hotel attack. Within 30 days police in Iowa identified the person in the photo as a known offender resident in their area. The man in question was arrested and subsequent DNA testing linked him to both 1999 attacks.

Subsequently, an Ohio police detective noticed similarities to the 1995 attack, and the DNA from that crime was submitted for testing and was found to match that of the offender.

The offender's extensive criminal history dates back to a 1979 felony burglary conviction, and included two separate burglary convictions in Iowa in 1989 (for which he was sentenced to a maximum of 25 years) plus a 1993 misdemeanor indecent exposure conviction and several acquittals on sexual assault. If Iowa had required DNA prior to releasing the offender for the 1989 felony burglary charge, and if the DNA evidence in the Ohio case had been submitted and analyzed in a more timely manner, *at least two sexual assaults in Nebraska could have been prevented*.

This case is an excellent example of the importance of each state's DNA program to solving and preventing crimes in other states. It is also worth noting that Iowa expanded its offender DNA database to include all convicted felons in 2002, but Nebraska still does not require DNA from felony burglary charges.

### THREE PREVENTABLE RAPES

In July of 2000, a woman was raped in her home after answering a knock at her door, but no DNA was collected. Between September of 2001 and September of 2002, four more rapes were committed by a perpetrator using the same modus operandi – but in these cases DNA was collected and linked to the same unidentified perpetrator.

Acting on a tip, police tracked down a suspect in November of 2002 and the man was subsequently convicted on all four rape cases for which DNA was collected. He is eligible for parole in 140 years.

The rapist in question had several convictions for misdemeanor thefts from 1996 through 1998. Moreover, he was convicted as a juvenile on felony theft charges in 1996. If a DNA sample had been required for the felony theft adjudication, *at least three rapes could have been prevented*.

**NEW JERSEY** 

#### THREE PREVENTABLE RAPES

In mid-May 1998, a woman was raped at a public library in Essex County. One month later, a second woman was raped in a library. DNA samples were taken from both cases and loaded into the DNA database, where it was discovered that the two rapes were linked. Importantly, in matching these cases to each other, DNA also excluded two men being held by police as possible suspects.

In late June of 1998, a child was abducted and raped. A DNA sample was collected and sent to a private laboratory for testing, and the victim provided a description of the attacker to police.

The police eventually identified a suspect, and DNA testing subsequently tied him to the child attack. After the sample was loaded into the state database, it was discovered that the same offender was the library rapist. Additionally, the same DNA profile was eventually matched to an October 1999 rape in New York City.

The offender had a prior felony arrest on a weapons possession charge. The charge was later reduced to a misdemeanor crime to which the offender pled guilty. If a DNA sample had been required of felony charges that result in misdemeanor convictions, the perpetrator could have been identified after the first rape, *thereby preventing the following three rapes*.

## SEVEN PREVENTABLE ROBBERIES, FIVE PREVENTABLE RAPES

In October of 2002, the first in a series of ten robberies began. The incidents included seven aggravated sexual assaults, spanned four months and included victims in at least nine different New Jersey cities.

A person known to the police became a strong suspect in the first offense, and a DNA sample was collected from him four days later. A DNA match was finally made in January of 2003 to the third offense, but only after a unit commander requested expedited testing of the evidence.

The offender in question was convicted in the 1980's on federal felony robberies charges, and was released from a federal prison in 1999. Unfortunately, the federal government did not begin requiring DNA from felony robbery convictions until 2000. Moreover, law enforcement had custody of all the DNA information that they needed to arrest this offender within only a few weeks of the first offense. The omission of his DNA from the federal DNA database, along with the backlog delay in processing DNA evidence, allowed the offender to remain on the streets. With stronger federal statutes and shorter DNA testing delays, the perpetrator could have been identified after the third attack, *thereby preventing the subsequent seven robberies and five rapes*.

#### FOUR PREVENTABLE RAPES

Between April of 2002 and May of 2003, five women were raped in the Trenton area. DNA testing linked all five offenses to the same unknown perpetrator. After police released a composite sketch of the suspect in 2003, nearly 75 tips were called in identifying the same person. In June 2003, U.S. Marshals eventually arrested the suspect in Pennsylvania on a parole violation warrant that was issued in July of 2002. Trenton Police obtained a DNA sample from the suspect through a court order, and thanks to expedited testing at the state laboratory the man was linked to the crimes within a few days. The charges on 16 counts involving five victims are pending as the suspect awaits extradition to New Jersey from Pennsylvania.

The suspect's criminal record included two felony convictions for theft and forgery related offenses in New Jersey, and nine felony convictions for theft, forgery, and receiving stolen

property in Pennsylvania. If the suspect had been required to give a DNA sample for any of these crimes *in either state*, he could have been identified after the first assault, *thereby* preventing the subsequent four rapes.

#### **NEW YORK**

#### THIRTEEN PREVENTABLE RAPES

In August of 1993 a young woman was raped in the Bronx in what was to be the first of up to 51 rapes attributed to the same offender over a five-year period. The perpetrator was dubbed the "Bronx Rapist" by the media.

A person known to the police became a suspect when he was identified in a transaction involving a victim's jewelry at a pawnshop. He was arrested and subsequent DNA testing linked him to several of the rapes. He has been convicted on fourteen counts of rape in the Bronx, six counts of sexual abuse, nineteen counts of robbery, and two counts of criminal possession of a weapon. He has been sentenced to two life sentences.

This offender had a prior conviction in 1989 for felony robbery and assault, for which he received a seven-year sentence. If the State of New York had begun requiring DNA from all convicted felons in 1990 this offender would have been on the DNA database prior to the first rape in 1993, and *at least thirteen rapes could have been prevented.* Moreover, when New York's database was established in 1994, an inclusion of all convicted felons and retroactive application to persons previously convicted but still under supervision would have captured this offender's DNA sample much earlier in the investigation.

### TWO PREVENTABLE SEXUAL ASSAULTS, ONE PREVENTABLE DEATH

From 1991 to 1999, three young women were murdered in New York City and four others were raped. The youngest victim was 13, and several of the crimes were noted for their brutality. During the course of the investigation, police identified a man who had just been released from jail for a sex crime in the same area in which a victim had been raped. He had been seen in the neighborhood just before and after the rape, and was picked out of a lineup. The man was jailed for four months, but DNA testing subsequently eliminated the man as a suspect.

Another person known to the police became a suspect in these crimes in 1999 and was placed under surveillance by police. He was eventually arrested on petty theft charges and DNA testing later linked him to evidence from the crimes. This person had been released from custody pending the DNA testing, and was arrested again in Miami after the DNA match was made. He was found with a young woman who may have been his next victim. This offender was found guilty on twenty-two counts, and sentenced to 400 years in prison.

This offender had been convicted of felony robbery in 1992 at a time when New York did not collect DNA samples from criminals convicted of felony robbery. Moreover, if a 1996

expansion of the database to include robbery convictions had been applied retroactively, he could have been required to provide a DNA sample at this point. The sample would have been linked to one of the previous crimes, *thereby preventing at least two sexual assaults against juveniles and one death.* It is also worth noting that a sooner DNA match would have prevented an innocent man from spending four months in prison.

### SEVEN PREVENTABLE RAPES AND ROBBERIES

In 2001 it was revealed that New York City had between 14,000 and 16,000 unanalyzed rape kits that were sitting in a storage rooms. Through a focused backlog reduction program, the City has been analyzing the rape kits and loading them into the state DNA database system.

In 2002, two unsolved rapes that were part of the backlog reduction project were connected to the same offender. The offender's criminal history included five prior arrests which resulted in two separate felony convictions – in 1991 for robbery and sexual abuse, and in 1997 for armed robbery. Although New York was not collecting DNA from robbery convictions in 1997, a 2000 law expanded the database to include robbery and included offenders who were still incarcerated for previous convictions. Upon release in 2001, the offender in question was required to give a DNA sample for the database.

This offender was arrested in December 2001 for a series of rapes and robberies (seven separate incidents). If the 1996 rape kits had been tested sooner, this person would have been linked to these assaults in 2001 prior to his release, *thereby preventing the subsequent 7 attacks occurring after his release*.

### **NORTH CAROLINA**

#### TWO PREVENTABLE RAPES

A series of three undetected rapes occurred in the Charlotte-Mecklenberg County area — one in February of 2000, and two more in May of 2002.

Police identified a suspect after the first attack in May, and positively identified the perpetrator through fingerprints found at the scene. Subsequent DNA testing also linked all three rapes to the suspect.

The offender was convicted in December of 2000 on felony breaking and entering and larceny charges. However, the North Carolina DNA database statute did not include either felony crime as a qualifying offense for the DNA database. If the suspect had been required to give a DNA sample for his December 2000 conviction, he could have been connected to the February 2000 rape, *thereby possibly preventing two rapes in May of 2002*.

### NORTH DAKOTA

### ONE PREVENTABLE RAPE

In June 2000, a woman in Fargo was sexually assaulted in her home by an unknown intruder. DNA evidence was collected from the crime scene, but was not matched to an offender. Police had identified a man as a possible suspect, but did not have sufficient cause to obtain a warrant for a DNA sample.

In August 2002, a teenager was sexually assaulted by an intruder in an apartment where she was babysitting. Several days later, police were finally able to obtain a warrant for a DNA sample from the suspect, and he was linked to both rapes.

The offender was convicted in Minnesota for felony possession of stolen property and burglary in September 1996, and for felony burglary in June 1997. At the time Minnesota did not require DNA from convicted burglars. North Dakota still does not require DNA from convicted burglars. If a DNA sample had been collected from the offender for either the 1996 or 1997 felony conviction, the June 2000 sexual assault could have been solved and the perpetrator apprehended, *thereby preventing one rape*. The offender also had a 1996 domestic assault misdemeanor conviction, which is also a DNA database crime in some states.

**OHIO** 

#### ONE PREVENTABLE RAPE

In early April of 2002 a man who was hired for yard work at a Columbus residence returned in the evening and burglarized the home where he had worked. In gaining entry, the burglar left blood that was collected and analyzed for DNA. The man had given the owner of the home a fictitious name and false credentials. However, through diligent detective work investigators assigned to the case identified a possible suspect and requested a DNA sample from him. The suspect complied with the request and the police returned to arrest the suspect in question after DNA forensic analysis matched the suspect's DNA and the DNA collected at the scene of the crime. Unfortunately, the suspect had disappeared in the interim.

In mid May of 2002, the offender in question raped a young girl who was able identify him. Investigators arrested him and found that his fingerprints matched felony prints on file noting that he was wanted in the April burglary. DNA testing subsequently confirmed that the suspect in question raped the young girl, and that he was the missing suspect in the home burglary.

The perpetrator in this case had a prior conviction in 1996 on felony drug trafficking and drug conspiracy charges. If the state of Ohio had required DNA from convicted felony drug offenders, police could have arrested this man after the burglary offense rather than asking him for a DNA sample which prompted his disappearance. This offender would have been

incarcerated for the burglary rather than being allowed to go into hiding, and the rape of a young girl could have been prevented.

#### TWO PREVENTABLE RAPES

Between 1993 and 1994, three rapes were committed by the same unknown perpetrator in the Hamilton County area. After the third rape, the victim was able to notify police officers promptly and they were able to apprehend a suspect in short order. DNA testing linked the suspect in question to all three rapes.

The perpetrator had a 1993 felony conviction for carrying a concealed weapon and served a sentence in state prison. He also had a 1984 conviction in California on felony sexual battery. If Ohio had required DNA from the 1993 concealed weapon conviction, the 1993 rape could have been solved, *thereby preventing two subsequent rapes*.

## ONE PREVENTABLE RAPE, ONE PREVENTABLE PROPERTY CRIME

In 1999, two rapes occurred several months apart in nearby counties. Both rapes were linked through DNA, and the database was used to identify a suspect in 2001.

The perpetrator's criminal history includes a 1994 burglary for which he was required to give a DNA sample for the database. The DNA sample was collected in 1997, and he was released from prison in 1999 on "shock probation" -- three months prior to the first rape. Unfortunately, the offender biological sample was in a backlog and remained unanalyzed until 2001 when the "cold hit" was made to both rapes. If not for the backlog, the man would have been identified and arrested as a suspect in the first rape, thereby preventing the subsequent rape. The offender in question was also convicted in late 1999 of a home burglary and theft of prescription drugs – another crime he would not have been free to commit if the DNA match had been made sooner.

#### **PENNSYLVANIA**

# TWO PREVENTABLE RAPES

In 1994, an elderly woman was raped in her home by a stranger. This case was followed in 1998 by two similar attacks on elderly women — one of whom lived independently before the attack but was particularly traumatized and moved to a nursing home thereafter. She died before the case was solved. DNA linked all three cases to one unidentified offender.

A person known to police became a suspect in the case when detectives realized the modus operandi was similar to a local known repeat offender who targeted elderly victims. The offender's parole officer supplied detectives with an envelope that had been sent by the suspected person, and DNA from the saliva was determined to be consistent with that of the attacker.

Detectives subsequently procured a search warrant for a blood sample, and that lead to a match between the three crimes and this specific offender.

The offender's criminal record included two separate felony convictions for a series of home invasion burglary crimes committed against elderly victims. His first sentence began in 1987, and he was released on parole in 1992. The second sentence began in 1995. If Pennsylvania had required DNA from felony burglaries in either 1992 or 1995, this repeat offender could have been linked to the 1994 rape, *thereby preventing two rapes in 1998*.

#### FIVE PREVENTABLE RAPES

In 2002, an unknown assailant in the Pittsburgh-area raped or attempted to rape six women and assaulted two other women. The youngest victim among these women was 14. In each case, the suspect approached the victim from behind, attempted to make conversation, and either made threats with a gun or displayed a firearm.

The suspect was eventually apprehended when his last victim shot him. The woman, who worked as a security guard, was approached by the offender in question in a park and she narrowly escaped him after he threatened her life. After reporting the incident to police, she continued her walk and was again approached by the same man. This time, instead of running away she shot the assailant twice in the stomach and then notified police. The offender in question has been linked to several of the crimes noted through DNA testing, and he is currently awaiting trial on multiple charges.

The offender's felony record included convictions in 1997 and 2002 for felony receiving of stolen property, and a 2001 conviction for felony theft. This offender's most recent release from incarceration occurred just weeks before the first rape. If the state had required DNA from all convicted felons at any point since 1997, the offender's DNA profile would have been on record and potentially could have identified him after the first rape, *thereby preventing the subsequent 5 rapes*. Pennsylvania still does not collect DNA from these felons.

### **RHODE ISLAND**

#### ONE PREVENTABLE SEXUAL ASSAULT

In Providence in 2000 an assailant kidnapped at gunpoint a male college student who was attempting to enter his apartment. The offender forced the student withdraw funds from two banks, then returned to the student's apartment and burglarized it.

In a second incident, a female and a male college student were both kidnapped and the two victims were robbed at gunpoint. The male student was forced to withdraw money from a bank, and both students were sexually assaulted. DNA was collected from this crime scene.

In a third incident, a female college student was kidnapped at gunpoint while she was entering her apartment. She was robbed of her money and her apartment was burglarized. The woman was then forced to drive to a bank in an attempt to have her withdraw additional money from the ATM. The offender then drove the female student to a secluded area and sexually assaulted her.

The offender in this case was ultimately identified through a photograph taken by an ATM Bank surveillance machine during commission of the second crime. The offender was arrested by police one day after the third crime, and he eventually received a 70-year sentence for his violent gun-involved crimes.

The offender was charged of felony possession of a stolen vehicle in 1999, which is not a crime included in the Rhode Island DNA database. With a DNA sample from this conviction, the crime laboratory could have linked the second incident to the offender, thereby possibly preventing the third crime involving a sexual assault.

### **SOUTH CAROLINA**

#### UNKNOWN PREVENTABLE CRIMES

In August of 2000 a teenager in Aiken was raped and murdered in her home. There was no suspect in the case, but DNA evidence from the crime was sent to the laboratory for testing and a profile was sent to the national database. Meanwhile, police actively pursued the investigation, conducting between 300 and 400 interviews and collecting 94 DNA samples for comparison.

In August of 2002 an offender's DNA sample was submitted to the Georgia DNA database after he was returned to state prison for violating his parole on a felony theft by taking conviction. When the sample was compared against the national database, it was matched to the South Carolina murder. The person in question had worked for a package delivery service in 2000. During one of his deliveries, the offender met the teenager and later he returned to kill her.

The offense which resulted in this offender's inclusion in Georgia's DNA database, and which eventually lead to the closure of the teenager's murder, is a crime that is excluded from the South Carolina database. Had the offender committed his prior crimes in South Carolina instead of Georgia, he would likely still be undetected and living unsuspected in the community. In addition to the theft by taking conviction, the repeat offender in question also had convictions for numerous other crimes. Among those crimes are escape, possession of a firearm by a convicted felon, theft of a motor vehicle, and entering a vehicle – none of which are qualifying offenses for the DNA database in the State of South Carolina.

# THREE PREVENTABLE RAPES

In September 1995, a woman was raped by a stranger and a rape kit was collected at the scene of the crime and submitted to the crime laboratory. The DNA profile was loaded into the state database system, but no match was made. In July, September and November of 1996 three more

women were raped, but no rape kits were collected from these crime scenes. However, all three women identified the same suspect as their attacker through mug shot photographs. Investigators interviewed a suspect at his home and made an arrest in early 1998. A DNA sample was collected from the suspect and was matched to the DNA profile from the first rape. The offender was tried and convicted for rape and burglary, and he was sentenced to life over 30 years in prison.

The offender was convicted in 1980 on two separate rape and burglary charges. He completed his sentence in June of 1995. Unfortunately, while the South Carolina database statute was enacted in 1994, it did not contain an emergency clause and therefore it did not come into effect until July of 1995 – one month too late to capture this offender's DNA sample. If the South Carolina statute had become effective immediately upon enactment, this offender could have been identified after the first rape, *thereby preventing the following three rapes*.

#### Conclusion

The case studies reported here serve their purpose well – namely, they illustrate quite clearly the concrete benefits to crime prevention of DNA databases. Many violent and property crimes committed by repeat offenders could have been prevented in these cases. Of course, the few cases set forth here only represent the tip of the iceberg; many more exemplary cases could be listed for each of these states. The following section moves on to the cost factors associated with gearing up to use DNA databases for proactive crime prevention and the clearance of unsolved cases.

## VII. COST AND BENEFIT QUESTIONS

A full cost-benefit analysis of forensic DNA testing and DNA databases is problematic since many of the potential benefits do not lend themselves to direct quantification. This Report did not attempt to develop this type of analysis, but presents the following discussion as a listing of relevant matters for consideration and possible further study.

The costs of DNA testing at crime laboratories are generally known, and have been discussed to some extent in other sections of this Report. In addition to the costs of equipment and reagents needed to complete DNA testing, an ever-increasing demand for DNA testing means a corresponding increase in the need for more capacity. Increased capacity, in turn, means more personnel, more equipment, and occasionally more space. Personnel costs, as demonstrated elsewhere in this Report, pose a considerable hurdle to DNA testing at many crime laboratories. The fact that these costs are relatively easy to calculate, and nonetheless have not been addressed adequately in some jurisdictions, raises an interesting question of funding priorities geared properly to community needs.

Beyond the quantifiable crime laboratory costs, there are also costs associated with the proper training of criminal justice professionals. Law enforcement training is needed to ensure the identification and collection of appropriate DNA evidence, as well timely collection of evidence from offenders. In order to ensure that DNA analysts are receiving quality DNA samples, there is also a need to ensure that medical professionals, particularly those responsible for rape kit collection, are adequately trained in evidence gathering and storage. Another group sometimes overlooked in this equation are the victim service providers who could prove instrumental in integrating the victim into the evidence gathering process and ensuring his/her informed participation so that DNA analysis is not performed needlessly.

Of course, once a DNA match has been made and a suspect arrested, criminal investigators, prosecutors and defense lawyers must devote their limited resources to determining whether to bring the case to trial. The training of prosecutors, criminal defense lawyers, and judges poses problems for legal professionals who must now be versed in the scientific issues of DNA analysis and related population probability statistics. Beyond training is the cost of simply having additional cases to be investigated, prosecuted and defended in an already over-burdened criminal justice system. Newspaper accounts quite frequently report on offenders who, when faced with DNA evidence, become inclined to accept plea bargains or provide admissions of guilt. Although this is certainly true for some cases, this study did not undertake to determine if the additional justice system caseload resulting from cases due to DNA matches is offset by increasing plea bargains. Moreover, some attorneys have argued that prosecutors actually may be less inclined to allow plea bargains once they have the advantage of DNA evidence, thus resulting in a heavier caseload.

Some of the most obvious benefits to forensic DNA testing are the potential to bring criminals to justice and closure to victims and their families. These, again, are largely immeasurable benefits. The "preventable crimes" case studies presented in this Report begin to paint a picture of potential added benefits to public safety from investment in DNA programs, but they do not provide a calculable measurement. There are certainly benefits to victims in DNA cases that bring an offender to justice, and individual stories can again give a glimpse into this world, but these cannot be computed in a manner that can be compared easily against costs. The question has also been raised as to whether DNA testing and DNA matches have an indirect

effect in encouraging victims of violent crimes to come forward. Again, anecdotally there are victims who will say this is the case, but further study of this question might be useful in providing reliable cost efficiency data.

Moreover, every match of an offender to a crime on the DNA database represents a case that an investigator may soon consider "cleared." Clearing these cases in turn allows the investigator to turn his or her attention to other unsolved crimes. To understand cost efficiencies to investigations, data including man-hours involved in a "typical" investigation in cases without DNA, would need to be compared to "typical" case with DNA – and/or cases in which DNA matches have named potential suspects. Without this type of data, it is difficult to have a full understanding of potential cost efficiencies.

And last, but by no means least, benefits are also derived from ensuring that the appropriate person faces justice for a crime. Post conviction DNA testing in the United States has been instrumental in the release of over 100 persons who have been wrongly convicted. Increasingly, DNA testing is now being used to exonerate persons <u>prior</u> to conviction. Indeed, crime laboratories estimate that in cases requesting a match of DNA evidence from an unsolved crime to a DNA sample from a known potential suspect, the analysis exonerates the suspect identified by investigators in approximately a third of the cases. Such exonerations not only have a significant benefit to the misidentified suspect, but also allow law enforcement to refocus their efforts and avoid wasting time investigating the wrong person. In addition to the direct benefit supplied to both the accused and investigators, DNA exonerations provide the general public with a greater level of confidence in the criminal justice system.

#### VIII. CONCLUSIONS AND POLICY IMPLICATIONS

In the relatively brief amount of time that forensic DNA has been available to the criminal justice system, its impact has already proven remarkable – from the conviction of rapists and murderers to the exoneration of wrongly convicted inmates on death row. However, the proven effectiveness of forensic DNA has resulted in a tremendous demand for additional testing which has not been met by a corresponding increase in available supply. The purpose of this study was to provide a basis for extrapolating the potential number of DNA cases sitting in law enforcement evidence rooms and storage areas throughout the country. In developing these figures, crime laboratories were also asked to supply data regarding their backlogged cases in order to arrive at a more precise national sum. It was also important in this study to have a discussion as to the reasons for the unanalyzed DNA cases found both at the law enforcement agencies and the crime laboratories. This discussion included capacity issues facing both law enforcement and crime laboratories as well as a discussion of how forensic DNA is viewed by law enforcement. Finally, the study also assembled an overview of DNA database matches in the United States along with a discussion on the potential of DNA to not just solve crimes, but also to prevent crimes. The following summarizes the major conclusions drawn from the Report.

1. The backlog of unsolved rapes and homicides in the U.S. is massive.

When unsolved property offenses with possible biological evidence are added into the equation, there are an estimated 700,000 of these unsolved cases nationally. Furthermore, of the 400,000+ unsolved rapes and homicides nationally law enforcement agencies reported that roughly half of such cases are likely to contain biological evidence for DNA testing. Thus, when including the extant unsolved property offenses with biological evidence, there may be a national total of 553,821 unsolved rapes, homicides, and property offenses that may be amenable to DNA testing.

2. The majority of law enforcement agencies have insufficient evidence storage capacity.

This issue is critical since, as demonstrated in Section III, pressures on storage space can result in biological evidence being maintained under improper conditions or, even worse, discarded or not collected at all. This issue is also significant given the finding that the overwhelming majority of law enforcement agencies (over 70 percent) reported their storage capacity limitations to be either "critical" or "highly critical."

3. A significant proportion of law enforcement agencies continue to misunderstand the potential benefits of DNA testing.

As reported by nearly one-fourth of all law enforcement agencies, one of the primary reasons for not sending DNA evidence to a crime laboratory is due to the lack of a suspect. To be sure, this is exactly the kind of situation where an offender DNA database would be most useful.

As a particularly notable example of this lack of understanding of the potential utility of DNA testing, during a telephone conversation with researchers the sex assault unit commander of the largest local law enforcement agency in a particular state reported that it had *never* submitted a DNA sample for testing.

4. Both state and local crime laboratories are overworked, understaffed, and insufficiently funded.

These agencies simply do not have the capacity to handle their current caseloads—let alone their anticipated backlogs for the near future—in a timely manner. This problem would appear to be broadly present across the country.

5. The role of the federal government in funding forensic DNA analysis has been, up to this point, important but rather minimal.

Indeed, the majority of funding for local crime laboratories is derived from local sources, and the majority of funding for state crime laboratories comes from state sources. Only one-in-five state crime laboratories, and fewer than one-in-twenty local crime laboratories, report that the majority of their funding comes from federal sources. Clearly, not only is there a greater potential role for the federal government to play in assisting with the current backlog problem, but there is also a certain need for state and local governments to re-evaluate their degree of investment in their forensic crime laboratories given that demand for services has far outstripped availability of analysis over the past several years.

6. Forensic DNA databases are important tools in solving a variety of crimes, committed by a variety of criminals.

The DNA database system has been used by an increasing number of law enforcement agencies throughout the nation to identify recidivist criminals and to provide investigatory leads to detectives, particularly in cases involving serial crimes. While the DNA database, to date, has been limited largely to violent offenders and violent unsolved crimes, this trend is changing. As demonstrated by states such as Virginia, the potential for DNA to play a vital role in solving non-violent crimes is considerable.

7. There are numerous crimes that are potentially preventable through better, more efficient use of forensic DNA analysis.

Although it is extremely difficult to document the number of preventable crimes associated with the timely availability of DNA evidence, a look at specific case studies from a variety of states clearly demonstrates the potential for DNA to prevent repeat offender criminals from victimizing multiple citizens.

8. The cost, and the offsetting benefits associated with DNA testing present several questions relating to return on investment which merit further study.

Many of the costs associated with DNA testing are discussed in this study, but a more comprehensive study of the costs of DNA forensic evidence collection and processing is needed to provide policy makers with additional valuable information required to make informed decisions during budgetary discussions. The benefits of DNA analysis to investigators, prosecutors, victims and the falsely accused are an important part of the cost/benefit puzzle that budget writers must take into consideration in allocating resources to the national DNA collection and analysis system.

<sup>&</sup>lt;sup>i</sup> Where a state forensic DNA laboratory does not exist, the local crime laboratory appointed as the FBI CODIS point of contact is treated as the state laboratory in this study. This is the case for Nevada (Washoe County Sheriff's Office laboratory), and Hawaii (Honolulu Police Department Crime Laboratory). The local laboratory in Albuquerque, New Mexico serves as the CODIS point of contact instead of the DNA state laboratory. As such, Albuquerque assisted in completing two assessment forms – one on behalf of the local laboratory, and the other calculating state laboratory figures.

ii Since we are dealing with extrapolating to the total population of law enforcement agencies (18,000) with our sample (over 1,500), researchers have taken the most methodologically rigorous (and therefore conservative) approach possible so that they could avoid any potential charge that these estimates are inflated. Accordingly, there is a possibility that the true numbers of unsolved homicides and rapes are actually higher than what is reported here.

The assessment form specified that law enforcement agencies should only count those cases that "were still in your control (i.e., had not been sent to a laboratory for testing)." Similarly, crime laboratories were asked for the number of cases "backlogged at your laboratory." For this reason, double counting of cases by law enforcement agencies and crime laboratories was not an issue.

iv "Critical" was defined as "available storage is quickly disappearing."

v "Highly Critical" was defined as "no additional storage space available."

vi A "median" figure of 0.0 results from a severely skewed distribution featuring many zero values.

<sup>&</sup>quot;Other," as defined by state laboratories, included expert software for CODIS review, IT support/informatics, in house offender analysis, software for case management, equipment and reagents used to address suspect/no suspect backlogs, and personnel (3).

viii "Other," as defined by local laboratories included personnel (2) and validation projects.

<sup>&</sup>lt;sup>ix</sup> The non-violent offenses include property offenses (theft, forgery, breaking and entering, burglary), drug offenses, and supervision violations. The violent offenses include murder/homicide, rape and sex offenses, assaults, and robbery.

<sup>&</sup>lt;sup>x</sup> Gilmer, James A., van Alstyne, David J., *The First 100 Hits — Forensic-Offender Matches on the New York State DNA Data Bank*, Research Note, New York State Division of Criminal Justice Services, Bureau of Research and Evaluation: 1.

xi Ibid.. 1.

xii Langan, Patrick A., Levin, David D., *Recidivism of Prisoners Released in 1994*, Special Report, Washington, DC: U.S. Department of Justice, Bureau of Justice Statistics, January 2002, NCJ 191191: 1.

xiii Ibid., Gilmer, The First 100 Hits, 1.

# **APPENDIX 1a**

Forensic DNA Assessment Form Cover Letter



Office of the Director

April 4, 2003

Washington, D.C. 20531

Dear Law Enforcement Administrator:

U.S. Attorney General John Aschroft recently announced President Bush's Initiative for Advancing Justice Through DNA Technology. The Initiative proposes to invest \$1 billion in forensic DNA over the next five years in order to improve the use of DNA in the criminal justice system. The much needed funding will be directed at such purposes as reducing DNA evidence backlogs, encouraging expanded offender DNA databases in solving crimes, and DNA training of law enforcement officers and other criminal justice professionals.

The National Institute of Justice has asked Smith Alling Lane and Washington State University to conduct a nationwide assessment regarding forensic DNA issues facing law enforcement. The information collected will be an important tool in crafting appropriate Department of Justice programs and policies to address these issues. In the past, such information has also been used by the U.S. Congress and many states and local communities in making budget and policy decisions. The scope of attention this matter is likely to draw in the coming year makes clear the need for accurate and comprehensive data on the current state of forensic DNA use by law enforcement nationwide. We hope that you will assist us in compiling this information.

Although your participation in the assessment is completely voluntary, the cooperation of each identified agency is essential to ensuring that the assessment results are representative and thorough. In the process of developing this assessment various experts in the legal, law enforcement, and forensic communities were consulted to ensure the collection of useful information without undue disruption to your work. The number of questions being asked has been limited to an absolute minimum. To provide timely results for the criminal justice community and for the public policy process, we ask that you complete the enclosed assessment questionnaire *by June 1, 2003*.

Enclosed with this correspondence you will find a set of instructions, a glossary of key terms, and a brief questionnaire. Also enclosed is a preaddressed, postage-prepaid envelope for your convenience. The final report presenting these data will be provided to you as soon as it is published.

Thank you in advance for your cooperation in this important national data collection effort. If you have any questions regarding this assessment or any particular questions on the questionnaire, please contact the project manager, Lisa Hurst of Smith Alling Lane at (520) 664-0806, or via e-mail at <a href="mailto:lhurst@smithallinglane.com">lhurst@smithallinglane.com</a>. Should you wish to discuss this project with someone at the National Institute of Justice, please contact Dr. Lisa Forman at (202) 307-6608, or Nick Viggiani at (202) 307-5847.

Sincerely,

Sarah V. Hart

Director

# **APPENDIX 1b**

Local Law Enforcement Agency Form and Instructions

### **GENERAL INSTRUCTIONS**

The person designated to respond to this assessment should have an extensive understanding of the agency's infrastructure and be knowledgeable about the system for processing evidence in criminal cases. Additionally, it would help if the respondant is positioned to assist with *some coordination between various units whose input may be needed in completing the assessment*. We appreciate that the organization of some agencies may make such coordination extremely difficult – if this is the case with your agency, please help us by answering what you can and use the "contact" line to give the title and phone number of the appropriate department with which we should follow up.

Please note – if your agency also operates a forensic crime laboratory, the laboratory director will receive a separate and distinct set of questions regarding forensic DNA testing. We have purposely split the assessment into two separate versions because the crime laboratories may not be best-suited to respond to general questions regarding continuing investigations and unsolved cases.

Respondants should make every attempt to use the dates specified. However, should you need to refer to Calendar Year 2001 statistics, please do so rather than answering "unknown."

- The enclosed form is meant as a general assessment and not a complete census of specific issues facing crime laboratories performing DNA analysis. Where exact numbers are unknown, *your best educated guess is always an appropriate response.* Please do not leave any responses blank. *Where necessary, please indicate NA for "not applicable."*
- REMINDER: The results of this assessment will be used by local, state, and federal policy makers in determining how to strengthen policies and funding for law enforcement's effective use of DNA evidence. Responses that indicate what your agency should be doing instead of what your agency is doing will result in continued gaps in important issues such as training and funding. Please be candid in your responses individual agency responses will not be identified in the published report.

Please complete the assessment and mail it in the enclosed PRE-PAID envelope.

#### **GLOSSARY OF TERMS**

**Biological Evidence** – bodily fluids such as blood, semen, and saliva, as well as bodily fluid stains or trace biological materials on clothing, weapons, cigarette butts, etc, which might be collected as part of a criminal investigation.

Case – in general, cases should be counted per criminal investigation – a multiple homicide at the same house should be counted as one case. However, in cases that are investigated as serial crimes, each crime should be counted as an individual case.

**DNA** – deoxyribonucleic acid, the chemical found in blood and other biological evidence which can be analyzed and used for individual identification purposes.

**Homicide** – to include all crimes of murder and manslaughter, as well as attempts of these crimes.

**Named Suspect Case** – case has at least one individual identified who has been identified by name for possible comparison to biological evidence.

**Rape** – to include attempted rape and other definitions of rape as supported by your state statute (such as 1st degree sexual assault). Statutory rape should be included.

**Rape Kit** – Rape kits are typically collected at hospitals by specially trained nurse examiners. The kits will contain evidence such as vaginal swabs and fingernail scrapings, but do not contain other crime scene materials with potential DNA evidence such as bedsheets or cigarette butts.

**Unnamed Suspect Case** – case has no individuals who have been identified by name for possible comparison to biological evidence.



#### Forensic DNA Assessment PMB #317 120 S. Houghton Road, Suite 138 Tucson, AZ 85748

# DUE DATE: June 1, 2003

					DA	ГА СО	MPLETED BY	,					
NAME				TI	TLE			AGE	NCY				
PHONE				FA	$X_{\underline{}}$			E-M.	AIL				
												_	
The person designated to respond to this questionnaire should have an extensive understanding of the agency's infrastructure and be knowledgeable about outstanding criminal cases. Some responses may require coordination with other units within the agency. Please make every attempt to answer						ency's riminal other	5. Of the rape of Question 4), I many have of could be teste	now many her evide	have an nce (not in	associated	d rape kit?	How	\
these questions a	s comp	letely as	possibl	e, but u	nderstan	d that		0% - 10%	11% - 25%	26% - 50%	51% - 75%	75% - 100%	
exact numbers are is an appropriate			your	best ed	lucated (	guess	RAPE KITS ONLY						
SECTIO	NI: DN	NA Case					RAPE KITS AND OTHER DNA EVIDENCE						
Where are cases     STATE AGENCY	LAB [	FBI LA	ıΒ	PRIN	/ATE LAB	l LAB	OTHER DNA EVIDENCE ONLY						
LOCAL AGENCY	LAB [	OTHER	R (Explai	n)			Educated gu	ess is not p	ossible with	out a compr	ehensive re	view	
	-						UNABLE TO ANSWER	COMPLETEL	Y. CONTAC	т			
2. For all rapes and July 1, 2002 to J will have DNA projection - you o	une 30, eviden	, 2003, <u>a</u> ce colle	pproxim	<u>nately</u> wl This is	hat perce asking	entage for a	6. Of the total r Question 4, h limitations as have exceed possible, plea suspect cases	ow many of Novem ed the sta ase indica	of these ber 1, 200 tute of lim	are <b>still v</b> 02? How itations by	vithin the many of June 30,	statute of these <i>will</i> 2003? (If	
HOMICIDES WITH							WITHIN THE STA	TUTE OF L	IMITATIONS	3			
POSSIBLE DNA							UNNAMED SI	JSPECT RA	APES				
Educated guess is	s not pos	sible with	out a con	nprehens	ive review		WILL EXCEED THE STATUTE BY 06/30/2003						
UNABLE TO ANSWER COI	MPLETEL	Y. CONTA	CT				UNNAMED SI	JSPECT RA	APES				
3. As of June 30, 20 and homicide committed within	ases v	were in	your ju	ollowing urisdictic	unsolve on (for o	d rape crimes	Educated UNABLE TO ANSWER	-			-		
		TOTAL	CASES	CASE	S WITH D	ONA	7. For all cases,	whore is a	ınanalyzor	l ovidonco	storod? C	heck all the	a f
UNSOLVED HOMIC	IDES						apply.	wilele is t	ilialialy2 <del>6</del> 0	eviderice	stored: C	TIECK AII LITE	זנ
UNSOLVED RAPES	}						CENTRALIZED	STORAGE	AREA	PRO	SECUTOR	'S FACILITY	
Educated gues	s is not p	possible w	vithout a	comprehe	ensive revi	ew	DECENTRALIZ	ED STORA	GE AREAS	/ □ CRI	ME LAB FA	CILITY	
UNABLE TO ANSWER C	OMPLETE	ELY. CONT	TACT				VARIOUS DIST	RICT LOCA	ATIONS	_			
<ol> <li>As of June 30, cases with poss control (ie, had possible, please</li> </ol>	sible be	<i>iologicia</i> en sent t	al evide to a lab	nce we	re still <i>in</i> for testing	<b>your</b> g)? <b>If</b>	8. Does your age storage?	ency have	sufficient	☐ OTH		evidence	_
unnamed suspec	ts.			BBOBE	DTV		SUFFICIENT	STORAGE	: C	INSUFFIC	CIENT STOP	RAGE	
TOTAL CASES	RAPE HOMICIDE PROPERTY CRIMES OTHER						If storage sp additional space					gaining	
UNNAMED							HIGHLY CRI	TICAL - N	O ADDITIO	NAL STORA	AGE SPACE	AVAILABLE	Ξ.
SUSPECTS							CRITICAL -	AVAILABLE	STORAGE	IS QUICK	LY DISAPPE	EARING	/
UNABLE TO ANSWER		•					PRESSING I SPACE TOD						

<sup>1</sup>Please indicate in the margin if you must use 2001 Calendar Year statistics instead of the dates given.

Your best educated guess is always an appropriate response if exact numbers are not known. Where necessary, please indicate NA for "not applicable."

9. If DNA evidence for either unsolved rapes or unsolved homicides has not been sent to a laboratory, please identify the reasons. Please select all that apply and rate the top three (with "1" being the most significant).  APPLIES RATE  (1.3)  NOT APPLICABLE  A SUSPECT HAS NOT YET BEEN IDENTIFIED	13. IF YOU ANSWERED "NO" TO QUESTION 11 – Does your agency have a plan to review old, unsolved cases for new investigatory leads, such as DNA testing? How many cases would be included in such a review, and approximately how many of these cases are thought to potentially contain biological evidence for DNA testing?  Agency has a plan to review cases for biological testing?
A SUSPECT HAS BEEN IDENTIFIED BUT HAS NOT BEEN FORMALLY	YES NO (explain)
CHARGED IN THE CASE  A SUSPECT HAS BEEN CHARGED AND A GUILTY PLEA IS	
ANTICIPATED WHICH WILL OBVIATE THE NEED FOR DNA TESTING	
UNCERTAIN HOW DNA ANALYSIS MAY BE USEFUL IN THE CASE	CASES ELIGIBLE FOR REVIEW
LACK OF FUNDING FOR DNA ANALYSIS	CASES THOUGHT TO POTENTIALLY HAVE BIOLOGICIAL EVIDENCE FOR TESTING
☐ INABILITY OF LABORATORIES TO PRODUCE TIMELY RESULTS	BIOLOGICIAL EVIDENCE FOR TESTING
LABORATORY IS NOT PROCESSING REQUESTS FOR DNA TESTING	Educated guess is not possible without a comprehensive review
UNCERTAIN WHERE TO SEND THE CASE FOR DNA ANALYSIS	UNABLE TO ANSWER COMPLETELY. CONTACT
ANALYSIS NOT REQUESTED BY PROSECUTORS	
OTHER (EXPLAIN)	14. Would your agency be interested in using forensic DNA
· · · ·	databases more frequently if there was a reasonable expectation that an unnamed suspect could be quickly identified through the
10. Does your department routinely collect DNA evidence for property crimes? Does your local/regional/state laboratory accept and process such evidence for DNA analysis if the suspect is unnamed?	database?  NO (please explain)
ROUTINE DNA COLLECTION YES NO	
LABORATORY ACCEPTS <b>AND</b> PROCESSES UNNAMED SUSPECT DNA YES NO EVIDENCE FOR PROPERTY CRIMES.	<b>15.</b> Has your agency used the DNA database system to aid investigations in any way that did <i>not</i> involve a <u>direct match</u> on the database (such as, in determining whether a person's DNA profile was already available through the database)?
LABORATORY ACCEPTS <b>BUT DOES NOT</b> PROCESS UNNAMED SUSPECT DNA YES NO EVIDENCE FOR PROPERTY CRIMES.  UNABLE TO ANSWER COMPLETELY. CONTACT	YES (please explain) NO
11. Does your agency have a "cold case" squad or other policy for reviewing unsolved violent crimes for fresh investigatory leads, such as new DNA testing?	SECTION III: Comments
☐ YES ☐ NO	Please use the remaining space on this form (or attach a sheet) to register any concerns or comments you wish to record regarding
If "YES" – Go to question 12. If "NO" – Go to question 13.	forensic DNA issues affecting your operations.
UNABLE TO ANSWER COMPLETELY. CONTACT	
12. IF YOU ANSWERED "YES" TO QUESTION 11 – Approximatley how many cases are potentially eligible for review? How many cases have been reviewed? How many reviewed cases were sent for DNA testing? How many additional cases do you believe may have biological evidence for DNA testing?	
ELIGIBLE FOR REVIEW	
CASES REVIEWED	
REVIEWED CASES SENT FOR DNA TESTING	
ADDITIONAL CASES POTENTIALLY ELIGIBLE FOR DNA TESTING	
Educated guess is not possible without a comprehensive review  UNABLE TO ANSWER COMPLETELY. CONTACT	

<sup>1</sup>Please indicate in the margin if you must use 2001 Calendar Year statistics instead of the dates given.

Your best educated guess is always an appropriate response if exact numbers are not known. Where necessary, please indicate NA for "not applicable."

# **APPENDIX 1c**

Local Laboratory Form and Instructions

#### **GENERAL INSTRUCTIONS**

If your laboratory does **not** perform any DNA testing, please write "No DNA testing" on the front of the enclosed assessment form and return it in the enclosed envelope.

This assessment would be best completed by the crime laboratory's DNA section, although some coordination with the agency investigatory units may be necessary. Please note – if your agency is connected to a police department or sheriffs office, the head of the agency will receive a separate and distinct assessment regarding forensic DNA evidence. We have purposely split the assessment into two separate versions because the crime laboratories may not be best-suited to respond to certain questions regarding continuing investigations and unsolved cases.

The enclosed form is meant as a general assessment, and not a complete census, of specific issues facing crime laboratories performing DNA analysis. Where exact numbers are unknown, *your best educated guess is always an appropriate response*.

Please do not leave any responses unanswered. Where necessary, please indicate NA for "not applicable."

Please complete the assessment and mail it in the enclosed PRE-PAID envelope.

#### GLOSSARY OF TERMS

Backlogged cases – unreported forensic cases in possession of the laboratory for more than 30 days.

**Backlogged offender samples** – unanalyzed offender samples in possession of the laboratory for *more than 30 days*.

**Biological Evidence** – bodily fluids such as blood, semen, and saliva, as well as bodily fluid stains or trace biological materials on clothing, weapons, cigarette butts, etc, which might be collected as part of a criminal investigation.

Case – in general, cases should be counted per criminal investigation – a multiple homicide at the same house should be counted as one case. However, in cases that are investigated as serial crimes, each crime should be counted as an individual case.

**DNA** – deoxyribonucleic acid, the chemical found in blood and other biological evidence which can be analyzed and used for individual identification purposes.

**Homicide** – to include all crimes of murder and manslaughter, as well as attempts of these crimes.

**Forensic hit** – occurs when two or more forensic samples are linked on the state, local or national DNA database system.

**Offender hit** – occurs when one or more forensic samples are linked to a convicted offender sample on the state or national DNA database system.

Named Suspect Case – case has at least one individual who has been identified by name for possible comparison to biological evidence.

**Rape** – to include attempted rape and other definitions of rape as supported by your state statute (such as 1st degree sexual assault).

**Rape Kit** – Rape kits are typically collected at hospitals by specially trained nurse examiners. The kits will contain evidence such as vaginal swabs and fingernail scrapings, but do not contain other crime scene materials with potential DNA evidence such as sheets or cigarette butts.

**Unnamed Suspect Case** – case has no individuals who have been identified by name for possible comparison to biological evidence.



2.

Please return to
Forensic DNA Assessment
PMB #317
120 S. Houghton Road, Suite 138
Tucson, AZ 85748

# **DUE DATE: May 30, 2003**

				ASS	ESS	MENT (	COME	PLETED 1	BY				
NAME						ΓΙΟΝ							
PHONE				F	AX_				E-N	IAIL_			
This form is not a comp unknown, y an appropri	s mear lete ce	nt as a nsus. <b>est e</b> d	Where exiducated g	ssessm act num	nent, a	are	1	Of the <u>backlo</u> many have a evidence (not DNA?	in associa	ted rape	kit? Hov	v many h	ave other
ан арргорі	rate re	эроп	<b>36</b> .						0% - 10%	11% - 25%	26% - 50%	51% - 75%	75% - 100%
As of November							RAP	E KITS <u>ONLY</u>					
were backlogged waiting to be scre selected for DN	ened for	or DN. alysis,	A materials and how	s, how r	many h	nad been orted? <u><b>If</b></u>	ОТН	E KITS <u>AND</u> ER DNA DENCE					
possible, please unnamed suspect		ate ho	ow many	of thes	se cas	ses have		ER DNA DENCE <u>ONLY</u>					
	ТО	TAL	SCREEN QUEUE		YSIS	REPORT QUEUE		Educated	guess is no	t possible w	ithout a cor	nprehensive	review
RAPE					-		_						
UNNAMED SUSPECT	Γ							Of your <u>backl</u> <i>within</i> the sta					
IOMICIDE								many of these	e will hav	e exceede	ed the sta	tute of lim	itations by
UNNAMED SUSPECT	Т							June 30, 2003 are unnamed				how man	y of these
				l		<u> </u>	-	THIN THE STA				1	
JNNAMED SUSPECT	-						Į Į	UNNAMED SI					
NAVAMED COCI ECT							14/1	LL EXCEED TH			/2002		
THER							I VVI	1			72003		
UNNAMED SUSPEC	Т							UNNAMED SI	JSPECT RA	APES			
For all rapes an agencies service percentage will ha APES WITH OSSIBLE DNA	d by	your	laboratory, dence colle	appro		ely what		By June 30, 2 packlog of rappossible, pleasuspect rape	2003, wha ape cases ase indica	t do you a awaiting te how m	inticipate v DNA rev	will be the iew and a	size of you analysis? oe unname
OMICIDES WITH								ANTICIPATE	D <b>RAPE</b> CA	SE BACKL	OG		
OSSIBLE DNA								UNNAME	ED SUSPEC	T RAPES			
Educated guess	s is not p	ossible	e without a c	ompreh	ensive i	review		ANTICIPATE	D BACKLO	G OF ALL C	CASES		
What percentage typically yield test cases (including of will yield testable)	table Dones wi	NA e	vidence? \	What pe	ercenta	age of all	7.	Educated  What is the (non-priority)	approxim	ate length	of time f	or analysi	s of a typ
	0% - 10%	11% 25%		50%- 75%	75% 100%			swab with or the rape kit is					
RAPE KITS WITH DNA								NON PRIORITY					
ALL CASES WITH DNA							L				vithout a cor	nprehensive	review
	1		sible without					Luucaleu	90000 10 110	r hossinie M	itilout a col	inprenensive	TOVICAN

8. What is currently the output capacity for your labor terms of completed DNA analysis on forensic cas averages.  SAMPLES per MONTH / YEAR (choose SAMPLES per CASE)	es? <i>Use</i>	13. If your laboratory does not process DNA evidence for <u>unnamed suspect property crimes</u> , what are the major obstacles to performing such tests? Please select all that apply <u>and rate the top three</u> ("1" being the most significant obstacle).  APPLIES RATE (() (1-3) LACK OF STORAGE SPACE					
9. What is the likelihod that your local agency(ies) has revidence storage that are unaccounted for by y laboratory (10 cases or more)?  A CERTAINTY VERY LIKELY SOMEWHAND SOMEWHAND UNKNOWN	our crime HAT LIKELY HAT UNLIKELY	MORE PERSONNEL NEEDED  LACK OF OPERATIONAL SPACE PROSECUTION MUST REQUEST					
<b>10.</b> If DNA evidence for either unsolved rapes or homicides is not in the queue at your laboratory, what to be the <u>major</u> obstacles? Select all that apply, and return three ("1" being the most significant obstacle.)	t are likely	SECTION II: Mitochondrial DNA					
NOT APPLICABLE  A SUSPECT HAS NOT YET BEEN IDENTIFIED  A SUSPECT HAS BEEN IDENTIFIED BUT HAS NOT YET FORMALLY CHARGED IN THE CASE	BEEN	14. For calendar year 2002, approximately how many forensic cases analyzed by your laboratory will fail to yield testable amounts of DNA for STR analysis, but could potentially benefit from mitochondrial DNA testing?					
A SUSPECT HAS BEEN CHARGED AND A GUILTY PLEA ANTICIPATED WHICH WILL OBVIATE THE NEED FOR DI		TOTAL CASES					
AGENCIES UNCERTAIN HOW DNA MAY BE USEFUL IN LACK OF FUNDING FOR DNA ANALYSIS  BACKLOG AT LABORATORY PREVENTS TIMELY RESULT		Educated guess is not possible without a comprehensive review  15. For calendar year 2002, approximately how many hair analysis cases will be received by your laboratory that could benefit from					
LABORATORY IS NOT PROCESSING REQUESTS FOR D		additional mitochondrial DNA testing?					
AGENCIES UNCERTAIN WHERE TO SEND THE CASE FOR ANALYSIS NOT REQUESTED BY PROSECUTION	OR ANALYSIS	HAIR ANALYSIS CASES					
OTHER (EXPLAIN)		Educated guess is not possible without a comprehensive review					
The state of the suspects of the forension of the suspects of the forension of the suspect of the	oratory was case profile, ed as likely who were	SECTION IV: Costs / Funding  16. What is the approximate average cost of processing an unnamed suspect rape kit? Assume vaginal swab with one perpetrator and one victim, and that time runs from date the rape kit is received by the laboratory until results are reported.  COST OF RAPE KIT ANALYSIS  Educated guess is not possible without a comprehensive review					
NAMED SUSPECTS	5.10	Factors included in the rape kit cost estimate. Check all that apply.					
Educated guess is not possible without a comprehensive	review	☐REAGENTS ☐OVERHEAD ☐EQUIPMENT ☐OTHER					
12. In your opinion, do the law enforcement agenci jurisdiction routinely collect DNA evidence from property crime? Does your laboratory accept and previdence for DNA analysis if the suspect is unnamed?      ROUTINE DNA COLLECTION FROM PROPERTY CRIMES.	scenes of	17. Over the past 5 years, approximately what percentage of your budget for DNA analysis has come from local funding? State funding? Federal funding?					
LABORATORY ACCEPTS <b>AND</b> PROCESSES UNNAMED SUSPECT DNA EVIDENCE FOR PROPERTY CRIMES.  YES	□ NO	0%-10% 11% - 25% 25% - 50% 50% - 75% 75% - 100%  LOCAL					
YES	□ NO	STATE FEDERAL /					
		Educated guess is not possible without a comprehensive review					

18. If supplemental funding was available for your laboratory's DNA program, how would you hope to use such funds? Please select all that apply, and rate the top three (using "1" as the most	
significant). APPLIES RATE	
SALARIES	
OVERTIME	
☐ TRAINING	
EQUIPMENT	
□ ROBOTICS	
☐ REAGENTS ☐ CONSTRUCTION / LEASE	
□ □ OFFENDER COLLECTION	
SUSPECT CASE BACKLOG	
□ NO SUSPECT CASE BACKLOG	
OUTSOURCING	
☐ OTHER	
SECTION V: DNA Database Matches	
Has your laboratory made any DNA database matches, as of October 1, 2002 that would not have been recorded at the state level? If so, please attach a sheet with the following information	
for each match:  Type(s) of match (forensic hit, or offender hit)	
Type(s) of crimes solved or linked with the match	
Offender's criminal history** Investigations aided	
Disposition of case(s) (if known)	
** If criminal histories are not readily available, please contact Lisa Hurst at (520) 664-0806 or <a href="mailto:lhurst@smithallinglane.com">lhurst@smithallinglane.com</a> .	
Note: the information being requested from you does not need to identify either the offender or the victim of a crime. If you provide information that does contain such personally identifying information, this information will be deleted from our records during the course of data assembly and analysis. No personally identifying information will be published, and this research project will abide by all federal human subjects rules and procedures. A copy of the project's federal privacy certificate can be obtained if necessary for lawful release of this information. If there are any problems related to the transfer of the requested information, please contact Lisa Hurst at (520) 664-0806 or <a href="mailto:lnust@smithallinglane.com">lnust@smithallinglane.com</a> .	
When in doubt, please use CODIS Hit Counting Guidelines as prepared by the Federal Bureau of Investigation.	
SECTION VI: Comments	
Please use the remaining space (or attach an additional sheet) to register any concerns or comments you wish to record regarding forensic DNA issues affecting the operation of your laboratory.	
	,

# **APPENDIX 1d**

State Laboratory Form and Instructions

#### **GENERAL INSTRUCTIONS**

If your laboratory does **not** perform any DNA testing, please write "No DNA testing" on the front of the enclosed questionnaire and return it in the enclosed envelope.

This assessment would be best completed by the crime laboratory's DNA section, although some coordination with the state's criminal investigations unit may be necessary.

The enclosed form is meant as a general assessment – and not a complete census – of specific issues facing crime laboratories performing DNA analysis. Where exact numbers are unknown, *your best educated guess is always an appropriate response*.

Please do not leave any responses unanswered. Where necessary, please indicate NA for "not applicable."

Please complete the assessment and mail it in the enclosed PRE-PAID envelope.

### **GLOSSARY OF TERMS**

Backlogged cases – unreported forensic cases in possession of the laboratory for *more than 30 days*.

**Backlogged offender samples** – unanalyzed offender samples in possession of the laboratory for *more than 30 days*.

**Biological Evidence** – bodily fluids such as blood, semen, and saliva, as well as bodily fluid stains or trace biological materials on clothing, weapons, cigarette butts, etc, which might be collected as part of a criminal investigation.

Case – in general, cases should be counted per criminal investigation – a multiple homicide at the same house should be counted as one case. However, in cases that are investigated as serial crimes, each crime should be counted as an individual case.

**DNA** – deoxyribonucleic acid, the chemical found in blood and other biological evidence which can be analyzed and used for individual identification purposes.

**Homicide** – to include all crimes of murder and manslaughter, as well as attempts of these crimes.

**Forensic hit** – occurs when two or more forensic samples are linked on the state, local or national DNA database system.

**Offender hit** – occurs when one or more forensic samples are linked to a convicted offender sample on the state or national DNA database system.

Named Suspect Case – case has at least one individual who has been identified by name for possible comparison to biological evidence.

**Rape** – to include attempted rape and other definitions of rape as supported by your state statute (such as 1st degree sexual assault).

**Rape Kit** – Rape kits are typically collected at hospitals by specially trained nurse examiners. The kits will contain evidence such as vaginal swabs and fingernail scrapings, but do not contain other crime scene materials with potential DNA evidence such as sheets or cigarette butts.

**Unnamed Suspect Case** – case has no individuals who have been identified by name for possible comparison to biological evidence.



Please return to
Forensic DNA Assessment
PMB #317
120 S. Houghton Road, Suite 138
Tucson, AZ 85748

# DUE DATE: December 15, 2002

NAME			TITL	.E		AGENO	CY				
PHONE			FAX	· ·	AGENCYE-MAIL						
SECTION I: DNA Casework Information  This form is meant as a general assessment, and not a complete census. Where exact numbers are unknown, your best educated guess is an					cases (incl	entage of geld testable uding ones stable DNA?	DNA evidowith rape	ence? kits) su	Wha bmitte	t percen ed for D	tage of a NA revie
	iate response.	euucate	u yuess is i	ali					26% - 50%	50%- 75%	75% - 100%
. As of November 1, 2002, how many forensic cases were in your state DNA database? <i>If your laboratory can track the types of crimes these samples represent</i> , please include		track the	RAPE DNA ALL C WITH								
those numbers		1		7	Educated	guess is not p	ossible with	out a co	mpreh	nensive re	view
# OF FOREN					5. Of the bac	dogged rape	cases (di	scussed	d in Q	uestion	2), how n
HOMIC					have an as	sociated rap	e kit? Ho	w man	y hav	e other	
RAPES	ERTY CRIMES			=	included in	пе гаре ки)	triat could	be lest	ed ioi	I DINA!	
OTHE				+		0% - 10%	11% - 25%	26% - 5	0% 5	51% - 75%	75% - 100
OTTL	`	1			RAPE KITS <u>ONI</u>	<u>Y</u>					
<ul> <li>As of Novemb were backlogge</li> </ul>	ed at your laborate				RAPE KITS <u>AND</u> OTHER DNA	4					
selected for [	creened for DNA DNA analysis, a se indicate how	materials and how	s, how many many rep	had been orted? <u>If</u>	EVIDENCE OTHER DNA	Y					
selected for [	DNA analysis, a se indicate how	materials and how	s, how many many rep	had been orted? <u>If</u>	OTHER DNA EVIDENCE ONL	Y guess is not p	ossible with	out a co	mpreh	nensive re	view
selected for I possible, plea	DNA analysis, a se indicate howests.	materials and how many  SCREEN	s, how many report these cas	had been orted? <u>If</u> ses have	OTHER DNA EVIDENCE ONL	_	ossible with	out a co	mpreh	nensive re	view
selected for I possible, plea	DNA analysis, a se indicate howests.	materials and how many	s, how many many rep of these cas	had been orted? <u>If</u> ses have	OTHER DNA EVIDENCE ONL  Educated  6. Of your be	guess is not p	oe cases,	how ma	any c	of these	were <i>stil</i>
selected for I possible, plea unnamed susp	DNA analysis, a se indicate howests.	materials and how many  SCREEN	s, how many report these cas	had been orted? <u>If</u> ses have	OTHER DNA EVIDENCE ONL  Educated  6. Of your ba within the many of th	guess is not p  cklogged rag statute of linese will hav	oe cases, nitations a: e exceed	how mands of No	any c vemb	of these per 1, 20 te of limi	were <b>stil</b> 02? How tations by
selected for I possible, plea unnamed suspendent please unnamed unnamed suspendent please unnamed	DNA analysis, a se indicate howests.	materials and how many  SCREEN	s, how many report these cas	had been orted? <u>If</u> ses have	EVIDENCE  OTHER DNA EVIDENCE ONL  Educated  6. Of your ba within the many of th June 30, 2	guess is not p  cklogged rap statute of lim ese will hav 003? (If pos.	ne cases, nitations are exceed sible, plea	how mass of No ed the s	any c vemb	of these per 1, 20 te of limi	were <i>stil</i> 02? How tations by
selected for I possible, plea unnamed suspe	DNA analysis, ase indicate howests.  TOTAL SECT.	materials and how many  SCREEN	s, how many report these cas	had been orted? <u>If</u> ses have	EVIDENCE  OTHER DNA EVIDENCE ONL  Educated  6. Of your ba within the many of th June 30, 2	guess is not p  cklogged rag statute of linese will hav	ne cases, nitations are exceed sible, plea	how mass of No ed the s	any c vemb	of these per 1, 20 te of limi	were <i>stil</i> 02? How tations by
selected for I possible, plea unnamed suspendent suspen	DNA analysis, ase indicate howests.  TOTAL SECT.	materials and how many  SCREEN	s, how many report these cas	had been orted? <u>If</u> ses have	EVIDENCE  OTHER DNA EVIDENCE ONL  Educated  6. Of your ba within the many of th June 30, 2	guess is not p cklogged rap statute of lin ese will hav 003? (If pos-	pe cases, nitations as e exceed sible, plea ape cases.	how makes of No ed the see indical	any c vemb	of these per 1, 20 te of limi	were <i>stil</i> 02? How tations by
RAPE UNNAMED SUSPE UNNAMED SUSPE UNNAMED SUSPE PROPERTY CRIME	DNA analysis, ase indicate howests.  TOTAL  ECT	materials and how many  SCREEN	s, how many report these cas	had been orted? <u>If</u> ses have	EVIDENCE  OTHER DNA EVIDENCE ONL  Educated  6. Of your ba within the many of th June 30, 2 are unnam  WITHIN THE S	guess is not p cklogged rap statute of lin ese will hav 003? (If pos-	ne cases, nitations are exceeds sible, plea ape cases.	how makes of No ed the see indical	any c vemb	of these per 1, 20 te of limi	were <i>stil</i> 02? How tations by
RAPE UNNAMED SUSPE	DNA analysis, ase indicate howests.  TOTAL  ECT	materials and how many  SCREEN	s, how many report these cas	had been orted? <u>If</u> ses have	EVIDENCE  OTHER DNA EVIDENCE ONL  Educated  6. Of your ba within the many of th June 30, 2 are unnam  WITHIN THE S	guess is not p  cklogged rap statute of lin ese will hav 003? (If pos. ed suspect ra  TATUTE OF L	pe cases, nitations are exceed sible, plea ape cases.  IMITATION	how mass of No ed the see indicate)	any c vemb	of these per 1, 20 te of limi	were <i>stil</i> 02? How tations by
RAPE UNNAMED SUSPE UNNAMED SUSPE PROPERTY CRIME	DNA analysis, ase indicate howests.  TOTAL  ECT	materials and how many  SCREEN	s, how many report these cas	had been orted? <u>If</u> ses have	EVIDENCE  OTHER DNA EVIDENCE ONL  Educated  6. Of your ba within the many of th June 30, 2 are unnam  WITHIN THE S  UNNAMED  WILL EXCEED	guess is not p  cklogged rap statute of lin ese will hav 003? (If pos. ed suspect ra  TATUTE OF L	pe cases, nitations are exceeding pleasible, pleasible, pleasible, pleasible cases.  IMITATION  APES  E BY 06/30	how mass of No ed the see indicate)	any c vemb	of these per 1, 20 te of limi	were <i>stil</i> 02? How tations by
RAPE UNNAMED SUSPE UNNAMED SUSPE UNNAMED SUSPE UNNAMED SUSPE UNNAMED SUSPE	DNA analysis, ase indicate howests.  TOTAL  SECT  CT	materials and how many  SCREEN	s, how many report these cas	had been orted? <u>If</u> ses have	EVIDENCE  OTHER DNA EVIDENCE ONL  Educated  6. Of your ba within the many of th June 30, 2 are unnam  WITHIN THE S  UNNAMED  UNNAMED	guess is not p  cklogged rap statute of lim ese will hav 1003? (If pos. ed suspect ra  TATUTE OF L 0 SUSPECT R	pe cases, nitations are exceed sible, plea ape cases.  IMITATION  APES  E BY 06/30	how makes of No ed the see indical)	any covemb	of these er 1, 20 te of limi now man	were <b>still</b> 02? How tations by y of these
RAPE UNNAMED SUSPE STATEMENT SUSPE UNNAMED SUSPE UNNAMED SUSPE STATEMENT SUSPE STATEMENT SUSPE STATEMENT SUSPE STATEMENT SUSPE UNNAMED SUSPE STATEMENT SUSPE STATEME	DNA analysis, ase indicate howests.  TOTAL  SCT  CT  ECT  ECT  ECT  ECT  ECT  ECT	materials and how many  SCREEN QUEUE  nvestigat r labor	ANALYSIS QUEUE  ed by the aatory, on	REPORT QUEUE  agency or average evidence	EVIDENCE  OTHER DNA EVIDENCE ONL  Educated  6. Of your ba within the many of th June 30, 2 are unnam  WITHIN THE S  UNNAMED  UNNAMED  Educated  7. By June 3 backlog of possible,	guess is not p  cklogged rap statute of lim ese will hav 003? (If pos. ed suspect ra  TATUTE OF L  SUSPECT RA THE STATUT	pe cases, nitations are exceedsible, plea ape cases.  IMITATION  APES  TE BY 06/30  APES  at do you sawaiting ate how in the same cases.	how may of No ed the see indical)  Solution of the see indical of the see indicates of	comp	of these over 1, 20 the of limit ow many of the service of limit ow many of the service of the s	were still 02? How tations by y of these review size of ye analysis? be unnam
RAPE UNNAMED SUSPE OTHER UNNAMED SUSPE agencies ser approximately collected?	DNA analysis, ase indicate howests.  TOTAL  TOTAL  CT  CT  ECT  and homicides inviced by you what percentage	materials and how reals	ANALYSIS QUEUE  ed by the a atory, on have DNA	REPORT QUEUE  REPORT QUEUE  agency or average evidence	EVIDENCE  OTHER DNA EVIDENCE ONL  Educated  6. Of your ba within the many of th June 30, 2 are unnam  WITHIN THE S  UNNAMED  UNNAMED  Educated  7. By June 3 backlog of possible, suspect re	guess is not p  cklogged rap statute of lin ese will hav 003? (If pos. ed suspect ra  TATUTE OF L  SUSPECT R  THE STATUT  SUSPECT R  ted guess is not 10, 2003, whi of rape case please indic	pe cases, nitations are exceeds ape cases.  IMITATION  APES  TE BY 06/30  APES  at do you at do you s awaiting ate how in of all cases	how many of No many of Cinclude (included)	comp	of these over 1, 20 the of limit ow many of the service of limit ow many of the service of the s	were still 02? How tations by y of these review size of ye analysis? be unnam
RAPE UNNAMED SUSPE S. For all rapes agencies ser approximately	DNA analysis, ase indicate howests.  TOTAL  TOTAL  CT  CT  and homicides inviced by you what percentage  0% - 11% -	materials and how r many  SCREEN QUEUE  nivestigat r labor ge will  26% -	ed by the a atory, on have DNA	REPORT QUEUE  REPORT QUEUE  agency or average evidence	EVIDENCE  OTHER DNA EVIDENCE ONL  Educated  6. Of your ba within the many of th June 30, 2 are unnam  WITHIN THE S  UNNAMED  UNNAMED  Educated  7. By June 3 backlog of possible, suspect re ANTICIPA	guess is not p  cklogged rap statute of lin ese will hav 003? (If pos. ed suspect ra  TATUTE OF L  SUSPECT R  THE STATUT  SUSPECT R  ded guess is not 10, 2003, what for rape case please indic spe cases) C	pe cases, nitations are exceeds sible, plea ape cases.  IMITATION APES  E BY 06/30  APES  at do you s awaiting ate how in of all cases.	how many of No many of Cinclude (included)	comp	of these over 1, 20 the of limit ow many of the service of limit ow many of the service of the s	were still 02? How tations by y of these review size of ye analysis? be unnam
RAPE UNNAMED SUSPE	DNA analysis, ase indicate howests.  TOTAL  TOTAL  CT  CT  and homicides inviced by you what percentage  0% - 11% -	materials and how r many  SCREEN QUEUE  nivestigat r labor ge will  26% -	ed by the a atory, on have DNA	REPORT QUEUE  REPORT QUEUE  agency or average evidence	6. Of your bawithin the many of th June 30, 2 are unnam  WITHIN THE SUNNAMED  UNNAMED  UNNAMED  The Bucan of the June 30, 2 are unnam  WITHIN THE SUNNAMED  UNNAMED  ANTICIPA  UNNAMED  UNNAMED	guess is not p  cklogged rap statute of lin ese will hav 003? (If pos. ed suspect ra  TATUTE OF L  SUSPECT R  THE STATUT  SUSPECT R  ted guess is not for rape case please indic ape cases) C  TED RAPE CA	pe cases, nitations are exceeds sible, plea ape cases.  IMITATION  APES  TE BY 06/30  APES  at do you at do you s awaiting ate how in of all cases.  SE BACKL  CT RAPES	how may so of No ed the see indice )  See indice )  See indice in	comp	of these over 1, 20 the of limit ow many of the service of limit ow many of the service of the s	were still 02? How tations by y of these review size of y analysis? be unnan

8. What is the approximate length of time for analysis of a typical (non-priority) unnamed suspect case rape kit? Assume vaginal swab with one perpetrator, one victim, and that time runs from date the rape kit is received by the laboratory until results are reported.	13. For cases with <u>named suspects</u> in which the laboratory was asked to match the suspect profile to the forensic case profile, what is the number of suspects who were included as likely matches, and what is the number of suspects who were excluded?
NON-PRIORITY ANALYSIS TIME	INCLUSIONS EXCLUSIONS
Educated guess is not possible without a comprehensive review	NAMED SUSPECT
<ol><li>What is currently the output capacity for your laboratory in terms of completed DNA analysis on forensic cases? Use averages.</li></ol>	Educated guess is not possible without a comprehensive review
SAMPLES per MONTH / YEAR (choose one).	SECTION II: Mitochondrial DNA (mtDNA)
SAMPLES per CASE  If you have multiple laboratories, please use an average. If there is an aberration or gross difference at one (or more) of these laboratories, please omit that laboratory from the average, and explain below.	14. For calendar year 2002, approximately how many forensic cases analyzed by your laboratory will fail to yield testable amounts of DNA for STR analysis, but could <u>potentially</u> benefit from mitochondrial DNA testing?
explain below.	TOTAL CASES
	Educated guess is not possible without a comprehensive review
10. What is the likelihood that the agencies in your jurisdiction have rape kits in evidence lockers that are unaccounted for by your crime laboratory (10 cases or more)?	15. For calendar year 2002, approximately how many hair analysis cases will be received by your laboratory that could benefit from additional mitochondrial DNA testing?
☐ A CERTAINTY ☐ VERY LIKELY ☐ SOMEWHAT LIKELY	HAIR ANALYSIS CASES
□NOT POSSIBLE □ VERY UNLIKELY □ SOMEWHAT UNLIKELY	Educated guess is not possible without a comprehensive review
UNKNOWN	
11. In your opinion, do a majority of the local jurisdictions that are serviced by your laboratory(ies) routinely collect DNA evidence from scenes of property crime? Do the laboratories in your state system accept and process such evidence for DNA analysis if the suspect is unnamed?	SECTION III: Offender Profile Information  16. How many offender profiles were in your DNA database as of November 1, 2002? How many additional offender profiles do you expect to add by June 30, 2003?
ROUTINE DNA COLLECTION YES NO FROM PROPERTY CRIMES.	CURRENT OFFENDER PROFILES
LABORATORY/JEC\ ACCEPTO AND	EXPECTED OFFENDER PROFILES
PROCESSES UNNAMED SUSPECT DNA EVIDENCE FOR PROPERTY CRIMES.	Educated guess is not possible without a comprehensive review
LABORATORY(IES) ACCEPTS <b>BUT DOES</b> NOT PROCESS UNNAMED SUSPECT DNA EVIDENCE FOR PROPERTY CRIMES.  If you have multiple laboratories, and there is an aberration or gross	<b>17.</b> What was the size of your offender sample backlog as of November 1, 2002? What is your projected offender backlog as of June 30, 2003?
difference at one (or more) of these laboratories, please omit that laboratory from the answers above, and explain below (consider	CURRENT OFFENDER BACKLOG
elements in Question 12). Attach an additional sheet if needed.	PROJECTED OFFENDER BACKLOG
	Educated guess is not possible without a comprehensive review
	40 IF STATE IS NOT CURRENTLY AUTHORIZED TO COLLECT
12. If your laboratory(ies) does not process DNA evidence for unnamed suspect property crimes, what are the major obstacles to performing such tests? Please select all that apply and rate the top three ("1" being the most significant obstacle).	18. IF STATE IS NOT CURRENTLY AUTHORIZED TO COLLECT FROM ALL FELONS: Approximately how many additional samples would such legislation entail in the first year (assuming retroactive to currently incarcerated)? In subsequent years?
IES RATE APPLIES RATE	NEW SAMPLES 1st YEAR
LACK OF STORAGE SPACE LACK OF OPERATIONAL SPACE	ANNUAL SAMPLES IN FUTURE YEARS
■ MORE PERSONNEL NEEDED ■ PROSECUTION MUST REQUEST	Educated guess is not possible without a comprehensive review
■ BACKLOGS / OTHER PRIORITY ■ INVESTIGATOR MUST REQUEST CASES	19. Are offender DNA samples in your state taken at the time of
REAGENTS AND EQUIPMENT OTHER (Describe)	sentencing or intake into the corrections system, or some other time prior to release from prison? <i>Please check one.</i>
	SAMPLES TAKEN AT SENTENCING OR INTAKE SAMPLES TAKEN PRIOR TO RELEASE

20. From the time an offender DNA samp how long loops it take for the sample		25. Does your s fee) for DN mechanism?	NA <b>collec</b> Approxi	ction cos mately ho	ts? Wh w much w	at is the	funding ed in the
into CODIS?		most recent costs does the				n of your o	collection
OFFENDER PROCESSING TIME		FUNDING MEC	HANISM?		YES	NO	
Educated guess is not possible with	out a comprehensive review	MECHANISM DESCRIPTION					
SECTION IV: Costs / F	_	AMOUNT COLLECTED					
21. What is the approximate average cos suspect rape kit? Assume vaginal sw one victim.		% OF COLLECT FUNDED	TON				
COST OF RAPE KIT ANALYSIS		Educated gu	ess is not p	ossible with	nout a comp	rehensive re	eview
Educated guess is not possible without	a comprehensive review	26. Over the par	st 5 years	, approxir	nately who	at percenta	age of your
Factors included in the rape kit cost est	imate. Check all that apply.	Federal fund		ne nom	iocai iuni	iliy? Stat	e fullding?
REAGENTS OVERHEAD			0%-10%	11% - 25%	25% - 50%	50% - 75%	75% - 100%
EQUIPMENT DOTHER		LOCAL					
SALARIES		STATE					
		FEDERAL					
22. What general type of DNA sample of		TEDERAL					
offender DNA testing?  BUCCAL BLOOD  23. Where possible, please give a per following costs for processing offender  COLLECTION		_ = =	would you and rank to ARIES ERTIME	u hope to	use such	our labora funds? <i>Pl</i>	ntory's DNA ease select
EQUIPMENT	\$		JIPMENT				
REAGENTS	\$	ROE					
SALARIES	\$		AGENTS NSTRUCTION	N / I EASE	=		
STORAGE	\$	l	ENDER CO				
OTHER (Explain below)			SPECT CAS				
Educated guess is not possible without	a comprehensive review		TSOURCIN				
24. Does your state have a dedicated fund for DNA analysis? What is the me how much was collected in the most proportion of your DNA caseload does FUNDING MECHANISM?  MECHANISM DESCRIPTION  AMOUNT COLLECTED  % CASELOAD	echanism? Approximately recent fiscal year? What	A centerpiece of thorough docum through the use order to conduct the following criti  The number of Please include Offender's Offender's Offender's	f this asse- nentation of of existing this study ical inform of offende the follow qualifying of criminal his	essment, a soft success state crim we require ation.  er hits wing inform offense	and the restul suspense laborate your coo	ct identific ory DNA da operation in	ort, will be a ations made atabases. In

#### The number of forensic hits

Please include the following information for each hit Types of crimes linked

[Please indicate which matches involved other states, and which states those were.]

#### The number of investigations aided

Types of cases or investigations aided through CODIS

<u>Please use October 1, 2002 as your cut-off date</u> for compiling this information (ie, do not consider hits made after October 1). When in doubt, please use CODIS Hit Counting Guidelines as prepared by the Federal Bureau of Investigation.

\*\* If criminal histories are not readily available, please contact Lisa Hurst at (520) 664-0806 or <a href="mailto:lhurst@smithallinglane.com">lhurst@smithallinglane.com</a>.

Note: the information being requested from you does not need to identify either the offender or the victim of a crime. If you provide information that does contain such personally identifying information, this information will be deleted from our records during the course of data assembly and analysis. No personally identifying information will be published, and this research project will abide by all federal human subjects rules and procedures. A copy of the project's federal privacy certificate can be obtained if necessary for lawful release of this information. If there are any problems related to the transfer of the requested information, please contact Lisa Hurst as soon as possible at (520) 664-0806 or <a href="mailto:lhurst@smithallinglane.com">lhurst@smithallinglane.com</a>.

Please enclose this information on "DNA Matches" along with your completed assessment questionnaire.

#### **SECTION VI: Comments**

Please attach a sheet to this questionnaire to register any concerns or comments you wish to record regarding forensic DNA issues affecting the operation of your laboratory.

# **APPENDIX 2**

Calculations of Local Law Enforcement Backlog

	Α	В	С	D	E	F	G	Н	I	J	K	L	M
1	Agency Size		Sample Size	Responses		Murder Mean	( /	Murder/DNA Mean	Total Murders w/DNA	Unsolved Rape Mean		Rape/DNA Mean	Total Rapes w/DNA
2													
3	Over 100	984	984	586	59	61.11	60,132	17.69	17,407	255.02	250,940	117.02	115,148
4	50-100	780	512	287	52	2.89	2,254	1.19	928	17.73	13,829	11.64	9,079
5	25-50	1,575	968	476	48	1.81	2,851	0.82	1,292	37.52	59,094	2.58	4,064
6	Under 25	7,794	492	170	34	0.63	4,910	0.39	3,040	1.55	12,081	1.27	9,898
7	Unspecified	4,184	188	99	53	23.63	98,868	8.98	37,572	60.51	253,174	69.31	289,993
8	Tribal	275	194	54	28	0.84	231	0.39	107	0.97	267	0.53	146
9	Total	15,592	3,338	1,672			169,246		60,346		589,384		428,328
10													
	UCR Reports							Clearance Rate	Unsolved Murders		Clearance Rate	Unsolved Rapes	
12		1999					15,522	69	4,812	89,411	49	45,600	
13		2000					15,586	69.1	4,816	90,1`78	49.5	45,540	
14		2001					15,980	62.4	5,997	90,491	44.3	50,403	
15		Three-year Average					15,696	66.8	4,929	90,027	47.6	47,174	
16		10 Year Estimate							49,290			471,740	

# **APPENDIX 3a**

Local Law Enforcement Response Frequencies by Question

# <u>Appendix 3a – Law Enforcement Response Frequencies by Question</u>

### Agency size strata

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Over 100 Officers	586	34.6	35.0	35.0
	50-100 Officers	287	17.0	17.1	52.2
	25-50 Officers	476	28.1	28.4	80.6
	UNder 25 Officers	170	10.0	10.2	90.7
	Unspecified Size	99	5.8	5.9	96.7
	Tribal	56	3.3	3.3	100.0
	Total	1674	98.9	100.0	
Missing	999	19	1.1		
Total		1693	100.0		

### STATE

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	CO	1	.1	.1	.1
	999	3	.2	.2	.2
	AK	1	.1	.1	.3
	AL	27	1.6	1.6	1.9
	AR	19	1.1	1.1	3.0
	ΑZ	33	1.9	1.9	5.0
	CA	113	6.7	6.7	11.6
	CO	34	2.0	2.0	13.6
	CT	26	1.5	1.5	15.2
	DE	2	.1	.1	15.3
	FL	84	5.0	5.0	20.3
	GA	66	3.9	3.9	24.2
	HI	3	.2	.2	24.3
	IA	17	1.0	1.0	25.3
	ID	13	.8	.8	26.1
	IL	89	5.3	5.3	31.4
	IN	34	2.0	2.0	33.4
	KS	34	2.0	2.0	35.4
	KY	15	.9	.9	36.3
	LA	34	2.0	2.0	38.3
	MA	29	1.7	1.7	40.0
	MD	18	1.1	1.1	41.1
	ME	7	.4	.4	41.5
	MI	66	3.9	3.9	45.4
	MN	40	2.4	2.4	47.7
	MO	43	2.5	2.5	50.3

		•	•	•
MS	15	.9	.9	51.2
MT	9	.5	.5	51.7
NC	69	4.1	4.1	55.8
ND	6	.4	.4	56.1
NE	11	.6	.6	56.8
NH	5	.3	.3	57.1
NJ	72	4.3	4.3	61.3
NM	14	.8	.8	62.1
NV	7	.4	.4	62.6
NY	70	4.1	4.1	66.7
ОН	98	5.8	5.8	72.5
OH?	1	.1	.1	72.5
OK	30	1.8	1.8	74.3
OR	23	1.4	1.4	75.7
PA	46	2.7	2.7	78.4
RI	8	.5	.5	78.9
SC	23	1.4	1.4	80.2
SD	10	.6	.6	80.8
TN	32	1.9	1.9	82.7
TX	110	6.5	6.5	89.2
UT	15	.9	.9	90.1
VA	54	3.2	3.2	93.3
VT	5	.3	.3	93.6
WA	45	2.7	2.7	96.2
Wi	1	.1	.1	96.3
WI	47	2.8	2.8	99.1
WV	9	.5	.5	99.6
Wy	1	.1	.1	99.6
WY	6	.4	.4	100.0
Total	1693	100.0	100.0	

## Where are cases primarily sent for DNA analysis?

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	State Agency Lab	1331	78.6	80.1	80.1
	Local Agency Lab	195	11.5	11.7	91.8
	FBI Lab	19	1.1	1.1	93.0
	Other	69	4.1	4.2	97.1
	Private Lab/Commercial Lab	48	2.8	2.9	100.0
	Total	1662	98.2	100.0	
Missing	999	31	1.8		
Total		1693	100.0		

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	224	13.2	14.5	14.5
	11-25%	88	5.2	5.7	20.2
	25-50%	180	10.6	11.7	31.8
	50-75%	309	18.3	20.0	51.8
	75-100%	744	43.9	48.2	100.0
	Total	1545	91.3	100.0	
Missing	999	148	8.7		
Total		1693	100.0		

## % homicides with possible DNA

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	278	16.4	19.8	19.8
	11-25%	68	4.0	4.8	24.6
	25-50%	136	8.0	9.7	34.3
	50-75%	163	9.6	11.6	45.9
	75-100%	760	44.9	54.1	100.0
	Total	1405	83.0	100.0	
Missing	999	288	17.0		
Total		1693	100.0		

### educated guess not possible

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	151	8.9	9.0	9.0
	No	1535	90.7	91.0	100.0
	Total	1686	99.6	100.0	
Missing	999	7	.4		
Total		1693	100.0		

### unable to answer completely

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	38	2.2	2.3	2.3
	No	1649	97.4	97.7	100.0
	Total	1687	99.6	100.0	
Missing	999	6	.4		
Total		1693	100.0		

## unsolved homicides--total cases

	Frequency	Percent	Valid Percent	Cumulative Percent
Valid 0	395	23.3	31.7	31.7
1	216	12.8	17.3	49.1
2	150	8.9	12.0	61.1
3	87	5.1	7.0	68.1
4	63	3.7	5.1	73.2
5	46	2.7	3.7	76.9
6	38	2.2	3.1	79.9
7	22	1.3	1.8	81.7
8	18	1.1	1.4	83.1
9	8	.5	.6	83.8
10	28	1.7	2.2	86.0
11	9	.5	.7	86.7
12	13	.8	1.0	87.8
13	3	.2	.2	88.0
14	5	.3	.4	88.4
15	8	.5	.6	89.1
16	4	.2	.3	89.4
17	4	.2	.3	89.7
18	7	.4	.6	90.3
20	11	.6	.9	91.2
21	1	.1	.1	91.2
22	2	.1	.2	91.4
23	1	.1	.1	91.5
24	2	.1	.2	91.6
25	3	.2	.2	91.9
26	1	.1	.1	92.0
27	2	.1	.2	92.1
28	3	.2	.2	92.4
29	1	.1	.1	92.4
30	7	.4	.6	93.0
31	3	.2	.2	93.3
33	2	.1	.2	93.4
34	1	.1	.1	93.5
35	1	.1	.1	93.6
36	1	.1	.1	93.7
40	4	.2	.3	94.0
41	1	.1	.1	94.1
42	2	.1	.2	94.2
43	1	.1	.1	94.3
47	1	.1	.1	94.4
50	1	.1	.1	94.5
51	2	.1	.2	94.6
52	3	.2	.2	94.9
55	1	.1	.1	94.9
56	1	.1	.1	95.0
59	2	.1	.2	95.2

				. <u>-</u>
65	1	.1	.1	95.3
70	1	.1	.1	95.3
72	1	.1	.1	95.4
73	1	.1	.1	95.5
75	4	.2	.3	95.8
77	2	.1	.2	96.0
78	1	.1	.1	96.1
80	1	.1	.1	96.1
83	1	.1	.1	96.2
88	1	.1	.1	96.3
90	1	.1	.1	96.4
99	2	.1	.2	96.5
100	2	.1	.2	96.7
105	1	.1	.1	96.8
111	1	.1	.1	96.9
113	1	.1	.1	96.9
115	1	.1	.1	97.0
118	1	.1	.1	97.1
120	2	.1	.2	97.3
134	1	.1	.1	97.3
148	1	.1	.1	97.4
150	1	.1	.1	97.5
158	1	.1	.1	97.6
168	1	.1	.1	97.0
180	1	.1	.1	97.7
183	1	.1	.1	97.8 97.8
197				
200	1	.1	.1	97.9
201	2 1	.1	.2 .1	98.1
207		.1		98.2
210	1	.1	.1	98.2
216	1	.1	.1	98.3
220	1	.1	.1	98.4
250	1	.1	.1	98.5
256	2	.1	.2	98.6
	1	.1	.1	98.7
261 280	1	.1	.1	98.8
	1	.1	.1	98.9
300	2	.1	.2	99.0
339	1	.1	.1	99.1
350	1	.1	.1	99.2
380	1	.1	.1	99.3
500	1	.1	.1	99.4
650 650	1	.1	.1	99.4
653	1	.1	.1	99.5
721	1	.1	.1	99.6
750	1	.1	.1	99.7
800	1	.1	.1	99.8
880	1	.1	.1	99.8
1000	1	.1	.1	99.9

Ī	8500	1	.1	.1	100.0
	Total	1245	73.5	100.0	
Missing	999	447	26.4		
	System	1	.1		
	Total	448	26.5		
Total		1693	100.0		

## unsolved homicides--cases with DNA

				Cumulative
	Frequency	Percent	Valid Percent	Percent
Valid 0	430	25.4	46.1	46.1
1	201	11.9	21.6	67.7
2	87	5.1	9.3	77.0
3	39	2.3	4.2	81.2
4	27	1.6	2.9	84.1
5	22	1.3	2.4	86.5
6	13	.8	1.4	87.9
7	12	.7	1.3	89.2
8	7	.4	.8	89.9
9	5	.3	.5	90.5
10	20	1.2	2.1	92.6
11	1	.1	.1	92.7
12	7	.4	.8	93.5
13	1	.1	.1	93.6
14	2	.1	.2	93.8
15	5	.3	.5	94.3
16	4	.2	.4	94.7
18	4	.2	.4	95.2
20	5	.3	.5	95.7
23	1	.1	.1	95.8
24	1	.1	.1	95.9
25	3	.2	.3	96.2
30	8	.5	.9	97.1
33	1	.1	.1	97.2
35	1	.1	.1	97.3
40	2	.1	.2	97.5
50	3	.2	.3	97.9
55	1	.1	.1	98.0
80	1	.1	.1	98.1
85	1	.1	.1	98.2
99	1	.1	.1	98.3
100	2	.1	.2	98.5
120	1	.1	.1	98.6
126	1	.1	.1	98.7
130	1	.1	.1	98.8
150	2	.1	.2	99.0
155	1	.1	.1	99.1
170	1	.1	.1	99.2

	200	2	.1	.2	99.5
	352	1	.1	.1	99.6
	390	1	.1	.1	99.7
	400	2	.1	.2	99.9
	850	1	.1	.1	100.0
	Total	932	55.1	100.0	
Missing	999	761	44.9		
Total		1693	100.0		

## unsolved rapes--total cases

					Cumulative
		Frequency	Percent	Valid Percent	Percent
Valid	0	376	22.2	38.2	38.2
	1	92	5.4	9.3	47.5
	2	76	4.5	7.7	55.2
	3	43	2.5	4.4	59.6
	4	23	1.4	2.3	61.9
	5	35	2.1	3.6	65.5
	6	22	1.3	2.2	67.7
	7	9	.5	.9	68.6
	8	10	.6	1.0	69.6
	9	7	.4	.7	70.4
	10	44	2.6	4.5	74.8
	11	6	.4	.6	75.4
	12	10	.6	1.0	76.4
	13	2	.1	.2	76.6
	14	2	.1	.2	76.9
	15	17	1.0	1.7	78.6
	16	3	.2	.3	78.9
	17	2	.1	.2	79.1
	18	4	.2	.4	79.5
	20	27	1.6	2.7	82.2
	21	1	.1	.1	82.3
	22	2	.1	.2	82.5
	23	1	.1	.1	82.6
	24	2	.1	.2	82.8
	25	5	.3	.5	83.4
	27	2	.1	.2	83.6
	28	1	.1	.1	83.7
	30	10	.6	1.0	84.7
	31	2	.1	.2	84.9
	34	2	.1	.2	85.1
	35	2	.1	.2	85.3
	39	1	.1	.1	85.4
	40	5	.3	.5	85.9
	46	1	.1	.1	86.0
	48	1	.1	.1	86.1
	49	2	.1	.2	86.3

50	13	.8	1.3	87.6
54	1	.1	.1	87.7
55	1	.1	.1	87.8
59	1	.1	.1	87.9
60	2	.1	.2	88.1
62	1	.1	.1	88.2
65	3	.2	.3	88.5
66	2	.1	.2	88.7
67	1	.1	.1	88.8
70	3	.2	.3	89.1
75	3	.2	.3	89.4
80	7	.4	.7	90.2
83	1	.1	.1	90.3
90	2	.1	.2	90.5
99	2	.1	.2	90.7
100	12	.7	1.2	91.9
106	2	.1	.2	92.1
110	1	.1	.1	92.2
111	1	.1	.1	92.3
116	1	.1	.1	92.4
120	2	.1	.2	92.6
122	1	.1	.1	92.7
123	1	.1	.1	92.8
125	1	.1	.1	92.9
127	2	.1	.2	93.1
134	1	.1	.1	93.2
136	1	.1	.1	93.3
150	2	.1	.2	93.5
159	1	.1	.1	93.6
160	2	.1	.2	93.8
165	1	.1	.1	93.9
167	1	.1	.1	94.0
175	1	.1	.1	94.1
178	1	.1	.1	94.2
180	2	.1	.2	94.4
200	4	.2	.4	94.8
217	1	.1	.1	94.9
221	1	.1	.1	95.0
228	1	.1	.1	95.1
230	1	.1	.1	95.2
237	1	.1	.1	95.3
250	4	.2	.4	95.7
261	1	.1	.1	95.8
270	2	.1	.2	96.0
280	2	.1	.2	96.2
300	3	.1	.3	96.2
320	1	.1	.3	
400	1	.1	.1	96.6 96.8
440	1	.1	.1	96.9
I	'	.'	.'	90.9

	490	1	.1	.1	97.0
	539	1	.1	.1	97.1
	540	1	.1	.1	97.2
	578	1	.1	.1	97.3
	579	1	.1	.1	97.4
	600	2	.1	.2	97.6
	660	1	.1	.1	97.7
	683	1	.1	.1	97.8
	700	1	.1	.1	97.9
	711	1	.1	.1	98.0
	900	1	.1	.1	98.1
	923	1	.1	.1	98.2
	980	1	.1	.1	98.3
	1000	1	.1	.1	98.4
	1200	2	.1	.2	98.6
	1230	1	.1	.1	98.7
	1300	1	.1	.1	98.8
	1400	1	.1	.1	98.9
	2000	1	.1	.1	99.0
	2121	1	.1	.1	99.1
	2534	1	.1	.1	99.2
	2809	1	.1	.1	99.3
	3000	1	.1	.1	99.4
	4000	1	.1	.1	99.5
	5000	1	.1	.1	99.6
	5600	1	.1	.1	99.7
	7000	1	.1	.1	99.8
	9999	1	.1	.1	99.9
	10000	1	.1	.1	100.0
	Total	985	58.2	100.0	
Missing	999	708	41.8		
Total		1693	100.0		

### unsolved rapes--cases with DNA

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	275	16.2	36.8	36.8
	1	104	6.1	13.9	50.7
	2	62	3.7	8.3	59.0
	3	37	2.2	4.9	63.9
	4	25	1.5	3.3	67.2
	5	36	2.1	4.8	72.1
	6	16	.9	2.1	74.2
	7	5	.3	.7	74.9
	8	9	.5	1.2	76.1
	9	4	.2	.5	76.6
	10	19	1.1	2.5	79.1
	11	4	.2	.5	79.7

i ippenum su Ec	•			
12	4	.2	.5	80.2
13	2	.1	.3	80.5
14	4	.2	.5	81.0
15	13	.8	1.7	82.8
16	4	.2	.5	83.3
17	3	.2	.4	83.7
18	3	.2	.4	84.1
19	1	.1	.1	84.2
20	18	1.1	2.4	86.6
21	1	.1	.1	86.8
22	2	.1	.3	87.0
23	1	.1	.1	87.2
24	1	.1	.1	87.3
25	9	.5	1.2	88.5
26	1	.1	.1	88.6
30	3	.2	.4	89.0
31	1	.1	.1	89.2
35	1	.1	.1	89.3
37	1	.1	.1	89.4
40	6	.4	.8	90.2
43	1	.1	.1	90.4
46	1	.1	.1	90.5
50	7		.9	91.4
55		.4		
57	2	.1	.3	91.7
59	1	.1	.1	91.8
60	1	.1	.1	92.0
66	4	.2	.5	92.5
67	2	.1	.3	92.8
75	1	.1	.1	92.9
	2	.1	.3	93.2
80	1	.1	.1	93.3
87	1	.1	.1	93.4
88	1	.1	.1	93.6
90	2	.1	.3	93.9
98	1	.1	.1	94.0
99	6	.4	.8	94.8
100	5	.3	.7	95.5
120	1	.1	.1	95.6
150	2	.1	.3	95.9
175	2	.1	.3	96.1
180	1	.1	.1	96.3
192	1	.1	.1	96.4
200	2	.1	.3	96.7
215	1	.1	.1	96.8
220	1	.1	.1	96.9
250	3	.2	.4	97.3
300	1	.1	.1	97.5
360	1	.1	.1	97.6
450	2	.1	.3	97.9

Ī	478	1	.1	.1	98.0
	490	1	.1	.1	98.1
	492	1	.1	.1	98.3
	500	1	.1	.1	98.4
	749	1	.1	.1	98.5
	750	1	.1	.1	98.7
	850	1	.1	.1	98.8
	902	1	.1	.1	98.9
	1000	1	.1	.1	99.1
	1020	1	.1	.1	99.2
	1200	1	.1	.1	99.3
	1500	1	.1	.1	99.5
	1600	1	.1	.1	99.6
	2000	1	.1	.1	99.7
	3800	1	.1	.1	99.9
	5400	1	.1	.1	100.0
	Total	748	44.2	100.0	100.0
Missing	999	945	55.8	100.0	
Total	000	1693			
i Otal		1093	100.0		

### educated guess not possible

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	565	33.4	33.6	33.6
	No	1116	65.9	66.4	100.0
	Total	1681	99.3	100.0	
Missing	999	12	.7		
Total		1693	100.0		

### unable to answer completely

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	102	6.0	6.0	6.0
	No	1586	93.7	94.0	100.0
	Total	1688	99.7	100.0	
Missing	999	5	.3		
Total		1693	100.0		

## total unsolved rape cases with possible biological evidence

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	722	42.6	66.6	66.6
	1	69	4.1	6.4	73.0
	2	42	2.5	3.9	76.8
	3	26	1.5	2.4	79.2

11		1	1	3 2
4	24	1.4	2.2	81.5
5	27	1.6	2.5	83.9
6	16	.9	1.5	85.4
7	10	.6	.9	86.3
8	11	.6	1.0	87.4
9	7	.4	.6	88.0
10	19	1.1	1.8	89.8
11	6	.4	.6	90.3
12	3	.2	.3	90.6
13	1	.1	.1	90.7
15	8	.5	.7	91.4
16	3	.2	.3	91.7
17	2	.1	.2	91.9
18	2	.1	.2	92.1
19	1	.1	.1	92.2
20	11	.6	1.0	93.2
21	2	.1	.2	93.4
22	1	.1	.1	93.5
23	1	.1	.1	93.5
24	2	.1	.2	93.7
25	5	.3	.5	94.2
27	2	.1	.2	94.4
28	1	.1	.1	94.5
30	5	.3	.5	94.9
31	1	.1	.1	95.0
35	5	.3	.5	95.5
40	2	.1	.2	95.7
49	1	.1	.1	95.8
50	4	.2	.4	96.1
54	1	.1	.1	96.2
55	1	.1	.1	96.3
58	1	.1	.1	96.4
59	1	.1	.1	96.5
60	3	.2	.3	96.8
61	1	.1	.1	96.9
66	1	.1	.1	97.0
74	1	.1	.1	97.0
80	1	.1	.1	97.1
90	1	.1	.1	97.2
92	1	.1	.1	97.3
95	1	.1	.1	97.4
99	1	.1	.1	97.5
100	4	.2	.4	97.9
116	1	.1	.1	98.0
120	2	.1	.2	98.2
127	1	.1	.1	98.2
150	1	.1	.1	98.3
167	1	.1	.1	98.4
180	2	.1	.2	98.6
	. "			

		_	1	i.	
	192	1	.1	.1	98.7
	200	1	.1	.1	98.8
	250	1	.1	.1	98.9
	336	1	.1	.1	99.0
	350	1	.1	.1	99.1
	439	1	.1	.1	99.2
	573	1	.1	.1	99.3
	596	1	.1	.1	99.4
	600	2	.1	.2	99.5
	749	1	.1	.1	99.6
	1000	1	.1	.1	99.7
	1230	1	.1	.1	99.8
	3800	1	.1	.1	99.9
	7000	1	.1	.1	100.0
	Total	1084	64.0	100.0	
Missing	999	609	36.0		
Total		1693	100.0		

# total unsolved homicide cases with possible biological evidence

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	782	46.2	73.5	73.5
	1	104	6.1	9.8	83.3
	2	42	2.5	3.9	87.2
	3	30	1.8	2.8	90.0
	4	19	1.1	1.8	91.8
	5	15	.9	1.4	93.2
	6	9	.5	.8	94.1
	7	3	.2	.3	94.4
	8	3	.2	.3	94.6
	9	3	.2	.3	94.9
	10	8	.5	.8	95.7
	11	2	.1	.2	95.9
	12	4	.2	.4	96.2
	14	1	.1	.1	96.3
	15	3	.2	.3	96.6
	17	2	.1	.2	96.8
	20	3	.2	.3	97.1
	21	1	.1	.1	97.2
	25	3	.2	.3	97.5
	27	2	.1	.2	97.7
	30	2	.1	.2	97.8
	39	2	.1	.2	98.0
	40	1	.1	.1	98.1
	45	1	.1	.1	98.2
	49	1	.1	.1	98.3
	50	1	.1	.1	98.4
	60	1	.1	.1	98.5

	64	1	.1	.1	98.6
	75	1	.1	.1	98.7
	87	1	.1	.1	98.8
	99	2	.1	.2	99.0
	113	1	.1	.1	99.1
	115	1	.1	.1	99.2
	120	1	.1	.1	99.2
	157	1	.1	.1	99.3
	200	1	.1	.1	99.4
	250	1	.1	.1	99.5
	300	1	.1	.1	99.6
	400	1	.1	.1	99.7
	1000	1	.1	.1	99.8
	2000	1	.1	.1	99.9
	9999	1	.1	.1	100.0
	Total	1064	62.8	100.0	
Missing	999	629	37.2		
Total		1693	100.0		

## total unsolved property crimes with possible biological evidence

					Cumulative
		Frequency	Percent	Valid Percent	Percent
Valid	0	592	35.0	73.7	73.7
	1	27	1.6	3.4	77.1
	2	31	1.8	3.9	80.9
	3	16	.9	2.0	82.9
	4	7	.4	.9	83.8
	5	16	.9	2.0	85.8
	6	9	.5	1.1	86.9
	7	1	.1	.1	87.0
	8	1	.1	.1	87.2
	9	1	.1	.1	87.3
	10	25	1.5	3.1	90.4
	12	5	.3	.6	91.0
	14	1	.1	.1	91.2
	15	5	.3	.6	91.8
	17	1	.1	.1	91.9
	18	1	.1	.1	92.0
	20	10	.6	1.2	93.3
	25	5	.3	.6	93.9
	26	1	.1	.1	94.0
	30	4	.2	.5	94.5
	34	1	.1	.1	94.6
	35	1	.1	.1	94.8
	36	1	.1	.1	94.9
	40	2	.1	.2	95.1
	45	1	.1	.1	95.3
	50	11	.6	1.4	96.6

Ī	52	1	.1	.1	96.8
	60				
		1	.1	.1	96.9
	71	1	.1	.1	97.0
	90	2	.1	.2	97.3
	100	5	.3	.6	97.9
	200	3	.2	.4	98.3
	205	1	.1	.1	98.4
	250	1	.1	.1	98.5
	500	2	.1	.2	98.8
	576	1	.1	.1	98.9
	800	1	.1	.1	99.0
	1200	1	.1	.1	99.1
	1666	1	.1	.1	99.3
	2000	1	.1	.1	99.4
	2183	1	.1	.1	99.5
	2800	1	.1	.1	99.6
	3447	1	.1	.1	99.8
	6896	1	.1	.1	99.9
	150000	1	.1	.1	100.0
	Total	803	47.4	100.0	
Missing	999	889	52.5		
	System	1	.1		
	Total	890	52.6		
Total	- 1	1693	100.0		

# total unsolved "other" cases with possible biological evidence

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	426	25.2	90.3	90.3
	1	17	1.0	3.6	93.9
	2	7	.4	1.5	95.3
	3	1	.1	.2	95.6
	5	3	.2	.6	96.2
	6	1	.1	.2	96.4
	9	1	.1	.2	96.6
	10	4	.2	.8	97.5
	12	1	.1	.2	97.7
	20	1	.1	.2	97.9
	25	1	.1	.2	98.1
	30	1	.1	.2	98.3
	50	1	.1	.2	98.5
	99	1	.1	.2	98.7
	100	1	.1	.2	98.9
	104	1	.1	.2	99.2
	280	1	.1	.2	99.4
	396	1	.1	.2	99.6
	403	1	.1	.2	99.8
	100000	1	.1	.2	100.0

	Total	472	27.9	100.0	
Missing	999	1220	72.1		
	System	1	.1		
	Total	1221	72.1		
Total		1693	100.0		

# total unsolved unnamed rape cases with possible biological evidence

	Frequency	Percent	Valid Percent	Cumulative Percent
Valid 0	352	20.8	53.2	53.2
1	65	3.8	9.8	63.0
2	53	3.1	8.0	71.0
3	24	1.4	3.6	74.6
4	24	1.4	3.6	78.2
5	29	1.7	4.4	82.6
6	10	.6	1.5	84.1
7	3	.2	.5	84.6
8	6	.4	.9	85.5
9	1	.1	.2	85.6
10	19	1.1	2.9	88.5
11	4	.2	.6	89.1
12	3	.2	.5	89.6
13	6	.4	.9	90.5
14	1	.1	.2	90.6
15	7	.4	1.1	91.7
17	1	.1	.2	91.8
19	1	.1	.2	92.0
20	6	.4	.9	92.9
21	2	.1	.3	93.2
22	2	.1	.3	93.5
23	2	.1	.3	93.8
24	1	.1	.2	94.0
25	5	.3	.8	94.7
27	1	.1	.2	94.9
30	4	.2	.6	95.5
35	1	.1	.2	95.6
40	2	.1	.3	95.9
50	3	.2	.5	96.4
58	1	.1	.2	96.5
60	2	.1	.3	96.8
68	1	.1	.2	97.0
70	1	.1	.2	97.1
80	3	.2	.5	97.6
85	1	.1	.2	97.7
90	1	.1	.2	97.9
99	1	.1	.2	98.0
100	1	.1	.2	98.2
120	2	.1	.3	98.5

	150	2	.1	.3	98.8
	183	1	.1	.2	98.9
	200	1	.1	.2	99.1
	300	1	.1	.2	99.2
	315	1	.1	.2	99.4
	350	1	.1	.2	99.5
	749	1	.1	.2	99.7
	1000	1	.1	.2	99.8
	3000	1	.1	.2	100.0
	Total	662	39.1	100.0	
Missing	999	1030	60.8		
	System	1	.1		
	Total	1031	60.9		
Total		1693	100.0		

# total unsolved unnamed homicide cases with possible biological evidence

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	382	22.6	60.5	60.5
	1	95	5.6	15.1	75.6
	2	34	2.0	5.4	81.0
	3	26	1.5	4.1	85.1
	4	18	1.1	2.9	88.0
	5	14	.8	2.2	90.2
	6	6	.4	1.0	91.1
	7	7	.4	1.1	92.2
	8	3	.2	.5	92.7
	9	2	.1	.3	93.0
	10	9	.5	1.4	94.5
	12	3	.2	.5	94.9
	13	1	.1	.2	95.1
	15	3	.2	.5	95.6
	16	1	.1	.2	95.7
	19	1	.1	.2	95.9
	20	3	.2	.5	96.4
	21	1	.1	.2	96.5
	22	1	.1	.2	96.7
	24	1	.1	.2	96.8
	25	2	.1	.3	97.1
	30	1	.1	.2	97.3
	35	1	.1	.2	97.5
	39	2	.1	.3	97.8
	40	1	.1	.2	97.9
	50	1	.1	.2	98.1
	54	1	.1	.2	98.3
	60	1	.1	.2	98.4
	63	1	.1	.2	98.6
	64	1	.1	.2	98.7

	65	1	.1	.2	98.9
	80	1	.1	.2	99.0
	120	1	.1	.2	99.2
	150	1	.1	.2	99.4
	175	1	.1	.2	99.5
	200	3	.2	.5	100.0
	Total	631	37.3	100.0	
Missing	999	1062	62.7		
Total		1693	100.0		

# total unsolved unnamed property crimes with possible biological evidence

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	295	17.4	61.7	61.7
valia	1	293	17.4	4.6	66.3
	2	28	1.3	5.9	72.2
	3	14	.8	2.9	72.2 75.1
	4	7	.o .4	1.5	76.6
	5	15	.9	3.1	70.0
	6	6	.4	1.3	81.0
	7	2	.1	.4	81.4
	8	1	.1	.2	81.6
	9	3	.2	.6	82.2
	10	21	1.2	4.4	86.6
	12	4	.2	.8	87.4
	14	1	.1	.2	87.7
	15	9	.5	1.9	89.5
	18	1	.1	.2	89.7
	20	6	.4	1.3	91.0
	25	3	. <del>-</del> .2	.6	91.6
	26	1	.1	.2	91.8
	30	4	.2	.8	92.7
	36	1	.1	.2	92.9
	40	3	.2	.6	93.5
	45	1	.1	.2	93.7
	50	5	.3	1.0	94.8
	54	1	.1	.2	95.0
	60	1	.1	.2	95.2
	70	1	.1	.2	95.4
	80	1	.1	.2	95.6
	99	1	.1	.2	95.8
	100	6	.4	1.3	97.1
	102	1	.1	.2	97.3
	200	3	.2	.6	97.9
	250	1	.1	.2	98.1
	450	1	.1	.2	98.3
	479	1	.1	.2	98.5
	700	1	.1	.2	98.7

	1265	1	.1	.2	99.0
	1381	1	.1	.2	99.2
	1479	1	.1	.2	99.4
	2500	1	.1	.2	99.6
	3010	1	.1	.2	99.8
	135000	1	.1	.2	100.0
	Total	478	28.2	100.0	
Missing	999	1215	71.8		
Total		1693	100.0		

# total unsolved unnamed "other" cases with possible biological evidence

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	239	14.1	85.1	85.1
	1	11	.6	3.9	89.0
	2	7	.4	2.5	91.5
	4	2	.1	.7	92.2
	5	4	.2	1.4	93.6
	8	1	.1	.4	94.0
	9	1	.1	.4	94.3
	10	2	.1	.7	95.0
	15	1	.1	.4	95.4
	25	1	.1	.4	95.7
	50	2	.1	.7	96.4
	99	3	.2	1.1	97.5
	100	2	.1	.7	98.2
	158	1	.1	.4	98.6
	365	1	.1	.4	98.9
	9999	2	.1	.7	99.6
	50000	1	.1	.4	100.0
	Total	281	16.6	100.0	
Missing	999	1412	83.4		
Total		1693	100.0		

# educated guess not possible

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	538	31.8	31.9	31.9
	No	1149	67.9	68.1	100.0
	Total	1687	99.6	100.0	
Missing	999	6	.4		
Total		1693	100.0		

unable to answer completely

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	87	5.1	5.2	5.2
	No	1602	94.6	94.8	100.0
	Total	1689	99.8	100.0	
Missing	999	4	.2		
Total		1693	100.0		

# % cases with rape kits only

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	399	23.6	46.3	46.3
	11-25%	40	2.4	4.6	51.0
	25-50%	61	3.6	7.1	58.1
	50-75%	97	5.7	11.3	69.3
	75-100%	264	15.6	30.7	100.0
	Total	861	50.9	100.0	
Missing	999	831	49.1		
	System	1	.1		
	Total	832	49.1		
Total		1693	100.0		

# % cases with rape kits and other DNA evidence

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	434	25.6	54.7	54.7
	11-25%	73	4.3	9.2	63.9
	25-50%	71	4.2	9.0	72.9
	50-75%	81	4.8	10.2	83.1
	75-100%	133	7.9	16.8	99.9
	9999	1	.1	.1	100.0
	Total	793	46.8	100.0	
Missing	999	900	53.2		
Total		1693	100.0		

% cases	with	other	DNA	evidence	only
---------	------	-------	-----	----------	------

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	519	30.7	81.6	81.6
	11-25%	42	2.5	6.6	88.2
	25-50%	27	1.6	4.2	92.5
	50-75%	15	.9	2.4	94.8
	75-100%	33	1.9	5.2	100.0
	Total	636	37.6	100.0	
Missing	999	1057	62.4		
Total		1693	100.0		

# educated guess not possible

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	399	23.6	23.7	23.7
	No	1286	76.0	76.3	100.0
	Total	1685	99.5	100.0	
Missing	999	8	.5		
Total		1693	100.0		

#### unable to answer completely

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	66	3.9	3.9	3.9
	No	1621	95.7	96.1	100.0
	Total	1687	99.6	100.0	
Missing	999	6	.4		
Total		1693	100.0		

#### rape cases within statute of limitations

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	301	17.8	46.0	46.0
	1	93	5.5	14.2	60.2
	2	53	3.1	8.1	68.3
	3	24	1.4	3.7	72.0
	4	18	1.1	2.8	74.8
	5	26	1.5	4.0	78.7
	6	16	.9	2.4	81.2
	7	9	.5	1.4	82.6
	8	6	.4	.9	83.5
	9	4	.2	.6	84.1
	10	19	1.1	2.9	87.0

- ipponus		_			
	11	5	.3	.8	87.8
	12	1	.1	.2	87.9
	13	3	.2	.5	88.4
	14	2	.1	.3	88.7
	15	7	.4	1.1	89.8
	16	3	.2	.5	90.2
	17	1	.1	.2	90.4
	18	1	.1	.2	90.5
	19	1	.1	.2	90.7
	20	4	.2	.6	91.3
	21	4	.2	.6	91.9
	22	1	.1	.2	92.0
	23	1	.1	.2	92.2
	24	2	.1	.3	92.5
	25	3	.2	.5	93.0
	27	1	.1	.2	93.1
	30	5	.3	.8	93.9
	35	3	.2	.5	94.3
	36	1	.1	.2	94.5
	40	2	.1	.3	94.8
	50	5	.3	.8	95.6
	59	1	.1	.2	95.7
	61	1	.1	.2	95.9
	70	1	.1	.2	96.0
	80	1	.1	.2	96.2
	92	1	.1	.2	96.3
	95	2	.1	.3	96.6
	99	1	.1	.2	96.8
	100	2	.1	.3	97.1
	111	1	.1	.2	97.1
	116	1	.1	.2	97.4
	120	3	.2	.5	97.9
	140	1	.1	.2	98.0
	190	1	.1	.2	98.2
	250	1	.1	.2	98.3
	350	1	.1	.2	98.5
	500				
	520	1	.1	.2	98.6 98.8
	525	1 1	.1	.2	
	530		.1		98.9
	533	1	.1	.2	99.1
	600	1	.1	.2	99.2
	738	1	.1	.2	99.4
	1000	1	.1	.2	99.5
	5000	1	.1	.2	99.7
	7000	1	.1	.2	99.8
	Total	1	.1	.2	100.0
Micoina		654	38.6	100.0	
Missing Total	999	1039	61.4		
TOLAI		1693	100.0		

unnamed rape cases within statute of limitations

	Frequency	Percent	Valid Percent	Cumulative Percent
Valid 0	262	15.5	48.3	48.3
1	80	4.7	14.7	63.0
2	51	3.0	9.4	72.4
3	25	1.5	4.6	77.0
4	14	.8	2.6	79.6
5	20	1.2	3.7	83.2
6	10	.6	1.8	85.1
7	7	.4	1.3	86.4
8	7	.4	1.3	87.7
9	1	.1	.2	87.8
10	11	.6	2.0	89.9
11	2	.1	.4	90.2
12	2	.1	.4	90.6
13	5	.3	.9	91.5
14	2	.1	.4	91.9
15	7	.4	1.3	93.2
17	1	.1	.2	93.4
20	2	.1	.4	93.7
21	2	.1	.4	94.1
22	1	.1	.2	94.3
23	1	.1	.2	94.5
24	1	.1	.2	94.7
25	3	.2	.6	95.2
27	1	.1	.2	95.4
30	3	.2	.6	95.9
40	1	.1	.2	96.1
42	1	.1	.2	96.3
50	2	.1	.4	96.7
58	1	.1	.2	96.9
60	1	.1	.2	97.1
70	1	.1	.2	97.2
80	1	.1	.2	97.4
100	1	.1	.2	97.6
120	1	.1	.2	97.8
130	1	.1	.2	98.0
140	1	.1	.2	98.2
180	1	.1	.2	98.3
200	2	.1	.4	98.7
250	1	.1	.2	98.9
500	1	.1	.2	99.1
533	1	.1	.2	99.3
600	1	.1	.2	99.4
738	1	.1	.2	99.6
3000	1	.1	.2	99.8

	7000	1	.1	.2	100.0
	Total	543	32.1	100.0	
Missing	999	1150	67.9		
Total		1693	100.0		

# rape cases to exceed statute of limitations

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	413	24.4	76.3	76.3
	1	31	1.8	5.7	82.1
	2	22	1.3	4.1	86.1
	3	11	.6	2.0	88.2
	4	5	.3	.9	89.1
	5	14	.8	2.6	91.7
	6	4	.2	.7	92.4
	7	1	.1	.2	92.6
	8	2	.1	.4	93.0
	9	2	.1	.4	93.3
	10	6	.4	1.1	94.5
	11	3	.2	.6	95.0
	12	1	.1	.2	95.2
	13	1	.1	.2	95.4
	15	5	.3	.9	96.3
	18	2	.1	.4	96.7
	20	1	.1	.2	96.9
	30	2	.1	.4	97.2
	33	1	.1	.2	97.4
	40	1	.1	.2	97.6
	45	1	.1	.2	97.8
	46	1	.1	.2	98.0
	58	1	.1	.2	98.2
	70	2	.1	.4	98.5
	100	1	.1	.2	98.7
	106	1	.1	.2	98.9
	110	1	.1	.2	99.1
	375	1	.1	.2	99.3
	705	1	.1	.2	99.4
	738	1	.1	.2	99.6
	9999	2	.1	.4	100.0
	Total	541	32.0	100.0	
Missing	999	1152	68.0		
Total		1693	100.0		

unnamed rape cases to exceed statute of limitations

					Cumulative
		Frequency	Percent	Valid Percent	Percent
Valid	0	351	20.7	76.1	76.1
	1	21	1.2	4.6	80.7
	2	23	1.4	5.0	85.7
	3	11	.6	2.4	88.1
	4	4	.2	.9	88.9
	5	9	.5	2.0	90.9
	6	4	.2	.9	91.8
	8	3	.2	.7	92.4
	10	5	.3	1.1	93.5
	11	2	.1	.4	93.9
	12	1	.1	.2	94.1
	15	3	.2	.7	94.8
	17	1	.1	.2	95.0
	24	1	.1	.2	95.2
	26	1	.1	.2	95.4
	30	2	.1	.4	95.9
	35	1	.1	.2	96.1
	36	1	.1	.2	96.3
	45	1	.1	.2	96.5
	55	1	.1	.2	96.7
	70	1	.1	.2	97.0
	99	12	.7	2.6	99.6
	200	1	.1	.2	99.8
	738	1	.1	.2	100.0
	Total	461	27.2	100.0	
Missing	999	1231	72.7		
	System	1	.1		
	Total	1232	72.8		
Total		1693	100.0		

# educated guess not possible

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	502	29.7	29.8	29.8
	No	1182	69.8	70.2	100.0
	Total	1684	99.5	100.0	
Missing	999	9	.5		
Total		1693	100.0		

# unable to answer completely

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	73	4.3	4.3	4.3
	No	1613	95.3	95.7	100.0
	Total	1686	99.6	100.0	
Missing	999	7	.4		
Total		1693	100.0		

#### unanalyzed evidence stored--central storage area

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	1332	78.7	79.0	79.0
	No	355	21.0	21.0	100.0
	Total	1687	99.6	100.0	
Missing	999	6	.4		
Total		1693	100.0		

#### unanalyzed evidence stored--decentralized storage area

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	52	3.1	3.1	3.1
	No	1636	96.6	96.9	100.0
	Total	1688	99.7	100.0	
Missing	999	5	.3		
Total		1693	100.0		

#### unanalyzed evidence stored--prosecutor's facility

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	34	2.0	2.0	2.0
	No	1654	97.7	98.0	100.0
	Total	1688	99.7	100.0	
Missing	999	5	.3		
Total		1693	100.0		

# unanalyzed evidence stored--crime lab facility

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	374	22.1	22.2	22.2
	No	1314	77.6	77.8	100.0
	Total	1688	99.7	100.0	
Missing	999	5	.3		
Total		1693	100.0		

#### unanalyzed evidence stored--other

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	94	5.6	5.6	5.6
	No	1592	94.0	94.4	100.0
	Total	1686	99.6	100.0	
Missing	999	6	.4		
	System	1	.1		
	Total	7	.4		
Total		1693	100.0		

# sufficient space for long-term evidence storage?

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Sufficient Storage	649	38.3	39.0	39.0
	Insufficient Storage	1013	59.8	61.0	100.0
	Total	1662	98.2	100.0	
Missing	999	31	1.8		
Total		1693	100.0		

# If insufficient, how critical is additional space?

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Highly Critical	241	14.2	18.0	18.0
	Critical	700	41.3	52.3	70.3
	Pressing But Not Critical	397	23.4	29.7	100.0
	Total	1338	79.0	100.0	
Missing	999	355	21.0		
Total		1693	100.0		

# DNA not sent to lab applicable--not applicable

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	626	37.0	37.2	37.2
	No	1055	62.3	62.8	100.0
	Total	1681	99.3	100.0	
Missing	999	12	.7		
Total		1693	100.0		

# DNA not sent to lab rank--not applicable

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	340	20.1	89.2	89.2
Second Most Significant	Second Most Significant	17	1.0	4.5	93.7
	Third Most Significant	24	1.4	6.3	100.0
	Total	381	22.5	100.0	
Missing	999	1312	77.5		
Total		1693	100.0		

# DNA not sent to lab applicable--no suspect identified

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	406	24.0	24.1	24.1
	No	1278	75.5	75.9	100.0
	Total	1684	99.5	100.0	
Missing	999	9	.5		
Total		1693	100.0		

# DNA not sent to lab rank--no suspect identified

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	234	13.8	65.9	65.9
	Second Most Significant	75	4.4	21.1	87.0
	Third Most Significant	46	2.7	13.0	100.0
	Total	355	21.0	100.0	
Missing	999	1338	79.0		
Total		1693	100.0		

# DNA not sent to lab applicable--no suspect charged

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	135	8.0	8.0	8.0
	No	1549	91.5	92.0	100.0
	Total	1684	99.5	100.0	
Missing	999	9	.5		
Total		1693	100.0		

# DNA not sent to lab rank--no suspect charged

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	36	2.1	33.6	33.6
	Second Most Significant	47	2.8	43.9	77.6
	Third Most Significant	24	1.4	22.4	100.0
	Total	107	6.3	100.0	
Missing	999	1586	93.7		
Total		1693	100.0		

# DNA not sent to lab applicable--guilty plea anticipated

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	194	11.5	11.5	11.5
	No	1491	88.1	88.5	100.0
	Total	1685	99.5	100.0	
Missing	999	8	.5		
Total		1693	100.0		

# DNA not sent to lab rank--guilty plea anticipated

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	54	3.2	34.0	34.0
	Second Most Significant	54	3.2	34.0	67.9
Third Most Significant	51	3.0	32.1	100.0	
	Total	159	9.4	100.0	
Missing	999	1534	90.6		
Total		1693	100.0		

# DNA not sent to lab applicable--DNA use uncertain

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	93	5.5	5.5	5.5
	No	1592	94.0	94.5	100.0
	Total	1685	99.5	100.0	
Missing	999	8	.5		
Total		1693	100.0		

#### DNA not sent to lab rank--DNA use uncertain

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	15	.9	19.7	19.7
	Second Most Significant	29	1.7	38.2	57.9
	Third Most Significant	32	1.9	42.1	100.0
	Total	76	4.5	100.0	
Missing	999	1617	95.5		
Total		1693	100.0		

# DNA not sent to lab applicable--lack of \$

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	112	6.6	6.6	6.6
	No	1573	92.9	93.4	100.0
	Total	1685	99.5	100.0	
Missing	999	8	.5		
Total		1693	100.0		

#### DNA not sent to lab rank--lack of \$

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	45	2.7	51.7	51.7
	Second Most Significant	21	1.2	24.1	75.9
	Third Most Significant	21	1.2	24.1	100.0
	Total	87	5.1	100.0	
Missing	999	1606	94.9		
Total		1693	100.0		

DNA not sent to lab	applicableinability	to aet timely	results
---------------------	---------------------	---------------	---------

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	130	7.7	7.7	7.7
	No	1555	91.8	92.3	100.0
	Total	1685	99.5	100.0	
Missing	999	8	.5		
Total		1693	100.0		

# DNA not sent to lab rank--inability to get timely results

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	35	2.1	32.1	32.1
	Second Most Significant	50	3.0	45.9	78.0
Third Most Significant		24	1.4	22.0	100.0
	Total	109	6.4	100.0	
Missing	999	1584	93.6		
Total		1693	100.0		

# DNA not sent to lab applicable--lab not processing requests

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	49	2.9	2.9	2.9
	No	1636	96.6	97.1	100.0
	Total	1685	99.5	100.0	
Missing	999	8	.5		
Total		1693	100.0		

# DNA not sent to lab rank--lab not processing requests

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	16	.9	40.0	40.0
	Second Most Significant	10	.6	25.0	65.0
Th	Third Most Significant	14	.8	35.0	100.0
	Total	40	2.4	100.0	
Missing	999	1653	97.6		
Total		1693	100.0		

# DNA not sent to lab applicable--uncertain where to send DNA

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	16	.9	.9	.9
	No	1669	98.6	99.1	100.0
	Total	1685	99.5	100.0	
Missing	999	8	.5		
Total		1693	100.0		

#### DNA not sent to lab rank--uncertain where to send DNA

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	5	.3	50.0	50.0
Second Most Significant		2	.1	20.0	70.0
	Third Most Significant	3	.2	30.0	100.0
	Total	10	.6	100.0	
Missing	999	1683	99.4		
Total		1693	100.0		

# DNA not sent to lab applicable--analysis not requested by prosecutor

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	120	7.1	7.1	7.1
	No	1565	92.4	92.9	100.0
	Total	1685	99.5	100.0	
Missing	999	8	.5		
Total		1693	100.0		

# DNA not sent to lab rank--analysis not requested by prosecutor

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	18	1.1	18.6	18.6
	Second Most Significant	34	2.0	35.1	53.6
	Third Most Significant	44	2.6	45.4	99.0
	9999	1	.1	1.0	100.0
	Total	97	5.7	100.0	
Missing	999	1596	94.3		
Total		1693	100.0		

# DNA not sent to lab applicable--other

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	124	7.3	7.4	7.4
	No	1559	92.1	92.6	100.0
	Total	1683	99.4	100.0	
Missing	999	10	.6		
Total		1693	100.0		

#### DNA not sent to lab rank--other

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	59	3.5	61.5	61.5
	Second Most Significant	24	1.4	25.0	86.5
T	Third Most Significant	13	.8	13.5	100.0
	Total	96	5.7	100.0	
Missing	999	1597	94.3		
Total		1693	100.0		

# Department routinely collect DNA from property crimes?

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	582	34.4	35.7	35.7
	No	1046	61.8	64.3	100.0
	Total	1628	96.2	100.0	
Missing	999	65	3.8		
Total		1693	100.0		

# Lab accept AND process unnamed suspect DNA cases for prop. crimes?

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	666	39.3	47.4	47.4
	No	738	43.6	52.6	100.0
	Total	1404	82.9	100.0	
Missing	999	289	17.1		
Total		1693	100.0		

# Lab accept but NOT process unnamed suspect DNA cases for prop. crimes?

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	311	18.4	27.3	27.3
	No	829	49.0	72.7	100.0
	Total	1140	67.3	100.0	
Missing	999	553	32.7		
Total		1693	100.0		

#### unable to answer completely

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	105	6.2	6.2	6.2
	No	1579	93.3	93.8	100.0
	Total	1684	99.5	100.0	
Missing	999	9	.5		
Total		1693	100.0		

# agency have cold case squad?

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	377	22.3	22.8	22.8
	No	1278	75.5	77.2	100.0
	Total	1655	97.8	100.0	
Missing	999	38	2.2		
Total		1693	100.0		

# unable to answer completely

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	17	1.0	1.0	1.0
	No	1666	98.4	99.0	100.0
	Total	1683	99.4	100.0	
Missing	999	10	.6		
Total		1693	100.0		

# # of cold cases eligible for review

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	29	1.7	9.4	9.4
	1	35	2.1	11.3	20.7
	2	33	1.9	10.7	31.4
	3	26	1.5	8.4	39.8

4         16         .9         5.2         45.0           5         16         .9         5.2         50.2           6         13         .8         4.2         54.4           7         7         4         2.3         56.6           8         14         .8         4.5         61.2           9         5         .3         1.6         62.8           10         14         .8         4.5         67.3           11         2         .1         .6         68.0           12         5         .3         1.6         69.6           13         1         .1         .3         70.2           14         1         .1         .3         70.2           15         4         .2         1.3         71.5           16         3         .2         1.0         72.5           17         2         .1         .6         73.1           18         2         .1         .6         73.8           20         6         .4         1.9         75.7           22         2         .1         .6         76.4     <	i ippenant su Ec	Emoreem	ent respon	ise i requeirer	es of Question
6         13         .8         4.2         54.4           7         7         .4         2.3         56.6           8         14         .8         4.5         66.2           9         .5         .3         1.6         62.8           10         14         .8         4.5         67.3           11         2         .1         .6         68.0           12         5         .3         1.6         69.6           13         1         .1         .3         69.9           13         1         .1         .3         70.2           15         4         .2         1.3         71.5           16         3         .2         1.0         72.5           17         2         .1         .6         73.1           18         2         .1         .6         73.1           18         2         .1         .6         73.1           18         2         .1         .6         76.4           23         1         .1         .3         76.7           24         1         .1         .3         78.3 </td <td></td> <td>16</td> <td>.9</td> <td>5.2</td> <td>45.0</td>		16	.9	5.2	45.0
7         7         4         2.3         56.6           8         14         8         4.5         61.2           9         5         .3         1.6         62.8           10         14         .8         4.5         67.3           11         2         .1         .6         68.0           12         5         .3         1.6         69.6           13         1         .1         .3         70.2           15         4         .2         1.3         71.5           16         3         .2         1.0         72.5           17         2         .1         .6         73.1           18         2         .1         .6         73.8           20         6         .4         1.9         75.7           22         2         .1         .6         76.4           23         1         .1         .3         76.7           24         1         .1         .3         78.3           28         1         .1         .3         78.0           30         4         .2         1.3         78.9 <td>5</td> <td>16</td> <td>.9</td> <td>5.2</td> <td>50.2</td>	5	16	.9	5.2	50.2
8         14         .8         4.5         61.2           9         5         .3         1.6         62.8           10         14         .8         4.5         67.3           11         2         .1         .6         68.0           12         5         .3         1.6         69.6           13         1         .1         .3         69.9           14         1         .1         .3         70.2           15         4         .2         1.3         71.5           16         3         .2         1.0         72.5           17         .2         1         .6         73.1           18         .2         .1         .6         73.8           20         .6         .4         1.9         75.7           22         .2         .1         .6         76.4           23         .1         .1         .3         76.7           24         .1         .1         .3         78.3           30         .4         .2         .1.3         79.9           33         .1         .1         .3         80.3 <td>6</td> <td>13</td> <td>.8</td> <td>4.2</td> <td>54.4</td>	6	13	.8	4.2	54.4
9         5         .3         1.6         62.8           10         14         .8         4.5         67.3           11         2         .1         .6         68.0           12         5         .3         1.6         69.0           13         1         .1         .3         69.9           14         1         .1         .3         70.2           15         4         .2         1.3         71.5           16         3         .2         1.0         72.5           17         2         .1         .6         73.1           18         2         .1         .6         73.1           18         2         .1         .6         73.1           18         2         .1         .6         73.1           18         2         .1         .6         73.1           18         2         .1         .6         73.1           18         2         .1         .6         76.4           23         1         .1         .3         76.7           24         1         .1         .3         80.3		7	.4	2.3	56.6
10         14         .8         4.5         67.3           11         2         .1         .6         68.0           12         5         .3         1.6         69.6           13         1         .1         .3         69.9           14         1         .1         .3         70.2           15         4         .2         1.3         71.5           16         3         .2         1.0         72.5           17         2         .1         .6         73.1           18         2         .1         .6         73.8           20         6         .4         1.9         .75.7           22         2         .1         .6         .76.4           23         1         .1         .3         .76.7           24         1         .1         .3         .76.7           24         1         .1         .3         .78.6           30         4         .2         .1.3         .78.3           33         1         .1         .3         .80.3           34         1         .1         .3         .80.3 <td>8</td> <td>14</td> <td>.8</td> <td>4.5</td> <td>61.2</td>	8	14	.8	4.5	61.2
111         2         .1         .6         68.0           12         5         .3         1.6         69.6           13         1         .1         .3         69.9           14         1         .1         .3         70.2           15         4         .2         1.3         71.5           16         3         .2         1.0         72.5           17         2         .1         .6         73.1           18         2         .1         .6         .73.8           20         6         .4         1.9         .75.7           22         2         .1         .6         .73.8           20         6         .4         1.9         .75.7           22         2         .1         .6         .76.4           23         1         .1         .3         .76.0           24         1         .1         .3         .78.3           28         1         .1         .3         .80.3           30         4         .2         .1.3         .79.9           33         1         .1         .3         .80.6 </td <td>9</td> <td>5</td> <td>.3</td> <td>1.6</td> <td>62.8</td>	9	5	.3	1.6	62.8
11         2         .1         .6         68.0           12         5         .3         1.6         69.6           13         1         .1         .3         69.9           15         4         2         1.3         71.5           16         3         .2         1.0         72.5           17         2         .1         .6         73.1           18         2         .1         .6         73.8           20         6         .4         1.9         75.7           22         2         .1         .6         .76.4           23         1         .1         .3         .76.7           24         1         .1         .3         .76.7           24         1         .1         .3         .78.3           28         1         .1         .3         .78.6           30         4         .2         1.3         .78.3           38         1         .1         .3         .80.6           35         1         .1         .3         .80.6           35         1         .1         .3         .80.9	10	14		4.5	
12         5         .3         1.6         69.6           13         1         .1         .3         69.9           14         1         .1         .3         70.2           15         4         .2         1.3         71.5           16         3         .2         1.0         72.5           17         2         .1         .6         73.1           18         2         .1         .6         73.8           20         6         .4         1.9         75.7           22         2         .1         .6         76.4           23         1         .1         .3         76.7           24         1         .1         .3         77.0           25         4         .2         1.3         78.3           30         4         .2         1.3         79.9           33         1         .1         .3         80.3           34         1         .1         .3         80.6           35         1         .1         .3         80.6           40         1         .1         .3         82.8 <td>11</td> <td></td> <td></td> <td></td> <td></td>	11				
13         1         .1         .3         69.9           14         1         .1         .3         70.2           15         4         .2         1.3         71.5           16         3         .2         1.0         72.5           17         2         .1         .6         73.1           18         2         .1         .6         73.1           18         2         .1         .6         73.1           20         .6         .4         .1.9         75.7           22         .2         .1         .6         .76.4           23         .1         .1         .3         .76.7           24         .1         .1         .3         .77.0           25         .4         .2         .1.3         .78.3           28         .1         .1         .3         .78.6           30         .4         .2         .1.3         .79.9           33         .1         .1         .3         .80.6           35         .1         .1         .3         .80.6           35         .1         .1         .3 <t< td=""><td>12</td><td>5</td><td></td><td></td><td></td></t<>	12	5			
14         1         .1         .3         70.2           15         4         .2         1.3         71.5           16         3         .2         1.0         72.5           17         2         .1         .6         73.1           18         2         .1         .6         73.8           20         .6         .4         .19         75.7           22         .2         .1         .6         .76.4           23         .1         .1         .3         .76.7           24         .1         .1         .3         .77.0           25         .4         .2         .1.3         .78.3           28         .1         .1         .3         .78.6           30         .4         .2         .1.3         .79.9           33         .1         .1         .3         .80.6           35         .1         .1         .3         .80.9           36         .2         .1         .6         .81.6           40         .1         .1         .3         .82.5           43         .1         .1         .3	13				
15         4         .2         1.3         71.5           16         3         .2         1.0         72.5           17         2         .1         .6         .73.1           18         2         .1         .6         .73.8           20         6         .4         1.9         .75.7           22         2         .1         .6         .76.4           23         1         .1         .3         .76.7           24         1         .1         .3         .77.0           25         4         .2         .1.3         .78.3           28         1         .1         .3         .78.6           30         4         .2         .1.3         .79.9           33         1         .1         .3         .80.3           34         1         .1         .3         .80.6           35         1         .1         .3         .80.6           35         1         .1         .3         .80.9           40         1         .1         .3         .80.9           45         1         .1         .3         .82	14				
16         3         .2         1.0         72.5           17         2         .1         .6         73.1           18         2         .1         .6         .73.8           20         6         .4         1.9         .75.7           22         2         .1         .6         .76.4           23         1         .1         .3         .76.7           24         1         .1         .3         .77.0           25         4         .2         1.3         .78.6           30         4         .2         1.3         .79.9           33         1         .1         .3         .80.3           34         1         .1         .3         .80.3           35         1         .1         .3         .80.9           36         2         .1         .6         .81.6           40         1         .1         .3         .81.9           45         1         .1         .3         .82.5           43         1         .1         .3         .83.2           45         1         .1         .3         .85.4<	15	4			
17       2       .1       .6       73.1         18       2       .1       .6       .73.8         20       6       .4       1.9       .75.7         22       2       .1       .6       .76.4         23       1       .1       .3       .76.7         24       1       .1       .3       .77.0         25       4       .2       1.3       .78.3         38       1       .1       .3       .78.3         30       4       .2       .13       .79.9         33       1       .1       .3       .80.3         34       1       .1       .3       .80.9         36       2       .1       .6       .81.6         40       1       .1       .3       .80.9         36       2       .1       .6       .82.5         43       1       .1       .3       .83.2         45       1       .1       .3       .83.2         46       1       .1       .3       .85.1         50       3       .2       .10       .84.5         51	16				
18         2         .1         .6         73.8           20         6         .4         1.9         75.7           22         2         .1         .6         .76.4           23         1         .1         .3         .76.7           24         1         .1         .3         .77.0           25         4         .2         1.3         .78.6           30         4         .2         1.3         .79.9           33         1         .1         .3         .80.3           34         1         .1         .3         .80.3           35         1         .1         .3         .80.9           36         2         .1         .6         .81.6           40         1         .1         .3         .80.9           36         2         .1         .6         .82.5           43         1         .1         .3         .82.8           45         1         .1         .3         .82.8           45         1         .1         .3         .83.2           46         1         .1         .3         .84.5 </td <td>17</td> <td></td> <td></td> <td></td> <td></td>	17				
20         6         .4         1.9         75.7           22         2         .1         .6         76.4           23         1         .1         .3         77.0           24         1         .1         .3         77.0           25         4         .2         1.3         78.3           28         1         .1         .3         78.6           30         4         .2         1.3         79.9           33         1         .1         .3         80.3           34         1         .1         .3         80.3           35         1         .1         .3         80.9           36         2         .1         .6         81.6           40         1         .1         .3         81.9           42         2         .1         .6         82.5           43         1         .1         .3         83.2           45         1         .1         .3         83.5           50         3         .2         1.0         84.5           51         1         .1         .3         85.1	18				
22         2         .1         .6         76.4           23         1         .1         .3         76.7           24         1         .1         .3         77.0           25         4         .2         1.3         78.3           28         1         .1         .3         79.9           33         1         .1         .3         80.3           34         1         .1         .3         80.3           35         1         .1         .3         80.6           35         1         .1         .3         80.9           36         2         .1         .6         81.6           40         1         .1         .3         81.9           42         2         .1         .6         82.5           43         1         .1         .3         82.8           45         1         .1         .3         83.2           46         1         .1         .3         85.5           50         3         .2         1.0         84.5           51         1         .1         .3         85.4	20				
23       1       .1       .3       76.7         24       1       .1       .3       77.0         25       4       .2       1.3       78.3         28       1       .1       .3       78.6         30       4       .2       1.3       79.9         33       1       .1       .3       80.3         34       1       .1       .3       80.6         35       1       .1       .3       80.9         36       2       .1       .6       81.6         40       1       .1       .3       81.9         42       2       .1       .6       82.5         43       1       .1       .3       83.2         45       1       .1       .3       83.2         46       1       .1       .3       83.5         50       3       .2       1.0       84.5         51       1       .1       .3       85.1         52       1       .1       .3       85.4         56       1       .1       .3       86.1         60       1       <	22				
24       1       .1       .3       77.0         25       4       .2       1.3       78.3         28       1       .1       .3       78.6         30       4       .2       1.3       79.9         33       1       .1       .3       80.3         34       1       .1       .3       80.6         35       1       .1       .3       80.9         36       2       .1       .6       81.6         40       1       .1       .3       81.9         42       2       .1       .6       82.5         43       1       .1       .3       83.2         45       1       .1       .3       83.2         46       1       .1       .3       83.5         50       3       .2       1.0       84.5         51       1       .1       .3       85.1         52       1       .1       .3       85.1         53       1       .1       .3       86.1         60       1       .1       .3       86.1         60       1       <	23				
25         4         .2         1.3         78.3           28         1         .1         .3         78.6           30         4         .2         1.3         79.9           33         1         .1         .3         80.3           34         1         .1         .3         80.9           36         2         .1         .6         81.6           40         1         .1         .3         81.9           42         2         .1         .6         82.5           43         1         .1         .3         83.2           46         1         .1         .3         83.5           50         3         .2         1.0         84.5           51         1         .1         .3         85.1           52         1         .1         .3         85.4           52         1         .1         .3         85.8           59         1         .1         .3         86.1           60         1         .1         .3         86.7           70         2         .1         .6         87.4	24				
28       1       .1       .3       78.6         30       4       .2       1.3       79.9         33       1       .1       .3       80.3         34       1       .1       .3       80.9         36       2       .1       .6       81.6         40       1       .1       .3       81.9         42       2       .1       .6       82.5         43       1       .1       .3       83.2         45       1       .1       .3       83.2         46       1       .1       .3       83.5         50       3       .2       1.0       84.5         51       1       .1       .3       85.1         52       1       .1       .3       85.1         53       1       .1       .3       85.8         59       1       .1       .3       86.4         60       1       .1       .3       86.7         70       2       .1       .6       87.4         73       1       .1       .3       89.0         90       2 <t< td=""><td>25</td><td></td><td></td><td></td><td></td></t<>	25				
30       4       .2       1.3       79.9         33       1       .1       .3       80.3         34       1       .1       .3       80.6         35       1       .1       .3       80.9         36       2       .1       .6       81.6         40       1       .1       .3       81.9         42       2       .1       .6       82.5         43       1       .1       .3       82.8         45       1       .1       .3       83.2         46       1       .1       .3       83.5         50       3       .2       1.0       84.5         51       1       .1       .3       85.1         52       1       .1       .3       85.1         53       1       .1       .3       85.8         59       1       .1       .3       86.1         60       1       .1       .3       86.7         70       2       .1       .6       87.4         73       1       .1       .3       89.7         75       2 <t< td=""><td>28</td><td></td><td></td><td></td><td></td></t<>	28				
33       1       .1       .3       80.3         34       1       .1       .3       80.6         35       1       .1       .3       80.9         36       2       .1       .6       81.6         40       1       .1       .3       81.9         42       2       .1       .6       82.5         43       1       .1       .3       82.8         45       1       .1       .3       83.2         46       1       .1       .3       83.5         50       3       .2       1.0       84.5         51       1       .1       .3       85.1         52       1       .1       .3       85.1         53       1       .1       .3       85.4         56       1       .1       .3       86.1         60       1       .1       .3       86.1         60       1       .1       .3       86.7         70       2       .1       .6       87.4         73       1       .1       .3       89.0         90       2 <td< td=""><td>30</td><td></td><td></td><td></td><td></td></td<>	30				
34       1       .1       .3       80.6         35       1       .1       .3       80.9         36       2       .1       .6       81.6         40       1       .1       .3       81.9         42       2       .1       .6       82.5         43       1       .1       .3       82.8         45       1       .1       .3       83.2         46       1       .1       .3       83.5         50       3       .2       1.0       84.5         51       1       .1       .3       85.1         52       1       .1       .3       85.4         56       1       .1       .3       85.8         59       1       .1       .3       86.4         64       1       .1       .3       86.7         70       2       .1       .6       87.4         73       1       .1       .3       88.7         75       2       .1       .6       88.3         85       1       .1       .3       89.0         90       2 <td< td=""><td>33</td><td></td><td></td><td></td><td></td></td<>	33				
35         1         .1         .3         80.9           36         2         .1         .6         81.6           40         1         .1         .3         81.9           42         2         .1         .6         82.5           43         1         .1         .3         82.8           45         1         .1         .3         83.2           46         1         .1         .3         83.5           50         3         .2         1.0         84.5           51         1         .1         .3         85.1           52         1         .1         .3         85.4           52         1         .1         .3         85.4           52         1         .1         .3         85.4           59         1         .1         .3         86.1           60         1         .1         .3         86.4           64         1         .1         .3         87.7           75         2         .1         .6         88.3           78         1         .1         .3         89.6					
36       2       .1       .6       81.6         40       1       .1       .3       81.9         42       2       .1       .6       82.5         43       1       .1       .3       82.8         45       1       .1       .3       83.2         46       1       .1       .3       83.5         50       3       .2       1.0       84.5         51       1       .1       .3       85.1         52       1       .1       .3       85.4         56       1       .1       .3       85.8         59       1       .1       .3       86.1         60       1       .1       .3       86.4         64       1       .1       .3       86.7         70       2       .1       .6       87.4         73       1       .1       .3       89.7         75       2       .1       .6       88.3         78       1       .1       .3       89.0         90       2       .1       .6       89.6         100       3 <t< td=""><td></td><td></td><td></td><td></td><td></td></t<>					
40       1       .1       .3       81.9         42       2       .1       .6       82.5         43       1       .1       .3       82.8         45       1       .1       .3       83.2         46       1       .1       .3       83.5         50       3       .2       1.0       84.5         51       1       .1       .3       85.1         52       1       .1       .3       85.4         56       1       .1       .3       85.8         59       1       .1       .3       86.1         60       1       .1       .3       86.4         64       1       .1       .3       86.7         70       2       .1       .6       87.4         73       1       .1       .3       88.3         78       1       .1       .3       89.0         90       2       .1       .6       89.6         100       3       .2       1.0       90.6         113       1       .1       .3       90.9         115       1					
42       2       .1       .6       82.5         43       1       .1       .3       82.8         45       1       .1       .3       83.2         46       1       .1       .3       83.5         50       3       .2       1.0       84.5         51       1       .1       .3       85.1         52       1       .1       .3       85.1         53       1       .1       .3       85.8         59       1       .1       .3       86.1         60       1       .1       .3       86.7         70       2       .1       .6       87.4         73       1       .1       .3       88.7         78       1       .1       .3       89.0         90       2       .1       .6       89.6         100       3       .2       1.0       90.6         113       1       .1       .3       90.9         115       1       .1       .3       91.3					
43       1       .1       .3       82.8         45       1       .1       .3       83.2         46       1       .1       .3       83.5         50       3       .2       1.0       84.5         51       1       .1       .3       84.8         52       1       .1       .3       85.1         53       1       .1       .3       85.4         56       1       .1       .3       86.1         60       1       .1       .3       86.4         64       1       .1       .3       86.7         70       2       .1       .6       87.4         73       1       .1       .3       88.7         75       2       .1       .6       88.3         78       1       .1       .3       89.0         90       2       .1       .6       89.6         100       3       .2       1.0       90.6         113       1       .1       .3       90.9         115       1       .1       .3       91.3	42				
45       1       .1       .3       83.2         46       1       .1       .3       83.5         50       3       .2       1.0       84.5         51       1       .1       .3       84.8         52       1       .1       .3       85.1         53       1       .1       .3       85.8         59       1       .1       .3       86.1         60       1       .1       .3       86.4         64       1       .1       .3       86.7         70       2       .1       .6       87.4         73       1       .1       .3       87.7         75       2       .1       .6       88.3         78       1       .1       .3       89.0         90       2       .1       .6       89.6         100       3       .2       1.0       90.6         113       1       .1       .3       90.9         115       1       .1       .3       91.3	43				
46       1       .1       .3       83.5         50       3       .2       1.0       84.5         51       1       .1       .3       84.8         52       1       .1       .3       85.1         53       1       .1       .3       85.4         56       1       .1       .3       86.1         60       1       .1       .3       86.1         60       1       .1       .3       86.7         70       2       .1       .6       87.4         73       1       .1       .3       87.7         75       2       .1       .6       88.3         78       1       .1       .3       89.0         90       2       .1       .6       89.6         100       3       .2       1.0       90.6         113       1       .1       .3       90.9         115       1       .1       .3       91.3	45				
50       3       .2       1.0       84.5         51       1       .1       .3       84.8         52       1       .1       .3       85.1         53       1       .1       .3       85.4         56       1       .1       .3       86.8         59       1       .1       .3       86.1         60       1       .1       .3       86.7         70       2       .1       .6       87.4         73       1       .1       .3       87.7         75       2       .1       .6       88.3         78       1       .1       .3       89.0         90       2       .1       .6       89.6         100       3       .2       1.0       90.6         113       1       .1       .3       90.9         115       1       .1       .3       91.3	46				
51       1       .1       .3       84.8         52       1       .1       .3       85.1         53       1       .1       .3       85.4         56       1       .1       .3       86.8         59       1       .1       .3       86.1         60       1       .1       .3       86.7         70       2       .1       .6       87.4         73       1       .1       .3       87.7         75       2       .1       .6       88.3         78       1       .1       .3       89.0         90       2       .1       .6       89.6         100       3       .2       1.0       90.6         113       1       .1       .3       90.9         115       1       .1       .3       91.3					
52       1       .1       .3       85.1         53       1       .1       .3       85.4         56       1       .1       .3       85.8         59       1       .1       .3       86.1         60       1       .1       .3       86.7         70       2       .1       .6       87.4         73       1       .1       .3       87.7         75       2       .1       .6       88.3         78       1       .1       .3       89.0         90       2       .1       .6       89.6         100       3       .2       1.0       90.6         113       1       .1       .3       90.9         115       1       .1       .3       91.3					
53       1       .1       .3       85.4         56       1       .1       .3       85.8         59       1       .1       .3       86.1         60       1       .1       .3       86.4         64       1       .1       .3       86.7         70       2       .1       .6       87.4         73       1       .1       .3       87.7         75       2       .1       .6       88.3         78       1       .1       .3       89.0         90       2       .1       .6       89.6         100       3       .2       1.0       90.6         113       1       .1       .3       90.9         115       1       .1       .3       91.3					
56       1       .1       .3       85.8         59       1       .1       .3       86.1         60       1       .1       .3       86.4         64       1       .1       .3       86.7         70       2       .1       .6       87.4         73       1       .1       .3       87.7         75       2       .1       .6       88.3         78       1       .1       .3       89.0         90       2       .1       .6       89.6         100       3       .2       1.0       90.6         113       1       .1       .3       90.9         115       1       .1       .3       91.3					
59       1       .1       .3       86.1         60       1       .1       .3       86.4         64       1       .1       .3       86.7         70       2       .1       .6       87.4         73       1       .1       .3       87.7         75       2       .1       .6       88.3         78       1       .1       .3       89.0         90       2       .1       .6       89.6         100       3       .2       1.0       90.6         113       1       .1       .3       90.9         115       1       .1       .3       91.3	56				
60       1       .1       .3       86.4         64       1       .1       .3       86.7         70       2       .1       .6       87.4         73       1       .1       .3       87.7         75       2       .1       .6       88.3         78       1       .1       .3       89.7         85       1       .1       .3       89.0         90       2       .1       .6       89.6         100       3       .2       1.0       90.6         113       1       .1       .3       90.9         115       1       .1       .3       91.3					
64       1       .1       .3       86.7         70       2       .1       .6       87.4         73       1       .1       .3       87.7         75       2       .1       .6       88.3         78       1       .1       .3       88.7         85       1       .1       .3       89.0         90       2       .1       .6       89.6         100       3       .2       1.0       90.6         113       1       .1       .3       90.9         115       1       .1       .3       91.3					
70       2       .1       .6       87.4         73       1       .1       .3       87.7         75       2       .1       .6       88.3         78       1       .1       .3       88.7         85       1       .1       .3       89.0         90       2       .1       .6       89.6         100       3       .2       1.0       90.6         113       1       .1       .3       90.9         115       1       .1       .3       91.3					
73       1       .1       .3       87.7         75       2       .1       .6       88.3         78       1       .1       .3       88.7         85       1       .1       .3       89.0         90       2       .1       .6       89.6         100       3       .2       1.0       90.6         113       1       .1       .3       90.9         115       1       .1       .3       91.3					
75       2       .1       .6       88.3         78       1       .1       .3       88.7         85       1       .1       .3       89.0         90       2       .1       .6       89.6         100       3       .2       1.0       90.6         113       1       .1       .3       90.9         115       1       .1       .3       91.3					
78       1       .1       .3       88.7         85       1       .1       .3       89.0         90       2       .1       .6       89.6         100       3       .2       1.0       90.6         113       1       .1       .3       90.9         115       1       .1       .3       91.3					
85     1     .1     .3     89.0       90     2     .1     .6     89.6       100     3     .2     1.0     90.6       113     1     .1     .3     90.9       115     1     .1     .3     91.3					
90     2     .1     .6     89.6       100     3     .2     1.0     90.6       113     1     .1     .3     90.9       115     1     .1     .3     91.3					
100     3     .2     1.0     90.6       113     1     .1     .3     90.9       115     1     .1     .3     91.3					
113     1     .1     .3     90.9       115     1     .1     .3     91.3					
115 1 .1 .3 91.3					
1					
	l	•	• •	.0	

	125	1	.1	.3	91.9
	168	1	.1	.3	92.2
	200	4	.2	1.3	93.5
	210	1	.1	.3	93.9
	220	1	.1	.3	94.2
	250	1	.1	.3	94.5
	252	1	.1	.3	94.8
	300	1	.1	.3	95.1
	350	2	.1	.6	95.8
	375	1	.1	.3	96.1
	377	1	.1	.3	96.4
	380	1	.1	.3	96.8
	400	2	.1	.6	97.4
	450	1	.1	.3	97.7
	500	3	.2	1.0	98.7
	700	1	.1	.3	99.0
	900	1	.1	.3	99.4
	1200	1	.1	.3	99.7
	8500	1	.1	.3	100.0
	Total	309	18.3	100.0	
Missing	999	1384	81.7		
Total		1693	100.0		

# # of cold cases reviewed

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	20	1.2	6.5	6.5
	1	53	3.1	17.3	23.8
	2				
		47	2.8	15.3	39.1
	3	28	1.7	9.1	48.2
	4	18	1.1	5.9	54.1
	5	21	1.2	6.8	60.9
	6	13	.8	4.2	65.1
	7	7	.4	2.3	67.4
	8	13	.8	4.2	71.7
	9	4	.2	1.3	73.0
	10	15	.9	4.9	77.9
	11	2	.1	.7	78.5
	12	6	.4	2.0	80.5
	14	1	.1	.3	80.8
	15	3	.2	1.0	81.8
	16	4	.2	1.3	83.1
	17	1	.1	.3	83.4
	18	1	.1	.3	83.7
	20	8	.5	2.6	86.3
	22	1	.1	.3	86.6
	24	1	.1	.3	87.0
	25	4	.2	1.3	88.3

	33	1	.1	.3	88.6
	36	1	.1	.3	88.9
	40	1	.1	.3	89.3
	45	2	.1	.7	89.9
	47	1	.1	.3	90.2
	50	4	.2	1.3	91.5
	52	2	.1	.7	92.2
	54	1	.1	.3	92.5
	55	1	.1	.3	92.8
	60	2	.1	.7	93.5
	64	1	.1	.3	93.8
	66	1	.1	.3	94.1
	70	1	.1	.3	94.5
	75	2	.1	.7	95.1
	80	1	.1	.3	95.4
	90	1	.1	.3	95.8
	100	1	.1	.3	96.1
	125	1	.1	.3	96.4
	150	1	.1	.3	96.7
	165	1	.1	.3	97.1
	168	1	.1	.3	97.4
	200	4	.2	1.3	98.7
	326	1	.1	.3	99.0
	375	1	.1	.3	99.3
	600	1	.1	.3	99.7
	1350	1	.1	.3	100.0
	Total	307	18.1	100.0	
Missing	999	1386	81.9		
Total		1693	100.0		

# # of cold cases sent for DNA testing

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	79	4.7	26.6	26.6
	1	79	4.7	26.6	53.2
	2	37	2.2	12.5	65.7
	3	25	1.5	8.4	74.1
	4	20	1.2	6.7	80.8
	5	9	.5	3.0	83.8
	6	6	.4	2.0	85.9
	7	3	.2	1.0	86.9
	8	2	.1	.7	87.5
	9	2	.1	.7	88.2
	10	9	.5	3.0	91.2
	11	1	.1	.3	91.6
	12	2	.1	.7	92.3
	14	1	.1	.3	92.6
	15	4	.2	1.3	93.9

Ī	18	1	.1	.3	94.3
	20	1	.1	.3	94.6
	22				
		1	.1	.3	94.9
	25	1	.1	.3	95.3
	30	3	.2	1.0	96.3
	35	1	.1	.3	96.6
	38	1	.1	.3	97.0
	40	1	.1	.3	97.3
	45	1	.1	.3	97.6
	50	2	.1	.7	98.3
	52	1	.1	.3	98.7
	75	1	.1	.3	99.0
	80	1	.1	.3	99.3
	200	1	.1	.3	99.7
	375	1	.1	.3	100.0
	Total	297	17.5	100.0	
Missing	999	1396	82.5		
Total		1693	100.0		

# # of additional cold cases potentially eligible for DNA testing

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	139	8.2	56.0	56.0
	1	17	1.0	6.9	62.9
	2	15	.9	6.0	69.0
	3	4	.2	1.6	70.6
	4	7	.4	2.8	73.4
	5	11	.6	4.4	77.8
	6	4	.2	1.6	79.4
	7	4	.2	1.6	81.0
	8	2	.1	.8	81.9
	9	1	.1	.4	82.3
	10	4	.2	1.6	83.9
	12	2	.1	.8	84.7
	15	4	.2	1.6	86.3
	16	1	.1	.4	86.7
	20	4	.2	1.6	88.3
	25	2	.1	.8	89.1
	27	1	.1	.4	89.5
	30	1	.1	.4	89.9
	36	1	.1	.4	90.3
	42	1	.1	.4	90.7
	45	1	.1	.4	91.1
	50	3	.2	1.2	92.3
	55	1	.1	.4	92.7
	64	1	.1	.4	93.1
	87	1	.1	.4	93.5
	99	3	.2	1.2	94.8

	135	1	.1	.4	95.2
	140	1	.1	.4	95.6
	142	1	.1	.4	96.0
	160	1	.1	.4	96.4
	250	1	.1	.4	96.8
	300	4	.2	1.6	98.4
	400	1	.1	.4	98.8
	850	1	.1	.4	99.2
	900	1	.1	.4	99.6
	1000	1	.1	.4	100.0
	Total	248	14.6	100.0	
Missing	999	1445	85.4		
Total		1693	100.0		

# educated guess is not possible

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	110	6.5	6.6	6.6
	No	1569	92.7	93.4	100.0
	Total	1679	99.2	100.0	
Missing	999	14	.8		
Total		1693	100.0		

# unable to answer completely

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	25	1.5	1.5	1.5
	No	1661	98.1	98.5	100.0
	Total	1686	99.6	100.0	
Missing	999	7	.4		
Total		1693	100.0		

# agency have plan to review [cold] cases for biological testing?

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	461	27.2	37.9	37.9
	No	755	44.6	62.1	100.0
	Total	1216	71.8	100.0	
Missing	999	477	28.2		
Total		1693	100.0		

# of cases eligible for review

	Frequency	Percent	Valid Percent	Cumulative Percent
Valid 0	169	10.0	32.8	32.8
1	81	4.8	15.7	48.5
2	76	4.5	14.8	63.3
3	34	2.0	6.6	69.9
4	20	1.2	3.9	73.8
5	23	1.4	4.5	78.3
6	8	.5	1.6	79.8
7	10	.6	1.9	81.7
8 9	6	.4	1.2	82.9
10	2	.1	.4	83.3
11	19	1.1	3.7	87.0
12	4	.2	.8 1.2	87.8 88.9
13	6 2	.4 .1	.4	89.3
15	2	.1	.4	89.7
17	1	.1	.2	89.7 89.9
18	3	.2	.6	90.5
19	1	.1	.2	90.7
20	12	.7	2.3	93.0
24	1	.1	.2	93.2
25	2	.1	.4	93.6
26	1	.1	.2	93.8
30	3	.2	.6	94.4
34	1	.1	.2	94.6
40	2	.1	.4	95.0
41	1	.1	.2	95.1
45	1	.1	.2	95.3
50	2	.1	.4	95.7
55	1	.1	.2	95.9
59	1	.1	.2	96.1
61	1	.1	.2	96.3
67	1	.1	.2	96.5
75 77	1	.1	.2	96.7
77 99	2	.1	.4	97.1
130	2	.1	.4	97.5
148	1	.1	.2	97.7
150	1	.1	.2	97.9
158	3	.2 .1	.6 .2	98.4 98.6
197	1	.1	.2	98.8
250	1	.1	.2	99.0
270	1	.1	.2	99.2
280	1	.1	.2	99.4
300	1	.1	.2	99.6
400	1	.1	.2	99.8

ĺ		600	1	.1	.2	100.0
		Total	515	30.4	100.0	
	Missing	999	1178	69.6		
	Total		1693	100.0		

# # of cases with potential for biological testing

		Fraguency	Percent	Valid Percent	Cumulative Percent
Valid	0	Frequency 209	12.3	45.9	45.9
Valla	1	81	4.8	17.8	63.7
	2	61	3.6	13.4	77.1
	3	25	1.5	5.5	82.6
	4	14	.8	3.1	85.7
	5	12	.7	2.6	88.4
	6	6	.4	1.3	89.7
	7	4	.2	.9	90.5
	8	1	.1	.2	90.8
	9	1	.1	.2	91.0
	10	6	.4	1.3	92.3
	11	1	.1	.2	92.5
	12	1	.1	.2	92.7
	15	1	.1	.2	93.0
	16	2	.1	.4	93.4
	18	2	.1	.4	93.8
	20	8	.5	1.8	95.6
	24	1	.1	.2	95.8
	25	3	.2	.7	96.5
	30	1	.1	.2	96.7
	50	1	.1	.2	96.9
	60	1	.1	.2	97.1
	61	1	.1	.2	97.4
	65	2	.1	.4	97.8
	67	1	.1	.2	98.0
	99	4	.2	.9	98.9
	100	2	.1	.4	99.3
	130	1	.1	.2	99.6
	150	1	.1	.2	99.8
	400	1	.1	.2	100.0
l	Total	455	26.9	100.0	
Missing	999	1238	73.1		
Total		1693	100.0		

# educated guess is not possible

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	290	17.1	17.3	17.3
	No	1389	82.0	82.7	100.0
	Total	1679	99.2	100.0	
Missing	999	14	.8		
Total		1693	100.0		

#### unable to answer completely

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	48	2.8	2.9	2.9
	No	1634	96.5	97.1	100.0
	Total	1682	99.4	100.0	
Missing	999	11	.6		
Total		1693	100.0		

# agency interested in using DNA database?

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	1595	94.2	96.9	96.9
	No	51	3.0	3.1	100.0
	Total	1646	97.2	100.0	
Missing	999	47	2.8		
Total		1693	100.0		

# has agency used DNA database for cases with no direct match?

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	300	17.7	18.4	18.4
	No	1330	78.6	81.6	100.0
	Total	1630	96.3	100.0	
Missing	999	63	3.7		
Total		1693	100.0		

# **APPENDIX 3b**

Local Law Enforcement Response Frequencies by Question (By Size Strata)

Due to the size of this file, a hard copy is not included. The file has been supplied to the Natoinal Institute of Justice in electronic form, or contact <a href="mailto:lhurst@smithallinglane.com">lhurst@smithallinglane.com</a> for a copy.

# <u>Appendix 3b – Law Enforcement Responses Split by Agency Size</u> Agency size strata = Over 100 Officers

# STATE(a)

				Cumulative
	Frequency	Percent	Valid Percent	Percent
Valid AL	9	1.5	1.5	1.5
AR	5	.9	.9	2.4
AZ	10	1.7	1.7	4.1
CA	59	10.1	10.1	14.2
CO	17	2.9	2.9	17.1
СТ	12	2.0	2.0	19.1
FL	56	9.6	9.6	28.7
GA	24	4.1	4.1	32.8
HI	3	.5	.5	33.3
IA	7	1.2	1.2	34.5
ID	3	.5	.5	35.0
IL	20	3.4	3.4	38.4
IN	13	2.2	2.2	40.6
KS	10	1.7	1.7	42.3
KY	2	.3	.3	42.7
LA	16	2.7	2.7	45.4
MA	6	1.0	1.0	46.4
MD	12	2.0	2.0	48.5
ME	1	.2	.2	48.6
MI	24	4.1	4.1	52.7
MN	6	1.0	1.0	53.8
MO	12	2.0	2.0	55.8
MS	4	.7	.7	56.5
MT	1	.2	.2	56.7
NC	27	4.6	4.6	61.3
ND	2	.3	.3	61.6
NE	3	.5	.5	62.1
NJ	16	2.7	2.7	64.8
NM	3	.5	.5	65.4
NV	2	.3	.3	65.7
NY	33	5.6	5.6	71.3
ОН	14	2.4	2.4	73.7
OK	5	.9	.9	74.6
OR	9	1.5	1.5	76.1
PA	10	1.7	1.7	77.8
RI	3	.5	.5	78.3
SC	10	1.7	1.7	80.0
SD	2	.3	.3	80.4
TN	13	2.2	2.2	82.6

TX	46	7.8	7.8	90.4
UT	3	.5	.5	91.0
VA	25	4.3	4.3	95.2
WA	13	2.2	2.2	97.4
WI	15	2.6	2.6	100.0
Total	586	100.0	100.0	

a Agency size strata = Over 100 Officers

# Where are cases primarily sent for DNA analysis?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	State Agency Lab	420	71.7	72.4	72.4
	Local Agency Lab	100	17.1	17.2	89.7
	FBI Lab	3	.5	.5	90.2
	Other	33	5.6	5.7	95.9
	Private Lab/Commercial Lab	24	4.1	4.1	100.0
	Total	580	99.0	100.0	
Missing	999	6	1.0		
Total		586	100.0		

a Agency size strata = Over 100 Officers

# % rapes with possible DNA(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	23	3.9	4.3	4.3
	11-25%	33	5.6	6.1	10.4
	25-50%	73	12.5	13.6	24.0
	50-75%	123	21.0	22.9	46.8
	75-100%	286	48.8	53.2	100.0
	Total	538	91.8	100.0	
Missing	999	48	8.2		
Total		586	100.0		

a Agency size strata = Over 100 Officers

% homicides with possible DNA(a	%	homicides	with	possibl	e Di	NA(	a)
---------------------------------	---	-----------	------	---------	------	-----	----

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	54	9.2	10.3	10.3
	11-25%	35	6.0	6.7	17.0
	25-50%	71	12.1	13.6	30.7
	50-75%	63	10.8	12.1	42.7
	75-100%	299	51.0	57.3	100.0
	Total	522	89.1	100.0	
Missing	999	64	10.9		
Total		586	100.0		

a Agency size strata = Over 100 Officers

# educated guess not possible(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	64	10.9	10.9	10.9
	No	521	88.9	89.1	100.0
	Total	585	99.8	100.0	
Missing	999	1	.2		
Total		586	100.0		

a Agency size strata = Over 100 Officers

# unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	19	3.2	3.2	3.2
	No	566	96.6	96.8	100.0
	Total	585	99.8	100.0	
Missing	999	1	.2		
Total		586	100.0		

a Agency size strata = Over 100 Officers

# unsolved homicides--total cases(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	32	5.5	8.1	8.1
	1	32	5.5	8.1	16.2
	2	40	6.8	10.1	26.3
	3	24	4.1	6.1	32.4
	4	22	3.8	5.6	38.0
	5	19	3.2	4.8	42.8
	6	24	4.1	6.1	48.9

8       17       2.9       4.3       55.7         9       3       .5       .8       56.5         10       21       3.6       5.3       61.8         11       6       1.0       1.5       63.3         12       11       1.9       2.8       66.1         13       2       .3       .5       66.6         14       4       .7       1.0       67.6         15       5       .9       1.3       68.9         16       3       .5       .8       69.6         17       4       .7       1.0       70.6         18       6       1.0       1.5       72.2         20       11       1.9       2.8       74.9         21       1       .2       .3       .5       75.7         23       1       .2       .3       .5       76.5         24       2       .3       .5       77.0         26       1       .2       .3       .5       77.7         28       3       .5       .8       78.5         29       1       .2       .3       .5<	7	40	4 7	0.5	
9		10	1.7	2.5	51.4
10         21         3.6         5.3         61.8           11         6         1.0         1.5         63.3           12         11         1.9         2.8         66.1           13         2         3         .5         66.6           14         4         .7         1.0         67.6           15         5         .9         1.3         68.9           16         3         .5         .8         69.6           17         4         .7         1.0         70.6           18         6         1.0         1.5         72.2           20         11         1.9         2.8         74.9           21         1         .2         .3         .75.9           23         1         .2         .3         .5         75.7           23         1         .2         .3         .5         77.0           24         2         .3         .5         77.0           28         3         .5         .8         78.5           29         1         .2         .3         .5         77.7           28         3					
11         6         1.0         1.5         63.3           12         11         1.9         2.8         66.1           13         2         .3         .5         66.6           14         4         .7         1.0         67.6           15         5         .9         1.3         68.9           16         3         .5         .8         69.6           17         4         .7         1.0         70.6           18         6         1.0         1.5         72.2           20         11         1.9         2.8         74.9           21         1         .2         .3         .5         75.2           22         2         .3         .5         75.2           23         1         .2         .3         .5         77.0           24         2         .3         .5         77.0         72.2           25         2         .3         .5         77.7         72.2           28         3         .5         .8         78.5         77.7           28         3         .5         .8         78.1         80.					
12         11         1.9         2.8         66.1           13         2         .3         .5         66.6           14         4         .7         1.0         67.6           15         5         .9         1.3         68.9           16         3         .5         .8         69.6           17         4         .7         1.0         70.6           18         6         1.0         1.5         72.2           20         11         1.9         2.8         74.9           21         1         .2         .3         .75.2           22         2         .3         .5         .75.7           23         1         .2         .3         .5         .75.7           24         2         .3         .5         .77.0         .7           25         2         .3         .5         .77.7         .7           28         3         .5         .8         .78.5           29         1         .2         .3         .5         .77.7           28         3         .5         .8         .81.5           31 </td <td></td> <td></td> <td></td> <td></td> <td></td>					
13         2         .3         .5         66.6           14         4         .7         1.0         67.6           15         5         .9         1.3         68.9           16         3         .5         .8         69.6           17         4         .7         1.0         70.6           18         6         1.0         1.5         72.2           20         11         1.9         2.8         74.9           21         1         .2         .3         .5         75.7           22         2         .3         .5         75.7           23         1         .2         .3         .5         76.5           24         2         .3         .5         77.7           28         3         .5         .8         78.5           29         1         .2         .3         .76.7           28         3         .5         .8         78.5           29         1         .2         .3         .76.7           28         3         .5         .8         81.3           33         1         .2					
14         4         .7         1.0         67.6           15         5         .9         1.3         68.9           16         3         .5         .8         69.6           17         4         .7         1.0         70.6           18         6         1.0         1.5         72.2           20         11         1.9         2.8         74.9           21         1         .2         .3         .5         75.7           22         2         .3         .5         75.7           23         1         .2         .3         .5         76.5           24         2         .3         .5         77.0           25         2         .3         .5         77.0           26         1         .2         .3         .5         77.7           28         3         .5         .8         87.8         78.7           29         1         .2         .3         .5         77.7         2         .8         .8         .8         .78.7         .8         .8         .8         .78.5         .77.7         .2         .3         .8 </td <td></td> <td></td> <td></td> <td></td> <td></td>					
15         5         .9         1.3         68.9           16         3         .5         .8         69.6           17         4         .7         1.0         70.6           18         6         1.0         1.5         72.2           20         11         1.9         2.8         74.9           21         1         .2         .3         .75.7           22         2         .3         .5         .75.7           23         1         .2         .3         .5         .75.9           24         2         .3         .5         .76.5           25         2         .3         .5         .77.0           26         1         .2         .3         .5         .77.7           28         3         .5         .8         .78.5           29         1         .2         .3         .5         .77.7           28         3         .5         .8         .81.3           33         1         .2         .3         .81.5           30         7         1.2         1.8         80.5           31         .2 </td <td></td> <td></td> <td></td> <td></td> <td></td>					
16         3         .5         .8         69.6           17         4         .7         1.0         70.6           18         6         1.0         1.5         72.2           20         11         1.9         2.8         74.9           21         1         .2         .3         .55         75.7           22         2         .3         .5         75.7           23         1         .2         .3         .5         76.5           24         2         .3         .5         77.0           26         1         .2         .3         .5         77.7           28         3         .5         .8         78.5           29         1         .2         .3         .75.7           28         3         .5         .8         78.5           30         7         1.2         1.8         80.5           31         3         .5         .8         81.3           33         1         .2         .3         81.5           34         1         .2         .3         82.0           36         1					
17         4         .7         1.0         70.6           18         6         1.0         1.5         72.2           20         11         1.9         2.8         74.9           21         1         2         .3         .75.2           22         2         2         .3         .5         .75.7           23         1         .2         .3         .5         .76.5           24         2         .3         .5         .77.0           26         1         .2         .3         .5         .77.7           28         3         .5         .8         .78.5           29         1         .2         .3         .78.7           28         3         .5         .8         .78.5           29         1         .2         .3         .78.7           28         3         .5         .8         .78.5           30         7         1.2         1.8         .80.5           31         3         .5         .8         .81.3           33         1         .2         .3         .82.3           40         4 <td></td> <td></td> <td></td> <td></td> <td></td>					
18       6       1.0       1.5       72.2         20       11       1.9       2.8       74.9         21       1       2       .3       .75.2         22       2       2.3       .5       .75.7         23       1       .2       .3       .5       .76.5         24       2       .3       .5       .77.0         26       1       .2       .3       .5       .77.2         27       2       .3       .5       .77.7         28       3       .5       .8       .78.5         29       1       .2       .3       .78.7         30       7       1.2       1.8       .80.5         31       3       .5       .8       .81.3         33       1       .2       .3       .81.5         34       1       .2       .3       .82.0         36       1       .2       .3       .82.3         40       4       .7       1.0       .83.3         41       1       .2       .3       .84.1         43       1       .2       .3       .5       <					
20         11         1.9         2.8         74.9           21         1         2         .3         75.2           22         2         .3         .5         75.7           23         1         .2         .3         .5         76.5           24         2         .3         .5         77.0           26         1         .2         .3         .5         77.7           28         3         .5         .8         78.5           29         1         .2         .3         78.7           30         7         1.2         1.8         80.5           31         3         .5         .8         81.3           33         1         .2         .3         81.5           34         1         .2         .3         82.0           36         1         .2         .3         82.0           36         1         .2         .3         82.0           36         1         .2         .3         82.3           40         4         .7         1.0         83.3           42         2         .3 <td< td=""><td></td><td></td><td></td><td></td><td></td></td<>					
21       1       .2       .3       .75.2         22       2       .3       .5       .75.7         23       1       .2       .3       .5       .75.9         24       2       .3       .5       .76.5         25       2       .3       .5       .77.0         26       1       .2       .3       .5       .77.7         28       3       .5       .8       .78.5         29       1       .2       .3       .78.7         30       7       1.2       1.8       80.5         31       3       .5       .8       81.3         33       1       .2       .3       81.5         34       1       .2       .3       82.0         36       1       .2       .3       82.0         36       1       .2       .3       82.3         40       4       .7       1.0       83.3         42       2       .3       .5       84.1         43       1       .2       .3       84.3         47       1       .2       .3       84.8 <tr< td=""><td></td><td></td><td></td><td></td><td></td></tr<>					
22       2       3       .5       75.7         23       1       .2       .3       .5       76.5         24       2       .3       .5       .76.5         25       2       .3       .5       .77.0         26       1       .2       .3       .5       .77.7         28       3       .5       .8       .78.5         29       1       .2       .3       .78.7         30       7       1.2       1.8       80.5         31       3       .5       .8       81.3         33       1       .2       .3       81.5         34       1       .2       .3       81.8         35       1       .2       .3       82.0         36       1       .2       .3       82.3         40       4       .7       1.0       83.3         41       1       .2       .3       84.1         42       2       .3       .5       84.1         43       1       .2       .3       84.8         51       2       .3       .5       85.3					
23       1       .2       .3       .75.9         24       2       .3       .5       .76.5         25       2       .3       .5       .77.0         26       1       .2       .3       .77.2         27       2       .3       .5       .77.7         28       3       .5       .8       .78.5         29       1       .2       .3       .78.7         30       7       1.2       1.8       80.5         31       3       .5       .8       81.3         33       1       .2       .3       81.5         34       1       .2       .3       82.0         36       1       .2       .3       82.3         40       4       .7       1.0       83.3         41       1       .2       .3       83.5         42       2       .3       .5       84.1         43       1       .2       .3       84.6         50       1       .2       .3       84.6         51       2       .3       .5       85.3         52       3		1	.2	.3	75.2
24       2       .3       .5       76.5         25       2       .3       .5       77.0         26       1       .2       .3       .77.2         27       2       .3       .5       .8       78.7         28       3       .5       .8       78.7         30       7       1.2       1.8       80.5         31       3       .5       .8       81.3         33       1       .2       .3       81.5         34       1       .2       .3       81.8         35       1       .2       .3       82.0         36       1       .2       .3       82.0         36       1       .2       .3       82.3         40       4       .7       1.0       83.3         41       1       .2       .3       84.1         43       1       .2       .3       84.6         50       1       .2       .3       84.8         51       2       .3       .5       85.3         52       3       .5       .8       86.1         55			.3	.5	75.7
25       2       3       5       77.0         26       1       .2       .3       .77.2         27       2       .3       .5       .8       .78.5         28       3       .5       .8       .78.5         29       1       .2       .3       .78.7         30       7       1.2       1.8       .80.5         31       3       .5       .8       .81.3         33       1       .2       .3       .81.5         34       1       .2       .3       .81.8         35       1       .2       .3       .82.0         36       1       .2       .3       .82.3         40       4       .7       1.0       .83.3         42       2       .3       .5       .84.1         43       1       .2       .3       .84.6         50       1       .2       .3       .84.6         51       2       .3       .5       .85.3         52       3       .5       .8       .86.1         55       1       .2       .3       .5       .86.8		1		.3	75.9
26       1       .2       .3       .77.2         27       2       .3       .5       .77.7         28       3       .5       .8       .78.5         29       1       .2       .3       .78.7         30       7       1.2       1.8       .80.5         31       3       .5       .8       .81.3         33       1       .2       .3       .81.5         34       1       .2       .3       .81.8         35       1       .2       .3       .82.0         36       1       .2       .3       .82.3         40       4       .7       1.0       .83.3         41       1       .2       .3       .84.3         42       2       .3       .5       .84.1         43       1       .2       .3       .84.6         50       1       .2       .3       .84.8         51       2       .3       .5       .85.3         52       3       .5       .8       .86.1         55       1       .2       .3       .87.1         70		2	.3	.5	76.5
27       2       3       .5       77.7         28       3       .5       .8       78.5         29       1       .2       .3       78.7         30       7       1.2       1.8       80.5         31       3       .5       .8       81.3         33       1       .2       .3       81.5         34       1       .2       .3       82.0         35       1       .2       .3       82.0         36       1       .2       .3       82.3         40       4       .7       1.0       83.3         41       1       .2       .3       83.5         42       2       .3       .5       84.1         43       1       .2       .3       84.6         50       1       .2       .3       84.8         51       2       .3       .5       85.3         52       3       .5       .8       86.1         55       1       .2       .3       87.1         70       1       .2       .3       87.6         73       1 <t< td=""><td></td><td>2</td><td>.3</td><td>.5</td><td>77.0</td></t<>		2	.3	.5	77.0
28       3       .5       .8       78.5         29       1       .2       .3       78.7         30       7       1.2       1.8       80.5         31       3       .5       .8       81.3         33       1       .2       .3       81.5         34       1       .2       .3       82.0         36       1       .2       .3       82.3         40       4       .7       1.0       83.3         41       1       .2       .3       84.3         42       2       .3       .5       84.1         43       1       .2       .3       84.6         50       1       .2       .3       84.8         51       2       .3       .5       85.3         52       3       .5       .8       86.1         55       1       .2       .3       86.8         65       1       .2       .3       87.1         70       1       .2       .3       87.6         73       1       .2       .3       87.6         75       4       <	26	1	.2	.3	77.2
29       1       2       .3       78.7         30       7       1.2       1.8       80.5         31       3       .5       .8       81.3         33       1       .2       .3       81.5         34       1       .2       .3       82.0         36       1       .2       .3       82.3         40       4       .7       1.0       83.3         41       1       .2       .3       84.3         42       2       .3       .5       84.1         43       1       .2       .3       84.6         50       1       .2       .3       84.8         51       2       .3       .5       85.3         52       3       .5       .8       86.1         55       1       .2       .3       86.3         59       2       .3       .5       86.8         65       1       .2       .3       87.1         70       1       .2       .3       87.6         73       1       .2       .3       87.6         75       4 <t< td=""><td>27</td><td>2</td><td>.3</td><td>.5</td><td>77.7</td></t<>	27	2	.3	.5	77.7
30       7       1.2       1.8       80.5         31       3       .5       .8       81.3         33       1       .2       .3       81.5         34       1       .2       .3       82.0         36       1       .2       .3       82.3         40       4       .7       1.0       83.3         41       1       .2       .3       84.1         43       1       .2       .3       84.3         47       1       .2       .3       84.6         50       1       .2       .3       84.6         51       2       .3       .5       85.3         52       3       .5       .8       86.1         55       1       .2       .3       .5       86.8         65       1       .2       .3       .5       86.8         65       1       .2       .3       .5       86.8         65       1       .2       .3       .5       86.8         72       1       .2       .3       .3       87.8         75       4       .7 <td< td=""><td>28</td><td>3</td><td>.5</td><td>.8</td><td>78.5</td></td<>	28	3	.5	.8	78.5
31       3       .5       .8       81.3         33       1       .2       .3       81.5         34       1       .2       .3       82.0         35       1       .2       .3       82.3         40       4       .7       1.0       83.3         41       1       .2       .3       84.1         42       2       .3       .5       84.1         43       1       .2       .3       84.3         47       1       .2       .3       84.6         50       1       .2       .3       84.8         51       2       .3       .5       85.3         52       3       .5       .8       86.1         55       1       .2       .3       86.3         59       2       .3       .5       86.8         65       1       .2       .3       87.6         70       1       .2       .3       87.6         73       1       .2       .3       87.6         75       4       .7       1.0       88.9         78       1 <t< td=""><td>29</td><td>1</td><td>.2</td><td>.3</td><td>78.7</td></t<>	29	1	.2	.3	78.7
33       1       .2       .3       81.5         34       1       .2       .3       82.0         35       1       .2       .3       82.3         40       4       .7       1.0       83.3         41       1       .2       .3       83.5         42       2       .3       .5       84.1         43       1       .2       .3       84.8         50       1       .2       .3       84.8         51       2       .3       .5       85.3         52       3       .5       .8       86.1         55       1       .2       .3       86.3         59       2       .3       .5       86.8         65       1       .2       .3       87.1         70       1       .2       .3       87.6         73       1       .2       .3       87.8         75       4       .7       1.0       88.9         78       1       .2       .3       89.4         80       1       .2       .3       89.6	30	7	1.2	1.8	80.5
33       1       .2       .3       81.5         34       1       .2       .3       81.8         35       1       .2       .3       82.0         36       1       .2       .3       82.3         40       4       .7       1.0       83.3         41       1       .2       .3       83.5         42       2       .3       .5       84.1         43       1       .2       .3       84.8         50       1       .2       .3       84.8         51       2       .3       .5       85.3         52       3       .5       .8       86.1         55       1       .2       .3       .5       86.8         65       1       .2       .3       87.3         70       1       .2       .3       87.8         73       1       .2       .3       87.8         75       4       .7       1.0       88.9         78       1       .2       .3       89.4         80       1       .2       .3       89.6	31	3	.5	.8	81.3
34       1       .2       .3       81.8         35       1       .2       .3       82.0         36       1       .2       .3       82.3         40       4       .7       1.0       83.3         41       1       .2       .3       .5       84.1         42       2       .3       .5       84.1         43       1       .2       .3       84.8         50       1       .2       .3       .5       85.3         51       2       .3       .5       86.1         52       3       .5       .8       86.1         55       1       .2       .3       .5       86.8         65       1       .2       .3       .5       86.8         65       1       .2       .3       87.6         72       1       .2       .3       87.8         75       4       .7       1.0       88.9         78       1       .2       .3       89.4         80       1       .2       .3       89.4         83       1       .2       .3	33	1			81.5
35       1       .2       .3       82.0         36       1       .2       .3       82.3         40       4       .7       1.0       83.3         41       1       .2       .3       83.5         42       2       .3       .5       84.1         43       1       .2       .3       84.8         50       1       .2       .3       .5       85.3         51       2       .3       .5       86.1         55       1       .2       .3       .5       86.3         59       2       .3       .5       86.8         65       1       .2       .3       87.1         70       1       .2       .3       87.6         73       1       .2       .3       87.8         75       4       .7       1.0       88.9         78       1       .2       .3       89.1         80       1       .2       .3       89.4         83       1       .2       .3       89.6	34	1			81.8
36       1       .2       .3       82.3         40       4       .7       1.0       83.3         41       1       .2       .3       83.5         42       2       .3       .5       84.1         43       1       .2       .3       84.8         50       1       .2       .3       84.8         51       2       .3       .5       85.3         52       3       .5       .8       86.1         55       1       .2       .3       .5       86.8         65       1       .2       .3       87.3         70       1       .2       .3       87.6         73       1       .2       .3       87.8         75       4       .7       1.0       88.9         78       1       .2       .3       89.4         80       1       .2       .3       89.4         83       1       .2       .3       89.6	35	1			
40       4       .7       1.0       83.3         41       1       .2       .3       83.5         42       2       .3       .5       84.1         43       1       .2       .3       84.6         50       1       .2       .3       84.8         51       2       .3       .5       85.3         52       3       .5       .8       86.1         55       1       .2       .3       .5       86.8         65       1       .2       .3       87.6         70       1       .2       .3       87.6         73       1       .2       .3       87.6         75       4       .7       1.0       88.9         78       1       .2       .3       89.4         80       1       .2       .3       89.4         83       1       .2       .3       89.6	36	1			
41       1       .2       .3       83.5         42       2       .3       .5       84.1         43       1       .2       .3       84.6         50       1       .2       .3       84.6         51       2       .3       .5       85.3         52       3       .5       .8       86.1         55       1       .2       .3       .5       86.8         65       1       .2       .3       87.1         70       1       .2       .3       87.3         72       1       .2       .3       87.8         75       4       .7       1.0       88.9         78       1       .2       .3       89.4         80       1       .2       .3       89.4         83       1       .2       .3       89.6	40	4			
42       2       .3       .5       84.1         43       1       .2       .3       84.8         47       1       .2       .3       84.6         50       1       .2       .3       .5       85.3         51       2       .3       .5       .8       86.1         52       3       .5       .8       86.1         55       1       .2       .3       .5       86.8         65       1       .2       .3       87.1         70       1       .2       .3       87.3         72       1       .2       .3       87.8         75       4       .7       1.0       88.9         78       1       .2       .3       89.4         80       1       .2       .3       89.4         83       1       .2       .3       89.6	41				
43       1       .2       .3       84.3         47       1       .2       .3       84.6         50       1       .2       .3       84.8         51       2       .3       .5       85.3         52       3       .5       .8       86.1         55       1       .2       .3       .5       86.8         65       1       .2       .3       87.1         70       1       .2       .3       87.3         72       1       .2       .3       87.8         75       4       .7       1.0       88.9         78       1       .2       .3       89.4         80       1       .2       .3       89.4         83       1       .2       .3       89.6	42				
47       1       .2       .3       84.6         50       1       .2       .3       84.8         51       2       .3       .5       85.3         52       3       .5       .8       86.1         55       1       .2       .3       .5       86.8         65       1       .2       .3       87.1         70       1       .2       .3       87.8         72       1       .2       .3       87.8         75       4       .7       1.0       88.9         78       1       .2       .3       89.4         80       1       .2       .3       89.4         83       1       .2       .3       89.6	43				
50       1       .2       .3       84.8         51       2       .3       .5       85.3         52       3       .5       .8       86.1         55       1       .2       .3       .5       86.8         65       1       .2       .3       87.1         70       1       .2       .3       87.3         72       1       .2       .3       87.8         75       4       .7       1.0       88.9         78       1       .2       .3       89.1         80       1       .2       .3       89.4         83       1       .2       .3       89.6					
51       2       .3       .5       85.3         52       3       .5       .8       86.1         55       1       .2       .3       .86.3         59       2       .3       .5       86.8         65       1       .2       .3       87.1         70       1       .2       .3       87.3         72       1       .2       .3       87.8         75       4       .7       1.0       88.9         78       1       .2       .3       89.4         80       1       .2       .3       89.4         83       1       .2       .3       89.6	50				
52       3       .5       .8       86.1         55       1       .2       .3       86.3         59       2       .3       .5       86.8         65       1       .2       .3       87.1         70       1       .2       .3       87.3         72       1       .2       .3       87.8         73       1       .2       .3       87.8         75       4       .7       1.0       88.9         78       1       .2       .3       89.1         80       1       .2       .3       89.4         83       1       .2       .3       89.6	51				
55       1       .2       .3       86.3         59       2       .3       .5       86.8         65       1       .2       .3       87.1         70       1       .2       .3       87.3         72       1       .2       .3       87.6         73       1       .2       .3       87.8         75       4       .7       1.0       88.9         78       1       .2       .3       89.1         80       1       .2       .3       89.4         83       1       .2       .3       89.6	52				
59       2       .3       .5       86.8         65       1       .2       .3       87.1         70       1       .2       .3       87.3         72       1       .2       .3       87.6         73       1       .2       .3       87.8         75       4       .7       1.0       88.9         78       1       .2       .3       89.1         80       1       .2       .3       89.4         83       1       .2       .3       89.6					
65       1       .2       .3       87.1         70       1       .2       .3       87.3         72       1       .2       .3       87.6         73       1       .2       .3       87.8         75       4       .7       1.0       88.9         78       1       .2       .3       89.1         80       1       .2       .3       89.4         83       1       .2       .3       89.6					
70       1       .2       .3       87.3         72       1       .2       .3       87.6         73       1       .2       .3       87.8         75       4       .7       1.0       88.9         78       1       .2       .3       89.1         80       1       .2       .3       89.4         83       1       .2       .3       89.6					
72       1       .2       .3       87.6         73       1       .2       .3       87.8         75       4       .7       1.0       88.9         78       1       .2       .3       89.1         80       1       .2       .3       89.4         83       1       .2       .3       89.6					
73       1       .2       .3       87.8         75       4       .7       1.0       88.9         78       1       .2       .3       89.1         80       1       .2       .3       89.4         83       1       .2       .3       89.6					
75       4       .7       1.0       88.9         78       1       .2       .3       89.1         80       1       .2       .3       89.4         83       1       .2       .3       89.6					
78       1       .2       .3       89.1         80       1       .2       .3       89.4         83       1       .2       .3       89.6					
80 1 .2 .3 89.4 83 1 .2 .3 89.6					
83 1 .2 .3 89.6					
.3   .2   .3   69.8					
		'	.2	.5	09.9

		•	1	II.	ı
	90	1	.2	.3	90.1
	99	1	.2	.3	90.4
	100	2	.3	.5	90.9
	105	1	.2	.3	91.1
	111	1	.2	.3	91.4
	113	1	.2	.3	91.6
	115	1	.2	.3	91.9
	120	2	.3	.5	92.4
	134	1	.2	.3	92.7
	148	1	.2	.3	92.9
	150	1	.2	.3	93.2
	158	1	.2	.3	93.4
	168	1	.2	.3	93.7
	180	1	.2	.3	93.9
	183	1	.2	.3	94.2
	197	1	.2	.3	94.4
	200	1	.2	.3	94.7
	201	1	.2	.3	94.9
	207	1	.2	.3	95.2
	210	1	.2	.3	95.4
	216	1	.2	.3	95.7
	220	1	.2	.3	95.9
	250	2	.3	.5	96.5
	256	1	.2	.3	96.7
	280	1	.2	.3	97.0
	300	2	.3	.5	97.5
	339	1	.2	.3	97.7
	380	1	.2	.3	98.0
	500	1	.2	.3	98.2
	650	1	.2	.3	98.5
	721	1	.2	.3	98.7
	750	1	.2	.3	99.0
	800	1	.2	.3	99.2
	880	1	.2	.3	99.5
	1000	1	.2	.3	99.7
	8500	1	.2	.3	100.0
	Total	395	67.4	100.0	
Missing	999	191	32.6		
Total		586	100.0		

a Agency size strata = Over 100 Officers

unsolved homicides--cases with DNA(a)

					Cumulative
		Frequency	Percent	Valid Percent	Percent
Valid 0		71	12.1	22.0	22.0
1		59	10.1	18.3	40.4
2		42	7.2	13.0	53.4
3		17	2.9	5.3	58.7
4		14	2.4	4.3	63.0
5		14	2.4	4.3	67.4
6		7	1.2	2.2	69.6
7		8	1.4	2.5	72.0
8		6	1.0	1.9	73.9
9		5	.9	1.6	75.5
1		19	3.2	5.9	81.4
1		1	.2	.3	81.7
1		5	.9	1.6	83.2
1		2	.3	.6	83.9
1		4	.7	1.2	85.1
1		4	.7	1.2	86.3
1		3	.5	.9	87.3
2		5	.9	1.6	88.8
2		1	.2	.3	89.1
2		3	.5	.9	90.1
3		7	1.2	2.2	92.2
3		1	.2	.3	92.5
3		1	.2	.3	92.9
4		2	.3	.6	93.5
5		3	.5	.9	94.4
5		1	.2	.3	94.7
8		1	.2	.3	95.0
8		1	.2	.3	95.3
9		1	.2	.3	95.7
1	00	2	.3	.6	96.3
	20	1	.2	.3	96.6
	26	1	.2	.3	96.9
1	30	1	.2	.3	97.2
	55	1	.2	.3	97.5
1	70	1	.2	.3	97.8
2	00	2	.3	.6	98.4
	52	1	.2	.3	98.8
	90	1	.2	.3	99.1
4	00	2	.3	.6	99.7
8	50	1	.2	.3	100.0
Т	otal	322	54.9	100.0	
Missing 9	99	264	45.1		
Total		586	100.0		

a Agency size strata = Over 100 Officers

# unsolved rapes--total cases(a)

	Frequency	Percent	Valid Percent	Cumulative Percent
Valid 0	39	6.7	14.3	14.3
1	11	1.9	4.0	18.4
2	13	2.2	4.8	23.2
3	6	1.0	2.2	25.4
4	6	1.0	2.2	27.6
5	11	1.9	4.0	31.6
6	6	1.0	2.2	33.8
7	5	.9	1.8	35.7
8	4	.7	1.5	37.1
9	5	.9	1.8	39.0
10	12	2.0	4.4	43.4
11	2	.3	.7	44.1
12	3	.5	1.1	45.2
13	2	.3	.7	46.0
14	1	.2	.4	46.3
15	5	.9	1.8	48.2
16	3	.5	1.1	49.3
17	1	.2	.4	49.6
18	3	.5	1.1	50.7
20	7	1.2	2.6	53.3
21	1	.2	.4	53.7
22	2	.3	.7	54.4
23	1	.2	.4	54.8
24	2	.3	.7	55.5
25	2	.3	.7	56.2
27	1	.2	.4	56.6
28	1	.2	.4	57.0
30	6	1.0	2.2	59.2
31 34	1	.2	.4	59.6
40	1	.2	.4	59.9
46	2	.3	.7	60.7
49	1	.2	.4	61.0
50	1	.2	.4	61.4
54	9	1.5	3.3	64.7
55	1	.2	.4	65.1
60	1	.2	.4	65.4
62	2	.3	.7	66.2
65	1	.2	.4	66.5 67.6
66	2	.5	1.1	67.6
70	3	.3 .5	.7 1.1	68.4 69.5
75	1	.5 .2		69.5
l '3	l '	.2	.4	6.80

80	4	.7	1.5	71.3
83	1	.2	.4	71.7
90	2	.3	.7	72.4
99	1	.2	.4	72.4
100	9			
106		1.5	3.3	76.1
110	2	.3	.7	76.8
	1	.2	.4	77.2
111	1	.2	.4	77.6
120	1	.2	.4	77.9
122	1	.2	.4	78.3
123	1	.2	.4	78.7
125	1	.2	.4	79.0
127	1	.2	.4	79.4
134	1	.2	.4	79.8
136	1	.2	.4	80.1
150	2	.3	.7	80.9
159	1	.2	.4	81.3
160	2	.3	.7	82.0
175	1	.2	.4	82.4
178	1	.2	.4	82.7
200	3	.5	1.1	83.8
217	1	.2	.4	84.2
221	1	.2	.4	84.6
230	1	.2	.4	84.9
237	1	.2	.4	85.3
250	4	.7	1.5	86.8
261	1	.2	.4	87.1
270	2			
280		.3	.7	87.9
300	1	.2	.4	88.2
320	3	.5	1.1	89.3
	1	.2	.4	89.7
400	1	.2	.4	90.1
440	1	.2	.4	90.4
490	1	.2	.4	90.8
540	1	.2	.4	91.2
578	1	.2	.4	91.5
579	1	.2	.4	91.9
600	2	.3	.7	92.6
660	1	.2	.4	93.0
683	1	.2	.4	93.4
700	1	.2	.4	93.8
711	1	.2	.4	94.1
900	1	.2	.4	94.5
923	1	.2	.4	94.9
980	1	.2	.4	95.2
1200	1	.2	.4	95.6
1230	1	.2	.4	96.0
<u> </u>		·-	1	1

	1400	1	.2	.4	96.3
	2000	1	.2	.4	96.7
	2121	1	.2	.4	97.1
	2534	1	.2	.4	97.4
	2809	1	.2	.4	97.8
	3000	1	.2	.4	98.2
	4000	1	.2	.4	98.5
	5000	1	.2	.4	98.9
	5600	1	.2	.4	99.3
	7000	1	.2	.4	99.6
	10000	1	.2	.4	100.0
	Total	272	46.4	100.0	
Missing	999	314	53.6		
Total		586	100.0		

a Agency size strata = Over 100 Officers

# unsolved rapes--cases with DNA(a)

		_			Cumulative
		Frequency	Percent	Valid Percent	Percent
Valid	0	36	6.1	15.4	15.4
	1	17	2.9	7.3	22.6
	2	16	2.7	6.8	29.5
	3	11	1.9	4.7	34.2
	4	15	2.6	6.4	40.6
	5	8	1.4	3.4	44.0
	6	7	1.2	3.0	47.0
	7	4	.7	1.7	48.7
	8	4	.7	1.7	50.4
	9	2	.3	.9	51.3
	10	6	1.0	2.6	53.8
	11	1	.2	.4	54.3
	12	3	.5	1.3	55.6
	13	2	.3	.9	56.4
	14	3	.5	1.3	57.7
	15	6	1.0	2.6	60.3
	16	2	.3	.9	61.1
	18	2	.3	.9	62.0
	19	1	.2	.4	62.4
	20	12	2.0	5.1	67.5
	21	1	.2	.4	67.9
	22	1	.2	.4	68.4
	23	1	.2	.4	68.8
	24	1	.2	.4	69.2
	25	6	1.0	2.6	71.8
	26	1	.2	.4	72.2
	30	2	.3	.9	73.1

35	1	.2	.4	73.5
40	4	.7	1.7	75.2
43	1	.2	.4	75.6
46	1	.2	.4	76.1
50	5	.9	2.1	78.2
55	2	.3	.9	79.1
57	1	.2	.4	79.5
59	1	.2	.4	79.9
60	4	.7	1.7	81.6
66	1	.2	.4	82.1
67	1	.2	.4	82.5
75	1	.2	.4	82.9
80	1	.2	.4	83.3
87	1	.2	.4	83.8
88	1	.2	.4	84.2
90	2	.3	.9	85.0
99	2	.3	.9	85.9
100	5	.9	2.1	88.0
120	1	.2	.4	88.5
150	2	.3	.9	89.3
175	1	.2	.4	89.7
180	1	.2	.4	90.2
192	1	.2	.4	90.6
200	2	.3	.9	91.5
215	1	.2	.4	91.9
250	3	.5	1.3	93.2
360	1	.2	.4	93.6
450	2	.3	.9	94.4
478	1	.2	.4	94.9
490	1	.2	.4	95.3
492	1	.2	.4	95.7
500	1	.2	.4	96.2
749	1	.2	.4	96.6
1000	1	.2	.4	97.0
1020	1	.2	.4	97.4
1200	1	.2	.4	97.9
1500	1	.2	.4	98.3
1600	1	.2	.4	98.7
2000	1	.2	.4	99.1
3800	1	.2	.4	99.6
5400	1	.2	.4	100.0
Total	234	39.9	100.0	
Missing 999 Total	352	60.1		
a Agency size strata	586	100.0		

a Agency size strata = Over 100 Officers

## educated guess not possible(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	271	46.2	46.5	46.5
	No	312	53.2	53.5	100.0
	Total	583	99.5	100.0	
Missing	999	3	.5		
Total		586	100.0		

a Agency size strata = Over 100 Officers

## unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	61	10.4	10.4	10.4
	No	525	89.6	89.6	100.0
	Total	586	100.0	100.0	

a Agency size strata = Over 100 Officers

## total unsolved rape cases with possible biological evidence(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	171	29.2	53.4	53.4
Valla	1	14			
	2		2.4	4.4	57.8
	3	10	1.7	3.1	60.9
		5	.9	1.6	62.5
	4	13	2.2	4.1	66.6
	5	6	1.0	1.9	68.4
	6	7	1.2	2.2	70.6
	7	5	.9	1.6	72.2
	8	5	.9	1.6	73.8
	9	2	.3	.6	74.4
	10	12	2.0	3.8	78.1
	11	2	.3	.6	78.8
	12	2	.3	.6	79.4
	13	1	.2	.3	79.7
	15	2	.3	.6	80.3
	16	2	.3	.6	80.9
	18	2	.3	.6	81.6
	19	1	.2	.3	81.9
	20	6	1.0	1.9	83.8
	21	1	.2	.3	84.1
	22	1	.2	.3	84.4
	24	2	.3	.6	85.0
	25	5	.9	1.6	86.6

l 2	27	1	.2	.3	86.9
	30	3	.5	.9	87.8
	35	1	.2	.3	88.1
	19	1	.2	.3	88.4
	50	2	.3	.6	89.1
	54	1	.2	.3	89.4
	55	1	.2	.3	89.7
	59	1	.2	.3	90.0
	30	3	.5	.9	90.9
	61	1	.2	.3	91.3
	66	1	.2	.3	91.6
	74	1	.2	.3	91.9
	30	1	.2	.3	92.2
	90	1	.2	.3	92.5
g	92	1	.2	.3	92.8
g	95	1	.2	.3	93.1
	99	1	.2	.3	93.4
	100	3	.5	.9	94.4
1	116	1	.2	.3	94.7
1	120	1	.2	.3	95.0
1	127	1	.2	.3	95.3
1	150	1	.2	.3	95.6
1	180	1	.2	.3	95.9
1	192	1	.2	.3	96.3
2	200	1	.2	.3	96.6
2	250	1	.2	.3	96.9
3	336	1	.2	.3	97.2
3	350	1	.2	.3	97.5
5	573	1	.2	.3	97.8
5	596	1	.2	.3	98.1
6	800	2	.3	.6	98.8
	<b>7</b> 49	1	.2	.3	99.1
1	1230	1	.2	.3	99.4
3	3800	1	.2	.3	99.7
7	7000	1	.2	.3	100.0
	Γotal	320	54.6	100.0	
	999	266	45.4		
Total		586	100.0		

a Agency size strata = Over 100 Officers

total unsolved homicide cases with possible biological evidence(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	204	34.8	62.8	62.8
	1	22	3.8	6.8	69.5
	2	21	3.6	6.5	76.0
	3	10	1.7	3.1	79.1
	4	10	1.7	3.1	82.2
	5	8	1.4	2.5	84.6
	6	5	.9	1.5	86.2
	7	2	.3	.6	86.8
	8	2	.3	.6	87.4
	9	1	.2	.3	87.7
	10	5	.9	1.5	89.2
	11	1	.2	.3	89.5
	12	3	.5	.9	90.5
	14	1	.2	.3	90.8
	15	3	.5	.9	91.7
	17	1	.2	.3	92.0
	20	3	.5	.9	92.9
	21	1	.2	.3	93.2
	25	2	.3	.6	93.8
	27	2	.3	.6	94.5
	30	2	.3	.6	95.1
	40	1	.2	.3	95.4
	49	1	.2	.3	95.7
	50	1	.2	.3	96.0
	60	1	.2	.3	96.3
	64	1	.2	.3	96.6
	75	1	.2	.3	96.9
	87	1	.2	.3	97.2
	99	1	.2	.3	97.5
	113	1	.2	.3	97.8
	115	1	.2	.3	98.2
	120	1	.2	.3	98.5
	200	1	.2	.3	98.8
	300	1	.2	.3	99.1
	400	1	.2	.3	99.4
	1000	1	.2	.3	99.7
	9999	1	.2	.3	100.0
	Total	325	55.5	100.0	
Missing	999	261	44.5		
Total		586	100.0		

a Agency size strata = Over 100 Officers

total unsolved property crimes with possible biological evidence(a)

					Cumulative
		Frequency	Percent	Valid Percent	Percent
Valid	0	135	23.0	64.3	64.3
	1	8	1.4	3.8	68.1
	2	6	1.0	2.9	71.0
	3	1	.2	.5	71.4
	4	2	.3	1.0	72.4
	5	6	1.0	2.9	75.2
	6	4	.7	1.9	77.1
	7	1	.2	.5	77.6
	10	8	1.4	3.8	81.4
	12	1	.2	.5	81.9
	14	1	.2	.5	82.4
	15	2	.3	1.0	83.3
	18	1	.2	.5	83.8
	20	5	.9	2.4	86.2
	25	2	.3	1.0	87.1
	30	2	.3	1.0	88.1
	36	1	.2	.5	88.6
	40	1	.2	.5	89.0
	50	9	1.5	4.3	93.3
	71	1	.2	.5	93.8
	90	1	.2	.5	94.3
	100	3	.5	1.4	95.7
	205	1	.2	.5	96.2
	500	1	.2	.5	96.7
	576	1	.2	.5	97.1
	800	1	.2	.5	97.6
	1200	1	.2	.5	98.1
	2000	1	.2	.5	98.6
	2800	1	.2	.5	99.0
	3447	1	.2	.5	99.5
	150000	1	.2	.5	100.0
	Total	210	35.8	100.0	
Missing	999	376	64.2		
Total		586	100.0		

a Agency size strata = Over 100 Officers

## total unsolved "other" cases with possible biological evidence(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	91	15.5	85.0	85.0
	1	2	.3	1.9	86.9
	2	1	.2	.9	87.9
	3	1	.2	.9	88.8
	5	1	.2	.9	89.7
	6	1	.2	.9	90.7
	10	1	.2	.9	91.6
	20	1	.2	.9	92.5
	30	1	.2	.9	93.5
	50	1	.2	.9	94.4
	100	1	.2	.9	95.3
	104	1	.2	.9	96.3
	280	1	.2	.9	97.2
	396	1	.2	.9	98.1
	403	1	.2	.9	99.1
	100000	1	.2	.9	100.0
	Total	107	18.3	100.0	
Missing	999	479	81.7		
Total		586	100.0		

a Agency size strata = Over 100 Officers

total unsolved unnamed rape cases with possible biological evidence(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	71	12.1	34.8	34.8
	1	12	2.0	5.9	40.7
	2	18	3.1	8.8	49.5
	3	6	1.0	2.9	52.5
	4	12	2.0	5.9	58.3
	5	9	1.5	4.4	62.7
	6	3	.5	1.5	64.2
	7	1	.2	.5	64.7
	8	4	.7	2.0	66.7
	10	15	2.6	7.4	74.0
	12	3	.5	1.5	75.5
	13	5	.9	2.5	77.9
	14	1	.2	.5	78.4
	15	3	.5	1.5	79.9
	19	1	.2	.5	80.4
	20	1	.2	.5	80.9
	21	2	.3	1.0	81.9
	22	2	.3	1.0	82.8
	23	1	.2	.5	83.3
	24	1	.2	.5	83.8
	25	5	.9	2.5	86.3
	30	2	.3	1.0	87.3
	40	1	.2	.5	87.7
	50	3	.5	1.5	89.2
	58	1	.2	.5	89.7
	60	2	.3	1.0	90.7
	68	1	.2	.5	91.2
	70	1	.2	.5	91.7
	80	3	.5	1.5	93.1
	85	1	.2	.5	93.6
	90	1	.2	.5	94.1
	100	1	.2	.5	94.6
	120	2	.3	1.0	95.6
	150	2	.3	1.0	96.6
	183	1	.2	.5	97.1
	200	1	.2	.5	97.5
	300	1	.2	.5	98.0
	315	1	.2	.5	98.5
	350	1	.2	.5	99.0
	749	1	.2	.5	99.5
	3000	1	.2	.5	100.0
N 4: .	Total	204	34.8	100.0	
Missing	999	382	65.2		
Total		586	100.0		

total unsolved unnamed homicide cases with possible biological evidence(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	93	15.9	45.6	45.6
Valla	1	22	3.8	10.8	56.4
	2	18	3.1	8.8	65.2
	3	9	1.5	4.4	69.6
	4	9	1.5	4.4	74.0
	5	8	1.4	3.9	74.0
	6	5	.9	2.5	80.4
	7	3	.5	1.5	81.9
	8	3	.5	1.5	83.3
	9	1	.2	.5	83.8
	10	7	1.2	3.4	87.3
	12	3	.5	1.5	88.7
	15	2	.3	1.0	89.7
	16	1	.2	.5	90.2
	20	2	.3	1.0	91.2
	21	1	.2	.5	91.7
	22	1	.2	.5	92.2
	24	1	.2	.5	92.6
	25	2	.3	1.0	93.6
	30	1	.2	.5	94.1
	35	1	.2	.5	94.6
	40	1	.2	.5	95.1
	50	1	.2	.5	95.6
	54	1	.2	.5	96.1
	60	1	.2	.5	96.6
	63	1	.2	.5	97.1
	64	1	.2	.5	97.5
	65	1	.2	.5	98.0
	120	1	.2	.5	98.5
	175	1	.2	.5	99.0
	200	2	.3	1.0	100.0
	Total	204	34.8	100.0	
Missing	999	382	65.2		
Total		586	100.0		

a Agency size strata = Over 100 Officers

total unsolved unnamed property crimes with possible biological evidence(a)

					Cumulative
		Frequency	Percent	Valid Percent	Percent
Valid	0	70	11.9	51.1	51.1
	1	7	1.2	5.1	56.2
	2	4	.7	2.9	59.1
	3	2	.3	1.5	60.6
	4	1	.2	.7	61.3
	5	6	1.0	4.4	65.7
	6	1	.2	.7	66.4
	7	1	.2	.7	67.2
	9	1	.2	.7	67.9
	10	10	1.7	7.3	75.2
	12	1	.2	.7	75.9
	14	1	.2	.7	76.6
	15	5	.9	3.6	80.3
	20	4	.7	2.9	83.2
	25	1	.2	.7	83.9
	30	2	.3	1.5	85.4
	36	1	.2	.7	86.1
	40	1	.2	.7	86.9
	45	1	.2	.7	87.6
	50	4	.7	2.9	90.5
	54	1	.2	.7	91.2
	70	1	.2	.7	92.0
	100	4	.7	2.9	94.9
	102	1	.2	.7	95.6
	450	1	.2	.7	96.4
	479	1	.2	.7	97.1
	700	1	.2	.7	97.8
	2500	1	.2	.7	98.5
	3010	1	.2	.7	99.3
	135000	1	.2	.7	100.0
	Total	137	23.4	100.0	
Missing	999	449	76.6		
Total	oizo etrete -	586	100.0		

a Agency size strata = Over 100 Officers

total unsolved unnamed	"other"	cases with	nossible biolog	nical evidence(a)
total ulisolved ullilaliled	Othiel	Cases Willi	POSSIBLE DIGIO	ilicai evidelice(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	48	8.2	75.0	75.0
	1	1	.2	1.6	76.6
	2	1	.2	1.6	78.1
	5	2	.3	3.1	81.3
	9	1	.2	1.6	82.8
	10	1	.2	1.6	84.4
	15	1	.2	1.6	85.9
	50	2	.3	3.1	89.1
	99	2	.3	3.1	92.2
	100	1	.2	1.6	93.8
	158	1	.2	1.6	95.3
	365	1	.2	1.6	96.9
	9999	1	.2	1.6	98.4
	50000	1	.2	1.6	100.0
	Total	64	10.9	100.0	
Missing	999	522	89.1		
Total		586	100.0		

a Agency size strata = Over 100 Officers

## educated guess not possible(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	265	45.2	45.2	45.2
	No	321	54.8	54.8	100.0
	Total	586	100.0	100.0	

a Agency size strata = Over 100 Officers

## unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	50	8.5	8.5	8.5
	No	536	91.5	91.5	100.0
	Total	586	100.0	100.0	

a Agency size strata = Over 100 Officers

% cases	s with	rape	kits	only(a)
---------	--------	------	------	---------

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	98	16.7	33.1	33.1
	11-25%	15	2.6	5.1	38.2
	25-50%	26	4.4	8.8	47.0
	50-75%	51	8.7	17.2	64.2
	75-100%	106	18.1	35.8	100.0
	Total	296	50.5	100.0	
Missing	999	289	49.3		
	System	1	.2		
	Total	290	49.5		
Total		586	100.0		

a Agency size strata = Over 100 Officers

## % cases with rape kits and other DNA evidence(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	107	18.3	38.2	38.2
	11-25%	43	7.3	15.4	53.6
	25-50%	33	5.6	11.8	65.4
	50-75%	40	6.8	14.3	79.6
	75-100%	56	9.6	20.0	99.6
	9999	1	.2	.4	100.0
	Total	280	47.8	100.0	
Missing	999	306	52.2		
Total		586	100.0		

a Agency size strata = Over 100 Officers

#### % cases with other DNA evidence only(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	176	30.0	76.9	76.9
	11-25%	25	4.3	10.9	87.8
	25-50%	13	2.2	5.7	93.4
	50-75%	6	1.0	2.6	96.1
	75-100%	9	1.5	3.9	100.0
	Total	229	39.1	100.0	
Missing	999	357	60.9		
Total		586	100.0		

a Agency size strata = Over 100 Officers

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	190	32.4	32.5	32.5
	No	394	67.2	67.5	100.0
	Total	584	99.7	100.0	
Missing	999	2	.3		
Total		586	100.0		

a Agency size strata = Over 100 Officers

## unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	35	6.0	6.0	6.0
	No	551	94.0	94.0	100.0
	Total	586	100.0	100.0	

a Agency size strata = Over 100 Officers

# rape cases within statute of limitations(a)

					Cumulative
		Frequency	Percent	Valid Percent	Percent
Valid	0	46	7.8	25.4	25.4
	1	13	2.2	7.2	32.6
	2	13	2.2	7.2	39.8
	3	7	1.2	3.9	43.6
	4	8	1.4	4.4	48.1
	5	11	1.9	6.1	54.1
	6	6	1.0	3.3	57.5
	7	3	.5	1.7	59.1
	8	3	.5	1.7	60.8
	9	1	.2	.6	61.3
	10	10	1.7	5.5	66.9
	11	2	.3	1.1	68.0
	13	3	.5	1.7	69.6
	14	1	.2	.6	70.2
	15	4	.7	2.2	72.4
	16	2	.3	1.1	73.5
	18	1	.2	.6	74.0
	19	1	.2	.6	74.6
	20	4	.7	2.2	76.8
	21	2	.3	1.1	77.9
	22	1	.2	.6	78.5
	23	1	.2	.6	79.0

	24	2	.3	1.1	80.1
	25	2	.3	1.1	81.2
	30	4	.7	2.2	83.4
	35	2	.3	1.1	84.5
	36	1	.2	.6	85.1
	40	1	.2	.6	85.6
	50	3	.5	1.7	87.3
	59	1	.2	.6	87.8
	61	1	.2	.6	88.4
	70	1	.2	.6	89.0
	80	1	.2	.6	89.5
	92	1	.2	.6	90.1
	95	1	.2	.6	90.6
	100	2	.3	1.1	91.7
	111	1	.2	.6	92.3
	116	1	.2	.6	92.8
	120	1	.2	.6	93.4
	140	1	.2	.6	93.9
	250	1	.2	.6	94.5
	350	1	.2	.6	95.0
	500	1	.2	.6	95.6
	520	1	.2	.6	96.1
	525	1	.2	.6	96.7
	530	1	.2	.6	97.2
	533	1	.2	.6	97.8
	738	1	.2	.6	98.3
	1000	1	.2	.6	98.9
	5000	1	.2	.6	99.4
	7000	1	.2	.6	100.0
	Total	181	30.9	100.0	
Missing	999	405	69.1		
Total		586	100.0		

a Agency size strata = Over 100 Officers

unnamed rape cases within statute of limitations(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	42	7.2	28.4	28.4
	1	9	1.5	6.1	34.5
	2	12	2.0	8.1	42.6
	3	9	1.5	6.1	48.6
	4	9	1.5	6.1	54.7
	5	8	1.4	5.4	60.1
	6	5	.9	3.4	63.5
	7	2	.3	1.4	64.9
	8	4	.7	2.7	67.6
	9	1	.2	.7	68.2
	10	7	1.2	4.7	73.0
	11	1	.2	.7	73.6
	12	1	.2	.7	74.3
	13	4	.7	2.7	77.0
	14	2	.3	1.4	78.4
	15	3	.5	2.0	80.4
	20	1	.2	.7	81.1
	21	2	.3	1.4	82.4
	22	1	.2	.7	83.1
	23	1	.2	.7	83.8
	24	1	.2	.7	84.5
	25	2	.3	1.4	85.8
	30	1	.2	.7	86.5
	42	1	.2	.7	87.2
	50	2	.3	1.4	88.5
	58	1	.2	.7	89.2
	60	1	.2	.7	89.9
	70	1	.2	.7	90.5
	80	1	.2	.7	91.2
	100	1	.2	.7	91.9
	120	1	.2	.7	92.6
	130 140	1	.2	.7	93.2
	180	1	.2	.7	93.9
	200	1	.2	.7	94.6
	250	2	.3	1.4	95.9
	500	1	.2	.7	96.6
	533	1 1	.2	.7	97.3
	738	1	.2 .2	.7 .7	98.0 98.6
	3000	1	.2	. <i>1</i> .7	99.3
	7000		.2		100.0
	Total	1 148	.2 25.3	.7 100.0	100.0
Missing	999	438	25.3 74.7	100.0	
Total	300	438 586	100.0		

rape cases to exceed statute of limitations(a)

					Cumulative
		Frequency	Percent	Valid Percent	Percent
Valid	0	90	15.4	63.8	63.8
	1	8	1.4	5.7	69.5
	2	6	1.0	4.3	73.8
	3	4	.7	2.8	76.6
	4	2	.3	1.4	78.0
	5	3	.5	2.1	80.1
	6	3	.5	2.1	82.3
	7	1	.2	.7	83.0
	9	1	.2	.7	83.7
	10	2	.3	1.4	85.1
	11	2	.3	1.4	86.5
	13	1	.2	.7	87.2
	15	2	.3	1.4	88.7
	18	1	.2	.7	89.4
	20	1	.2	.7	90.1
	30	2	.3	1.4	91.5
	33	1	.2	.7	92.2
	40	1	.2	.7	92.9
	45	1	.2	.7	93.6
	46	1	.2	.7	94.3
	70	1	.2	.7	95.0
	100	1	.2	.7	95.7
	106	1	.2	.7	96.5
	110	1	.2	.7	97.2
	375	1	.2	.7	97.9
	705	1	.2	.7	98.6
	738	1	.2	.7	99.3
	9999	1	.2	.7	100.0
	Total	141	24.1	100.0	
Missing	999	445	75.9		
Total		586	100.0		

a Agency size strata = Over 100 Officers

unnamed rape cases to exceed	d statute of limitations(a)	)
------------------------------	-----------------------------	---

					Cumulative
		Frequency	Percent	Valid Percent	Percent
Valid	0	71	12.1	62.8	62.8
	1	5	.9	4.4	67.3
	2	6	1.0	5.3	72.6
	3	2	.3	1.8	74.3
	4	3	.5	2.7	77.0
	5	3	.5	2.7	79.6
	6	2	.3	1.8	81.4
	10	2	.3	1.8	83.2
	11	2	.3	1.8	85.0
	15	1	.2	.9	85.8
	24	1	.2	.9	86.7
	26	1	.2	.9	87.6
	30	2	.3	1.8	89.4
	36	1	.2	.9	90.3
	45	1	.2	.9	91.2
	55	1	.2	.9	92.0
	99	7	1.2	6.2	98.2
	200	1	.2	.9	99.1
	738	1	.2	.9	100.0
	Total	113	19.3	100.0	
Missing	999	473	80.7		
Total		586	100.0		

a Agency size strata = Over 100 Officers

# educated guess not possible(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	258	44.0	44.3	44.3
	No	325	55.5	55.7	100.0
	Total	583	99.5	100.0	
Missing	999	3	.5		
Total		586	100.0		

a Agency size strata = Over 100 Officers

unable	to	answer	comp	letelv(a)
ullable	w	aliswei	COLLID	ictely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	44	7.5	7.5	7.5
	No	541	92.3	92.5	100.0
	Total	585	99.8	100.0	
Missing	999	1	.2		
Total		586	100.0		

a Agency size strata = Over 100 Officers

#### unanalyzed evidence stored--central storage area(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	492	84.0	84.0	84.0
	No	94	16.0	16.0	100.0
	Total	586	100.0	100.0	

a Agency size strata = Over 100 Officers

#### unanalyzed evidence stored--decentralized storage area(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	23	3.9	3.9	3.9
	No	563	96.1	96.1	100.0
	Total	586	100.0	100.0	

a Agency size strata = Over 100 Officers

#### unanalyzed evidence stored--prosecutor's facility(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	8	1.4	1.4	1.4
	No	578	98.6	98.6	100.0
	Total	586	100.0	100.0	

a Agency size strata = Over 100 Officers

#### unanalyzed evidence stored--crime lab facility(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	143	24.4	24.4	24.4
	No	443	75.6	75.6	100.0
	Total	586	100.0	100.0	

a Agency size strata = Over 100 Officers

unanaly	zed	evidence	stored	other	a١
ullallal	<i>,</i> 25 u	CVIGCIICE	3101 Eu	OLITEIT	a,

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	17	2.9	2.9	2.9
	No	568	96.9	97.1	100.0
	Total	585	99.8	100.0	
Missing	999	1	.2		
Total		586	100.0		

a Agency size strata = Over 100 Officers

## sufficient space for long-term evidence storage?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Sufficient Storage	223	38.1	38.6	38.6
	Insufficient Storage	354	60.4	61.4	100.0
	Total	577	98.5	100.0	
Missing	999	9	1.5		
Total		586	100.0		

a Agency size strata = Over 100 Officers

## If insufficient, how critical is additional space?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Highly Critical	93	15.9	19.6	19.6
	Critical	266	45.4	56.1	75.7
	Pressing But Not Critical	115	19.6	24.3	100.0
	Total	474	80.9	100.0	
Missing	999	112	19.1		
Total		586	100.0		

a Agency size strata = Over 100 Officers

#### DNA not sent to lab applicable--not applicable(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	186	31.7	32.0	32.0
	No	396	67.6	68.0	100.0
	Total	582	99.3	100.0	
Missing	999	4	.7		
Total		586	100.0		

a Agency size strata = Over 100 Officers

DNA not sent	to lab ranknot	applicable(a)
--------------	----------------	---------------

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	91	15.5	82.0	82.0
	Second Most Significant	8	1.4	7.2	89.2
	Third Most Significant	12	2.0	10.8	100.0
	Total	111	18.9	100.0	
Missing	999	475	81.1		
Total		586	100.0		

a Agency size strata = Over 100 Officers

#### DNA not sent to lab applicable--no suspect identified(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	170	29.0	29.1	29.1
	No	415	70.8	70.9	100.0
	Total	585	99.8	100.0	
Missing	999	1	.2		
Total		586	100.0		

a Agency size strata = Over 100 Officers

#### DNA not sent to lab rank--no suspect identified(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	98	16.7	62.0	62.0
	Second Most Significant	35	6.0	22.2	84.2
	Third Most Significant	25	4.3	15.8	100.0
	Total	158	27.0	100.0	
Missing	999	428	73.0		
Total		586	100.0		

a Agency size strata = Over 100 Officers

#### DNA not sent to lab applicable--no suspect charged(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	65	11.1	11.1	11.1
	No	519	88.6	88.9	100.0
	Total	584	99.7	100.0	
Missing	999	2	.3		
Total		586	100.0		

a Agency size strata = Over 100 Officers

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	7	1.2	14.0	14.0
	Second Most Significant	29	4.9	58.0	72.0
	Third Most Significant	14	2.4	28.0	100.0
	Total	50	8.5	100.0	
Missing	999	536	91.5		
Total		586	100.0		

a Agency size strata = Over 100 Officers

## DNA not sent to lab applicable--guilty plea anticipated(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	95	16.2	16.2	16.2
	No	490	83.6	83.8	100.0
	Total	585	99.8	100.0	
Missing	999	1	.2		
Total		586	100.0		

a Agency size strata = Over 100 Officers

#### DNA not sent to lab rank--guilty plea anticipated(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	23	3.9	27.7	27.7
	Second Most Significant	27	4.6	32.5	60.2
	Third Most Significant	33	5.6	39.8	100.0
	Total	83	14.2	100.0	
Missing	999	503	85.8		
Total		586	100.0		

a Agency size strata = Over 100 Officers

## DNA not sent to lab applicable--DNA use uncertain(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	42	7.2	7.2	7.2
	No	543	92.7	92.8	100.0
	Total	585	99.8	100.0	
Missing	999	1	.2		
Total		586	100.0		

a Agency size strata = Over 100 Officers

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	4	.7	11.4	11.4
	Second Most Significant	15	2.6	42.9	54.3
	Third Most Significant	16	2.7	45.7	100.0
	Total	35	6.0	100.0	
Missing	999	551	94.0		
Total		586	100.0		

a Agency size strata = Over 100 Officers

## DNA not sent to lab applicable--lack of \$(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	56	9.6	9.6	9.6
	No	529	90.3	90.4	100.0
	Total	585	99.8	100.0	
Missing	999	1	.2		
Total		586	100.0		

a Agency size strata = Over 100 Officers

#### DNA not sent to lab rank--lack of \$(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	25	4.3	55.6	55.6
	Second Most Significant	10	1.7	22.2	77.8
	Third Most Significant	10	1.7	22.2	100.0
	Total	45	7.7	100.0	
Missing	999	541	92.3		
Total		586	100.0		

a Agency size strata = Over 100 Officers

## DNA not sent to lab applicable--inability to get timely results(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	72	12.3	12.3	12.3
	No	513	87.5	87.7	100.0
	Total	585	99.8	100.0	
Missing	999	1	.2		
Total		586	100.0		

DNA not sent to lab rankinability	to get timely	v results(a)
-----------------------------------	---------------	--------------

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	19	3.2	31.1	31.1
	Second Most Significant	28	4.8	45.9	77.0
	Third Most Significant	14	2.4	23.0	100.0
	Total	61	10.4	100.0	
Missing	999	525	89.6		
Total		586	100.0		

a Agency size strata = Over 100 Officers

#### DNA not sent to lab applicable--lab not processing requests(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	28	4.8	4.8	4.8
	No	557	95.1	95.2	100.0
	Total	585	99.8	100.0	
Missing	999	1	.2		
Total		586	100.0		

a Agency size strata = Over 100 Officers

#### DNA not sent to lab rank--lab not processing requests(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	8	1.4	32.0	32.0
	Second Most Significant	7	1.2	28.0	60.0
	Third Most Significant	10	1.7	40.0	100.0
	Total	25	4.3	100.0	
Missing	999	561	95.7		
Total		586	100.0		

a Agency size strata = Over 100 Officers

#### DNA not sent to lab applicable--uncertain where to send DNA(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	3	.5	.5	.5
	No	582	99.3	99.5	100.0
	Total	585	99.8	100.0	
Missing	999	1	.2		
Total		586	100.0		

a Agency size strata = Over 100 Officers

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	1	.2	33.3	33.3
	Third Most Significant	2	.3	66.7	100.0
	Total	3	.5	100.0	
Missing	999	583	99.5		
Total		586	100.0		

a Agency size strata = Over 100 Officers

## DNA not sent to lab applicable--analysis not requested by prosecutor(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	48	8.2	8.2	8.2
	No	537	91.6	91.8	100.0
	Total	585	99.8	100.0	
Missing	999	1	.2		
Total		586	100.0		

a Agency size strata = Over 100 Officers

# DNA not sent to lab rank--analysis not requested by prosecutor(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	6	1.0	18.2	18.2
	Second Most Significant	12	2.0	36.4	54.5
Third Most Significant		14	2.4	42.4	97.0
	9999	1	.2	3.0	100.0
	Total	33	5.6	100.0	
Missing	999	553	94.4		
Total		586	100.0		

a Agency size strata = Over 100 Officers

## DNA not sent to lab applicable--other(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	68	11.6	11.6	11.6
	No	517	88.2	88.4	100.0
	Total	585	99.8	100.0	
Missing	999	1	.2		
Total		586	100.0		

a Agency size strata = Over 100 Officers

DNA not sent to	lab rankother(a	a)
-----------------	-----------------	----

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	40	6.8	66.7	66.7
	Second Most Significant	11	1.9	18.3	85.0
Third Most Significant		9	1.5	15.0	100.0
	Total	60	10.2	100.0	
Missing	999	526	89.8		
Total		586	100.0		

a Agency size strata = Over 100 Officers

#### Department routinely collect DNA from property crimes?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	246	42.0	43.5	43.5
	No	319	54.4	56.5	100.0
	Total	565	96.4	100.0	
Missing	999	21	3.6		
Total		586	100.0		

a Agency size strata = Over 100 Officers

#### Lab accept AND process unnamed suspect DNA cases for prop. crimes?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	244	41.6	47.3	47.3
	No	272	46.4	52.7	100.0
	Total	516	88.1	100.0	
Missing	999	70	11.9		
Total		586	100.0		

a Agency size strata = Over 100 Officers

#### Lab accept but NOT process unnamed suspect DNA cases for prop. crimes?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	102	17.4	24.3	24.3
	No	318	54.3	75.7	100.0
	Total	420	71.7	100.0	
Missing	999	166	28.3		
Total		586	100.0		

a Agency size strata = Over 100 Officers

## unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	32	5.5	5.5	5.5
	No	553	94.4	94.5	100.0
	Total	585	99.8	100.0	
Missing	999	1	.2		
Total		586	100.0		

a Agency size strata = Over 100 Officers

## agency have cold case squad?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	221	37.7	38.6	38.6
	No	352	60.1	61.4	100.0
	Total	573	97.8	100.0	
Missing	999	13	2.2		
Total		586	100.0		

a Agency size strata = Over 100 Officers

## unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	12	2.0	2.1	2.1
	No	572	97.6	97.9	100.0
	Total	584	99.7	100.0	
Missing	999	2	.3		
Total		586	100.0		

a Agency size strata = Over 100 Officers

#### who to contact?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	999	575	98.1	98.1	98.1
	Gini Smart	1	.2	.2	98.3
	Hampton Police Department	1	.2	.2	98.5
	HPD-SIS	1	.2	.2	98.6
	Investigator MJ Ball	1	.2	.2	98.8
	J.R. Simpson	1	.2	.2	99.0
	Lt. Dominguez	1	.2	.2	99.1
	Lt. Myhre	1	.2	.2	99.3

Major Perez	1	.2	.2	99.5
Sat. Km Cavanah	1	.2	.2	99.7
Sergeant Byess	1	.2	.2	99.8
Sgt. Killilea	1	.2	.2	100.0
Total	586	100.0	100.0	

a Agency size strata = Over 100 Officers

# # of cold cases eligible for review(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	4	.7	2.5	2.5
	1	5	.9	3.1	5.7
	2	8	1.4	5.0	10.7
	3	8	1.4	5.0	15.7
	4	7	1.2	4.4	20.1
	5	7	1.2	4.4	24.5
	6	7	1.2	4.4	28.9
	7	5	.9	3.1	32.1
	8	11	1.9	6.9	39.0
	9	3	.5	1.9	40.9
	10	10	1.7	6.3	47.2
	11	2	.3	1.3	48.4
	12	4	.7	2.5	50.9
	14	1	.2	.6	51.6
	15	3	.5	1.9	53.5
	16	2	.3	1.3	54.7
	17	1	.2	.6	55.3
	18	1	.2	.6	56.0
	20	3	.5	1.9	57.9
	22	2	.3	1.3	59.1
	23	1	.2	.6	59.7
	24	1	.2	.6	60.4
	25	2	.3	1.3	61.6
	28	1	.2	.6	62.3
	30	4	.7	2.5	64.8
	33	1	.2	.6	65.4
	34	1	.2	.6	66.0
	35	1	.2	.6	66.7
	36	2	.3	1.3	67.9
	40	1	.2	.6	68.6
	42	2	.3	1.3	69.8
	43	1	.2	.6	70.4
	45	1	.2	.6	71.1
	46	1	.2	.6	71.7
	50	1	.2	.6	72.3

	51	1	.2	.6	73.0
	52	1	.2	.6	73.6
	53	1	.2	.6	74.2
	56	1	.2	.6	74.8
	59	1	.2	.6	75.5
	60	1	.2	.6	76.1
	64	1	.2	.6	76.7
	70	2	.3	1.3	78.0
	73	1	.2	.6	78.6
	75	2	.3	1.3	79.9
	78	1	.2	.6	80.5
	85	1	.2	.6	81.1
	90	2	.3	1.3	82.4
	100	3	.5	1.9	84.3
	113	1	.2	.6	84.9
	115	1	.2	.6	85.5
	120	1	.2	.6	86.2
	125	1	.2	.6	86.8
	168	1	.2	.6	87.4
	200	3	.5	1.9	89.3
	210	1	.2	.6	89.9
	220	1	.2	.6	90.6
	250	1	.2	.6	91.2
	252	1	.2	.6	91.8
	300	1	.2	.6	92.5
	350	1	.2	.6	93.1
	375	1	.2	.6	93.7
	377	1	.2	.6	94.3
	380	1	.2	.6	95.0
	400	2	.3	1.3	96.2
	500	3	.5	1.9	98.1
	700	1	.2	.6	98.7
	900	1	.2	.6	99.4
	8500	1	.2	.6	100.0
	Total	159	27.1	100.0	
Missing	999	427	72.9		
Total		586	100.0		

a Agency size strata = Over 100 Officers

# # of cold cases reviewed(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	2	.3	1.3	1.3
	1	10	1.7	6.3	7.5
	2	14	2.4	8.8	16.4
	3	13	2.2	8.2	24.5

Ī	4	10	1.7	6.3	30.8
	5	12	2.0	7.5	38.4
	6	9	1.5	5.7	44.0
	7	4	.7	2.5	46.5
	8	11	1.9	6.9	53.5
	9	3	.5	1.9	55.3
	10	14	2.4	8.8	64.2
	11	2	.3	1.3	65.4
	12	6	1.0	3.8	69.2
	15	3	.5	1.9	71.1
	16	3	.5	1.9	73.0
	17	1	.2	.6	73.6
	18	1	.2	.6	74.2
	20	5	.9	3.1	77.4
	22	1	.2	.6	78.0
	24	1	.2	.6	78.6
	25	2	.3	1.3	79.9
	33	1	.2	.6	80.5
	36	1	.2	.6	81.1
	40	1	.2	.6	81.8
	45	2	.3	1.3	83.0
	47	1	.2	.6	83.6
	50	3	.5	1.9	85.5
	52	2	.3	1.3	86.8
	54	1	.2	.6	87.4
	55	1	.2	.6	88.1
	60	2	.3	1.3	89.3
	64	1	.2	.6	89.9
	66	1	.2	.6	90.6
	70	1	.2	.6	91.2
	75	2	.3	1.3	92.5
	80	1	.2	.6	93.1
	90	1	.2	.6	93.7
	100	1	.2	.6	94.3
	125	1	.2	.6	95.0
	165	1	.2	.6	95.6
	168	1	.2	.6	96.2
	200	3	.5	1.9	98.1
	326	1	.2	.6	98.7
	375 1350	1	.2	.6	99.4
	1350 Total	1	.2	.6	100.0
Mincina	Total	159	27.1	100.0	
Missing	999	427	72.9		
Total		586	100.0		

a Agency size strata = Over 100 Officers

# of cold cases sent for DNA testing(a)

					Cumulative
		Frequency	Percent	Valid Percent	Percent
Valid	0	21	3.6	13.7	13.7
	1	32	5.5	20.9	34.6
	2	21	3.6	13.7	48.4
	3	17	2.9	11.1	59.5
	4	16	2.7	10.5	69.9
	5	8	1.4	5.2	75.2
	6	6	1.0	3.9	79.1
	7	3	.5	2.0	81.0
	8	1	.2	.7	81.7
	9	1	.2	.7	82.4
	10	9	1.5	5.9	88.2
	12	1	.2	.7	88.9
	15	3	.5	2.0	90.8
	18	1	.2	.7	91.5
	20	1	.2	.7	92.2
	22	1	.2	.7	92.8
	25	1	.2	.7	93.5
	30	1	.2	.7	94.1
	35	1	.2	.7	94.8
	38	1	.2	.7	95.4
	40	1	.2	.7	96.1
	45	1	.2	.7	96.7
	50	2	.3	1.3	98.0
	52	1	.2	.7	98.7
	75	1	.2	.7	99.3
	375	1	.2	.7	100.0
	Total	153	26.1	100.0	
Missing	999	433	73.9		
Total		586	100.0		

a Agency size strata = Over 100 Officers

# of additional cold cases potentially eligible for DNA testing(a)

					Cumulativa
		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	54	9.2	42.5	42.5
	1	6	1.0	4.7	47.2
	2	10	1.7	7.9	55.1
	3	2	.3	1.6	56.7
	4	4	.7	3.1	59.8
	5	9	1.5	7.1	66.9
	6	4	.7	3.1	70.1
	7	4	.7	3.1	73.2
	8	2	.3	1.6	74.8
	9	1	.2	.8	75.6
	10	2	.3	1.6	77.2
	15	2	.3	1.6	78.7
	16	1	.2	.8	79.5
	20	3	.5	2.4	81.9
	25	2	.3	1.6	83.5
	27	1	.2	.8	84.3
	30	1	.2	.8	85.0
	36	1	.2	.8	85.8
	42	1	.2	.8	86.6
	45	1	.2	.8	87.4
	50	3	.5	2.4	89.8
	55	1	.2	.8	90.6
	64	1	.2	.8	91.3
	87	1	.2	.8	92.1
	99	1	.2	.8	92.9
	135	1	.2	.8	93.7
	140	1	.2	.8	94.5
	160	1	.2	.8	95.3
	250	1	.2	.8	96.1
	300	2	.3	1.6	97.6
	850	1	.2	.8	98.4
	900	1	.2	.8	99.2
	1000 Total	1	.2	.8	100.0
Mincina	Total	127	21.7	100.0	
Missing	999	459	78.3		
Total		586	100.0		

a Agency size strata = Over 100 Officers

educated	auess	is not	possible	(a)
Caacatca	gucss	13 1101	POSSIBIO	141

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	79	13.5	13.6	13.6
	No	503	85.8	86.4	100.0
	Total	582	99.3	100.0	
Missing	999	4	.7		
Total		586	100.0		

a Agency size strata = Over 100 Officers

## unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	20	3.4	3.4	3.4
	No	565	96.4	96.6	100.0
	Total	585	99.8	100.0	
Missing	999	1	.2		
Total		586	100.0		

a Agency size strata = Over 100 Officers

## agency have plan to review [cold] cases for biological testing?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	169	28.8	49.4	49.4
	No	173	29.5	50.6	100.0
	Total	342	58.4	100.0	
Missing	999	244	41.6		
Total		586	100.0		

a Agency size strata = Over 100 Officers

## # of cases eligible for review(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	22	3.8	17.7	17.7
	1	5	.9	4.0	21.8
	2	17	2.9	13.7	35.5
	3	5	.9	4.0	39.5
	4	3	.5	2.4	41.9
	5	5	.9	4.0	46.0
	6	3	.5	2.4	48.4
	7	6	1.0	4.8	53.2
	8	3	.5	2.4	55.6
	9	2	.3	1.6	57.3

	10	7	1.2	5.6	62.9
	11	3	.5	2.4	65.3
	12	5	.9	4.0	69.4
	13	2	.3	1.6	71.0
	15	2	.3	1.6	72.6
	17	1	.2	.8	73.4
	18	1	.2	.8	74.2
	19	1	.2	.8	75.0
	20	7	1.2	5.6	80.6
	25	1	.2	.8	81.5
	26	1	.2	.8	82.3
	30	2	.3	1.6	83.9
	34	1	.2	.8	84.7
	40	1	.2	.8	85.5
	41	1	.2	.8	86.3
	50	2	.3	1.6	87.9
	55	1	.2	.8	88.7
	59	1	.2	.8	89.5
	61	1	.2	.8	90.3
	99	1	.2	.8	91.1
	130	1	.2	.8	91.9
	148	1	.2	.8	92.7
	150	3	.5	2.4	95.2
	158	1	.2	.8	96.0
	197	1	.2	.8	96.8
	270	1	.2	.8	97.6
	280	1	.2	.8	98.4
	400	1	.2	.8	99.2
	600	1	.2	.8	100.0
	Total	124	21.2	100.0	
Missing	999	462	78.8		
Total		586	100.0		

a Agency size strata = Over 100 Officers

# # of cases with potential for biological testing(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	29	4.9	26.1	26.1
	1	10	1.7	9.0	35.1
	2	19	3.2	17.1	52.3
	3	11	1.9	9.9	62.2
	4	7	1.2	6.3	68.5
	5	2	.3	1.8	70.3
	6	4	.7	3.6	73.9
	7	3	.5	2.7	76.6
	8	1	.2	.9	77.5

	9	1	.2	.9	78.4
	10	3	.5	2.7	81.1
	11	1	.2	.9	82.0
	12	1	.2	.9	82.9
	15	1	.2	.9	83.8
	20	5	.9	4.5	88.3
	24	1	.2	.9	89.2
	25	1	.2	.9	90.1
	30	1	.2	.9	91.0
	50	1	.2	.9	91.9
	60	1	.2	.9	92.8
	61	1	.2	.9	93.7
	99	2	.3	1.8	95.5
	100	2	.3	1.8	97.3
	130	1	.2	.9	98.2
	150	1	.2	.9	99.1
	400	1	.2	.9	100.0
	Total	111	18.9	100.0	
Missing	999	475	81.1		
Total		586	100.0		

a Agency size strata = Over 100 Officers

## educated guess is not possible(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	130	22.2	22.3	22.3
	No	452	77.1	77.7	100.0
	Total	582	99.3	100.0	
Missing	999	4	.7		
Total		586	100.0		

a Agency size strata = Over 100 Officers

## unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	31	5.3	5.3	5.3
	No	552	94.2	94.7	100.0
	Total	583	99.5	100.0	
Missing	999	3	.5		
Total		586	100.0		

a Agency size strata = Over 100 Officers

agency	interested	in usir	ng DNA	database?	(a)
--------	------------	---------	--------	-----------	-----

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	563	96.1	98.3	98.3
	No	10	1.7	1.7	100.0
	Total	573	97.8	100.0	
Missing	999	13	2.2		
Total		586	100.0		

a Agency size strata = Over 100 Officers

## has agency used DNA database for cases with no direct match?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	172	29.4	30.6	30.6
	No	391	66.7	69.4	100.0
	Total	563	96.1	100.0	
Missing	999	23	3.9		
Total		586	100.0		

a Agency size strata = Over 100 Officers

## Any comments?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	129	22.0	22.1	22.1
	No	456	77.8	77.9	100.0
	Total	585	99.8	100.0	
Missing	999	1	.2		
Total		586	100.0		

a Agency size strata = Over 100 Officers

# Agency size strata = 50-100 Officers

# STATE(a)

				Cumulative
	Frequency	Percent	Valid Percent	Percent
Valid AL	3	1.0	1.0	1.0
AR	2	.7	.7	1.7
AZ	9	3.1	3.1	4.9
CA	22	7.7	7.7	12.5
CO	3	1.0	1.0	13.6
CT	3	1.0	1.0	14.6
DE	1	.3	.3	15.0
FL	14	4.9	4.9	19.9
GA	12	4.2	4.2	24.0
ID	5	1.7	1.7	25.8
IL	18	6.3	6.3	32.1
IN	5	1.7	1.7	33.8
KS	3	1.0	1.0	34.8
KY	2	.7	.7	35.5
LA	4	1.4	1.4	36.9
MA	9	3.1	3.1	40.1
MD	4	1.4	1.4	41.5
ME	1	.3	.3	41.8
MI	16	5.6	5.6	47.4
MN	5	1.7	1.7	49.1
MO	6	2.1	2.1	51.2
MS	1	.3	.3	51.6
MT	1	.3	.3	51.9
NC	14	4.9	4.9	56.8
ND	1	.3	.3	57.1
NE	1	.3	.3	57.5
NJ	16	5.6	5.6	63.1
NM	3	1.0	1.0	64.1
NV	1	.3	.3	64.5
NY	8	2.8	2.8	67.2
ОН	19	6.6	6.6	73.9
OK	4	1.4	1.4	75.3
OR	4	1.4	1.4	76.7
PA	6	2.1	2.1	78.7
RI	2	.7	.7	79.4
SC	2	.7	.7	80.1
SD	1	.3	.3	80.5
TN	4	1.4	1.4	81.9
TX	20	7.0	7.0	88.9
UT	5	1.7	1.7	90.6
VA	10	3.5	3.5	94.1

WA	5	1.7	1.7	95.8
WI	10	3.5	3.5	99.3
WV	1	.3	.3	99.7
WY	1	.3	.3	100.0
Total	287	100.0	100.0	

a Agency size strata = 50-100 Officers

# Where are cases primarily sent for DNA analysis?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	State Agency Lab	238	82.9	83.5	83.5
Local Agency Lab		29	10.1	10.2	93.7
	FBI Lab	1	.3	.4	94.0
	Other	10	3.5	3.5	97.5
	Private Lab/Commercial Lab	7	2.4	2.5	100.0
	Total	285	99.3	100.0	
Missing	999	2	.7		
Total		287	100.0		

a Agency size strata = 50-100 Officers

# % rapes with possible DNA(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	28	9.8	10.1	10.1
	11-25%	17	5.9	6.1	16.2
	25-50%	41	14.3	14.8	31.0
	50-75%	60	20.9	21.7	52.7
	75-100%	131	45.6	47.3	100.0
	Total	277	96.5	100.0	
Missing	999	10	3.5		
Total		287	100.0		

a Agency size strata = 50-100 Officers

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	40	13.9	16.1	16.1
	11-25%	11	3.8	4.4	20.6
	25-50%	19	6.6	7.7	28.2
	50-75%	32	11.1	12.9	41.1
	75-100%	146	50.9	58.9	100.0
	Total	248	86.4	100.0	
Missing	999	39	13.6		
Total		287	100.0		

a Agency size strata = 50-100 Officers

# educated guess not possible(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	26	9.1	9.1	9.1
	No	261	90.9	90.9	100.0
	Total	287	100.0	100.0	

a Agency size strata = 50-100 Officers

# unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	2	.7	.7	.7
	No	284	99.0	99.3	100.0
	Total	286	99.7	100.0	
Missing	999	1	.3		
Total		287	100.0		

a Agency size strata = 50-100 Officers

# unsolved homicides--total cases(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	50	17.4	22.8	22.8
	1	46	16.0	21.0	43.8
	2	33	11.5	15.1	58.9
	3	23	8.0	10.5	69.4
	4	23	8.0	10.5	79.9
	5	16	5.6	7.3	87.2
	6	6	2.1	2.7	90.0
	7	8	2.8	3.7	93.6
	9	3	1.0	1.4	95.0

	10	3	1.0	1.4	96.3
	11	1	.3	.5	96.8
	12	1	.3	.5	97.3
	13	1	.3	.5	97.7
	15	3	1.0	1.4	99.1
	16	1	.3	.5	99.5
	33	1	.3	.5	100.0
	Total	219	76.3	100.0	
Missing	999	67	23.3		
	System	1	.3		
	Total	68	23.7		
Total		287	100.0		

a Agency size strata = 50-100 Officers

# unsolved homicides--cases with DNA(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	84	29.3	46.7	46.7
	1	49	17.1	27.2	73.9
	2	20	7.0	11.1	85.0
	3	10	3.5	5.6	90.6
	4	7	2.4	3.9	94.4
	5	3	1.0	1.7	96.1
	6	2	.7	1.1	97.2
	7	4	1.4	2.2	99.4
	13	1	.3	.6	100.0
	Total	180	62.7	100.0	
Missing	999	107	37.3		
Total		287	100.0		

a Agency size strata = 50-100 Officers

# unsolved rapes--total cases(a)

					Cumulative
		Frequency	Percent	Valid Percent	Percent
Valid	0	53	18.5	32.1	32.1
	1	16	5.6	9.7	41.8
	2	15	5.2	9.1	50.9
	3	3	1.0	1.8	52.7
	4	6	2.1	3.6	56.4
	5	8	2.8	4.8	61.2
	6	6	2.1	3.6	64.8
	7	1	.3	.6	65.5
	8	2	.7	1.2	66.7
	9	2	.7	1.2	67.9
	10	12	4.2	7.3	75.2
	11	3	1.0	1.8	77.0
	12	4	1.4	2.4	79.4
	15	3	1.0	1.8	81.2
	17	1	.3	.6	81.8
	20	10	3.5	6.1	87.9
	27	1	.3	.6	88.5
	30	2	.7	1.2	89.7
	39	1	.3	.6	90.3
	40	1	.3	.6	90.9
	49	1	.3	.6	91.5
	50	2	.7	1.2	92.7
	67	1	.3	.6	93.3
	75	1	.3	.6	93.9
	80	2	.7	1.2	95.2
	100	1	.3	.6	95.8
	116	1	.3	.6	96.4
	120	1	.3	.6	97.0
	127	1	.3	.6	97.6
	167	1	.3	.6	98.2
	200	1	.3	.6	98.8
	280	1	.3	.6	99.4
	539	1	.3	.6	100.0
	Total	165	57.5	100.0	
Missing	999	122	42.5		
Total		287	100.0		

a Agency size strata = 50-100 Officers

# unsolved rapes--cases with DNA(a)

		Fraguanay	Doroont	Valid Darsont	Cumulative
Valid	0	Frequency	Percent	Valid Percent	Percent
valiu	1	48	16.7	34.3	34.3
	2	14	4.9	10.0	44.3
	3	15	5.2	10.7	55.0
	4	7	2.4	5.0	60.0
		5	1.7	3.6	63.6
	5	11	3.8	7.9	71.4
	6	5	1.7	3.6	75.0
	7	1	.3	.7	75.7
	8	3	1.0	2.1	77.9
	9	1	.3	.7	78.6
	10	7	2.4	5.0	83.6
	11	2	.7	1.4	85.0
	15	3	1.0	2.1	87.1
	17	2	.7	1.4	88.6
	20	2	.7	1.4	90.0
	22	1	.3	.7	90.7
	30	1	.3	.7	91.4
	37	1	.3	.7	92.1
	40	1	.3	.7	92.9
	50	1	.3	.7	93.6
	66	1	.3	.7	94.3
	75	1	.3	.7	95.0
	98	1	.3	.7	95.7
	99	4	1.4	2.9	98.6
	175	1	.3	.7	99.3
	220	1	.3	.7	100.0
	Total	140	48.8	100.0	
Missing	999	147	51.2		
Total		287	100.0		

a Agency size strata = 50-100 Officers

# educated guess not possible(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	105	36.6	36.7	36.7
	No	181	63.1	63.3	100.0
	Total	286	99.7	100.0	
Missing	999	1	.3		
Total		287	100.0		

a Agency size strata = 50-100 Officers

#### unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	9	3.1	3.1	3.1
	No	278	96.9	96.9	100.0
	Total	287	100.0	100.0	

a Agency size strata = 50-100 Officers

# total unsolved rape cases with possible biological evidence(a)

					Cumulative
		Frequency	Percent	Valid Percent	Percent
Valid	0	114	39.7	60.0	60.0
	1	15	5.2	7.9	67.9
	2	12	4.2	6.3	74.2
	3	5	1.7	2.6	76.8
	4	3	1.0	1.6	78.4
	5	8	2.8	4.2	82.6
	6	3	1.0	1.6	84.2
	7	1	.3	.5	84.7
	8	3	1.0	1.6	86.3
	9	3	1.0	1.6	87.9
	10	4	1.4	2.1	90.0
	11	2	.7	1.1	91.1
	15	2	.7	1.1	92.1
	20	3	1.0	1.6	93.7
	21	1	.3	.5	94.2
	23	1	.3	.5	94.7
	27	1	.3	.5	95.3
	28	1	.3	.5	95.8
	30	1	.3	.5	96.3
	35	3	1.0	1.6	97.9
	50	1	.3	.5	98.4
	100	1	.3	.5	98.9
	167	1	.3	.5	99.5
	439	1	.3	.5	100.0
	Total	190	66.2	100.0	
Missing	999	97	33.8		
Total		287	100.0		

a Agency size strata = 50-100 Officers

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	126	43.9	67.4	67.4
	1	25	8.7	13.4	80.7
	2	8	2.8	4.3	85.0
	3	9	3.1	4.8	89.8
	4	6	2.1	3.2	93.0
	5	5	1.7	2.7	95.7
	6	1	.3	.5	96.3
	7	1	.3	.5	96.8
	8	1	.3	.5	97.3
	10	2	.7	1.1	98.4
	11	1	.3	.5	98.9
	17	1	.3	.5	99.5
	99	1	.3	.5	100.0
	Total	187	65.2	100.0	
Missing	999	100	34.8		
Total	-:44-	287	100.0		

a Agency size strata = 50-100 Officers

# total unsolved property crimes with possible biological evidence(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	88	30.7	67.2	67.2
	1	4	1.4	3.1	70.2
	2	5	1.7	3.8	74.0
	3	6	2.1	4.6	78.6
	5	3	1.0	2.3	80.9
	6	1	.3	.8	81.7
	8	1	.3	.8	82.4
	9	1	.3	.8	83.2
	10	5	1.7	3.8	87.0
	12	1	.3	.8	87.8
	15	2	.7	1.5	89.3
	17	1	.3	.8	90.1
	20	2	.7	1.5	91.6
	40	1	.3	.8	92.4
	45	1	.3	.8	93.1
	52	1	.3	.8	93.9
	90	1	.3	.8	94.7
	100	1	.3	.8	95.4
	200	3	1.0	2.3	97.7
	500	1	.3	.8	98.5
	2183	1	.3	.8	99.2

	6896	1	.3	.8	100.0
	Total	131	45.6	100.0	
Missing	999	156	54.4		
Total		287	100.0		

a Agency size strata = 50-100 Officers

# total unsolved "other" cases with possible biological evidence(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	67	23.3	83.8	83.8
	1	4	1.4	5.0	88.8
	2	2	.7	2.5	91.3
	5	2	.7	2.5	93.8
	10	2	.7	2.5	96.3
	12	1	.3	1.3	97.5
	25	1	.3	1.3	98.8
	99	1	.3	1.3	100.0
	Total	80	27.9	100.0	
Missing	999	207	72.1		
Total		287	100.0		

a Agency size strata = 50-100 Officers

# total unsolved unnamed rape cases with possible biological evidence(a)

					Cumulative
		Frequency	Percent	Valid Percent	Percent
Valid	0	53	18.5	45.3	45.3
	1	16	5.6	13.7	59.0
	2	10	3.5	8.5	67.5
	3	5	1.7	4.3	71.8
	4	4	1.4	3.4	75.2
	5	13	4.5	11.1	86.3
	6	3	1.0	2.6	88.9
	8	2	.7	1.7	90.6
	9	1	.3	.9	91.5
	10	2	.7	1.7	93.2
	11	2	.7	1.7	94.9
	15	2	.7	1.7	96.6
	20	1	.3	.9	97.4
	27	1	.3	.9	98.3
	30	1	.3	.9	99.1
	99	1	.3	.9	100.0
	Total	117	40.8	100.0	
Missing	999	170	59.2		
Total		287	100.0		

a Agency size strata = 50-100 Officers

total unsolved unnamed	homicido casos with	h nassibla bialagia	al avidanca(a)
totai unsoived unnamed	nomiciae cases witi	n bossible blolodic	ai evidence(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	61	21.3	53.0	53.0
	1	26	9.1	22.6	75.7
	2	4	1.4	3.5	79.1
	3	7	2.4	6.1	85.2
	4	6	2.1	5.2	90.4
	5	2	.7	1.7	92.2
	7	4	1.4	3.5	95.7
	10	2	.7	1.7	97.4
	15	1	.3	.9	98.3
	20	1	.3	.9	99.1
	80	1	.3	.9	100.0
	Total	115	40.1	100.0	
Missing	999	172	59.9		
Total		287	100.0		

a Agency size strata = 50-100 Officers

# total unsolved unnamed property crimes with possible biological evidence(a)

					Cumulative
		Frequency	Percent	Valid Percent	Percent
Valid	0	41	14.3	53.2	53.2
	1	3	1.0	3.9	57.1
	2	7	2.4	9.1	66.2
	3	4	1.4	5.2	71.4
	4	2	.7	2.6	74.0
	5	2	.7	2.6	76.6
	6	1	.3	1.3	77.9
	9	2	.7	2.6	80.5
	10	3	1.0	3.9	84.4
	12	1	.3	1.3	85.7
	15	2	.7	2.6	88.3
	20	1	.3	1.3	89.6
	40	1	.3	1.3	90.9
	80	1	.3	1.3	92.2
	100	1	.3	1.3	93.5
	200	3	1.0	3.9	97.4
	1381	1	.3	1.3	98.7
	1479	1	.3	1.3	100.0
	Total	77	26.8	100.0	
Missing	999	210	73.2		
Total		287	100.0		

a Agency size strata = 50-100 Officers

total unsolved	unnamed "	other" c	ases with	nossible l	hiological	evidence(a)
total ulisolveu	ullilallieu	ULITEI C	ases willi	ひしろろいいせょ	Jibibuicai	evidelice(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	41	14.3	77.4	77.4
	1	3	1.0	5.7	83.0
	2	2	.7	3.8	86.8
	4	2	.7	3.8	90.6
	5	1	.3	1.9	92.5
	10	1	.3	1.9	94.3
	25	1	.3	1.9	96.2
	100	1	.3	1.9	98.1
	9999	1	.3	1.9	100.0
	Total	53	18.5	100.0	
Missing	999	234	81.5		
Total		287	100.0		

a Agency size strata = 50-100 Officers

# educated guess not possible(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	100	34.8	35.1	35.1
	No	185	64.5	64.9	100.0
	Total	285	99.3	100.0	
Missing	999	2	.7		
Total		287	100.0		

a Agency size strata = 50-100 Officers

# unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	11	3.8	3.8	3.8
	No	276	96.2	96.2	100.0
	Total	287	100.0	100.0	

a Agency size strata = 50-100 Officers

# % cases with rape kits only(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	54	18.8	37.0	37.0
	11-25%	9	3.1	6.2	43.2
	25-50%	14	4.9	9.6	52.7
	50-75%	17	5.9	11.6	64.4
	75-100%	52	18.1	35.6	100.0
	Total	146	50.9	100.0	
Missing	999	141	49.1		
Total		287	100.0		

a Agency size strata = 50-100 Officers

# % cases with rape kits and other DNA evidence(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	66	23.0	51.6	51.6
	11-25%	14	4.9	10.9	62.5
	25-50%	15	5.2	11.7	74.2
	50-75%	14	4.9	10.9	85.2
	75-100%	19	6.6	14.8	100.0
	Total	128	44.6	100.0	
Missing	999	159	55.4		
Total		287	100.0		

a Agency size strata = 50-100 Officers

# % cases with other DNA evidence only(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	78	27.2	78.0	78.0
	11-25%	9	3.1	9.0	87.0
	25-50%	8	2.8	8.0	95.0
	50-75%	1	.3	1.0	96.0
	75-100%	4	1.4	4.0	100.0
	Total	100	34.8	100.0	
Missing	999	187	65.2		
Total		287	100.0		

a Agency size strata = 50-100 Officers

# educated guess not possible(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	74	25.8	25.8	25.8
	No	213	74.2	74.2	100.0
	Total	287	100.0	100.0	

a Agency size strata = 50-100 Officers

# unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	7	2.4	2.4	2.4
	No	280	97.6	97.6	100.0
	Total	287	100.0	100.0	

a Agency size strata = 50-100 Officers

rape cases within statute of limitations(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	44	15.3	37.0	37.0
valid	1				
		22	7.7	18.5	55.5
	2	11	3.8	9.2	64.7
	3	6	2.1	5.0	69.7
	4	3	1.0	2.5	72.3
	5	8	2.8	6.7	79.0
	6	3	1.0	2.5	81.5
	7	1	.3	.8	82.4
	8	2	.7	1.7	84.0
	9	2	.7	1.7	85.7
	10	3	1.0	2.5	88.2
	11	2	.7	1.7	89.9
	12	1	.3	.8	90.8
	14	1	.3	.8	91.6
	17	1	.3	.8	92.4
	21	1	.3	.8	93.3
	25	1	.3	.8	94.1
	27	1	.3	.8	95.0
	30	1	.3	.8	95.8
	35	1	.3	.8	96.6
	40	1	.3	.8	97.5
	50	1	.3	.8	98.3
	120	1	.3	.8	99.2
	190	1	.3	.8	100.0
	Total	119	41.5	100.0	
Missing	999	168	58.5		
Total		287	100.0		
		- 50 100 Office			

a Agency size strata = 50-100 Officers

unnamed rape cases	within	statute	of I	limitations(a	1)
--------------------	--------	---------	------	---------------	----

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	40	13.9	41.7	41.7
	1	18	6.3	18.8	60.4
	2	13	4.5	13.5	74.0
	3	5	1.7	5.2	79.2
	4	2	.7	2.1	81.2
	5	7			
	6	-	2.4	7.3	88.5
	7	1	.3	1.0	89.6
	8	1	.3	1.0	90.6
		2	.7	2.1	92.7
	10	1	.3	1.0	93.8
	12	1	.3	1.0	94.8
	15	1	.3	1.0	95.8
	17	1	.3	1.0	96.9
	25	1	.3	1.0	97.9
	27	1	.3	1.0	99.0
	30	1	.3	1.0	100.0
	Total	96	33.4	100.0	
Missing	999	191	66.6		
Total		287	100.0		

a Agency size strata = 50-100 Officers

# rape cases to exceed statute of limitations(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	78	27.2	77.2	77.2
	1	6	2.1	5.9	83.2
	2	2	.7	2.0	85.1
	3	3	1.0	3.0	88.1
	4	1	.3	1.0	89.1
	5	3	1.0	3.0	92.1
	8	2	.7	2.0	94.1
	9	1	.3	1.0	95.0
	10	1	.3	1.0	96.0
	11	1	.3	1.0	97.0
	15	2	.7	2.0	99.0
	58	1	.3	1.0	100.0
	Total	101	35.2	100.0	
Missing	999	186	64.8		
Total		287	100.0		

a Agency size strata = 50-100 Officers

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	61	21.3	73.5	73.5
	1	3	1.0	3.6	77.1
	2	5	1.7	6.0	83.1
	3	4	1.4	4.8	88.0
	5	2	.7	2.4	90.4
	8	3	1.0	3.6	94.0
	10	1	.3	1.2	95.2
	15	1	.3	1.2	96.4
	35	1	.3	1.2	97.6
	99	2	.7	2.4	100.0
	Total	83	28.9	100.0	
Missing	999	204	71.1		
Total		287	100.0		

a Agency size strata = 50-100 Officers

# educated guess not possible(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	93	32.4	32.4	32.4
	No	194	67.6	67.6	100.0
	Total	287	100.0	100.0	

a Agency size strata = 50-100 Officers

# unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	7	2.4	2.4	2.4
	No	280	97.6	97.6	100.0
	Total	287	100.0	100.0	

a Agency size strata = 50-100 Officers

# unanalyzed evidence stored--central storage area(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	244	85.0	85.3	85.3
	No	42	14.6	14.7	100.0
	Total	286	99.7	100.0	
Missing	999	1	.3		
Total		287	100.0		

a Agency size strata = 50-100 Officers

#### unanalyzed evidence stored--decentralized storage area(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	5	1.7	1.7	1.7
	No	282	98.3	98.3	100.0
	Total	287	100.0	100.0	

a Agency size strata = 50-100 Officers

### unanalyzed evidence stored--prosecutor's facility(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	5	1.7	1.7	1.7
	No	282	98.3	98.3	100.0
	Total	287	100.0	100.0	

a Agency size strata = 50-100 Officers

#### unanalyzed evidence stored--crime lab facility(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	56	19.5	19.5	19.5
	No	231	80.5	80.5	100.0
	Total	287	100.0	100.0	

a Agency size strata = 50-100 Officers

#### unanalyzed evidence stored--other(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	13	4.5	4.5	4.5
	No	273	95.1	95.5	100.0
	Total	286	99.7	100.0	
Missing	System	1	.3		
Total		287	100.0		

a Agency size strata = 50-100 Officers

sufficient space	for	long-term evid	dence storage?(a	a)	
------------------	-----	----------------	------------------	----	--

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Sufficient Storage	119	41.5	42.0	42.0
Insufficient Storage	164	57.1	58.0	100.0	
	Total	283	98.6	100.0	
Missing	999	4	1.4		
Total		287	100.0		

a Agency size strata = 50-100 Officers

# If insufficient, how critical is additional space?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Highly Critical	30	10.5	12.9	12.9
	Critical	127	44.3	54.5	67.4
	Pressing But Not Critical	76	26.5	32.6	100.0
	Total	233	81.2	100.0	
Missing	999	54	18.8		
Total		287	100.0		

a Agency size strata = 50-100 Officers

#### DNA not sent to lab applicable--not applicable(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	111	38.7	38.8	38.8
	No	175	61.0	61.2	100.0
	Total	286	99.7	100.0	
Missing	999	1	.3		
Total		287	100.0		

a Agency size strata = 50-100 Officers

### DNA not sent to lab rank--not applicable(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	61	21.3	92.4	92.4
	Second Most Significant	3	1.0	4.5	97.0
T 8	Third Most Significant	2	.7	3.0	100.0
	Total	66	23.0	100.0	
Missing	999	221	77.0		
Total		287	100.0		

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	89	31.0	31.1	31.1
	No	197	68.6	68.9	100.0
	Total	286	99.7	100.0	
Missing	999	1	.3		
Total		287	100.0		

a Agency size strata = 50-100 Officers

#### DNA not sent to lab rank--no suspect identified(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	46	16.0	64.8	64.8
	Second Most Significant	18	6.3	25.4	90.1
	Third Most Significant	7	2.4	9.9	100.0
	Total	71	24.7	100.0	
Missing	999	216	75.3		
Total		287	100.0		

a Agency size strata = 50-100 Officers

#### DNA not sent to lab applicable--no suspect charged(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	18	6.3	6.3	6.3
	No	269	93.7	93.7	100.0
	Total	287	100.0	100.0	

a Agency size strata = 50-100 Officers

#### DNA not sent to lab rank--no suspect charged(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	6	2.1	40.0	40.0
Significant Third Most	Second Most Significant	6	2.1	40.0	80.0
	Third Most Significant	3	1.0	20.0	100.0
	Total	15	5.2	100.0	
Missing	999	272	94.8		
Total		287	100.0		

a Agency size strata = 50-100 Officers

# DNA not sent to lab applicable--guilty plea anticipated(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	30	10.5	10.5	10.5
	No	257	89.5	89.5	100.0
	Total	287	100.0	100.0	

a Agency size strata = 50-100 Officers

# DNA not sent to lab rank--guilty plea anticipated(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	10	3.5	40.0	40.0
	Second Most Significant	8	2.8	32.0	72.0
	Third Most Significant	7	2.4	28.0	100.0
	Total	25	8.7	100.0	
Missing	999	262	91.3		
Total		287	100.0		

a Agency size strata = 50-100 Officers

#### DNA not sent to lab applicable--DNA use uncertain(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	18	6.3	6.3	6.3
	No	269	93.7	93.7	100.0
	Total	287	100.0	100.0	

a Agency size strata = 50-100 Officers

#### DNA not sent to lab rank--DNA use uncertain(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	3	1.0	18.8	18.8
	Second Most Significant	6	2.1	37.5	56.3
	Third Most Significant	7	2.4	43.8	100.0
	Total	16	5.6	100.0	
Missing	999	271	94.4		
Total		287	100.0		

a Agency size strata = 50-100 Officers

#### DNA not sent to lab applicable--lack of \$(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	25	8.7	8.7	8.7
	No	262	91.3	91.3	100.0
	Total	287	100.0	100.0	

a Agency size strata = 50-100 Officers

#### DNA not sent to lab rank--lack of \$(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	10	3.5	52.6	52.6
	Second Most Significant	5	1.7	26.3	78.9
	Third Most Significant	4	1.4	21.1	100.0
	Total	19	6.6	100.0	
Missing	999	268	93.4		
Total		287	100.0		

a Agency size strata = 50-100 Officers

#### DNA not sent to lab applicable--inability to get timely results(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	18	6.3	6.3	6.3
	No	269	93.7	93.7	100.0
	Total	287	100.0	100.0	

a Agency size strata = 50-100 Officers

#### DNA not sent to lab rank--inability to get timely results(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	7	2.4	38.9	38.9
	Second Most Significant	9	3.1	50.0	88.9
	Third Most Significant	2	.7	11.1	100.0
	Total	18	6.3	100.0	
Missing	999	269	93.7		
Total		287	100.0		

a Agency size strata = 50-100 Officers

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	5	1.7	1.7	1.7
	No	282	98.3	98.3	100.0
	Total	287	100.0	100.0	

a Agency size strata = 50-100 Officers

#### DNA not sent to lab rank--lab not processing requests(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	3	1.0	75.0	75.0
	Second Most Significant	1	.3	25.0	100.0
	Total	4	1.4	100.0	
Missing	999	283	98.6		
Total		287	100.0		

a Agency size strata = 50-100 Officers

# DNA not sent to lab applicable--uncertain where to send DNA(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	4	1.4	1.4	1.4
	No	283	98.6	98.6	100.0
	Total	287	100.0	100.0	

a Agency size strata = 50-100 Officers

#### DNA not sent to lab rank--uncertain where to send DNA(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	2	.7	50.0	50.0
	Second Most Significant	2	.7	50.0	100.0
	Total	4	1.4	100.0	
Missing	999	283	98.6		
Total		287	100.0		

a Agency size strata = 50-100 Officers

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	24	8.4	8.4	8.4
	No	263	91.6	91.6	100.0
	Total	287	100.0	100.0	

a Agency size strata = 50-100 Officers

### DNA not sent to lab rank--analysis not requested by prosecutor(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	6	2.1	28.6	28.6
	Second Most Significant	5	1.7	23.8	52.4
	Third Most Significant	10	3.5	47.6	100.0
	Total	21	7.3	100.0	
Missing	999	266	92.7		
Total		287	100.0		

a Agency size strata = 50-100 Officers

#### DNA not sent to lab applicable--other(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	15	5.2	5.2	5.2
	No	271	94.4	94.8	100.0
	Total	286	99.7	100.0	
Missing	999	1	.3		
Total		287	100.0		

a Agency size strata = 50-100 Officers

#### DNA not sent to lab rank--other(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	5	1.7	62.5	62.5
	Second Most Significant	3	1.0	37.5	100.0
	Total	8	2.8	100.0	
Missing	999	279	97.2		
Total		287	100.0		

a Agency size strata = 50-100 Officers

#### Department routinely collect DNA from property crimes?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	103	35.9	36.1	36.1
	No	182	63.4	63.9	100.0
	Total	285	99.3	100.0	
Missing	999	2	.7		
Total		287	100.0		

a Agency size strata = 50-100 Officers

#### Lab accept AND process unnamed suspect DNA cases for prop. crimes?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	121	42.2	47.5	47.5
	No	134	46.7	52.5	100.0
	Total	255	88.9	100.0	
Missing	999	32	11.1		
Total		287	100.0		

a Agency size strata = 50-100 Officers

#### Lab accept but NOT process unnamed suspect DNA cases for prop. crimes?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	59	20.6	28.1	28.1
	No	151	52.6	71.9	100.0
	Total	210	73.2	100.0	
Missing	999	77	26.8		
Total		287	100.0		

a Agency size strata = 50-100 Officers

#### unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	16	5.6	5.6	5.6
	No	271	94.4	94.4	100.0
	Total	287	100.0	100.0	

a Agency size strata = 50-100 Officers

agency	have	cold	case	squad?(	(a)	)
--------	------	------	------	---------	-----	---

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	46	16.0	16.1	16.1
	No	239	83.3	83.9	100.0
	Total	285	99.3	100.0	
Missing	999	2	.7		
Total		287	100.0		

a Agency size strata = 50-100 Officers

# unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	1	.3	.3	.3
	No	286	99.7	99.7	100.0
	Total	287	100.0	100.0	

a Agency size strata = 50-100 Officers

# who to contact?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	99	1	.3	.3	.3
	999	285	99.3	99.3	99.7
	Sgt. Ron Morgan	1	.3	.3	100.0
	Total	287	100.0	100.0	

a Agency size strata = 50-100 Officers

# # of cold cases eligible for review(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	2	.7	4.7	4.7
	1	5	1.7	11.6	16.3
	2	10	3.5	23.3	39.5
	3	9	3.1	20.9	60.5
	5	5	1.7	11.6	72.1
	6	1	.3	2.3	74.4
	7	1	.3	2.3	76.7
	8	2	.7	4.7	81.4
	10	2	.7	4.7	86.0
	13	1	.3	2.3	88.4
	15	1	.3	2.3	90.7
	16	1	.3	2.3	93.0

	20	1	.3	2.3	95.3
	25	1	.3	2.3	97.7
	50	1	.3	2.3	100.0
	Total	43	15.0	100.0	
Missing	999	244	85.0		
Total		287	100.0		

a Agency size strata = 50-100 Officers

# # of cold cases reviewed(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	2	.7	4.8	4.8
	1	8	2.8	19.0	23.8
	2	15	5.2	35.7	59.5
	3	6	2.1	14.3	73.8
	5	4	1.4	9.5	83.3
	6	1	.3	2.4	85.7
	7	1	.3	2.4	88.1
	14	1	.3	2.4	90.5
	16	1	.3	2.4	92.9
	20	1	.3	2.4	95.2
	25	1	.3	2.4	97.6
	50	1	.3	2.4	100.0
	Total	42	14.6	100.0	
Missing	999	245	85.4		
Total		287	100.0		

a Agency size strata = 50-100 Officers

# # of cold cases sent for DNA testing(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	10	3.5	25.0	25.0
	1	13	4.5	32.5	57.5
	2	7	2.4	17.5	75.0
	3	3	1.0	7.5	82.5
	4	1	.3	2.5	85.0
	5	1	.3	2.5	87.5
	9	1	.3	2.5	90.0
	11	1	.3	2.5	92.5
	12	1	.3	2.5	95.0
	14	1	.3	2.5	97.5
	30	1	.3	2.5	100.0
	Total	40	13.9	100.0	
Missing	999	247	86.1		
Total		287	100.0		

# # of additional cold cases potentially eligible for DNA testing(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	20	7.0	62.5	62.5
	1	4	1.4	12.5	75.0
	2	3	1.0	9.4	84.4
	4	1	.3	3.1	87.5
	5	1	.3	3.1	90.6
	12	2	.7	6.3	96.9
	99	1	.3	3.1	100.0
	Total	32	11.1	100.0	
Missing	999	255	88.9		
Total		287	100.0		

a Agency size strata = 50-100 Officers

#### educated guess is not possible(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	11	3.8	3.9	3.9
	No	274	95.5	96.1	100.0
	Total	285	99.3	100.0	
Missing	999	2	.7		
Total		287	100.0		

a Agency size strata = 50-100 Officers

#### unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	2	.7	.7	.7
	No	285	99.3	99.3	100.0
	Total	287	100.0	100.0	

a Agency size strata = 50-100 Officers

#### who to contact?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	999	286	99.7	99.7	99.7
	The 2 cases in question are being review	1	.3	.3	100.0
	Total	287	100.0	100.0	

a Agency size strata = 50-100 Officers

agency have plan to review [cold] cases for biological testing?(a)
--

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	87	30.3	38.7	38.7
	No	138	48.1	61.3	100.0
	Total	225	78.4	100.0	
Missing	999	62	21.6		
Total		287	100.0		

a Agency size strata = 50-100 Officers

# # of cases eligible for review(a)

					Cumulative
		Frequency	Percent	Valid Percent	Percent
Valid	0	20	7.0	21.1	21.1
	1	16	5.6	16.8	37.9
	2	15	5.2	15.8	53.7
	3	11	3.8	11.6	65.3
	4	7	2.4	7.4	72.6
	5	9	3.1	9.5	82.1
	6	3	1.0	3.2	85.3
	7	3	1.0	3.2	88.4
	8	2	.7	2.1	90.5
	10	3	1.0	3.2	93.7
	11	1	.3	1.1	94.7
	12	1	.3	1.1	95.8
	20	1	.3	1.1	96.8
	25	1	.3	1.1	97.9
	40	1	.3	1.1	98.9
	67	1	.3	1.1	100.0
	Total	95	33.1	100.0	
Missing	999	192	66.9		
Total		287	100.0		

a Agency size strata = 50-100 Officers

# of cases with	notential f	or biologic	cal testing(a)
m oi cases with	potentian	OI DIOIOGI	cai testiligia,

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	33	11.5	39.3	39.3
	1	18	6.3	21.4	60.7
	2	19	6.6	22.6	83.3
	3	5	1.7	6.0	89.3
	4	2	.7	2.4	91.7
	5	2	.7	2.4	94.0
	6	1	.3	1.2	95.2
	10	1	.3	1.2	96.4
	20	1	.3	1.2	97.6
	25	1	.3	1.2	98.8
	67	1	.3	1.2	100.0
	Total	84	29.3	100.0	
Missing	999	203	70.7		
Total		287	100.0		

a Agency size strata = 50-100 Officers

# educated guess is not possible(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	53	18.5	18.5	18.5
	No	234	81.5	81.5	100.0
	Total	287	100.0	100.0	

a Agency size strata = 50-100 Officers

# unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	4	1.4	1.4	1.4
	No	283	98.6	98.6	100.0
	Total	287	100.0	100.0	

a Agency size strata = 50-100 Officers

# agency interested in using DNA database?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	276	96.2	96.8	96.8
	No	9	3.1	3.2	100.0
	Total	285	99.3	100.0	
Missing	999	2	.7		
Total		287	100.0		

# has agency used DNA database for cases with no direct match?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	47	16.4	16.7	16.7
	No	235	81.9	83.3	100.0
	Total	282	98.3	100.0	
Missing	999	5	1.7		
Total		287	100.0		

a Agency size strata = 50-100 Officers

# Any comments?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	43	15.0	15.0	15.0
	No	244	85.0	85.0	100.0
	Total	287	100.0	100.0	

a Agency size strata = 50-100 Officers

# Agency size strata = 25-50 Officers

# STATE(a)

				Cumulative
	Frequency	Percent	Valid Percent	Percent
Valid CO	1	.2	.2	.2
999	1	.2	.2	.4
AK	1	.2	.2	.6
AL	8	1.7	1.7	2.3
AR	9	1.9	1.9	4.2
AZ	3	.6	.6	4.8
CA	21	4.4	4.4	9.2
CO	10	2.1	2.1	11.3
CT	10	2.1	2.1	13.4
FL	7	1.5	1.5	14.9
GA	21	4.4	4.4	19.3
IA	6	1.3	1.3	20.6
ID	3	.6	.6	21.2
IL	35	7.4	7.4	28.6
IN	10	2.1	2.1	30.7
KS	10	2.1	2.1	32.8
KY	5	1.1	1.1	33.8
LA	11	2.3	2.3	36.1
MA	9	1.9	1.9	38.0
MD	1	.2	.2	38.2
ME	3	.6	.6	38.9
MI	15	3.2	3.2	42.0
MN	14	2.9	2.9	45.0
MO	9	1.9	1.9	46.8
MS	5	1.1	1.1	47.9
MT	3	.6	.6	48.5
NC	18	3.8	3.8	52.3
ND	1	.2	.2	52.5
NE	4	.8	.8	53.4
NH	1	.2	.2	53.6
NJ	27	5.7	5.7	59.2
NM	3	.6	.6	59.9
NY	21	4.4	4.4	64.3
ОН	36	7.6	7.6	71.8
OK	7	1.5	1.5	73.3
OR	7	1.5	1.5	74.8
PA	17	3.6	3.6	78.4
RI	3	.6	.6	79.0
SC	11	2.3	2.3	81.3
SD	1	.2	.2	81.5
TN	6	1.3	1.3	82.8

TX	27	5.7	5.7	88.4
UT	5	1.1	1.1	89.5
VA	15	3.2	3.2	92.6
VT	3	.6	.6	93.3
WA	8	1.7	1.7	95.0
WI	17	3.6	3.6	98.5
WV	4	.8	.8	99.4
Wy	1	.2	.2	99.6
WY	2	.4	.4	100.0
Total	476	100.0	100.0	

a Agency size strata = 25-50 Officers

# Where are cases primarily sent for DNA analysis?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	State Agency Lab	407	85.5	86.6	86.6
	Local Agency Lab	33	6.9	7.0	93.6
	FBI Lab	3	.6	.6	94.3
	Other	19	4.0	4.0	98.3
	Private Lab/Commercial Lab	8	1.7	1.7	100.0
	Total	470	98.7	100.0	
Missing	999	6	1.3		
Total		476	100.0		

a Agency size strata = 25-50 Officers

# % rapes with possible DNA(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	83	17.4	18.4	18.4
	11-25%	26	5.5	5.8	24.2
	25-50%	40	8.4	8.9	33.1
	50-75%	86	18.1	19.1	52.2
	75-100%	215	45.2	47.8	100.0
	Total	450	94.5	100.0	
Missing	999	26	5.5		
Total		476	100.0		

a Agency size strata = 25-50 Officers

% homicides with possible DNA(a)	%	homicid	les with	possible	DNA(a)
----------------------------------	---	---------	----------	----------	--------

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	100	21.0	25.2	25.2
	11-25%	8	1.7	2.0	27.2
	25-50%	29	6.1	7.3	34.5
	50-75%	43	9.0	10.8	45.3
	75-100%	217	45.6	54.7	100.0
	Total	397	83.4	100.0	
Missing	999	79	16.6		
Total		476	100.0		

a Agency size strata = 25-50 Officers

# educated guess not possible(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	33	6.9	6.9	6.9
	No	443	93.1	93.1	100.0
	Total	476	100.0	100.0	

a Agency size strata = 25-50 Officers

#### unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	5	1.1	1.1	1.1
	No	471	98.9	98.9	100.0
	Total	476	100.0	100.0	

a Agency size strata = 25-50 Officers

#### unsolved homicides--total cases(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	153	32.1	39.1	39.1
	1	105	22.1	26.9	66.0
	2	58	12.2	14.8	80.8
	3	35	7.4	9.0	89.8
	4	13	2.7	3.3	93.1
	5	8	1.7	2.0	95.1
	6	6	1.3	1.5	96.7
	7	4	.8	1.0	97.7
	8	1	.2	.3	98.0
	9	1	.2	.3	98.2
	10	3	.6	.8	99.0

	11	1	.2	.3	99.2
	12	1	.2	.3	99.5
	56	1	.2	.3	99.7
	99	1	.2	.3	100.0
	Total	391	82.1	100.0	
Missing	999	85	17.9		
Total		476	100.0		

a Agency size strata = 25-50 Officers

# unsolved homicides--cases with DNA(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	173	36.3	59.0	59.0
	1	74	15.5	25.3	84.3
	2	21	4.4	7.2	91.5
	3	12	2.5	4.1	95.6
	4	4	.8	1.4	96.9
	5	4	.8	1.4	98.3
	6	3	.6	1.0	99.3
	12	1	.2	.3	99.7
	23	1	.2	.3	100.0
	Total	293	61.6	100.0	
Missing	999	183	38.4		
Total		476	100.0		

a Agency size strata = 25-50 Officers

unsolved rapes--total cases(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	126	26.5	39.4	39.4
Valla	1	46	9.7	14.4	53.8
	2	37	9.7 7.8	11.6	65.3
	3	20	7.0 4.2	6.3	71.6
	4	11			71.6 75.0
	5		2.3 2.7	3.4	
	6	13		4.1	79.1
	7	9	1.9	2.8	81.9
	8	3	.6	.9	82.8
	10	3	.6	.9	83.8
	11	14	2.9	4.4	88.1
	12	1	.2	.3	88.4
	12 14	2	.4	.6	89.1
		1	.2	.3	89.4
	15	9	1.9	2.8	92.2
	18	1	.2	.3	92.5
	20	7	1.5	2.2	94.7
	25	1	.2	.3	95.0
	30	2	.4	.6	95.6
	34	1	.2	.3	95.9
	35	2	.4	.6	96.6
	40	2	.4	.6	97.2
	50	2	.4	.6	97.8
	75	1	.2	.3	98.1
	99	1	.2	.3	98.4
	100	2	.4	.6	99.1
	165	1	.2	.3	99.4
	228	1	.2	.3	99.7
	9999	1	.2	.3	100.0
	Total	320	67.2	100.0	
Missing	999	156	32.8		
Total		476	100.0		

a Agency size strata = 25-50 Officers

unsolved	rapescases	with	DNA(	a١
uliouiveu	Tabes-cases	**   L	PINA	aı

		Fraguanay	Percent	Valid Percent	Cumulative Percent
Valid	0	Frequency 112	23.5	45.2	45.2
	1	53	11.1	21.4	66.5
	2	25	5.3	10.1	76.6
	3	14	2.9	5.6	82.3
	4	4	.8	1.6	83.9
	5	12	2.5	4.8	88.7
	6	4	.8	1.6	90.3
	8	1	.2	.4	90.7
	9	1	.2	.4	91.1
	10	6	1.3	2.4	93.5
	11	1	.2	.4	94.0
	12	1	.2	.4	94.0
	14	1	.2	.4	94.4
	15	-		.8	94.8 95.6
	16	2	.4		
	17	2	.4	.8	96.4
	18	1	.2	.4	96.8
	20	1	.2	.4	97.2
	25 25	3	.6	1.2	98.4
		3	.6	1.2	99.6
	50 Tatal	1	.2	.4	100.0
	Total	248	52.1	100.0	
Missing	999	228	47.9		
Total		476	100.0		

a Agency size strata = 25-50 Officers

# educated guess not possible(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	119	25.0	25.2	25.2
	No	353	74.2	74.8	100.0
	Total	472	99.2	100.0	
Missing	999	4	.8		
Total		476	100.0		

a Agency size strata = 25-50 Officers

# unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	16	3.4	3.4	3.4
	No	459	96.4	96.6	100.0
	Total	475	99.8	100.0	
Missing	999	1	.2		
Total		476	100.0		

a Agency size strata = 25-50 Officers

# total unsolved rape cases with possible biological evidence(a)

		_			Cumulative
		Frequency	Percent	Valid Percent	Percent
Valid	0	238	50.0	70.2	70.2
	1	33	6.9	9.7	79.9
	2	12	2.5	3.5	83.5
	3	12	2.5	3.5	87.0
	4	7	1.5	2.1	89.1
	5	8	1.7	2.4	91.4
	6	5	1.1	1.5	92.9
	7	4	.8	1.2	94.1
	8	2	.4	.6	94.7
	9	2	.4	.6	95.3
	10	3	.6	.9	96.2
	11	1	.2	.3	96.5
	12	1	.2	.3	96.8
	15	3	.6	.9	97.6
	16	1	.2	.3	97.9
	17	2	.4	.6	98.5
	20	1	.2	.3	98.8
	40	2	.4	.6	99.4
	50	1	.2	.3	99.7
	58	1	.2	.3	100.0
	Total	339	71.2	100.0	
Missing	999	137	28.8		
Total		476	100.0		

a Agency size strata = 25-50 Officers

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	263	55.3	79.5	79.5
	1	44	9.2	13.3	92.7
	2	9	1.9	2.7	95.5
	3	9	1.9	2.7	98.2
	4	1	.2	.3	98.5
	5	2	.4	.6	99.1
	6	1	.2	.3	99.4
	9	1	.2	.3	99.7
	45	1	.2	.3	100.0
	Total	331	69.5	100.0	
Missing	999	145	30.5		
Total		476	100.0		

a Agency size strata = 25-50 Officers

# total unsolved property crimes with possible biological evidence(a)

					Cumulative
		Frequency	Percent	Valid Percent	Percent
Valid	0	209	43.9	76.0	76.0
	1	8	1.7	2.9	78.9
	2	10	2.1	3.6	82.5
	3	9	1.9	3.3	85.8
	4	4	.8	1.5	87.3
	5	6	1.3	2.2	89.5
	6	4	.8	1.5	90.9
	10	8	1.7	2.9	93.8
	12	2	.4	.7	94.5
	20	3	.6	1.1	95.6
	25	3	.6	1.1	96.7
	30	1	.2	.4	97.1
	34	1	.2	.4	97.5
	35	1	.2	.4	97.8
	50	2	.4	.7	98.5
	60	1	.2	.4	98.9
	100	1	.2	.4	99.3
	250	1	.2	.4	99.6
	1666	1	.2	.4	100.0
	Total	275	57.8	100.0	
Missing	999	201	42.2		
Total		476	100.0		

a Agency size strata = 25-50 Officers

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	152	31.9	93.3	93.3
	1	7	1.5	4.3	97.5
	2	2	.4	1.2	98.8
	9	1	.2	.6	99.4
	10	1	.2	.6	100.0
	Total	163	34.2	100.0	
Missing	999	313	65.8		
Total		476	100.0		

a Agency size strata = 25-50 Officers

# total unsolved unnamed rape cases with possible biological evidence(a)

r					
					Cumulative
		Frequency	Percent	Valid Percent	Percent
Valid	0	126	26.5	59.2	59.2
	1	31	6.5	14.6	73.7
	2	17	3.6	8.0	81.7
	3	13	2.7	6.1	87.8
	4	6	1.3	2.8	90.6
	5	5	1.1	2.3	93.0
	6	3	.6	1.4	94.4
	7	1	.2	.5	94.8
	10	2	.4	.9	95.8
	11	2	.4	.9	96.7
	13	1	.2	.5	97.2
	15	1	.2	.5	97.7
	17	1	.2	.5	98.1
	20	2	.4	.9	99.1
	35	1	.2	.5	99.5
	40	1	.2	.5	100.0
	Total	213	44.7	100.0	
Missing	999	263	55.3		
Total		476	100.0		

a Agency size strata = 25-50 Officers

# total unsolved unnamed homicide cases with possible biological evidence(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	141	29.6	71.6	71.6
	1	33	6.9	16.8	88.3
	2	10	2.1	5.1	93.4
	3	8	1.7	4.1	97.5
	4	2	.4	1.0	98.5
	5	2	.4	1.0	99.5
	6	1	.2	.5	100.0
	Total	197	41.4	100.0	
Missing	999	279	58.6		
Total		476	100.0		

a Agency size strata = 25-50 Officers

# total unsolved unnamed property crimes with possible biological evidence(a)

		_			Cumulative
		Frequency	Percent	Valid Percent	Percent
Valid	0	108	22.7	66.7	66.7
	1	8	1.7	4.9	71.6
	2	8	1.7	4.9	76.5
	3	7	1.5	4.3	80.9
	4	3	.6	1.9	82.7
	5	5	1.1	3.1	85.8
	6	3	.6	1.9	87.7
	7	1	.2	.6	88.3
	8	1	.2	.6	88.9
	10	5	1.1	3.1	92.0
	12	1	.2	.6	92.6
	15	2	.4	1.2	93.8
	18	1	.2	.6	94.4
	25	2	.4	1.2	95.7
	30	2	.4	1.2	96.9
	50	1	.2	.6	97.5
	60	1	.2	.6	98.1
	100	1	.2	.6	98.8
	250	1	.2	.6	99.4
	1265	1	.2	.6	100.0
	Total	162	34.0	100.0	
Missing	999	314	66.0		
Total		476	100.0		

a Agency size strata = 25-50 Officers

# total unsolved unnamed "other" cases with possible biological evidence(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	89	18.7	91.8	91.8
	1	5	1.1	5.2	96.9
	2	1	.2	1.0	97.9
	5	1	.2	1.0	99.0
	8	1	.2	1.0	100.0
	Total	97	20.4	100.0	
Missing	999	379	79.6		
Total		476	100.0		

a Agency size strata = 25-50 Officers

#### educated guess not possible(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	116	24.4	24.4	24.4
	No	360	75.6	75.6	100.0
	Total	476	100.0	100.0	

a Agency size strata = 25-50 Officers

#### unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	15	3.2	3.2	3.2
	No	461	96.8	96.8	100.0
	Total	476	100.0	100.0	

a Agency size strata = 25-50 Officers

#### % cases with rape kits only(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	138	29.0	54.3	54.3
	11-25%	14	2.9	5.5	59.8
	25-50%	14	2.9	5.5	65.4
	50-75%	18	3.8	7.1	72.4
	75-100%	70	14.7	27.6	100.0
	Total	254	53.4	100.0	
Missing	999	222	46.6		
Total		476	100.0		

a Agency size strata = 25-50 Officers

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	153	32.1	65.1	65.1
	11-25%	10	2.1	4.3	69.4
	25-50%	14	2.9	6.0	75.3
	50-75%	18	3.8	7.7	83.0
	75-100%	40	8.4	17.0	100.0
	Total	235	49.4	100.0	
Missing	999	241	50.6		
Total		476	100.0		

a Agency size strata = 25-50 Officers

# % cases with other DNA evidence only(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	151	31.7	83.4	83.4
	11-25%	6	1.3	3.3	86.7
	25-50%	5	1.1	2.8	89.5
	50-75%	6	1.3	3.3	92.8
	75-100%	13	2.7	7.2	100.0
	Total	181	38.0	100.0	
Missing	999	295	62.0		
Total		476	100.0		

a Agency size strata = 25-50 Officers

# educated guess not possible(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	86	18.1	18.1	18.1
	No	389	81.7	81.9	100.0
	Total	475	99.8	100.0	
Missing	999	1	.2		
Total		476	100.0		

a Agency size strata = 25-50 Officers

# unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	13	2.7	2.7	2.7
	No	462	97.1	97.3	100.0
	Total	475	99.8	100.0	
Missing	999	1	.2		
Total		476	100.0		

a Agency size strata = 25-50 Officers

# rape cases within statute of limitations(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	1			
valid		102	21.4	48.6	48.6
	1	46	9.7	21.9	70.5
	2	21	4.4	10.0	80.5
	3	8	1.7	3.8	84.3
	4	5	1.1	2.4	86.7
	5	7	1.5	3.3	90.0
	6	3	.6	1.4	91.4
	7	4	.8	1.9	93.3
	9	1	.2	.5	93.8
	10	6	1.3	2.9	96.7
	11	1	.2	.5	97.1
	15	2	.4	1.0	98.1
	16	1	.2	.5	98.6
	50	1	.2	.5	99.0
	95	1	.2	.5	99.5
	99	1	.2	.5	100.0
	Total	210	44.1	100.0	
Missing	999	266	55.9		
Total		476	100.0		

a Agency size strata = 25-50 Officers

# unnamed rape cases within statute of limitations(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	84	17.6	47.7	47.7
	1	44	9.2	25.0	72.7
	2	19	4.0	10.8	83.5
	3	10	2.1	5.7	89.2
	4	2	.4	1.1	90.3
	5	4	.8	2.3	92.6
	6	2	.4	1.1	93.8
	7	3	.6	1.7	95.5
	10	3	.6	1.7	97.2
	11	1	.2	.6	97.7
	13	1	.2	.6	98.3
	15	2	.4	1.1	99.4
	40	1	.2	.6	100.0
	Total	176	37.0	100.0	
Missing	999	300	63.0		
Total		476	100.0		

a Agency size strata = 25-50 Officers

# rape cases to exceed statute of limitations(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	125	26.3	74.4	74.4
	1	13	2.7	7.7	82.1
	2	13	2.7	7.7	89.9
	3	3	.6	1.8	91.7
	4	2	.4	1.2	92.9
	5	7	1.5	4.2	97.0
	6	1	.2	.6	97.6
	10	2	.4	1.2	98.8
	15	1	.2	.6	99.4
	18	1	.2	.6	100.0
	Total	168	35.3	100.0	
Missing	999	308	64.7		
Total		476	100.0		

a Agency size strata = 25-50 Officers

unnamed rape cases to exceed statute	01	f limitations	(a)
--------------------------------------	----	---------------	-----

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	112	23.5	75.2	75.2
	1	11	2.3	7.4	82.6
	2	10	2.1	6.7	89.3
	3	4	.8	2.7	91.9
	4	1	.2	.7	92.6
	5	2	.4	1.3	94.0
	6	2	.4	1.3	95.3
	10	2	.4	1.3	96.6
	15	1	.2	.7	97.3
	17	1	.2	.7	98.0
	99	3	.6	2.0	100.0
	Total	149	31.3	100.0	
Missing	999	326	68.5		
	System	1	.2		
	Total	327	68.7		
Total		476	100.0		

a Agency size strata = 25-50 Officers

# educated guess not possible(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	96	20.2	20.2	20.2
	No	379	79.6	79.8	100.0
	Total	475	99.8	100.0	
Missing	999	1	.2		
Total		476	100.0		

a Agency size strata = 25-50 Officers

# unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	12	2.5	2.5	2.5
	No	463	97.3	97.5	100.0
	Total	475	99.8	100.0	
Missing	999	1	.2		
Total		476	100.0		

a Agency size strata = 25-50 Officers

#### unanalyzed evidence stored--central storage area(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	380	79.8	79.8	79.8
	No	96	20.2	20.2	100.0
	Total	476	100.0	100.0	

a Agency size strata = 25-50 Officers

#### unanalyzed evidence stored--decentralized storage area(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	10	2.1	2.1	2.1
	No	466	97.9	97.9	100.0
	Total	476	100.0	100.0	

a Agency size strata = 25-50 Officers

#### unanalyzed evidence stored--prosecutor's facility(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	13	2.7	2.7	2.7
	No	463	97.3	97.3	100.0
	Total	476	100.0	100.0	

a Agency size strata = 25-50 Officers

#### unanalyzed evidence stored--crime lab facility(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	89	18.7	18.7	18.7
	No	387	81.3	81.3	100.0
	Total	476	100.0	100.0	

a Agency size strata = 25-50 Officers

#### unanalyzed evidence stored--other(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	37	7.8	7.8	7.8
	No	439	92.2	92.2	100.0
	Total	476	100.0	100.0	

a Agency size strata = 25-50 Officers

# sufficient space for long-term evidence storage?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Sufficient Storage	188	39.5	39.9	39.9
	Insufficient Storage	283	59.5	60.1	100.0
	Total	471	98.9	100.0	
Missing	999	5	1.1		
Total		476	100.0		

a Agency size strata = 25-50 Officers

# If insufficient, how critical is additional space?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Highly Critical	71	14.9	18.7	18.7
	Critical	193	40.5	50.8	69.5
	Pressing But Not Critical	116	24.4	30.5	100.0
	Total	380	79.8	100.0	
Missing	999	96	20.2		
Total		476	100.0		

a Agency size strata = 25-50 Officers

#### DNA not sent to lab applicable--not applicable(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	190	39.9	39.9	39.9
	No	286	60.1	60.1	100.0
	Total	476	100.0	100.0	

a Agency size strata = 25-50 Officers

# DNA not sent to lab rank--not applicable(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	104	21.8	91.2	91.2
	Second Most Significant	3	.6	2.6	93.9
	Third Most Significant	7	1.5	6.1	100.0
	Total	114	23.9	100.0	
Missing	999	362	76.1		
Total		476	100.0		

a Agency size strata = 25-50 Officers

# DNA not sent to lab applicable--no suspect identified(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	99	20.8	20.8	20.8
	No	377	79.2	79.2	100.0
	Total	476	100.0	100.0	

a Agency size strata = 25-50 Officers

# DNA not sent to lab rank--no suspect identified(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	57	12.0	71.3	71.3
	Second Most Significant	16	3.4	20.0	91.3
	Third Most Significant	7	1.5	8.8	100.0
	Total	80	16.8	100.0	
Missing	999	396	83.2		
Total		476	100.0		

a Agency size strata = 25-50 Officers

# DNA not sent to lab applicable--no suspect charged(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	25	5.3	5.3	5.3
	No	451	94.7	94.7	100.0
	Total	476	100.0	100.0	

a Agency size strata = 25-50 Officers

#### DNA not sent to lab rank--no suspect charged(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	13	2.7	56.5	56.5
	Second Most Significant	7	1.5	30.4	87.0
	Third Most Significant	3	.6	13.0	100.0
	Total	23	4.8	100.0	
Missing	999	453	95.2		
Total		476	100.0		

a Agency size strata = 25-50 Officers

# DNA not sent to lab applicable--guilty plea anticipated(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	37	7.8	7.8	7.8
	No	439	92.2	92.2	100.0
	Total	476	100.0	100.0	

a Agency size strata = 25-50 Officers

#### DNA not sent to lab rank--guilty plea anticipated(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	12	2.5	46.2	46.2
	Second Most Significant	9	1.9	34.6	80.8
	Third Most Significant	5	1.1	19.2	100.0
	Total	26	5.5	100.0	
Missing	999	450	94.5		
Total		476	100.0		

a Agency size strata = 25-50 Officers

#### DNA not sent to lab applicable--DNA use uncertain(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	20	4.2	4.2	4.2
	No	456	95.8	95.8	100.0
	Total	476	100.0	100.0	

a Agency size strata = 25-50 Officers

#### DNA not sent to lab rank--DNA use uncertain(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	3	.6	21.4	21.4
	Second Most Significant	5	1.1	35.7	57.1
	Third Most Significant	6	1.3	42.9	100.0
	Total	14	2.9	100.0	
Missing	999	462	97.1		
Total		476	100.0		

a Agency size strata = 25-50 Officers

# DNA not sent to lab applicable--lack of \$(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	16	3.4	3.4	3.4
	No	460	96.6	96.6	100.0
	Total	476	100.0	100.0	

a Agency size strata = 25-50 Officers

# DNA not sent to lab rank--lack of \$(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	3	.6	25.0	25.0
	Second Most Significant	3	.6	25.0	50.0
	Third Most Significant	6	1.3	50.0	100.0
	Total	12	2.5	100.0	
Missing	999	464	97.5		
Total		476	100.0		

a Agency size strata = 25-50 Officers

# DNA not sent to lab applicable--inability to get timely results(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	21	4.4	4.4	4.4
	No	455	95.6	95.6	100.0
	Total	476	100.0	100.0	

a Agency size strata = 25-50 Officers

#### DNA not sent to lab rank--inability to get timely results(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	4	.8	30.8	30.8
	Second Most Significant	5	1.1	38.5	69.2
	Third Most Significant	4	.8	30.8	100.0
	Total	13	2.7	100.0	
Missing	999	463	97.3		
Total		476	100.0		

a Agency size strata = 25-50 Officers

#### DNA not sent to lab applicable--lab not processing requests(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	10	2.1	2.1	2.1
	No	466	97.9	97.9	100.0
	Total	476	100.0	100.0	

a Agency size strata = 25-50 Officers

#### DNA not sent to lab rank--lab not processing requests(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	3	.6	50.0	50.0
	Second Most Significant	2	.4	33.3	83.3
	Third Most Significant	1	.2	16.7	100.0
	Total	6	1.3	100.0	
Missing	999	470	98.7		
Total		476	100.0		

a Agency size strata = 25-50 Officers

#### DNA not sent to lab applicable--uncertain where to send DNA(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	5	1.1	1.1	1.1
	No	471	98.9	98.9	100.0
	Total	476	100.0	100.0	

a Agency size strata = 25-50 Officers

#### DNA not sent to lab rank--uncertain where to send DNA(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	1	.2	100.0	100.0
Missing	999	475	99.8		
Total		476	100.0		

a Agency size strata = 25-50 Officers

#### DNA not sent to lab applicable--analysis not requested by prosecutor(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	37	7.8	7.8	7.8
	No	439	92.2	92.2	100.0
	Total	476	100.0	100.0	

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	4	.8	11.8	11.8
	Second Most Significant	14	2.9	41.2	52.9
	Third Most Significant	16	3.4	47.1	100.0
	Total	34	7.1	100.0	
Missing	999	442	92.9		
Total		476	100.0		

a Agency size strata = 25-50 Officers

#### DNA not sent to lab applicable--other(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	20	4.2	4.2	4.2
	No	456	95.8	95.8	100.0
	Total	476	100.0	100.0	

a Agency size strata = 25-50 Officers

#### DNA not sent to lab rank--other(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	5	1.1	38.5	38.5
	Second Most Significant	5	1.1	38.5	76.9
	Third Most Significant	3	.6	23.1	100.0
	Total	13	2.7	100.0	
Missing	999	463	97.3		
Total		476	100.0		

a Agency size strata = 25-50 Officers

# Department routinely collect DNA from property crimes?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	158	33.2	33.8	33.8
	No	310	65.1	66.2	100.0
	Total	468	98.3	100.0	
Missing	999	8	1.7		
Total		476	100.0		

a Agency size strata = 25-50 Officers

# Lab accept AND process unnamed suspect DNA cases for prop. crimes?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	203	42.6	51.1	51.1
	No	194	40.8	48.9	100.0
	Total	397	83.4	100.0	
Missing	999	79	16.6		
Total		476	100.0		

a Agency size strata = 25-50 Officers

# Lab accept but NOT process unnamed suspect DNA cases for prop. crimes?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	90	18.9	29.4	29.4
	No	216	45.4	70.6	100.0
	Total	306	64.3	100.0	
Missing	999	170	35.7		
Total		476	100.0		

a Agency size strata = 25-50 Officers

#### unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	29	6.1	6.1	6.1
	No	447	93.9	93.9	100.0
	Total	476	100.0	100.0	

a Agency size strata = 25-50 Officers

#### agency have cold case squad?(a)

					Cumulative
		Frequency	Percent	Valid Percent	Percent
Valid	Yes	71	14.9	15.1	15.1
	No	398	83.6	84.9	100.0
	Total	469	98.5	100.0	
Missing	999	7	1.5		
Total		476	100.0		

a Agency size strata = 25-50 Officers

# unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	3	.6	.6	.6
	No	473	99.4	99.4	100.0
	Total	476	100.0	100.0	

a Agency size strata = 25-50 Officers

# # of cold cases eligible for review(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	·			
valiu		14	2.9	20.3	20.3
	1	16	3.4	23.2	43.5
	2	12	2.5	17.4	60.9
	3	8	1.7	11.6	72.5
	4	6	1.3	8.7	81.2
	5	4	.8	5.8	87.0
	6	3	.6	4.3	91.3
	7	1	.2	1.4	92.8
	9	1	.2	1.4	94.2
	10	2	.4	2.9	97.1
	12	1	.2	1.4	98.6
	25	1	.2	1.4	100.0
	Total	69	14.5	100.0	
Missing	999	407	85.5		
Total		476	100.0		

a Agency size strata = 25-50 Officers

# # of cold cases reviewed(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	7	1.5	10.0	10.0
	1	27	5.7	38.6	48.6
	2	15	3.2	21.4	70.0
	3	8	1.7	11.4	81.4
	4	5	1.1	7.1	88.6
	5	3	.6	4.3	92.9
	6	2	.4	2.9	95.7
	7	1	.2	1.4	97.1
	20	1	.2	1.4	98.6
	25	1	.2	1.4	100.0
	Total	70	14.7	100.0	
Missing	999	406	85.3		
Total		476	100.0		

a Agency size strata = 25-50 Officers

# of cold	CASES	sent for	$DN\Delta$	testing(a)
TOI COIG	Cases	36111101		testiliata,

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	27	5.7	40.3	40.3
	1	25	5.3	37.3	77.6
	2	8	1.7	11.9	89.6
	3	4	.8	6.0	95.5
	4	2	.4	3.0	98.5
	15	1	.2	1.5	100.0
	Total	67	14.1	100.0	
Missing	999	409	85.9		
Total		476	100.0		

a Agency size strata = 25-50 Officers

# # of additional cold cases potentially eligible for DNA testing(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	46	9.7	80.7	80.7
	1	3	.6	5.3	86.0
	2	2	.4	3.5	89.5
	3	2	.4	3.5	93.0
	4	2	.4	3.5	96.5
	10	1	.2	1.8	98.2
	99	1	.2	1.8	100.0
	Total	57	12.0	100.0	
Missing	999	419	88.0		
Total		476	100.0		

a Agency size strata = 25-50 Officers

# educated guess is not possible(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	11	2.3	2.3	2.3
	No	463	97.3	97.7	100.0
	Total	474	99.6	100.0	
Missing	999	2	.4		
Total		476	100.0		

a Agency size strata = 25-50 Officers

# unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	3	.6	.6	.6
	No	473	99.4	99.4	100.0
	Total	476	100.0	100.0	

a Agency size strata = 25-50 Officers

# agency have plan to review [cold] cases for biological testing?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	133	27.9	35.7	35.7
	No	240	50.4	64.3	100.0
	Total	373	78.4	100.0	
Missing	999	103	21.6		
Total		476	100.0		

a Agency size strata = 25-50 Officers

# # of cases eligible for review(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	54	11.3	31.0	31.0
	1	45	9.5	25.9	56.9
	2	33	6.9	19.0	75.9
	3	13	2.7	7.5	83.3
	4	8	1.7	4.6	87.9
	5	5	1.1	2.9	90.8
	6	2	.4	1.1	92.0
	8	1	.2	.6	92.5
	10	5	1.1	2.9	95.4
	18	2	.4	1.1	96.6
	20	2	.4	1.1	97.7
	24	1	.2	.6	98.3
	30	1	.2	.6	98.9
	75	1	.2	.6	99.4
	300	1	.2	.6	100.0
	Total	174	36.6	100.0	
Missing	999	302	63.4		
Total		476	100.0		

a Agency size strata = 25-50 Officers

# of cases with potential for biological testing(	(a	١
---	----	---

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	72	15.1	46.2	46.2
	1	41	8.6	26.3	72.4
	2	19	4.0	12.2	84.6
	3	7	1.5	4.5	89.1
	4	4	.8	2.6	91.7
	5	5	1.1	3.2	94.9
	6	1	.2	.6	95.5
	10	1	.2	.6	96.2
	16	2	.4	1.3	97.4
	18	2	.4	1.3	98.7
	25	1	.2	.6	99.4
	99	1	.2	.6	100.0
	Total	156	32.8	100.0	
Missing	999	320	67.2		
Total		476	100.0		

a Agency size strata = 25-50 Officers

# educated guess is not possible(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	66	13.9	13.9	13.9
	No	409	85.9	86.1	100.0
	Total	475	99.8	100.0	
Missing	999	1	.2		
Total		476	100.0		

a Agency size strata = 25-50 Officers

# unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	7	1.5	1.5	1.5
	No	469	98.5	98.5	100.0
	Total	476	100.0	100.0	

a Agency size strata = 25-50 Officers

# agency interested in using DNA database?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	458	96.2	97.4	97.4
	No	12	2.5	2.6	100.0
	Total	470	98.7	100.0	
Missing	999	6	1.3		
Total		476	100.0		

a Agency size strata = 25-50 Officers

# has agency used DNA database for cases with no direct match?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	60	12.6	12.9	12.9
	No	405	85.1	87.1	100.0
	Total	465	97.7	100.0	
Missing	999	11	2.3		
Total		476	100.0		

a Agency size strata = 25-50 Officers

#### Any comments?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	77	16.2	16.2	16.2
	No	399	83.8	83.8	100.0
	Total	476	100.0	100.0	

a Agency size strata = 25-50 Officers

# Agency size strata = Under 25 Officers

STATE(a)

		STATE(a)		
				Cumulative
 	Frequency	Percent	Valid Percent	Percent
AL	3	1.8	1.8	1.8
AR	1	.6	.6	2.4
AZ	2	1.2	1.2	3.5
CA	5	2.9	2.9	6.5
CO	1	.6	.6	7.1
DE	1	.6	.6	7.6
FL	2	1.2	1.2	8.8
GA	4	2.4	2.4	11.2
IA	4	2.4	2.4	13.5
ID	2	1.2	1.2	14.7
IL	8	4.7	4.7	19.4
IN	4	2.4	2.4	21.8
KS	9	5.3	5.3	27.1
KY	4	2.4	2.4	29.4
MA	2	1.2	1.2	30.6
MD	1	.6	.6	31.2
ME	1	.6	.6	31.8
MI	7	4.1	4.1	35.9
MN	7	4.1	4.1	40.0
MO	7	4.1	4.1	44.1
MS	2	1.2	1.2	45.3
MT	2	1.2	1.2	46.5
NC	8	4.7	4.7	51.2
ND	1	.6	.6	51.8
NE	3	1.8	1.8	53.5
NJ	8	4.7	4.7	58.2
NY	3	1.8	1.8	60.0
ОН	14	8.2	8.2	68.2
OH?	1	.6	.6	68.8
OK	6	3.5	3.5	72.4
OR	1	.6	.6	72.9
PA	10	5.9	5.9	78.8
SD	2	1.2	1.2	80.0
TN	9	5.3	5.3	85.3
TX	12	7.1	7.1	92.4
UT	1	.6	.6	92.9
VA	2	1.2	1.2	94.1
VT	1	.6	.6	94.7
WA	2	1.2	1.2	95.9
WI	2	1.2	1.2	97.1
WV	2	1.2	1.2	98.2
WY	3	1.8	1.8	100.0
Total	170	100.0	100.0	

Where are cases primarily sent for DNA analysis?(a)
---

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	State Agency Lab	143	84.1	88.3	88.3
	Local Agency Lab	8	4.7	4.9	93.2
	FBI Lab	3	1.8	1.9	95.1
	Other	1	.6	.6	95.7
L	Private Lab/Commercial Lab	7	4.1	4.3	100.0
	Total	162	95.3	100.0	
Missing	999	8	4.7		
Total		170	100.0		

a Agency size strata = Under 25 Officers

# % rapes with possible DNA(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	45	26.5	33.3	33.3
	11-25%	6	3.5	4.4	37.8
	25-50%	15	8.8	11.1	48.9
	50-75%	20	11.8	14.8	63.7
	75-100%	49	28.8	36.3	100.0
	Total	135	79.4	100.0	
Missing	999	35	20.6		
Total		170	100.0		

a Agency size strata = Under 25 Officers

# % homicides with possible DNA(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	44	25.9	38.9	38.9
	11-25%	4	2.4	3.5	42.5
	25-50%	7	4.1	6.2	48.7
	50-75%	11	6.5	9.7	58.4
	75-100%	47	27.6	41.6	100.0
	Total	113	66.5	100.0	
Missing	999	57	33.5		
Total		170	100.0		

a Agency size strata = Under 25 Officers

#### educated guess not possible(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	15	8.8	8.9	8.9
	No	154	90.6	91.1	100.0
	Total	169	99.4	100.0	
Missing	999	1	.6		
Total		170	100.0		

a Agency size strata = Under 25 Officers

# unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	5	2.9	3.0	3.0
	No	164	96.5	97.0	100.0
	Total	169	99.4	100.0	
Missing	999	1	.6		
Total		170	100.0		

a Agency size strata = Under 25 Officers

# unsolved homicides--total cases(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	89	52.4	72.4	72.4
	1	18	10.6	14.6	87.0
	2	9	5.3	7.3	94.3
	3	3	1.8	2.4	96.7
	4	1	.6	.8	97.6
	5	2	1.2	1.6	99.2
	18	1	.6	.8	100.0
	Total	123	72.4	100.0	
Missing	999	47	27.6		
Total		170	100.0		

a Agency size strata = Under 25 Officers

unsolved	homicidescase	es with DNA(a)
unsoivea	nomicidescase	es with DNA(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	53	31.2	82.8	82.8
	1	10	5.9	15.6	98.4
	15	1	.6	1.6	100.0
	Total	64	37.6	100.0	
Missing	999	106	62.4		
Total		170	100.0		

a Agency size strata = Under 25 Officers

# unsolved rapes--total cases(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	89	52.4	75.4	75.4
	1	8	4.7	6.8	82.2
	2	5	2.9	4.2	86.4
	3	6	3.5	5.1	91.5
	5	1	.6	.8	92.4
	10	4	2.4	3.4	95.8
	12	1	.6	.8	96.6
	20	2	1.2	1.7	98.3
	25	2	1.2	1.7	100.0
	Total	118	69.4	100.0	
Missing	999	52	30.6		
Total		170	100.0		

a Agency size strata = Under 25 Officers

# unsolved rapes--cases with DNA(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	39	22.9	65.0	65.0
	1	9	5.3	15.0	80.0
	2	4	2.4	6.7	86.7
	3	3	1.8	5.0	91.7
	5	3	1.8	5.0	96.7
	15	1	.6	1.7	98.3
	20	1	.6	1.7	100.0
	Total	60	35.3	100.0	
Missing	999	110	64.7		
Total		170	100.0		

a Agency size strata = Under 25 Officers

# educated guess not possible(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	32	18.8	18.9	18.9
	No	137	80.6	81.1	100.0
	Total	169	99.4	100.0	
Missing	999	1	.6		
Total		170	100.0		

a Agency size strata = Under 25 Officers

#### unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	4	2.4	2.4	2.4
	No	165	97.1	97.6	100.0
	Total	169	99.4	100.0	
Missing	999	1	.6		
Total		170	100.0		

a Agency size strata = Under 25 Officers

#### total unsolved rape cases with possible biological evidence(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	113	66.5	88.3	88.3
	1	5	2.9	3.9	92.2
	2	4	2.4	3.1	95.3
	3	2	1.2	1.6	96.9
	5	3	1.8	2.3	99.2
	8	1	.6	.8	100.0
	Total	128	75.3	100.0	
Missing	999	42	24.7		
Total		170	100.0		

a Agency size strata = Under 25 Officers

#### total unsolved homicide cases with possible biological evidence(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	108	63.5	90.8	90.8
	1	7	4.1	5.9	96.6
	2	3	1.8	2.5	99.2
	12	1	.6	.8	100.0
	Total	119	70.0	100.0	
Missing	999	51	30.0		
Total		170	100.0		

a Agency size strata = Under 25 Officers

total unsolved property crimes with possible biological evidence(a)	total unsolved	property cr	imes with r	oossible biolo	gical evidence(a)
---	----------------	-------------	-------------	----------------	-------------------

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	90	52.9	88.2	88.2
	1	5	2.9	4.9	93.1
	2	4	2.4	3.9	97.1
	5	1	.6	1.0	98.0
	10	1	.6	1.0	99.0
	15	1	.6	1.0	100.0
	Total	102	60.0	100.0	
Missing	999	68	40.0		
Total		170	100.0		

a Agency size strata = Under 25 Officers

# total unsolved "other" cases with possible biological evidence(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	67	39.4	95.7	95.7
	1	2	1.2	2.9	98.6
	2	1	.6	1.4	100.0
	Total	70	41.2	100.0	
Missing	999	100	58.8		
Total		170	100.0		

a Agency size strata = Under 25 Officers

# total unsolved unnamed rape cases with possible biological evidence(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	54	31.8	90.0	90.0
	1	2	1.2	3.3	93.3
	2	2	1.2	3.3	96.7
	5	1	.6	1.7	98.3
	6	1	.6	1.7	100.0
	Total	60	35.3	100.0	
Missing	999	110	64.7		
Total		170	100.0		

a Agency size strata = Under 25 Officers

# total unsolved unnamed homicide cases with possible biological evidence(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	43	25.3	82.7	82.7
	1	8	4.7	15.4	98.1
	13	1	.6	1.9	100.0
	Total	52	30.6	100.0	
Missing	999	118	69.4		
Total		170	100.0		

a Agency size strata = Under 25 Officers

#### total unsolved unnamed property crimes with possible biological evidence(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	37	21.8	80.4	80.4
	1	1	.6	2.2	82.6
	2	4	2.4	8.7	91.3
	3	1	.6	2.2	93.5
	5	1	.6	2.2	95.7
	10	1	.6	2.2	97.8
	20	1	.6	2.2	100.0
	Total	46	27.1	100.0	
Missing	999	124	72.9		
Total		170	100.0		

a Agency size strata = Under 25 Officers

# total unsolved unnamed "other" cases with possible biological evidence(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	31	18.2	93.9	93.9
	1	1	.6	3.0	97.0
	2	1	.6	3.0	100.0
	Total	33	19.4	100.0	
Missing	999	137	80.6		
Total		170	100.0		

a Agency size strata = Under 25 Officers

#### educated guess not possible(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	23	13.5	13.6	13.6
	No	146	85.9	86.4	100.0
	Total	169	99.4	100.0	
Missing	999	1	.6		
Total		170	100.0		

a Agency size strata = Under 25 Officers

#### unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	1	.6	.6	.6
	No	168	98.8	99.4	100.0
	Total	169	99.4	100.0	
Missing	999	1	.6		
Total		170	100.0		

a Agency size strata = Under 25 Officers

# % cases with rape kits only(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	59	34.7	78.7	78.7
	11-25%	1	.6	1.3	80.0
	50-75%	3	1.8	4.0	84.0
	75-100%	12	7.1	16.0	100.0
	Total	75	44.1	100.0	
Missing	999	95	55.9		
Total		170	100.0		

a Agency size strata = Under 25 Officers

#### % cases with rape kits and other DNA evidence(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	62	36.5	89.9	89.9
	11-25%	3	1.8	4.3	94.2
	50-75%	2	1.2	2.9	97.1
	75-100%	2	1.2	2.9	100.0
	Total	69	40.6	100.0	
Missing	999	101	59.4		
Total		170	100.0		

a Agency size strata = Under 25 Officers

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	61	35.9	96.8	96.8
	75-100%	2	1.2	3.2	100.0
	Total	63	37.1	100.0	
Missing	999	107	62.9		
Total		170	100.0		

a Agency size strata = Under 25 Officers

# educated guess not possible(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	24	14.1	14.3	14.3
	No	144	84.7	85.7	100.0
	Total	168	98.8	100.0	
Missing	999	2	1.2		
Total		170	100.0		

a Agency size strata = Under 25 Officers

# unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	4	2.4	2.4	2.4
	No	164	96.5	97.6	100.0
	Total	168	98.8	100.0	
Missing	999	2	1.2		
Total		170	100.0		

a Agency size strata = Under 25 Officers

# rape cases within statute of limitations(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	60	35.3	84.5	84.5
	1	4	2.4	5.6	90.1
	2	3	1.8	4.2	94.4
	3	2	1.2	2.8	97.2
	6	1	.6	1.4	98.6
	7	1	.6	1.4	100.0
	Total	71	41.8	100.0	
Missing	999	99	58.2		
Total		170	100.0		

a Agency size strata = Under 25 Officers

unnamed	rane cases	within	statute	οf	limitations(a)
ullilallicu	Tabe cases	***********	Statute	vı	IIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIII

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	46	27.1	83.6	83.6
	1	2	1.2	3.6	87.3
	2	4	2.4	7.3	94.5
	3	1	.6	1.8	96.4
	6	1	.6	1.8	98.2
	7	1	.6	1.8	100.0
	Total	55	32.4	100.0	
Missing	999	115	67.6		
Total		170	100.0		

a Agency size strata = Under 25 Officers

#### rape cases to exceed statute of limitations(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	57	33.5	95.0	95.0
	1	2	1.2	3.3	98.3
	2	1	.6	1.7	100.0
	Total	60	35.3	100.0	
Missing	999	110	64.7		
Total		170	100.0		

a Agency size strata = Under 25 Officers

# unnamed rape cases to exceed statute of limitations(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	49	28.8	96.1	96.1
	2	1	.6	2.0	98.0
	3	1	.6	2.0	100.0
	Total	51	30.0	100.0	
Missing	999	119	70.0		
Total		170	100.0		

a Agency size strata = Under 25 Officers

educated guess not possible(a)
--------------------------------

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	21	12.4	12.5	12.5
	No	147	86.5	87.5	100.0
	Total	168	98.8	100.0	
Missing	999	2	1.2		
Total		170	100.0		

a Agency size strata = Under 25 Officers

#### unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	2	1.2	1.2	1.2
	No	166	97.6	98.8	100.0
	Total	168	98.8	100.0	
Missing	999	2	1.2		
Total		170	100.0		

a Agency size strata = Under 25 Officers

# unanalyzed evidence stored--central storage area(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	110	64.7	65.1	65.1
	No	59	34.7	34.9	100.0
	Total	169	99.4	100.0	
Missing	999	1	.6		
Total		170	100.0		

a Agency size strata = Under 25 Officers

#### unanalyzed evidence stored--decentralized storage area(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	5	2.9	3.0	3.0
	No	164	96.5	97.0	100.0
	Total	169	99.4	100.0	
Missing	999	1	.6		
Total		170	100.0		

a Agency size strata = Under 25 Officers

#### unanalyzed evidence stored--prosecutor's facility(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	6	3.5	3.6	3.6
	No	163	95.9	96.4	100.0
	Total	169	99.4	100.0	
Missing	999	1	.6		
Total		170	100.0		

a Agency size strata = Under 25 Officers

#### unanalyzed evidence stored--crime lab facility(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	38	22.4	22.5	22.5
	No	131	77.1	77.5	100.0
	Total	169	99.4	100.0	
Missing	999	1	.6		
Total		170	100.0		

a Agency size strata = Under 25 Officers

#### unanalyzed evidence stored--other(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	16	9.4	9.5	9.5
	No	153	90.0	90.5	100.0
	Total	169	99.4	100.0	
Missing	999	1	.6		
Total		170	100.0		

a Agency size strata = Under 25 Officers

#### sufficient space for long-term evidence storage?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Sufficient Storage	64	37.6	39.3	39.3
	Insufficient Storage	99	58.2	60.7	100.0
	Total	163	95.9	100.0	
Missing	999	7	4.1		
Total		170	100.0		

a Agency size strata = Under 25 Officers

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Highly Critical	16	9.4	13.1	13.1
	Critical	61	35.9	50.0	63.1
	Pressing But Not Critical	45	26.5	36.9	100.0
	Total	122	71.8	100.0	
Missing	999	48	28.2		
Total		170	100.0		

a Agency size strata = Under 25 Officers

# DNA not sent to lab applicable--not applicable(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	71	41.8	42.3	42.3
	No	97	57.1	57.7	100.0
	Total	168	98.8	100.0	
Missing	999	2	1.2		
Total		170	100.0		

a Agency size strata = Under 25 Officers

# DNA not sent to lab rank--not applicable(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	46	27.1	93.9	93.9
	Second Most Significant	2	1.2	4.1	98.0
	Third Most Significant	1	.6	2.0	100.0
	Total	49	28.8	100.0	
Missing	999	121	71.2		
Total		170	100.0		

a Agency size strata = Under 25 Officers

# DNA not sent to lab applicable--no suspect identified(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	18	10.6	10.7	10.7
	No	151	88.8	89.3	100.0
	Total	169	99.4	100.0	
Missing	999	1	.6		
Total		170	100.0		

a Agency size strata = Under 25 Officers

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	15	8.8	83.3	83.3
	Second Most Significant	2	1.2	11.1	94.4
	Third Most Significant	1	.6	5.6	100.0
	Total	18	10.6	100.0	
Missing	999	152	89.4		
Total		170	100.0		

a Agency size strata = Under 25 Officers

#### DNA not sent to lab applicable--no suspect charged(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	9	5.3	5.3	5.3
	No	160	94.1	94.7	100.0
	Total	169	99.4	100.0	
Missing	999	1	.6		
Total		170	100.0		

a Agency size strata = Under 25 Officers

#### DNA not sent to lab rank--no suspect charged(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	5	2.9	71.4	71.4
	Second Most Significant	1	.6	14.3	85.7
	Third Most Significant	1	.6	14.3	100.0
	Total	7	4.1	100.0	
Missing	999	163	95.9		
Total		170	100.0		

a Agency size strata = Under 25 Officers

#### DNA not sent to lab applicable--guilty plea anticipated(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	13	7.6	7.7	7.7
	No	156	91.8	92.3	100.0
	Total	169	99.4	100.0	
Missing	999	1	.6		
Total		170	100.0		

a Agency size strata = Under 25 Officers

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	4	2.4	33.3	33.3
	Second Most Significant	5	2.9	41.7	75.0
_	Third Most Significant	3	1.8	25.0	100.0
	Total	12	7.1	100.0	
Missing	999	158	92.9		
Total		170	100.0		

a Agency size strata = Under 25 Officers

#### DNA not sent to lab applicable--DNA use uncertain(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	3	1.8	1.8	1.8
	No	166	97.6	98.2	100.0
	Total	169	99.4	100.0	
Missing	999	1	.6		
Total		170	100.0		

a Agency size strata = Under 25 Officers

#### DNA not sent to lab rank--DNA use uncertain(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	2	1.2	50.0	50.0
	Second Most Significant	1	.6	25.0	75.0
Third Mo	Third Most Significant	1	.6	25.0	100.0
	Total	4	2.4	100.0	
Missing	999	166	97.6		
Total		170	100.0		

a Agency size strata = Under 25 Officers

## DNA not sent to lab applicable--lack of \$(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	3	1.8	1.8	1.8
	No	166	97.6	98.2	100.0
	Total	169	99.4	100.0	
Missing	999	1	.6		
Total		170	100.0		

a Agency size strata = Under 25 Officers

DNA	not	sent	to	lab	ranklack	of \$(	a)
-----	-----	------	----	-----	----------	--------	----

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	1	.6	50.0	50.0
	Second Most Significant	1	.6	50.0	100.0
	Total	2	1.2	100.0	
Missing	999	168	98.8		
Total		170	100.0		

a Agency size strata = Under 25 Officers

#### DNA not sent to lab applicable--inability to get timely results(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	6	3.5	3.6	3.6
	No	163	95.9	96.4	100.0
	Total	169	99.4	100.0	
Missing	999	1	.6		
Total		170	100.0		

a Agency size strata = Under 25 Officers

## DNA not sent to lab rank--inability to get timely results(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	3	1.8	50.0	50.0
	Second Most Significant	2	1.2	33.3	83.3
	Third Most Significant	1	.6	16.7	100.0
	Total	6	3.5	100.0	
Missing	999	164	96.5		
Total		170	100.0		

a Agency size strata = Under 25 Officers

#### DNA not sent to lab applicable--lab not processing requests(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	1	.6	.6	.6
	No	168	98.8	99.4	100.0
	Total	169	99.4	100.0	
Missing	999	1	.6		
Total		170	100.0		

a Agency size strata = Under 25 Officers

DNA not sent to	lab ra	nklab no	processing	g requests	(a)	)
-----------------	--------	----------	------------	------------	-----	---

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	1	.6	50.0	50.0
	Third Most Significant	1	.6	50.0	100.0
	Total	2	1.2	100.0	
Missing	999	168	98.8		
Total		170	100.0		

a Agency size strata = Under 25 Officers

## DNA not sent to lab applicable--uncertain where to send DNA(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	1	.6	.6	.6
	No	168	98.8	99.4	100.0
	Total	169	99.4	100.0	
Missing	999	1	.6		
Total		170	100.0		

a Agency size strata = Under 25 Officers

#### DNA not sent to lab rank--uncertain where to send DNA(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	1	.6	50.0	50.0
	Third Most Significant	1	.6	50.0	100.0
	Total	2	1.2	100.0	
Missing	999	168	98.8		
Total		170	100.0		

a Agency size strata = Under 25 Officers

#### DNA not sent to lab applicable--analysis not requested by prosecutor(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	4	2.4	2.4	2.4
	No	165	97.1	97.6	100.0
	Total	169	99.4	100.0	
Missing	999	1	.6		
Total		170	100.0		

a Agency size strata = Under 25 Officers

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	1	.6	20.0	20.0
	Second Most Significant	2	1.2	40.0	60.0
Third Most Significant Total		2	1.2	40.0	100.0
	Total	5	2.9	100.0	
Missing	999	165	97.1		
Total		170	100.0		

a Agency size strata = Under 25 Officers

#### DNA not sent to lab applicable--other(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	8	4.7	4.7	4.7
	No	161	94.7	95.3	100.0
	Total	169	99.4	100.0	
Missing	999	1	.6		
Total		170	100.0		

a Agency size strata = Under 25 Officers

#### DNA not sent to lab rank--other(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	4	2.4	66.7	66.7
	Second Most Significant	1	.6	16.7	83.3
Third Most	Third Most Significant	1	.6	16.7	100.0
	Total	6	3.5	100.0	
Missing	999	164	96.5		
Total		170	100.0		

a Agency size strata = Under 25 Officers

## Department routinely collect DNA from property crimes?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	35	20.6	22.4	22.4
	No	121	71.2	77.6	100.0
	Total	156	91.8	100.0	
Missing	999	14	8.2		
Total		170	100.0		

a Agency size strata = Under 25 Officers

Lab accept AND	process unnamed	Suspect DNA	cases for i	nron crimes?(a	١
Lab accept Aito	process uninamed	Suspect DitA	Cases IOI	prop. Crimics : (a	

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	51	30.0	44.7	44.7
	No	63	37.1	55.3	100.0
	Total	114	67.1	100.0	
Missing	999	56	32.9		
Total		170	100.0		

a Agency size strata = Under 25 Officers

#### Lab accept but NOT process unnamed suspect DNA cases for prop. crimes?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	33	19.4	35.1	35.1
	No	61	35.9	64.9	100.0
	Total	94	55.3	100.0	
Missing	999	76	44.7		
Total		170	100.0		

a Agency size strata = Under 25 Officers

#### unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	15	8.8	8.9	8.9
	No	154	90.6	91.1	100.0
	Total	169	99.4	100.0	
Missing	999	1	.6		
Total		170	100.0		

a Agency size strata = Under 25 Officers

# agency have cold case squad?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	14	8.2	8.5	8.5
	No	150	88.2	91.5	100.0
	Total	164	96.5	100.0	
Missing	999	6	3.5		
Total		170	100.0		

a Agency size strata = Under 25 Officers

## unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	No	169	99.4	100.0	100.0
Missing	999	1	.6		
Total		170	100.0		

a Agency size strata = Under 25 Officers

#### who to contact?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	2	1	.6	.6	.6
	999	169	99.4	99.4	100.0
	Total	170	100.0	100.0	

a Agency size strata = Under 25 Officers

# # of cold cases eligible for review(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	6	3.5	40.0	40.0
	1	6	3.5	40.0	80.0
	2	1	.6	6.7	86.7
	3	1	.6	6.7	93.3
	18	1	.6	6.7	100.0
	Total	15	8.8	100.0	
Missing	999	155	91.2		
Total		170	100.0		

a Agency size strata = Under 25 Officers

#### # of cold cases reviewed(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	6	3.5	37.5	37.5
	1	6	3.5	37.5	75.0
	2	1	.6	6.3	81.3
	3	1	.6	6.3	87.5
	5	1	.6	6.3	93.8
	10	1	.6	6.3	100.0
	Total	16	9.4	100.0	
Missing	999	154	90.6		
Total		170	100.0		

a Agency size strata = Under 25 Officers

## # of cold cases sent for DNA testing(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	9	5.3	60.0	60.0
	1	5	2.9	33.3	93.3
	3	1	.6	6.7	100.0
	Total	15	8.8	100.0	
Missing	999	155	91.2		
Total		170	100.0		

a Agency size strata = Under 25 Officers

#### # of additional cold cases potentially eligible for DNA testing(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	10	5.9	83.3	83.3
	10	1	.6	8.3	91.7
	15	1	.6	8.3	100.0
	Total	12	7.1	100.0	
Missing	999	158	92.9		
Total		170	100.0		

a Agency size strata = Under 25 Officers

#### educated guess is not possible(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	2	1.2	1.2	1.2
	No	167	98.2	98.8	100.0
	Total	169	99.4	100.0	
Missing	999	1	.6		
Total		170	100.0		

a Agency size strata = Under 25 Officers

#### unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	No	169	99.4	100.0	100.0
Missing	999	1	.6		
Total		170	100.0		

a Agency size strata = Under 25 Officers

agency have plan to review [cold] cases for	' biologicai	testing ((a)
---	--------------	--------------

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	28	16.5	19.6	19.6
	No	115	67.6	80.4	100.0
	Total	143	84.1	100.0	
Missing	999	27	15.9		
Total		170	100.0		

a Agency size strata = Under 25 Officers

## # of cases eligible for review(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	44	25.9	71.0	71.0
	1	6	3.5	9.7	80.6
	2	6	3.5	9.7	90.3
	3	1	.6	1.6	91.9
	5	1	.6	1.6	93.5
	7	1	.6	1.6	95.2
	10	1	.6	1.6	96.8
	20	1	.6	1.6	98.4
	250	1	.6	1.6	100.0
	Total	62	36.5	100.0	
Missing	999	108	63.5		
Total		170	100.0		

a Agency size strata = Under 25 Officers

# # of cases with potential for biological testing(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	47	27.6	82.5	82.5
	1	4	2.4	7.0	89.5
	2	2	1.2	3.5	93.0
	3	1	.6	1.8	94.7
	5	1	.6	1.8	96.5
	10	1	.6	1.8	98.2
	20	1	.6	1.8	100.0
	Total	57	33.5	100.0	
Missing	999	113	66.5		
Total		170	100.0		

a Agency size strata = Under 25 Officers

## educated guess is not possible(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	16	9.4	9.5	9.5
	No	153	90.0	90.5	100.0
	Total	169	99.4	100.0	
Missing	999	1	.6		
Total		170	100.0		

a Agency size strata = Under 25 Officers

#### unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	3	1.8	1.8	1.8
	No	166	97.6	98.2	100.0
	Total	169	99.4	100.0	
Missing	999	1	.6		
Total		170	100.0		

a Agency size strata = Under 25 Officers

## agency interested in using DNA database?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	148	87.1	93.1	93.1
	No	11	6.5	6.9	100.0
	Total	159	93.5	100.0	
Missing	999	11	6.5		
Total		170	100.0		

a Agency size strata = Under 25 Officers

#### has agency used DNA database for cases with no direct match?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	3	1.8	1.9	1.9
	No	158	92.9	98.1	100.0
	Total	161	94.7	100.0	
Missing	999	9	5.3		
Total		170	100.0		

a Agency size strata = Under 25 Officers

# Any comments?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	39	22.9	23.1	23.1
	No	130	76.5	76.9	100.0
	Total	169	99.4	100.0	
Missing	999	1	.6		
Total		170	100.0		

a Agency size strata = Under 25 Officers

# Agency size strata = Unspecified Size

STATE(a)

		_			Cumulative
17.15.1	A.I.	Frequency	Percent	Valid Percent	Percent
Valid	AL	4	4.0	4.0	4.0
	AR	2	2.0	2.0	6.1
	AZ	1	1.0	1.0	7.1
	CA	5	5.1	5.1	12.1
	CO	1	1.0	1.0	13.1
	CT	1	1.0	1.0	14.1
	FL	3	3.0	3.0	17.2
	GA	4	4.0	4.0	21.2
	IL	8	8.1	8.1	29.3
	IN	1	1.0	1.0	30.3
	KY	1	1.0	1.0	31.3
	LA	3	3.0	3.0	34.3
	MA	3	3.0	3.0	37.4
	ME	1	1.0	1.0	38.4
	MI	1	1.0	1.0	39.4
	MN	7	7.1	7.1	46.5
	MO	9	9.1	9.1	55.6
	MS	3	3.0	3.0	58.6
	NC	2	2.0	2.0	60.6
	NH	4	4.0	4.0	64.6
	NJ	5	5.1	5.1	69.7
	NV	1	1.0	1.0	70.7
	NY	3	3.0	3.0	73.7
	OH	14	14.1	14.1	87.9
	PA	3	3.0	3.0	90.9
	SD	1	1.0	1.0	91.9
	TX	3	3.0	3.0	94.9
	UT	1	1.0	1.0	96.0
	VT	1	1.0	1.0	97.0
	WI	1	1.0	1.0	98.0
	WV	2	2.0	2.0	100.0
	Total	99	100.0	100.0	

a Agency size strata = Unspecified Size

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	State Agency Lab	74	74.7	79.6	79.6
	Local Agency Lab	16	16.2	17.2	96.8
	Other	3	3.0	3.2	100.0
	Total	93	93.9	100.0	
Missing	999	6	6.1		
Total		99	100.0		

a Agency size strata = Unspecified Size

## % rapes with possible DNA(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	24	24.2	29.6	29.6
	11-25%	3	3.0	3.7	33.3
	25-50%	6	6.1	7.4	40.7
	50-75%	12	12.1	14.8	55.6
	75-100%	36	36.4	44.4	100.0
	Total	81	81.8	100.0	
Missing	999	18	18.2		
Total		99	100.0		

a Agency size strata = Unspecified Size

## % homicides with possible DNA(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	20	20.2	29.4	29.4
	11-25%	7	7.1	10.3	39.7
	25-50%	4	4.0	5.9	45.6
	50-75%	7	7.1	10.3	55.9
	75-100%	30	30.3	44.1	100.0
	Total	68	68.7	100.0	
Missing	999	31	31.3		
Total		99	100.0		

a Agency size strata = Unspecified Size

## educated guess not possible(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	6	6.1	6.3	6.3
	No	89	89.9	93.7	100.0
	Total	95	96.0	100.0	
Missing	999	4	4.0		
Total		99	100.0		

a Agency size strata = Unspecified Size

## unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	1	1.0	1.0	1.0
	No	96	97.0	99.0	100.0
	Total	97	98.0	100.0	
Missing	999	2	2.0		
Total		99	100.0		

a Agency size strata = Unspecified Size

## unsolved homicides--total cases(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	47	47.5	66.2	66.2
	1	8	8.1	11.3	77.5
	2	3	3.0	4.2	81.7
	3	1	1.0	1.4	83.1
	4	1	1.0	1.4	84.5
	6	1	1.0	1.4	85.9
	9	1	1.0	1.4	87.3
	10	1	1.0	1.4	88.7
	11	1	1.0	1.4	90.1
	14	1	1.0	1.4	91.5
	25	1	1.0	1.4	93.0
	118	1	1.0	1.4	94.4
	200	1	1.0	1.4	95.8
	261	1	1.0	1.4	97.2
	350	1	1.0	1.4	98.6
	653	1	1.0	1.4	100.0
	Total	71	71.7	100.0	
Missing	999	28	28.3		
Total		99	100.0		

a Agency size strata = Unspecified Size

# unsolved homicides--cases with DNA(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	31	31.3	70.5	70.5
	1	2	2.0	4.5	75.0
	2	1	1.0	2.3	77.3
	4	2	2.0	4.5	81.8
	5	1	1.0	2.3	84.1
	8	1	1.0	2.3	86.4
	10	1	1.0	2.3	88.6
	12	1	1.0	2.3	90.9
	18	1	1.0	2.3	93.2
	30	1	1.0	2.3	95.5
	150	2	2.0	4.5	100.0
	Total	44	44.4	100.0	
Missing	999	55	55.6		
Total		99	100.0		

a Agency size strata = Unspecified Size

## unsolved rapes--total cases(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	41	41.4	61.2	61.2
	1	8	8.1	11.9	73.1
	2	4	4.0	6.0	79.1
	3	3	3.0	4.5	83.6
	5	1	1.0	1.5	85.1
	6	1	1.0	1.5	86.6
	20	1	1.0	1.5	88.1
	31	1	1.0	1.5	89.6
	48	1	1.0	1.5	91.0
	59	1	1.0	1.5	92.5
	180	2	2.0	3.0	95.5
	1000	1	1.0	1.5	97.0
	1200	1	1.0	1.5	98.5
	1300	1	1.0	1.5	100.0
	Total	67	67.7	100.0	
Missing	999	32	32.3		
Total		99	100.0		

a Agency size strata = Unspecified Size

unsolved rapescases	with	DNA(	a)
---------------------	------	------	----

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	23	23.2	54.8	54.8
	1	8	8.1	19.0	73.8
	2	1	1.0	2.4	76.2
	3	1	1.0	2.4	78.6
	5	2	2.0	4.8	83.3
	15	1	1.0	2.4	85.7
	31	1	1.0	2.4	88.1
	40	1	1.0	2.4	90.5
	300	1	1.0	2.4	92.9
	750	1	1.0	2.4	95.2
	850	1	1.0	2.4	97.6
	902	1	1.0	2.4	100.0
	Total	42	42.4	100.0	
Missing	999	57	57.6		
Total		99	100.0		

a Agency size strata = Unspecified Size

## educated guess not possible(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	20	20.2	20.6	20.6
	No	77	77.8	79.4	100.0
	Total	97	98.0	100.0	
Missing	999	2	2.0		
Total		99	100.0		

a Agency size strata = Unspecified Size

## unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	4	4.0	4.1	4.1
	No	93	93.9	95.9	100.0
	Total	97	98.0	100.0	
Missing	999	2	2.0		
Total		99	100.0		

a Agency size strata = Unspecified Size

# total unsolved rape cases with possible biological evidence(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	47	47.5	73.4	73.4
	1	2	2.0	3.1	76.6
	2	4	4.0	6.3	82.8
	4	1	1.0	1.6	84.4
	6	1	1.0	1.6	85.9
	11	1	1.0	1.6	87.5
	15	1	1.0	1.6	89.1
	20	1	1.0	1.6	90.6
	30	1	1.0	1.6	92.2
	31	1	1.0	1.6	93.8
	35	1	1.0	1.6	95.3
	120	1	1.0	1.6	96.9
	180	1	1.0	1.6	98.4
	1000	1	1.0	1.6	100.0
	Total	64	64.6	100.0	
Missing	999	35	35.4		
Total		99	100.0		

a Agency size strata = Unspecified Size

# total unsolved homicide cases with possible biological evidence(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	49	49.5	77.8	77.8
	1	4	4.0	6.3	84.1
	3	2	2.0	3.2	87.3
	4	1	1.0	1.6	88.9
	6	1	1.0	1.6	90.5
	9	1	1.0	1.6	92.1
	10	1	1.0	1.6	93.7
	25	1	1.0	1.6	95.2
	157	1	1.0	1.6	96.8
	250	1	1.0	1.6	98.4
	2000	1	1.0	1.6	100.0
	Total	63	63.6	100.0	
Missing	999	36	36.4		
Total		99	100.0		

a Agency size strata = Unspecified Size

## total unsolved property crimes with possible biological evidence(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	43	43.4	79.6	79.6
	1	2	2.0	3.7	83.3
	2	4	4.0	7.4	90.7
	4	1	1.0	1.9	92.6
	10	2	2.0	3.7	96.3
	26	1	1.0	1.9	98.1
	30	1	1.0	1.9	100.0
	Total	54	54.5	100.0	
Missing	999	45	45.5		
Total		99	100.0		

a Agency size strata = Unspecified Size

#### total unsolved "other" cases with possible biological evidence(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	31	31.3	96.9	96.9
	1	1	1.0	3.1	100.0
	Total	32	32.3	100.0	
Missing	999	66	66.7		
	System	1	1.0		
	Total	67	67.7		
Total		99	100.0		

a Agency size strata = Unspecified Size

#### total unsolved unnamed rape cases with possible biological evidence(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	28	28.3	65.1	65.1
	1	3	3.0	7.0	72.1
	2	5	5.1	11.6	83.7
	4	1	1.0	2.3	86.0
	7	1	1.0	2.3	88.4
	15	1	1.0	2.3	90.7
	20	2	2.0	4.7	95.3
	23	1	1.0	2.3	97.7
	1000	1	1.0	2.3	100.0
	Total	43	43.4	100.0	
Missing	999	56	56.6		
Total		99	100.0		

a Agency size strata = Unspecified Size

## total unsolved unnamed homicide cases with possible biological evidence(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	28	28.3	71.8	71.8
	1	3	3.0	7.7	79.5
	2	1	1.0	2.6	82.1
	3	2	2.0	5.1	87.2
	5	1	1.0	2.6	89.7
	9	1	1.0	2.6	92.3
	19	1	1.0	2.6	94.9
	150	1	1.0	2.6	97.4
	200	1	1.0	2.6	100.0
	Total	39	39.4	100.0	
Missing	999	60	60.6		
Total		99	100.0		

a Agency size strata = Unspecified Size

#### total unsolved unnamed property crimes with possible biological evidence(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	24	24.2	66.7	66.7
	1	2	2.0	5.6	72.2
	2	3	3.0	8.3	80.6
	4	1	1.0	2.8	83.3
	5	1	1.0	2.8	86.1
	6	1	1.0	2.8	88.9
	10	1	1.0	2.8	91.7
	26	1	1.0	2.8	94.4
	40	1	1.0	2.8	97.2
	99	1	1.0	2.8	100.0
	Total	36	36.4	100.0	
Missing	999	63	63.6		
Total		99	100.0		

a Agency size strata = Unspecified Size

#### total unsolved unnamed "other" cases with possible biological evidence(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	19	19.2	95.0	95.0
	99	1	1.0	5.0	100.0
	Total	20	20.2	100.0	
Missing	999	79	79.8		
Total		99	100.0		

a Agency size strata = Unspecified Size

educated	guess not	possibl	e(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	18	18.2	18.6	18.6
	No	79	79.8	81.4	100.0
	Total	97	98.0	100.0	
Missing	999	2	2.0		
Total		99	100.0		

a Agency size strata = Unspecified Size

## unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	3	3.0	3.1	3.1
	No	94	94.9	96.9	100.0
	Total	97	98.0	100.0	
Missing	999	2	2.0		
Total		99	100.0		

a Agency size strata = Unspecified Size

#### % cases with rape kits only(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	32	32.3	54.2	54.2
	11-25%	1	1.0	1.7	55.9
	25-50%	6	6.1	10.2	66.1
	50-75%	4	4.0	6.8	72.9
	75-100%	16	16.2	27.1	100.0
	Total	59	59.6	100.0	
Missing	999	40	40.4		
Total		99	100.0		

a Agency size strata = Unspecified Size

## % cases with rape kits and other DNA evidence(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	28	28.3	57.1	57.1
	11-25%	2	2.0	4.1	61.2
	25-50%	8	8.1	16.3	77.6
	50-75%	4	4.0	8.2	85.7
	75-100%	7	7.1	14.3	100.0
	Total	49	49.5	100.0	
Missing	999	50	50.5		
Total		99	100.0		

a Agency size strata = Unspecified Size

% cases	with	other	$DN\Delta$	evidence	only(	a١
/0 Cases	WILLI	ouiei	DINA	evidence	OHITY	a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	34	34.3	89.5	89.5
	11-25%	1	1.0	2.6	92.1
	25-50%	1	1.0	2.6	94.7
	75-100%	2	2.0	5.3	100.0
	Total	38	38.4	100.0	
Missing	999	61	61.6		
Total		99	100.0		

a Agency size strata = Unspecified Size

## educated guess not possible(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	17	17.2	17.3	17.3
	No	81	81.8	82.7	100.0
	Total	98	99.0	100.0	
Missing	999	1	1.0		
Total		99	100.0		

a Agency size strata = Unspecified Size

#### unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	2	2.0	2.0	2.0
	No	96	97.0	98.0	100.0
	Total	98	99.0	100.0	
Missing	999	1	1.0		
Total		99	100.0		

a Agency size strata = Unspecified Size

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	31	31.3	68.9	68.9
	1	4	4.0	8.9	77.8
	2	3	3.0	6.7	84.4
	6	2	2.0	4.4	88.9
	8	1	1.0	2.2	91.1
	15	1	1.0	2.2	93.3
	21	1	1.0	2.2	95.6
	120	1	1.0	2.2	97.8
	600	1	1.0	2.2	100.0
	Total	45	45.5	100.0	
Missing	999	54	54.5		
Total		99	100.0		

a Agency size strata = Unspecified Size

# unnamed rape cases within statute of limitations(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	31	31.3	70.5	70.5
	1	5	5.1	11.4	81.8
	2	1	1.0	2.3	84.1
	4	1	1.0	2.3	86.4
	5	1	1.0	2.3	88.6
	6	1	1.0	2.3	90.9
	8	1	1.0	2.3	93.2
	15	1	1.0	2.3	95.5
	20	1	1.0	2.3	97.7
	600	1	1.0	2.3	100.0
	Total	44	44.4	100.0	
Missing	999	55	55.6		
Total		99	100.0		

a Agency size strata = Unspecified Size

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	39	39.4	84.8	84.8
	1	2	2.0	4.3	89.1
	5	1	1.0	2.2	91.3
	10	1	1.0	2.2	93.5
	12	1	1.0	2.2	95.7
	70	1	1.0	2.2	97.8
	9999	1	1.0	2.2	100.0
	Total	46	46.5	100.0	
Missing	999	53	53.5		
Total		99	100.0		

a Agency size strata = Unspecified Size

## unnamed rape cases to exceed statute of limitations(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	38	38.4	90.5	90.5
	1	1	1.0	2.4	92.9
	5	1	1.0	2.4	95.2
	12	1	1.0	2.4	97.6
	70	1	1.0	2.4	100.0
	Total	42	42.4	100.0	
Missing	999	57	57.6		
Total		99	100.0		

a Agency size strata = Unspecified Size

## educated guess not possible(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	20	20.2	20.6	20.6
	No	77	77.8	79.4	100.0
	Total	97	98.0	100.0	
Missing	999	2	2.0		
Total		99	100.0		

a Agency size strata = Unspecified Size

unable to answer comp	letely(a	ı)
-----------------------	----------	----

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	2	2.0	2.1	2.1
	No	95	96.0	97.9	100.0
	Total	97	98.0	100.0	
Missing	999	2	2.0		
Total		99	100.0		

a Agency size strata = Unspecified Size

## unanalyzed evidence stored--central storage area(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	55	55.6	57.3	57.3
	No	41	41.4	42.7	100.0
	Total	96	97.0	100.0	
Missing	999	3	3.0		
Total		99	100.0		

a Agency size strata = Unspecified Size

## unanalyzed evidence stored--decentralized storage area(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	5	5.1	5.2	5.2
	No	91	91.9	94.8	100.0
	Total	96	97.0	100.0	
Missing	999	3	3.0		
Total		99	100.0		

a Agency size strata = Unspecified Size

## unanalyzed evidence stored--prosecutor's facility(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	2	2.0	2.1	2.1
	No	94	94.9	97.9	100.0
	Total	96	97.0	100.0	
Missing	999	3	3.0		
Total		99	100.0		

a Agency size strata = Unspecified Size

## unanalyzed evidence stored--crime lab facility(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	32	32.3	33.3	33.3
	No	64	64.6	66.7	100.0
	Total	96	97.0	100.0	
Missing	999	3	3.0		
Total		99	100.0		

a Agency size strata = Unspecified Size

#### unanalyzed evidence stored--other(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	5	5.1	5.2	5.2
	No	91	91.9	94.8	100.0
	Total	96	97.0	100.0	
Missing	999	3	3.0		
Total		99	100.0		

a Agency size strata = Unspecified Size

## sufficient space for long-term evidence storage?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Sufficient Storage	28	28.3	29.5	29.5
Insufficient Storage	Insufficient Storage	67	67.7	70.5	100.0
	Total	95	96.0	100.0	
Missing	999	4	4.0		
Total		99	100.0		

a Agency size strata = Unspecified Size

## If insufficient, how critical is additional space?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Highly Critical	16	16.2	21.6	21.6
	Critical	29	29.3	39.2	60.8
	Pressing But Not Critical	29	29.3	39.2	100.0
	Total	74	74.7	100.0	
Missing	999	25	25.3		
Total		99	100.0		

a Agency size strata = Unspecified Size

DNA not sent to lab applicablenot applicat	ole(a	)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	41	41.4	42.3	42.3
	No	56	56.6	57.7	100.0
	Total	97	98.0	100.0	
Missing	999	2	2.0		
Total		99	100.0		

a Agency size strata = Unspecified Size

#### DNA not sent to lab rank--not applicable(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	17	17.2	85.0	85.0
	Second Most Significant	1	1.0	5.0	90.0
	Third Most Significant	2	2.0	10.0	100.0
	Total	20	20.2	100.0	
Missing	999	79	79.8		
Total		99	100.0		

a Agency size strata = Unspecified Size

#### DNA not sent to lab applicable--no suspect identified(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	18	18.2	18.6	18.6
	No	79	79.8	81.4	100.0
	Total	97	98.0	100.0	
Missing	999	2	2.0		
Total		99	100.0		

a Agency size strata = Unspecified Size

## DNA not sent to lab rank--no suspect identified(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	9	9.1	64.3	64.3
	Second Most Significant	4	4.0	28.6	92.9
	Third Most Significant	1	1.0	7.1	100.0
	Total	14	14.1	100.0	
Missing	999	85	85.9		
Total		99	100.0		

a Agency size strata = Unspecified Size

DNA not sent to	lab applicable	no suspect c	harged(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	9	9.1	9.3	9.3
	No	88	88.9	90.7	100.0
	Total	97	98.0	100.0	
Missing	999	2	2.0		
Total		99	100.0		

a Agency size strata = Unspecified Size

# DNA not sent to lab rank--no suspect charged(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	3	3.0	50.0	50.0
	Second Most Significant	1	1.0	16.7	66.7
	Third Most Significant	2	2.0	33.3	100.0
	Total	6	6.1	100.0	
Missing	999	93	93.9		
Total		99	100.0		

a Agency size strata = Unspecified Size

## DNA not sent to lab applicable--guilty plea anticipated(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	10	10.1	10.3	10.3
	No	87	87.9	89.7	100.0
	Total	97	98.0	100.0	
Missing	999	2	2.0		
Total		99	100.0		

a Agency size strata = Unspecified Size

#### DNA not sent to lab rank--guilty plea anticipated(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	3	3.0	50.0	50.0
	Second Most Significant	2	2.0	33.3	83.3
	Third Most Significant	1	1.0	16.7	100.0
	Total	6	6.1	100.0	
Missing	999	93	93.9		
Total		99	100.0		

a Agency size strata = Unspecified Size

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	7	7.1	7.2	7.2
	No	90	90.9	92.8	100.0
	Total	97	98.0	100.0	
Missing	999	2	2.0		
Total		99	100.0		

a Agency size strata = Unspecified Size

## DNA not sent to lab rank--DNA use uncertain(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	3	3.0	50.0	50.0
	Second Most Significant	2	2.0	33.3	83.3
-	Third Most Significant	1	1.0	16.7	100.0
	Total	6	6.1	100.0	
Missing	999	93	93.9		
Total		99	100.0		

a Agency size strata = Unspecified Size

## DNA not sent to lab applicable--lack of \$(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	7	7.1	7.2	7.2
	No	90	90.9	92.8	100.0
	Total	97	98.0	100.0	
Missing	999	2	2.0		
Total		99	100.0		

a Agency size strata = Unspecified Size

#### DNA not sent to lab rank--lack of \$(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	2	2.0	40.0	40.0
	Second Most Significant	2	2.0	40.0	80.0
	Third Most Significant	1	1.0	20.0	100.0
	Total	5	5.1	100.0	
Missing	999	94	94.9		
Total		99	100.0		

a Agency size strata = Unspecified Size

DNA not sent to lab applicableinability to get ti	imely results(a)
---	------------------

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	7	7.1	7.2	7.2
	No	90	90.9	92.8	100.0
	Total	97	98.0	100.0	
Missing	999	2	2.0		
Total		99	100.0		

a Agency size strata = Unspecified Size

## DNA not sent to lab rank--inability to get timely results(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Second Most Significant	2	2.0	50.0	50.0
Third Most Significant	2	2.0	50.0	100.0	
	Total	4	4.0	100.0	
Missing	999	95	96.0		
Total		99	100.0		

a Agency size strata = Unspecified Size

#### DNA not sent to lab applicable--lab not processing requests(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	3	3.0	3.1	3.1
	No	94	94.9	96.9	100.0
	Total	97	98.0	100.0	
Missing	999	2	2.0		
Total		99	100.0		

a Agency size strata = Unspecified Size

## DNA not sent to lab rank--lab not processing requests(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	1	1.0	100.0	100.0
Missing	999	98	99.0		
Total		99	100.0		

a Agency size strata = Unspecified Size

## DNA not sent to lab applicable--uncertain where to send DNA(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	1	1.0	1.0	1.0
	No	96	97.0	99.0	100.0
	Total	97	98.0	100.0	
Missing	999	2	2.0		
Total		99	100.0		

a Agency size strata = Unspecified Size

#### DNA not sent to lab rank--uncertain where to send DNA(a)

		Frequency	Percent
Missing	999	99	100.0

a Agency size strata = Unspecified Size

#### DNA not sent to lab applicable--analysis not requested by prosecutor(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	5	5.1	5.2	5.2
	No	92	92.9	94.8	100.0
	Total	97	98.0	100.0	
Missing	999	2	2.0		
Total		99	100.0		

a Agency size strata = Unspecified Size

#### DNA not sent to lab rank--analysis not requested by prosecutor(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	1	1.0	33.3	33.3
	Second Most Significant	1	1.0	33.3	66.7
Third Most	Third Most Significant	1	1.0	33.3	100.0
	Total	3	3.0	100.0	
Missing	999	96	97.0		
Total		99	100.0		

a Agency size strata = Unspecified Size

## DNA not sent to lab applicable--other(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	6	6.1	6.3	6.3
	No	90	90.9	93.8	100.0
	Total	96	97.0	100.0	
Missing	999	3	3.0		
Total		99	100.0		

a Agency size strata = Unspecified Size

#### DNA not sent to lab rank--other(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	2	2.0	50.0	50.0
	Second Most Significant	2	2.0	50.0	100.0
	Total	4	4.0	100.0	
Missing	999	95	96.0		
Total		99	100.0		

a Agency size strata = Unspecified Size

## Department routinely collect DNA from property crimes?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	28	28.3	30.8	30.8
	No	63	63.6	69.2	100.0
	Total	91	91.9	100.0	
Missing	999	8	8.1		
Total		99	100.0		

a Agency size strata = Unspecified Size

## Lab accept AND process unnamed suspect DNA cases for prop. crimes?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	35	35.4	48.6	48.6
	No	37	37.4	51.4	100.0
	Total	72	72.7	100.0	
Missing	999	27	27.3		
Total		99	100.0		

a Agency size strata = Unspecified Size

## Lab accept but NOT process unnamed suspect DNA cases for prop. crimes?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	20	20.2	31.3	31.3
	No	44	44.4	68.8	100.0
	Total	64	64.6	100.0	
Missing	999	35	35.4		
Total		99	100.0		

a Agency size strata = Unspecified Size

#### unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	8	8.1	8.3	8.3
	No	88	88.9	91.7	100.0
	Total	96	97.0	100.0	
Missing	999	3	3.0		
Total		99	100.0		

a Agency size strata = Unspecified Size

## agency have cold case squad?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	20	20.2	21.1	21.1
	No	75	75.8	78.9	100.0
	Total	95	96.0	100.0	
Missing	999	4	4.0		
Total		99	100.0		

a Agency size strata = Unspecified Size

#### unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	No	96	97.0	100.0	100.0
Missing	999	3	3.0		
Total		99	100.0		

a Agency size strata = Unspecified Size

## who to contact?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	2	1	1.0	1.0	1.0
	99	1	1.0	1.0	2.0
	999	97	98.0	98.0	100.0
	Total	99	100.0	100.0	

a Agency size strata = Unspecified Size

# # of cold cases eligible for review(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	2	2.0	11.1	11.1
	1	2	2.0	11.1	22.2
	2	1	1.0	5.6	27.8
	4	2	2.0	11.1	38.9
	6	1	1.0	5.6	44.4
	8	1	1.0	5.6	50.0
	9	1	1.0	5.6	55.6
	17	1	1.0	5.6	61.1
	20	2	2.0	11.1	72.2
	50	1	1.0	5.6	77.8
	200	1	1.0	5.6	83.3
	350	1	1.0	5.6	88.9
	450	1	1.0	5.6	94.4
	1200	1	1.0	5.6	100.0
	Total	18	18.2	100.0	
Missing	999	81	81.8		
Total		99	100.0		

a Agency size strata = Unspecified Size

# of cold cases reviewed(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	2	2.0	12.5	12.5
	1	1	1.0	6.3	18.8
	2	1	1.0	6.3	25.0
	4	2	2.0	12.5	37.5
	5	1	1.0	6.3	43.8
	6	1	1.0	6.3	50.0
	7	1	1.0	6.3	56.3
	8	2	2.0	12.5	68.8
	9	1	1.0	6.3	75.0
	20	1	1.0	6.3	81.3
	150	1	1.0	6.3	87.5
	200	1	1.0	6.3	93.8
	600	1	1.0	6.3	100.0
	Total	16	16.2	100.0	
Missing	999	83	83.8		
Total	-:44-	99	100.0		

a Agency size strata = Unspecified Size

## # of cold cases sent for DNA testing(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	9	9.1	50.0	50.0
	1	3	3.0	16.7	66.7
	2	1	1.0	5.6	72.2
	4	1	1.0	5.6	77.8
	8	1	1.0	5.6	83.3
	30	1	1.0	5.6	88.9
	80	1	1.0	5.6	94.4
	200	1	1.0	5.6	100.0
	Total	18	18.2	100.0	
Missing	999	81	81.8		
Total		99	100.0		

a Agency size strata = Unspecified Size

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	6	6.1	42.9	42.9
	1	2	2.0	14.3	57.1
	15	1	1.0	7.1	64.3
	20	1	1.0	7.1	71.4
	142	1	1.0	7.1	78.6
	300	2	2.0	14.3	92.9
	400	1	1.0	7.1	100.0
	Total	14	14.1	100.0	
Missing	999	85	85.9		
Total		99	100.0		

a Agency size strata = Unspecified Size

#### educated guess is not possible(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	4	4.0	4.1	4.1
	No	93	93.9	95.9	100.0
	Total	97	98.0	100.0	
Missing	999	2	2.0		
Total		99	100.0		

a Agency size strata = Unspecified Size

#### unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	No	97	98.0	100.0	100.0
Missing	999	2	2.0		
Total		99	100.0		

a Agency size strata = Unspecified Size

#### who to contact?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	99	1	1.0	1.0	1.0
	999	98	99.0	99.0	100.0
	Total	99	100.0	100.0	

a Agency size strata = Unspecified Size

agency have plan to review	[cold] cases	for biologica	I testing?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	25	25.3	34.2	34.2
	No	48	48.5	65.8	100.0
	Total	73	73.7	100.0	
Missing	999	26	26.3		
Total		99	100.0		

a Agency size strata = Unspecified Size

## # of cases eligible for review(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	19	19.2	54.3	54.3
	1	6	6.1	17.1	71.4
	2	2	2.0	5.7	77.1
	3	1	1.0	2.9	80.0
	4	1	1.0	2.9	82.9
	5	1	1.0	2.9	85.7
	10	2	2.0	5.7	91.4
	20	1	1.0	2.9	94.3
	45	1	1.0	2.9	97.1
	99	1	1.0	2.9	100.0
	Total	35	35.4	100.0	
Missing	999	64	64.6		
Total		99	100.0		

a Agency size strata = Unspecified Size

# # of cases with potential for biological testing(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	19	19.2	65.5	65.5
	1	4	4.0	13.8	79.3
	3	1	1.0	3.4	82.8
	4	1	1.0	3.4	86.2
	5	1	1.0	3.4	89.7
	7	1	1.0	3.4	93.1
	20	1	1.0	3.4	96.6
	99	1	1.0	3.4	100.0
	Total	29	29.3	100.0	
Missing	999	70	70.7		
Total	ļ	99	100.0		

a Agency size strata = Unspecified Size

## educated guess is not possible(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	14	14.1	14.7	14.7
	No	81	81.8	85.3	100.0
	Total	95	96.0	100.0	
Missing	999	4	4.0		
Total		99	100.0		

a Agency size strata = Unspecified Size

#### unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	No	96	97.0	100.0	100.0
Missing	999	3	3.0		
Total		99	100.0		

a Agency size strata = Unspecified Size

## agency interested in using DNA database?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	88	88.9	95.7	95.7
	No	4	4.0	4.3	100.0
	Total	92	92.9	100.0	
Missing	999	7	7.1		
Total		99	100.0		

a Agency size strata = Unspecified Size

#### has agency used DNA database for cases with no direct match?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	11	11.1	12.0	12.0
	No	81	81.8	88.0	100.0
	Total	92	92.9	100.0	
Missing	999	7	7.1		
Total		99	100.0		

a Agency size strata = Unspecified Size

# Any comments?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	27	27.3	27.6	27.6
	No	71	71.7	72.4	100.0
	Total	98	99.0	100.0	
Missing	999	1	1.0		
Total		99	100.0		

a Agency size strata = Unspecified Size

# Agency size strata = Tribal

STATE(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	999	1	1.8	1.8	1.8
	AZ	8	14.3	14.3	16.1
	CO	2	3.6	3.6	19.6
	FL	1	1.8	1.8	21.4
	KS	2	3.6	3.6	25.0
	MI	2	3.6	3.6	28.6
	MN	1	1.8	1.8	30.4
	MT	2	3.6	3.6	33.9
	ND	1	1.8	1.8	35.7
	NM	5	8.9	8.9	44.6
	NV	3	5.4	5.4	50.0
	NY	1	1.8	1.8	51.8
	OK	8	14.3	14.3	66.1
	OR	2	3.6	3.6	69.6
	SD	3	5.4	5.4	75.0
	TX	1	1.8	1.8	76.8
	WA	10	17.9	17.9	94.6
	Wi	1	1.8	1.8	96.4
	WI	2	3.6	3.6	100.0
	Total	56	100.0	100.0	

a Agency size strata = Tribal

## Where are cases primarily sent for DNA analysis?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	State Agency Lab	35	62.5	64.8	64.8
	Local Agency Lab	8	14.3	14.8	79.6
	FBI Lab	9	16.1	16.7	96.3
	Other	2	3.6	3.7	100.0
	Total	54	96.4	100.0	
Missing	999	2	3.6		
Total		56	100.0		

a Agency size strata = Tribal

% rape	s with	possible	DNA(	a)
--------	--------	----------	------	----

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	18	32.1	36.0	36.0
	11-25%	2	3.6	4.0	40.0
	25-50%	4	7.1	8.0	48.0
	50-75%	5	8.9	10.0	58.0
	75-100%	21	37.5	42.0	100.0
	Total	50	89.3	100.0	
Missing	999	6	10.7		
Total		56	100.0		

a Agency size strata = Tribal

## % homicides with possible DNA(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	17	30.4	39.5	39.5
	11-25%	3	5.4	7.0	46.5
	25-50%	4	7.1	9.3	55.8
	50-75%	4	7.1	9.3	65.1
	75-100%	15	26.8	34.9	100.0
	Total	43	76.8	100.0	
Missing	999	13	23.2		
Total		56	100.0		

a Agency size strata = Tribal

## educated guess not possible(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	4	7.1	7.1	7.1
	No	52	92.9	92.9	100.0
	Total	56	100.0	100.0	

a Agency size strata = Tribal

## unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	3	5.4	5.4	5.4
	No	53	94.6	94.6	100.0
	Total	56	100.0	100.0	

a Agency size strata = Tribal

## unsolved homicides--total cases(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	22	39.3	59.5	59.5
	1	7	12.5	18.9	78.4
	2	4	7.1	10.8	89.2
	3	1	1.8	2.7	91.9
	4	2	3.6	5.4	97.3
	5	1	1.8	2.7	100.0
	Total	37	66.1	100.0	
Missing	999	19	33.9		
Total		56	100.0		

a Agency size strata = Tribal

## unsolved homicides--cases with DNA(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	16	28.6	69.6	69.6
	1	5	8.9	21.7	91.3
	2	2	3.6	8.7	100.0
	Total	23	41.1	100.0	
Missing	999	33	58.9		
Total		56	100.0		

a Agency size strata = Tribal

#### unsolved rapes--total cases(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	24	42.9	68.6	68.6
	1	3	5.4	8.6	77.1
	2	2	3.6	5.7	82.9
	3	4	7.1	11.4	94.3
	5	1	1.8	2.9	97.1
	10	1	1.8	2.9	100.0
	Total	35	62.5	100.0	
Missing	999	21	37.5		
Total		56	100.0		

a Agency size strata = Tribal

unsolved rapescases	with	DNA(	a)
---------------------	------	------	----

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	15	26.8	78.9	78.9
	1	1	1.8	5.3	84.2
	2	1	1.8	5.3	89.5
	3	1	1.8	5.3	94.7
	4	1	1.8	5.3	100.0
	Total	19	33.9	100.0	
Missing	999	37	66.1		
Total		56	100.0		

a Agency size strata = Tribal

## educated guess not possible(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	14	25.0	25.0	25.0
	No	42	75.0	75.0	100.0
	Total	56	100.0	100.0	

a Agency size strata = Tribal

## unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	6	10.7	10.7	10.7
	No	50	89.3	89.3	100.0
	Total	56	100.0	100.0	

a Agency size strata = Tribal

# total unsolved rape cases with possible biological evidence(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	32	57.1	91.4	91.4
	3	2	3.6	5.7	97.1
	5	1	1.8	2.9	100.0
	Total	35	62.5	100.0	
Missing	999	21	37.5		
Total		56	100.0		

a Agency size strata = Tribal

## total unsolved homicide cases with possible biological evidence(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	27	48.2	93.1	93.1
	1	2	3.6	6.9	100.0
	Total	29	51.8	100.0	
Missing	999	27	48.2		
Total		56	100.0		

a Agency size strata = Tribal

## total unsolved property crimes with possible biological evidence(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	24	42.9	88.9	88.9
	2	2	3.6	7.4	96.3
	10	1	1.8	3.7	100.0
	Total	27	48.2	100.0	
Missing	999	28	50.0		
	System	1	1.8		
	Total	29	51.8		
Total		56	100.0		

a Agency size strata = Tribal

#### total unsolved "other" cases with possible biological evidence(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	15	26.8	88.2	88.2
	1	1	1.8	5.9	94.1
	2	1	1.8	5.9	100.0
	Total	17	30.4	100.0	
Missing	999	39	69.6		
Total		56	100.0		

a Agency size strata = Tribal

## total unsolved unnamed rape cases with possible biological evidence(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	16	28.6	80.0	80.0
	1	1	1.8	5.0	85.0
	2	1	1.8	5.0	90.0
	4	1	1.8	5.0	95.0
	5	1	1.8	5.0	100.0
	Total	20	35.7	100.0	
Missing	999	35	62.5		
	System	1	1.8		
	Total	36	64.3		
Total		56	100.0		

a Agency size strata = Tribal

#### total unsolved unnamed homicide cases with possible biological evidence(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	13	23.2	76.5	76.5
	1	2	3.6	11.8	88.2
	2	1	1.8	5.9	94.1
	4	1	1.8	5.9	100.0
	Total	17	30.4	100.0	
Missing	999	39	69.6		
Total		56	100.0		

a Agency size strata = Tribal

### total unsolved unnamed property crimes with possible biological evidence(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	12	21.4	75.0	75.0
	1	1	1.8	6.3	81.3
	2	2	3.6	12.5	93.8
	10	1	1.8	6.3	100.0
	Total	16	28.6	100.0	
Missing	999	40	71.4		
Total		56	100.0		

a Agency size strata = Tribal

## total unsolved unnamed "other" cases with possible biological evidence(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	9	16.1	75.0	75.0
	1	1	1.8	8.3	83.3
	2	2	3.6	16.7	100.0
	Total	12	21.4	100.0	
Missing	999	44	78.6		
Total		56	100.0		

a Agency size strata = Tribal

#### educated guess not possible(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	9	16.1	16.1	16.1
	No	47	83.9	83.9	100.0
	Total	56	100.0	100.0	

a Agency size strata = Tribal

## unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	6	10.7	10.7	10.7
	No	50	89.3	89.3	100.0
	Total	56	100.0	100.0	

a Agency size strata = Tribal

#### % cases with rape kits only(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	16	28.6	64.0	64.0
	25-50%	1	1.8	4.0	68.0
	50-75%	3	5.4	12.0	80.0
	75-100%	5	8.9	20.0	100.0
	Total	25	44.6	100.0	
Missing	999	31	55.4		
Total		56	100.0		

a Agency size strata = Tribal

% cases with rape kits and other DNA evidence(a
---

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	16	28.6	64.0	64.0
	25-50%	1	1.8	4.0	68.0
	50-75%	2	3.6	8.0	76.0
	75-100%	6	10.7	24.0	100.0
	Total	25	44.6	100.0	
Missing	999	31	55.4		
Total		56	100.0		

a Agency size strata = Tribal

## % cases with other DNA evidence only(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	16	28.6	76.2	76.2
	11-25%	1	1.8	4.8	81.0
	50-75%	2	3.6	9.5	90.5
	75-100%	2	3.6	9.5	100.0
	Total	21	37.5	100.0	
Missing	999	35	62.5		
Total		56	100.0		

a Agency size strata = Tribal

## educated guess not possible(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	7	12.5	12.7	12.7
	No	48	85.7	87.3	100.0
	Total	55	98.2	100.0	
Missing	999	1	1.8		
Total		56	100.0		

a Agency size strata = Tribal

## unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	5	8.9	9.1	9.1
	No	50	89.3	90.9	100.0
	Total	55	98.2	100.0	
Missing	999	1	1.8		
Total		56	100.0		

a Agency size strata = Tribal

rape cases within statute of limitations(a	a)	)
--	----	---

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	17	30.4	63.0	63.0
	1	4	7.1	14.8	77.8
	2	2	3.6	7.4	85.2
	3	1	1.8	3.7	88.9
	4	2	3.6	7.4	96.3
	6	1	1.8	3.7	100.0
	Total	27	48.2	100.0	
Missing	999	29	51.8		
Total		56	100.0		

a Agency size strata = Tribal

## unnamed rape cases within statute of limitations(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	18	32.1	81.8	81.8
	1	2	3.6	9.1	90.9
	2	2	3.6	9.1	100.0
	Total	22	39.3	100.0	
Missing	999	34	60.7		
Total		56	100.0		

a Agency size strata = Tribal

#### rape cases to exceed statute of limitations(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	23	41.1	95.8	95.8
	3	1	1.8	4.2	100.0
	Total	24	42.9	100.0	
Missing	999	32	57.1		
Total		56	100.0		

a Agency size strata = Tribal

## unnamed rape cases to exceed statute of limitations(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	19	33.9	95.0	95.0
	2	1	1.8	5.0	100.0
	Total	20	35.7	100.0	
Missing	999	36	64.3		
Total		56	100.0		

a Agency size strata = Tribal

#### educated guess not possible(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	11	19.6	19.6	19.6
	No	45	80.4	80.4	100.0
	Total	56	100.0	100.0	

a Agency size strata = Tribal

#### unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	6	10.7	10.7	10.7
	No	50	89.3	89.3	100.0
	Total	56	100.0	100.0	

a Agency size strata = Tribal

#### unanalyzed evidence stored--central storage area(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	36	64.3	64.3	64.3
	No	20	35.7	35.7	100.0
	Total	56	100.0	100.0	

a Agency size strata = Tribal

#### unanalyzed evidence stored--decentralized storage area(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	3	5.4	5.4	5.4
	No	53	94.6	94.6	100.0
	Total	56	100.0	100.0	

a Agency size strata = Tribal

## unanalyzed evidence stored--prosecutor's facility(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	No	56	100.0	100.0	100.0

a Agency size strata = Tribal

## unanalyzed evidence stored--crime lab facility(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	11	19.6	19.6	19.6
	No	45	80.4	80.4	100.0
	Total	56	100.0	100.0	

a Agency size strata = Tribal

## unanalyzed evidence stored--other(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	6	10.7	10.7	10.7
	No	50	89.3	89.3	100.0
	Total	56	100.0	100.0	

a Agency size strata = Tribal

#### sufficient space for long-term evidence storage?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Sufficient Storage	19	33.9	34.5	34.5
Insufficie Storage	Insufficient Storage	36	64.3	65.5	100.0
	Total	55	98.2	100.0	
Missing	999	1	1.8		
Total		56	100.0		

a Agency size strata = Tribal

## If insufficient, how critical is additional space?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Highly Critical	13	23.2	29.5	29.5
	Critical	19	33.9	43.2	72.7
	Pressing But Not Critical	12	21.4	27.3	100.0
	Total	44	78.6	100.0	
Missing	999	12	21.4		
Total		56	100.0		

a Agency size strata = Tribal

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	22	39.3	40.7	40.7
	No	32	57.1	59.3	100.0
	Total	54	96.4	100.0	
Missing	999	2	3.6		
Total		56	100.0		

a Agency size strata = Tribal

#### DNA not sent to lab rank--not applicable(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	18	32.1	100.0	100.0
Missing	999	38	67.9		
Total		56	100.0		

a Agency size strata = Tribal

## DNA not sent to lab applicable--no suspect identified(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	10	17.9	18.5	18.5
	No	44	78.6	81.5	100.0
	Total	54	96.4	100.0	
Missing	999	2	3.6		
Total		56	100.0		

a Agency size strata = Tribal

## DNA not sent to lab rank--no suspect identified(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	9	16.1	81.8	81.8
	Third Most Significant	2	3.6	18.2	100.0
	Total	11	19.6	100.0	
Missing	999	45	80.4		
Total		56	100.0		

a Agency size strata = Tribal

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	6	10.7	11.1	11.1
	No	48	85.7	88.9	100.0
	Total	54	96.4	100.0	
Missing	999	2	3.6		
Total		56	100.0		

a Agency size strata = Tribal

#### DNA not sent to lab rank--no suspect charged(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	2	3.6	50.0	50.0
	Second Most Significant	2	3.6	50.0	100.0
	Total	4	7.1	100.0	
Missing	999	52	92.9		
Total		56	100.0		

a Agency size strata = Tribal

## DNA not sent to lab applicable--guilty plea anticipated(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	6	10.7	11.1	11.1
	No	48	85.7	88.9	100.0
	Total	54	96.4	100.0	
Missing	999	2	3.6		
Total		56	100.0		

a Agency size strata = Tribal

## DNA not sent to lab rank--guilty plea anticipated(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	1	1.8	20.0	20.0
	Second Most Significant	3	5.4	60.0	80.0
Third Most Significant		1	1.8	20.0	100.0
	Total	5	8.9	100.0	
Missing	999	51	91.1		
Total		56	100.0		

a Agency size strata = Tribal

#### DNA not sent to lab applicable--DNA use uncertain(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	1	1.8	1.9	1.9
	No	53	94.6	98.1	100.0
	Total	54	96.4	100.0	
Missing	999	2	3.6		
Total		56	100.0		

a Agency size strata = Tribal

#### DNA not sent to lab rank--DNA use uncertain(a)

		Frequency	Percent
Missing	999	56	100.0

a Agency size strata = Tribal

#### DNA not sent to lab applicable--lack of \$(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	2	3.6	3.7	3.7
	No	52	92.9	96.3	100.0
	Total	54	96.4	100.0	
Missing	999	2	3.6		
Total		56	100.0		

a Agency size strata = Tribal

#### DNA not sent to lab rank--lack of \$(a)

	Frequency	Percent
Missing 999	56	100.0

a Agency size strata = Tribal

#### DNA not sent to lab applicable--inability to get timely results(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	2	3.6	3.7	3.7
	No	52	92.9	96.3	100.0
	Total	54	96.4	100.0	
Missing	999	2	3.6		
Total		56	100.0		

a Agency size strata = Tribal

## DNA not sent to lab rank--inability to get timely results(a)

		Frequency	Percent
Missing	999	56	100.0

a Agency size strata = Tribal

#### DNA not sent to lab applicable--lab not processing requests(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	2	3.6	3.7	3.7
	No	52	92.9	96.3	100.0
	Total	54	96.4	100.0	
Missing	999	2	3.6		
Total		56	100.0		

a Agency size strata = Tribal

#### DNA not sent to lab rank--lab not processing requests(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Third Most Significant	1	1.8	100.0	100.0
Missing	999	55	98.2		
Total		56	100.0		

a Agency size strata = Tribal

#### DNA not sent to lab applicable--uncertain where to send DNA(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	1	1.8	1.9	1.9
	No	53	94.6	98.1	100.0
	Total	54	96.4	100.0	
Missing	999	2	3.6		
Total		56	100.0		

a Agency size strata = Tribal

#### DNA not sent to lab rank--uncertain where to send DNA(a)

		Frequency	Percent
Missing	999	56	100.0

a Agency size strata = Tribal

## DNA not sent to lab applicable--analysis not requested by prosecutor(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	2	3.6	3.7	3.7
	No	52	92.9	96.3	100.0
	Total	54	96.4	100.0	
Missing	999	2	3.6		
Total		56	100.0		

a Agency size strata = Tribal

#### DNA not sent to lab rank--analysis not requested by prosecutor(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Third Most Significant	1	1.8	100.0	100.0
Missing	999	55	98.2		
Total		56	100.0		

a Agency size strata = Tribal

#### DNA not sent to lab applicable--other(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	7	12.5	13.0	13.0
	No	47	83.9	87.0	100.0
	Total	54	96.4	100.0	
Missing	999	2	3.6		
Total		56	100.0		

a Agency size strata = Tribal

#### DNA not sent to lab rank--other(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	1	1.8	33.3	33.3
	Second Most Significant	2	3.6	66.7	100.0
	Total	3	5.4	100.0	
Missing	999	53	94.6		
Total		56	100.0		

a Agency size strata = Tribal

## Department routinely collect DNA from property crimes?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	9	16.1	18.8	18.8
	No	39	69.6	81.3	100.0
	Total	48	85.7	100.0	
Missing	999	8	14.3		
Total		56	100.0		

a Agency size strata = Tribal

## Lab accept AND process unnamed suspect DNA cases for prop. crimes?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	8	14.3	20.5	20.5
	No	31	55.4	79.5	100.0
	Total	39	69.6	100.0	
Missing	999	17	30.4		
Total		56	100.0		

a Agency size strata = Tribal

#### Lab accept but NOT process unnamed suspect DNA cases for prop. crimes?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	5	8.9	13.9	13.9
	No	31	55.4	86.1	100.0
	Total	36	64.3	100.0	
Missing	999	20	35.7		
Total		56	100.0		

a Agency size strata = Tribal

#### unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	5	8.9	9.3	9.3
	No	49	87.5	90.7	100.0
	Total	54	96.4	100.0	
Missing	999	2	3.6		
Total		56	100.0		

a Agency size strata = Tribal

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	2	3.6	3.8	3.8
	No	51	91.1	96.2	100.0
	Total	53	94.6	100.0	
Missing	999	3	5.4		
Total		56	100.0		

a Agency size strata = Tribal

## unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	1	1.8	1.9	1.9
	No	53	94.6	98.1	100.0
	Total	54	96.4	100.0	
Missing	999	2	3.6		
Total		56	100.0		

a Agency size strata = Tribal

## who to contact?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	999	55	98.2	98.2	98.2
	Lake Co. S.O.	1	1.8	1.8	100.0
	Total	56	100.0	100.0	

a Agency size strata = Tribal

## # of cold cases eligible for review(a)

					0
		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	1	1.8	33.3	33.3
	1	1	1.8	33.3	66.7
	4	1	1.8	33.3	100.0
	Total	3	5.4	100.0	
Missing	999	53	94.6		
Total		56	100.0		

a Agency size strata = Tribal

## # of cold cases reviewed(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	1	1.8	50.0	50.0
	4	1	1.8	50.0	100.0
	Total	2	3.6	100.0	
Missing	999	54	96.4		
Total		56	100.0		

a Agency size strata = Tribal

#### # of cold cases sent for DNA testing(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	2	3.6	100.0	100.0
Missing	999	54	96.4		
Total		56	100.0		

a Agency size strata = Tribal

#### # of additional cold cases potentially eligible for DNA testing(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	2	3.6	66.7	66.7
	1	1	1.8	33.3	100.0
	Total	3	5.4	100.0	
Missing	999	53	94.6		
Total		56	100.0		

a Agency size strata = Tribal

## educated guess is not possible(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	3	5.4	5.6	5.6
	No	51	91.1	94.4	100.0
	Total	54	96.4	100.0	
Missing	999	2	3.6		
Total		56	100.0		

a Agency size strata = Tribal

#### unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	No	54	96.4	100.0	100.0
Missing	999	2	3.6		
Total		56	100.0		

a Agency size strata = Tribal

agency have plan to review	coldl caese	for biologica	I toeting 2/a
adency have plan to review	icolai cases	Tor biologica	i testina (ta)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	15	26.8	31.9	31.9
	No	32	57.1	68.1	100.0
	Total	47	83.9	100.0	
Missing	999	9	16.1		
Total		56	100.0		

a Agency size strata = Tribal

## # of cases eligible for review(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	10	17.9	45.5	45.5
	1	3	5.4	13.6	59.1
	2	3	5.4	13.6	72.7
	3	2	3.6	9.1	81.8
	4	1	1.8	4.5	86.4
	5	2	3.6	9.1	95.5
	10	1	1.8	4.5	100.0
	Total	22	39.3	100.0	
Missing	999	34	60.7		
Total		56	100.0		

a Agency size strata = Tribal

## # of cases with potential for biological testing(a)

		Fraguanay	Percent	Valid Percent	Cumulative Percent
		Frequency	reiteiit	valiu Percent	reiteilt
Valid	0	8	14.3	53.3	53.3
	1	4	7.1	26.7	80.0
	2	2	3.6	13.3	93.3
	5	1	1.8	6.7	100.0
	Total	15	26.8	100.0	
Missing	999	41	73.2		
Total		56	100.0		

a Agency size strata = Tribal

## educated guess is not possible(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	8	14.3	14.8	14.8
	No	46	82.1	85.2	100.0
	Total	54	96.4	100.0	
Missing	999	2	3.6		
Total		56	100.0		

a Agency size strata = Tribal

## unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	3	5.4	5.6	5.6
	No	51	91.1	94.4	100.0
	Total	54	96.4	100.0	
Missing	999	2	3.6		
Total		56	100.0		

a Agency size strata = Tribal

## agency interested in using DNA database?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	46	82.1	90.2	90.2
	No	5	8.9	9.8	100.0
	Total	51	91.1	100.0	
Missing	999	5	8.9		
Total		56	100.0		

a Agency size strata = Tribal

## has agency used DNA database for cases with no direct match?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	3	5.4	5.8	5.8
	No	49	87.5	94.2	100.0
	Total	52	92.9	100.0	
Missing	999	4	7.1		
Total		56	100.0		

a Agency size strata = Tribal

# Any comments?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	18	32.1	32.1	32.1
	No	38	67.9	67.9	100.0
	Total	56	100.0	100.0	

a Agency size strata = Tribal

# Agency size strata = Unknown

STATE(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	999	1	5.3	5.3	5.3
	CA	1	5.3	5.3	10.5
	FL	1	5.3	5.3	15.8
	GA	1	5.3	5.3	21.1
	IN	1	5.3	5.3	26.3
	KY	1	5.3	5.3	31.6
	MI	1	5.3	5.3	36.8
	NY	1	5.3	5.3	42.1
	ОН	1	5.3	5.3	47.4
	TX	1	5.3	5.3	52.6
	VA	2	10.5	10.5	63.2
	WA	7	36.8	36.8	100.0
	Total	19	100.0	100.0	

a Agency size strata = 999

# Where are cases primarily sent for DNA analysis?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	State Agency Lab	14	73.7	77.8	77.8
	Local Agency Lab	1	5.3	5.6	83.3
	Other	1	5.3	5.6	88.9
L	Private Lab/Commercial Lab	2	10.5	11.1	100.0
	Total	18	94.7	100.0	
Missing	999	1	5.3		
Total		19	100.0		

a Agency size strata = 999

% rapes with	possible	DNA(	a)
--------------	----------	------	----

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	3	15.8	21.4	21.4
	11-25%	1	5.3	7.1	28.6
	25-50%	1	5.3	7.1	35.7
	50-75%	3	15.8	21.4	57.1
	75-100%	6	31.6	42.9	100.0
	Total	14	73.7	100.0	
Missing	999	5	26.3		
Total		19	100.0		

a Agency size strata = 999

## % homicides with possible DNA(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	3	15.8	21.4	21.4
	25-50%	2	10.5	14.3	35.7
	50-75%	3	15.8	21.4	57.1
	75-100%	6	31.6	42.9	100.0
	Total	14	73.7	100.0	
Missing	999	5	26.3		
Total		19	100.0		

a Agency size strata = 999

#### educated guess not possible(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	3	15.8	16.7	16.7
	No	15	78.9	83.3	100.0
	Total	18	94.7	100.0	
Missing	999	1	5.3		
Total		19	100.0		

a Agency size strata = 999

## unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	3	15.8	16.7	16.7
	No	15	78.9	83.3	100.0
	Total	18	94.7	100.0	
Missing	999	1	5.3		
Total		19	100.0		

a Agency size strata = 999

unsolved homicidestotal case
------------------------------

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	2	10.5	22.2	22.2
	2	3	15.8	33.3	55.6
	4	1	5.3	11.1	66.7
	6	1	5.3	11.1	77.8
	77	2	10.5	22.2	100.0
	Total	9	47.4	100.0	
Missing	999	10	52.6		
Total		19	100.0		

a Agency size strata = 999

## unsolved homicides--cases with DNA(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	2	10.5	33.3	33.3
	1	2	10.5	33.3	66.7
	2	1	5.3	16.7	83.3
	6	1	5.3	16.7	100.0
	Total	6	31.6	100.0	
Missing	999	13	68.4		
Total		19	100.0		

a Agency size strata = 999

## unsolved rapes--total cases(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	4	21.1	50.0	50.0
	3	1	5.3	12.5	62.5
	8	1	5.3	12.5	75.0
	10	1	5.3	12.5	87.5
	80	1	5.3	12.5	100.0
	Total	8	42.1	100.0	
Missing	999	11	57.9		
Total		19	100.0		

a Agency size strata = 999

unsolved rapescases w	ith DN	A(a)
-----------------------	--------	------

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	2	10.5	40.0	40.0
	1	2	10.5	40.0	80.0
	8	1	5.3	20.0	100.0
	Total	5	26.3	100.0	
Missing	999	14	73.7		
Total		19	100.0		

a Agency size strata = 999

#### educated guess not possible(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	4	21.1	22.2	22.2
	No	14	73.7	77.8	100.0
	Total	18	94.7	100.0	
Missing	999	1	5.3		
Total		19	100.0		

a Agency size strata = 999

## unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	2	10.5	11.1	11.1
	No	16	84.2	88.9	100.0
	Total	18	94.7	100.0	
Missing	999	1	5.3		
Total		19	100.0		

a Agency size strata = 999

## total unsolved rape cases with possible biological evidence(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	7	36.8	87.5	87.5
	5	1	5.3	12.5	100.0
	Total	8	42.1	100.0	
Missing	999	11	57.9		
Total		19	100.0		

a Agency size strata = 999

## total unsolved homicide cases with possible biological evidence(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	5	26.3	50.0	50.0
	2	1	5.3	10.0	60.0
	4	1	5.3	10.0	70.0
	6	1	5.3	10.0	80.0
	39	2	10.5	20.0	100.0
	Total	10	52.6	100.0	
Missing	999	9	47.4		
Total		19	100.0		

a Agency size strata = 999

#### total unsolved property crimes with possible biological evidence(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	3	15.8	75.0	75.0
	12	1	5.3	25.0	100.0
	Total	4	21.1	100.0	
Missing	999	15	78.9		
Total		19	100.0		

a Agency size strata = 999

#### total unsolved "other" cases with possible biological evidence(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	3	15.8	100.0	100.0
Missing	999	16	84.2		
Total		19	100.0		

a Agency size strata = 999

#### total unsolved unnamed rape cases with possible biological evidence(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	4	21.1	80.0	80.0
	30	1	5.3	20.0	100.0
	Total	5	26.3	100.0	
Missing	999	14	73.7		
Total		19	100.0		

a Agency size strata = 999

## total unsolved unnamed homicide cases with possible biological evidence(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	3	15.8	42.9	42.9
	1	1	5.3	14.3	57.1
	5	1	5.3	14.3	71.4
	39	2	10.5	28.6	100.0
	Total	7	36.8	100.0	
Missing	999	12	63.2		
Total		19	100.0		

a Agency size strata = 999

#### total unsolved unnamed property crimes with possible biological evidence(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	3	15.8	75.0	75.0
	12	1	5.3	25.0	100.0
	Total	4	21.1	100.0	
Missing	999	15	78.9		
Total		19	100.0		

a Agency size strata = 999

#### total unsolved unnamed "other" cases with possible biological evidence(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	2	10.5	100.0	100.0
Missing	999	17	89.5		
Total		19	100.0		

a Agency size strata = 999

#### educated guess not possible(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	7	36.8	38.9	38.9
	No	11	57.9	61.1	100.0
	Total	18	94.7	100.0	
Missing	999	1	5.3		
Total		19	100.0		

a Agency size strata = 999

unable	to a	nswer	comp	letely	(a)
--------	------	-------	------	--------	-----

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	1	5.3	5.6	5.6
	No	17	89.5	94.4	100.0
	Total	18	94.7	100.0	
Missing	999	1	5.3		
Total		19	100.0		

a Agency size strata = 999

## % cases with rape kits only(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	2	10.5	33.3	33.3
	50-75%	1	5.3	16.7	50.0
	75-100%	3	15.8	50.0	100.0
	Total	6	31.6	100.0	
Missing	999	13	68.4		
Total		19	100.0		

a Agency size strata = 999

## % cases with rape kits and other DNA evidence(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	2	10.5	28.6	28.6
	11-25%	1	5.3	14.3	42.9
	50-75%	1	5.3	14.3	57.1
	75-100%	3	15.8	42.9	100.0
	Total	7	36.8	100.0	
Missing	999	12	63.2		
Total		19	100.0		

a Agency size strata = 999

## % cases with other DNA evidence only(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	3	15.8	75.0	75.0
	75-100%	1	5.3	25.0	100.0
	Total	4	21.1	100.0	
Missing	999	15	78.9		
Total		19	100.0		

a Agency size strata = 999

## educated guess not possible(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	1	5.3	5.6	5.6
	No	17	89.5	94.4	100.0
	Total	18	94.7	100.0	
Missing	999	1	5.3		
Total		19	100.0		

a Agency size strata = 999

#### unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	No	18	94.7	100.0	100.0
Missing	999	1	5.3		
Total		19	100.0		

a Agency size strata = 999

#### rape cases within statute of limitations(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	1	5.3	100.0	100.0
Missing	999	18	94.7		
Total		19	100.0		

a Agency size strata = 999

#### unnamed rape cases within statute of limitations(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	1	5.3	50.0	50.0
	30	1	5.3	50.0	100.0
	Total	2	10.5	100.0	
Missing	999	17	89.5		
Total		19	100.0		

a Agency size strata = 999

## rape cases to exceed statute of limitations(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	1	5.3	100.0	100.0
Missing	999	18	94.7		
Total		19	100.0		

a Agency size strata = 999

unnamed rape cases	to exceed	statute of	limitations(	a)
--------------------	-----------	------------	--------------	----

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	1	5.3	33.3	33.3
	1	1	5.3	33.3	66.7
	5	1	5.3	33.3	100.0
	Total	3	15.8	100.0	
Missing	999	16	84.2		
Total		19	100.0		

a Agency size strata = 999

## educated guess not possible(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	3	15.8	16.7	16.7
	No	15	78.9	83.3	100.0
	Total	18	94.7	100.0	
Missing	999	1	5.3		
Total		19	100.0		

a Agency size strata = 999

## unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	No	18	94.7	100.0	100.0
Missing	999	1	5.3		
Total		19	100.0		

a Agency size strata = 999

#### unanalyzed evidence stored--central storage area(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	15	78.9	83.3	83.3
	No	3	15.8	16.7	100.0
	Total	18	94.7	100.0	
Missing	999	1	5.3		
Total		19	100.0		

a Agency size strata = 999

## unanalyzed evidence stored--decentralized storage area(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	1	5.3	5.6	5.6
	No	17	89.5	94.4	100.0
	Total	18	94.7	100.0	
Missing	999	1	5.3		
Total		19	100.0		

a Agency size strata = 999

#### unanalyzed evidence stored--prosecutor's facility(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	No	18	94.7	100.0	100.0
Missing	999	1	5.3		
Total		19	100.0		

a Agency size strata = 999

## unanalyzed evidence stored--crime lab facility(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	5	26.3	27.8	27.8
	No	13	68.4	72.2	100.0
	Total	18	94.7	100.0	
Missing	999	1	5.3		
Total		19	100.0		

a Agency size strata = 999

#### unanalyzed evidence stored--other(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	No	18	94.7	100.0	100.0
Missing	999	1	5.3		
Total		19	100.0		

a Agency size strata = 999

## sufficient space for long-term evidence storage?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Sufficient Storage	8	42.1	44.4	44.4
	Insufficient Storage	10	52.6	55.6	100.0
	Total	18	94.7	100.0	
Missing	999	1	5.3		
Total		19	100.0		

a Agency size strata = 999

## If insufficient, how critical is additional space?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Highly Critical	2	10.5	18.2	18.2
	Critical	5	26.3	45.5	63.6
	Pressing But Not Critical	4	21.1	36.4	100.0
	Total	11	57.9	100.0	
Missing	999	8	42.1		
Total		19	100.0		

a Agency size strata = 999

#### DNA not sent to lab applicable--not applicable(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	5	26.3	27.8	27.8
	No	13	68.4	72.2	100.0
	Total	18	94.7	100.0	
Missing	999	1	5.3		
Total		19	100.0		

a Agency size strata = 999

## DNA not sent to lab rank--not applicable(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	3	15.8	100.0	100.0
Missing	999	16	84.2		
Total		19	100.0		

a Agency size strata = 999

## DNA not sent to lab applicable--no suspect identified(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	2	10.5	11.8	11.8
	No	15	78.9	88.2	100.0
	Total	17	89.5	100.0	
Missing	999	2	10.5		
Total		19	100.0		

a Agency size strata = 999

#### DNA not sent to lab rank--no suspect identified(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Third Most Significant	3	15.8	100.0	100.0
Missing	999	16	84.2		
Total		19	100.0		

a Agency size strata = 999

## DNA not sent to lab applicable--no suspect charged(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	3	15.8	17.6	17.6
	No	14	73.7	82.4	100.0
	Total	17	89.5	100.0	
Missing	999	2	10.5		
Total		19	100.0		

a Agency size strata = 999

## DNA not sent to lab rank--no suspect charged(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Second Most Significant	1	5.3	50.0	50.0
Third Most Significant		1	5.3	50.0	100.0
	Total	2	10.5	100.0	
Missing	999	17	89.5		
Total		19	100.0		

a Agency size strata = 999

## DNA not sent to lab applicable--guilty plea anticipated(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	3	15.8	17.6	17.6
	No	14	73.7	82.4	100.0
	Total	17	89.5	100.0	
Missing	999	2	10.5		
Total		19	100.0		

a Agency size strata = 999

## DNA not sent to lab rank--guilty plea anticipated(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	1	5.3	50.0	50.0
	Third Most Significant	1	5.3	50.0	100.0
	Total	2	10.5	100.0	
Missing	999	17	89.5		
Total		19	100.0		

a Agency size strata = 999

## DNA not sent to lab applicable--DNA use uncertain(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	2	10.5	11.8	11.8
	No	15	78.9	88.2	100.0
	Total	17	89.5	100.0	
Missing	999	2	10.5		
Total		19	100.0		

a Agency size strata = 999

#### DNA not sent to lab rank--DNA use uncertain(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Third Most Significant	1	5.3	100.0	100.0
Missing	999	18	94.7		
Total		19	100.0		

a Agency size strata = 999

### DNA not sent to lab applicable--lack of \$(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	3	15.8	17.6	17.6
	No	14	73.7	82.4	100.0
	Total	17	89.5	100.0	
Missing	999	2	10.5		
Total		19	100.0		

a Agency size strata = 999

### DNA not sent to lab rank--lack of \$(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	4	21.1	100.0	100.0
Missing	999	15	78.9		
Total		19	100.0		

a Agency size strata = 999

### DNA not sent to lab applicable--inability to get timely results(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	4	21.1	23.5	23.5
	No	13	68.4	76.5	100.0
	Total	17	89.5	100.0	
Missing	999	2	10.5		
Total		19	100.0		

a Agency size strata = 999

### DNA not sent to lab rank--inability to get timely results(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	2	10.5	28.6	28.6
	Second Most Significant	4	21.1	57.1	85.7
	Third Most Significant	1	5.3	14.3	100.0
	Total	7	36.8	100.0	
Missing	999	12	63.2		
Total		19	100.0		

a Agency size strata = 999

### DNA not sent to lab applicable--lab not processing requests(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	No	17	89.5	100.0	100.0
Missing	999	2	10.5		
Total		19	100.0		

a Agency size strata = 999

#### DNA not sent to lab rank--lab not processing requests(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Third Most Significant	1	5.3	100.0	100.0
Missing	999	18	94.7		
Total		19	100.0		

a Agency size strata = 999

### DNA not sent to lab applicable--uncertain where to send DNA(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	1	5.3	5.9	5.9
	No	16	84.2	94.1	100.0
	Total	17	89.5	100.0	
Missing	999	2	10.5		
Total		19	100.0		

a Agency size strata = 999

#### DNA not sent to lab rank--uncertain where to send DNA(a)

		Frequency	Percent
Missing	999	19	100.0

a Agency size strata = 999

#### DNA not sent to lab applicable--analysis not requested by prosecutor(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	No	17	89.5	100.0	100.0
Missing	999	2	10.5		
Total		19	100.0		

a Agency size strata = 999

#### DNA not sent to lab rank--analysis not requested by prosecutor(a)

		Frequency	Percent
Missing	999	19	100.0

a Agency size strata = 999

### DNA not sent to lab applicable--other(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	No	17	89.5	100.0	100.0
Missing	999	2	10.5		
Total		19	100.0		

a Agency size strata = 999

### DNA not sent to lab rank--other(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	2	10.5	100.0	100.0
Missing	999	17	89.5		
Total		19	100.0		

a Agency size strata = 999

### Department routinely collect DNA from property crimes?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	3	15.8	20.0	20.0
	No	12	63.2	80.0	100.0
	Total	15	78.9	100.0	
Missing	999	4	21.1		
Total		19	100.0		

a Agency size strata = 999

### Lab accept AND process unnamed suspect DNA cases for prop. crimes?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	4	21.1	36.4	36.4
	No	7	36.8	63.6	100.0
	Total	11	57.9	100.0	
Missing	999	8	42.1		
Total		19	100.0		

a Agency size strata = 999

### Lab accept but NOT process unnamed suspect DNA cases for prop. crimes?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	2	10.5	20.0	20.0
	No	8	42.1	80.0	100.0
	Total	10	52.6	100.0	
Missing	999	9	47.4		
Total		19	100.0		

a Agency size strata = 999

### unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	No	17	89.5	100.0	100.0
Missing	999	2	10.5		
Total		19	100.0		

a Agency size strata = 999

### agency have cold case squad?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	3	15.8	18.8	18.8
	No	13	68.4	81.3	100.0
	Total	16	84.2	100.0	
Missing	999	3	15.8		
Total		19	100.0		

a Agency size strata = 999

### unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	No	17	89.5	100.0	100.0
Missing	999	2	10.5		
Total		19	100.0		

a Agency size strata = 999

### # of cold cases eligible for review(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	2	1	5.3	50.0	50.0
	6	1	5.3	50.0	100.0
	Total	2	10.5	100.0	
Missing	999	17	89.5		
Total		19	100.0		

a Agency size strata = 999

#	٥f	cold	caene	reviewed	<b>(2</b> )
#	OI	COIG	cases	reviewed	(4)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	1	1	5.3	50.0	50.0
	2	1	5.3	50.0	100.0
	Total	2	10.5	100.0	
Missing	999	17	89.5		
Total		19	100.0		

a Agency size strata = 999

### # of cold cases sent for DNA testing(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	1	5.3	50.0	50.0
	1	1	5.3	50.0	100.0
	Total	2	10.5	100.0	
Missing	999	17	89.5		
Total		19	100.0		

a Agency size strata = 999

### # of additional cold cases potentially eligible for DNA testing(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	1	5.3	33.3	33.3
	1	1	5.3	33.3	66.7
	5	1	5.3	33.3	100.0
	Total	3	15.8	100.0	
Missing	999	16	84.2		
Total		19	100.0		

a Agency size strata = 999

### educated guess is not possible(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	No	18	94.7	100.0	100.0
Missing	999	1	5.3		
Total		19	100.0		

a Agency size strata = 999

### unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	No	18	94.7	100.0	100.0
Missing	999	1	5.3		
Total		19	100.0		

a Agency size strata = 999

### agency have plan to review [cold] cases for biological testing?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	4	21.1	30.8	30.8
	No	9	47.4	69.2	100.0
	Total	13	68.4	100.0	
Missing	999	6	31.6		
Total		19	100.0		

a Agency size strata = 999

### # of cases eligible for review(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	3	1	5.3	33.3	33.3
	77	2	10.5	66.7	100.0
	Total	3	15.8	100.0	
Missing	999	16	84.2		
Total		19	100.0		

a Agency size strata = 999

### # of cases with potential for biological testing(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	1	5.3	33.3	33.3
	65	2	10.5	66.7	100.0
	Total	3	15.8	100.0	
Missing	999	16	84.2		
Total		19	100.0		

a Agency size strata = 999

### educated guess is not possible(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	3	15.8	17.6	17.6
	No	14	73.7	82.4	100.0
	Total	17	89.5	100.0	
Missing	999	2	10.5		
Total		19	100.0		

a Agency size strata = 999

### unable to answer completely(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	No	17	89.5	100.0	100.0
Missing	999	2	10.5		
Total		19	100.0		

a Agency size strata = 999

### agency interested in using DNA database?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	16	84.2	100.0	100.0
Missing	999	3	15.8		
Total		19	100.0		

a Agency size strata = 999

### has agency used DNA database for cases with no direct match?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	4	21.1	26.7	26.7
	No	11	57.9	73.3	100.0
	Total	15	78.9	100.0	
Missing	999	4	21.1		
Total		19	100.0		

a Agency size strata = 999

### Any comments?(a)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	4	21.1	22.2	22.2
	No	14	73.7	77.8	100.0
	Total	18	94.7	100.0	
Missing	999	1	5.3		
Total		19	100.0		

a Agency size strata = 999

# **APPENDIX 3c**

Local Laboratory Response Frequencies by Question

# Appendix 3C - Local Lab Response Frequencies by Question

STATE

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	999	2	3.4	3.4	3.4
	Arizona	4	6.9	6.9	10.3
	California	12	20.7	20.7	31.0
	Florida	2	3.4	3.4	34.5
	Illinois	2	3.4	3.4	37.9
	Indiana	1	1.7	1.7	39.7
	Kansas	2	3.4	3.4	43.1
	Louisiana	3	5.2	5.2	48.3
	Maryland	3	5.2	5.2	53.4
	Massachus etts	1	1.7	1.7	55.2
	Massachus sets	1	1.7	1.7	56.9
	Michigan	1	1.7	1.7	58.6
	Minnesota	1	1.7	1.7	60.3
	Missouri	3	5.2	5.2	65.5
	MIssouri	1	1.7	1.7	67.2
	New Mexico	1	1.7	1.7	69.0
	New York	5	8.6	8.6	77.6
	North Carolina	1	1.7	1.7	79.3
	Ohio	4	6.9	6.9	86.2
	Oklahoma	1	1.7	1.7	87.9
	Pennsylvani a	2	3.4	3.4	91.4
	Texas	5	8.6	8.6	100.0
	Total	58	100.0	100.0	

total rape

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	2	3.4	4.2	4.2
	2	1	1.7	2.1	6.3
	4	1	1.7	2.1	8.3
	8	1	1.7	2.1	10.4
	11	1	1.7	2.1	12.5
	12	2	3.4	4.2	16.7
	13	1	1.7	2.1	18.8
	26	1	1.7	2.1	20.8
	32	1	1.7	2.1	22.9
	35	1	1.7	2.1	25.0
	40	2	3.4	4.2	29.2
	44	1	1.7	2.1	31.2
	54	1	1.7	2.1	33.3
	59	1	1.7	2.1	35.4
	60	1	1.7	2.1	37.5
	69	1	1.7	2.1	39.6
	71	1	1.7	2.1	41.7
	75	3	5.2	6.3	47.9
	79	1	1.7	2.1	50.0
	85	1	1.7	2.1	52.1
	86	1	1.7	2.1	54.2
	88	1	1.7	2.1	56.3
	106	1	1.7	2.1	58.3
	115	1	1.7	2.1	60.4
	167	1	1.7	2.1	62.5
	172	1	1.7	2.1	64.6
	175	1	1.7	2.1	66.7
	183	1	1.7	2.1	68.8
	192	1	1.7	2.1	70.8
	200	1	1.7	2.1	72.9
	245	1	1.7	2.1	75.0
	247	1	1.7	2.1	77.1
	350	1	1.7	2.1	79.2
	514	1	1.7	2.1	81.2
	569	1	1.7	2.1	83.3
	600	1	1.7	2.1	85.4
	626	1	1.7	2.1	87.5
	781	1	1.7	2.1	89.6
	980	1	1.7	2.1	91.7
	1350	1	1.7	2.1	93.7
	1400	2	3.4	4.2	97.9
	3036	1	1.7	2.1	100.0
	Total	48	82.8	100.0	
Missing	999	10	17.2		
Total		58	100.0		

rape screen queue

					Cumulative
		Frequency	Percent	Valid Percent	Percent
Valid	0	5	8.6	12.2	12.2
	2	1	1.7	2.4	14.6
	4	1	1.7	2.4	17.1
	6	1	1.7	2.4	19.5
	8	1	1.7	2.4	22.0
	11	1	1.7	2.4	24.4
	12	1	1.7	2.4	26.8
	15	1	1.7	2.4	29.3
	18	1	1.7	2.4	31.7
	20	2	3.4	4.9	36.6
	22	2	3.4	4.9	41.5
	25	1	1.7	2.4	43.9
	28	1	1.7	2.4	46.3
	30	1	1.7	2.4	48.8
	31	1	1.7	2.4	51.2
	35	1	1.7	2.4	53.7
	39	1	1.7	2.4	56.1
	40	1	1.7	2.4	58.5
	47	2	3.4	4.9	63.4
	50	3	5.2	7.3	70.7
	66	1	1.7	2.4	73.2
	78	1	1.7	2.4	75.6
	80	1	1.7	2.4	78.0
	100	1	1.7	2.4	80.5
	106	1	1.7	2.4	82.9
	111	1	1.7	2.4	85.4
	170	1	1.7	2.4	87.8
	201	1	1.7	2.4	90.2
	402	1	1.7	2.4	92.7
	524	1	1.7	2.4	95.1
	840	1	1.7	2.4	97.6
	1328	1	1.7	2.4	100.0
	Total	41	70.7	100.0	
Missing	999	17	29.3		
Total		58	100.0		

rape analysis queue

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	2	3.4	4.8	4.8
	2	1	1.7	2.4	7.1
	4	3	5.2	7.1	14.3
	5	1	1.7	2.4	16.7
	6	1	1.7	2.4	19.0
	8	1	1.7	2.4	21.4
	10	1	1.7	2.4	23.8
	11	2	3.4	4.8	28.6
	12	2	3.4	4.8	33.3
	15	2	3.4	4.8	38.1
	19	1	1.7	2.4	40.5
	20	1	1.7	2.4	42.9
	21	1	1.7	2.4	45.2
	22	1	1.7	2.4	47.6
	24	1	1.7	2.4	50.0
	35	1	1.7	2.4	52.4
	40	1	1.7	2.4	54.8
	49	1	1.7	2.4	57.1
	50	1	1.7	2.4	59.5
	57	1	1.7	2.4	61.9
	67	1	1.7	2.4	64.3
	70	1	1.7	2.4	66.7
	76	1	1.7	2.4	69.0
	86	1	1.7	2.4	71.4
	100	1	1.7	2.4	73.8
	113	1	1.7	2.4	76.2
	120	1	1.7	2.4	78.6
	131	1	1.7	2.4	81.0
	136	1	1.7	2.4	83.3
	150	1	1.7	2.4	85.7
	153	1	1.7	2.4	88.1
	154	1	1.7	2.4	90.5
	167	1	1.7	2.4	92.9
	200	1	1.7	2.4	95.2
	524	1	1.7	2.4	97.6
	930	1	1.7	2.4	100.0
	Total	42	72.4	100.0	
Missing	999	16	27.6		
Total		58	100.0		

rape report queue

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	9	15.5	31.0	31.0
	1	1	1.7	3.4	34.5
	2	2	3.4	6.9	41.4
	3	2	3.4	6.9	48.3
	4	1	1.7	3.4	51.7
	5	2	3.4	6.9	58.6
	7	1	1.7	3.4	62.1
	8	1	1.7	3.4	65.5
	9	1	1.7	3.4	69.0
	10	1	1.7	3.4	72.4
	12	1	1.7	3.4	75.9
	13	1	1.7	3.4	79.3
	20	1	1.7	3.4	82.8
	25	1	1.7	3.4	86.2
	32	1	1.7	3.4	89.7
	61	1	1.7	3.4	93.1
	95	1	1.7	3.4	96.6
	930	1	1.7	3.4	100.0
	Total	29	50.0	100.0	
Missing	999	29	50.0		
Total		58	100.0		

total rape unnamed suspect

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	3	5.2	7.5	7.5
	1	1	1.7	2.5	10.0
	3	3	5.2	7.5	17.5
	10	1	1.7	2.5	20.0
	12	1	1.7	2.5	22.5
	14	1	1.7	2.5	25.0
	17	1	1.7	2.5	27.5
	19	1	1.7	2.5	30.0
	20	1	1.7	2.5	32.5
	23	1	1.7	2.5	35.0
	26	1	1.7	2.5	37.5
	29	1	1.7	2.5	40.0
	30	1	1.7	2.5	42.5
	32	1	1.7	2.5	45.0
	33	1	1.7	2.5	47.5
	39	1	1.7	2.5	50.0
	43	1	1.7	2.5	52.5
	48	1	1.7	2.5	55.0
	50	2	3.4	5.0	60.0
	60	1	1.7	2.5	62.5
	85	1	1.7	2.5	65.0
	91	1	1.7	2.5	67.5
	100	1	1.7	2.5	70.0
	108	1	1.7	2.5	72.5
	153	1	1.7	2.5	75.0
	200	1	1.7	2.5	77.5
	262	1	1.7	2.5	80.0
	490	1	1.7	2.5	82.5
	560	1	1.7	2.5	85.0
	612	1	1.7	2.5	87.5
	756	1	1.7	2.5	90.0
	781	1	1.7	2.5	92.5
	950	1	1.7	2.5	95.0
	1200	1	1.7	2.5	97.5
	1479	1	1.7	2.5	100.0
	Total	40	69.0	100.0	
Missing	999	18	31.0		
Total		58	100.0		

rape unnamed suspect queue

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	7	12.1	21.2	21.2
	1	1	1.7	3.0	24.2
	2	1	1.7	3.0	27.3
	3	1	1.7	3.0	30.3
	4	1	1.7	3.0	33.3
	7	1	1.7	3.0	36.4
	10	2	3.4	6.1	42.4
	11	1	1.7	3.0	45.5
	15	2	3.4	6.1	51.5
	19	2	3.4	6.1	57.6
	23	1	1.7	3.0	60.6
	25	1	1.7	3.0	63.6
	35	1	1.7	3.0	66.7
	50	1	1.7	3.0	69.7
	66	1	1.7	3.0	72.7
	71	1	1.7	3.0	75.8
	80	1	1.7	3.0	78.8
	100	1	1.7	3.0	81.8
	185	1	1.7	3.0	84.8
	276	1	1.7	3.0	87.9
	465	1	1.7	3.0	90.9
	590	1	1.7	3.0	93.9
	735	1	1.7	3.0	97.0
	756	1	1.7	3.0	100.0
	Total	33	56.9	100.0	
Missing	999	25	43.1		
Total		58	100.0		

rape unnamed suspect analysis queue

		Francis	Darsant	Valid Darsont	Cumulative
Valid	0	Frequency 4	Percent	Valid Percent	Percent
valiu	1	•	6.9	12.5	12.5
	2	1	1.7	3.1	15.6
	5	1	1.7	3.1	18.8
	8	3	5.2	9.4	28.1
		1	1.7	3.1	31.3
	9	2	3.4	6.3	37.5
	10	3	5.2	9.4	46.9
	11	1	1.7	3.1	50.0
	21	1	1.7	3.1	53.1
	23	1	1.7	3.1	56.3
	30	2	3.4	6.3	62.5
	32	1	1.7	3.1	65.6
	42	1	1.7	3.1	68.8
	46	1	1.7	3.1	71.9
	50	2	3.4	6.3	78.1
	71	1	1.7	3.1	81.3
	77	1	1.7	3.1	84.4
	100	1	1.7	3.1	87.5
	120	1	1.7	3.1	90.6
	153	1	1.7	3.1	93.8
	425	1	1.7	3.1	96.9
	631	1	1.7	3.1	100.0
	Total	32	55.2	100.0	
Missing	999	26	44.8		
Total		58	100.0		

# rape unnamed suspect report queue

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	10	17.2	40.0	40.0
	1	2	3.4	8.0	48.0
	2	4	6.9	16.0	64.0
	3	1	1.7	4.0	68.0
	5	2	3.4	8.0	76.0
	15	1	1.7	4.0	80.0
	20	1	1.7	4.0	84.0
	35	1	1.7	4.0	88.0
	261	1	1.7	4.0	92.0
	310	1	1.7	4.0	96.0
	425	1	1.7	4.0	100.0
	Total	25	43.1	100.0	
Missing	999	33	56.9		
Total		58	100.0		

total homicide

					Cumulativa
		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	3	5.2	6.5	6.5
	1	2	3.4	4.3	10.9
	2	4	6.9	8.7	19.6
	3	2	3.4	4.3	23.9
	4	2	3.4	4.3	28.3
	5	1	1.7	2.2	30.4
	6	1	1.7	2.2	32.6
	8	3	5.2	6.5	39.1
	10	1	1.7	2.2	41.3
	12	1	1.7	2.2	43.5
	14	1	1.7	2.2	45.7
	15	1	1.7	2.2	47.8
	20	1	1.7	2.2	50.0
	28	1	1.7	2.2	52.2
	29	1	1.7	2.2	54.3
	30	2	3.4	4.3	58.7
	32	1	1.7	2.2	60.9
	33	1	1.7	2.2	63.0
	35	1	1.7	2.2	65.2
	38	1	1.7	2.2	67.4
	45	1	1.7	2.2	69.6
	53	1	1.7	2.2	71.7
	66	1	1.7	2.2	73.9
	70	1	1.7	2.2	76.1
	80	1	1.7	2.2	78.3
	85	1	1.7	2.2	80.4
	100	1	1.7	2.2	82.6
	120	1	1.7	2.2	84.8
	175	1	1.7	2.2	87.0
	230	1	1.7	2.2	89.1
	256	1	1.7	2.2	91.3
	280	1	1.7	2.2	93.5
	325	1	1.7	2.2	95.7
	650	1	1.7	2.2	97.8
	700	1	1.7	2.2	100.0
	Total	46	79.3	100.0	
Missing	999	12	20.7		
Total		58	100.0		

# homicide screen queue

		Гиализанан	Percent	Valid Percent	Cumulative Percent
Valid	0	Frequency			
Vallu		6	10.3	15.4	15.4
	1	4	6.9	10.3	25.6
	3	2	3.4	5.1	30.8
	4	2	3.4	5.1	35.9
	5	1	1.7	2.6	38.5
	7	1	1.7	2.6	41.0
	8	1	1.7	2.6	43.6
	10	5	8.6	12.8	56.4
	11	1	1.7	2.6	59.0
	13	1	1.7	2.6	61.5
	15	2	3.4	5.1	66.7
	16	1	1.7	2.6	69.2
	20	3	5.2	7.7	76.9
	26	1	1.7	2.6	79.5
	37	1	1.7	2.6	82.1
	42	1	1.7	2.6	84.6
	68	1	1.7	2.6	87.2
	80	1	1.7	2.6	89.7
	184	1	1.7	2.6	92.3
	253	1	1.7	2.6	94.9
	333	1	1.7	2.6	97.4
	628	1	1.7	2.6	100.0
	Total	39	67.2	100.0	
Missing	999	19	32.8		
Total		58	100.0		

# homicide analysis queue

		T			
		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	5	8.6	12.2	12.2
	1	4	6.9	9.8	22.0
	3	2	3.4	4.9	26.8
	4	3	5.2	7.3	34.1
	5	1	1.7	2.4	36.6
	6	1	1.7	2.4	39.0
	7	1	1.7	2.4	41.5
	10	3	5.2	7.3	48.8
	12	1	1.7	2.4	51.2
	13	1	1.7	2.4	53.7
	14	1	1.7	2.4	56.1
	16	1	1.7	2.4	58.5
	17	1	1.7	2.4	61.0
	18	1	1.7	2.4	63.4
	20	2	3.4	4.9	68.3
	28	1	1.7	2.4	70.7
	29	2	3.4	4.9	75.6
	34	1	1.7	2.4	78.0
	37	1	1.7	2.4	80.5
	38	1	1.7	2.4	82.9
	40	1	1.7	2.4	85.4
	50	1	1.7	2.4	87.8
	60	1	1.7	2.4	90.2
	68	1	1.7	2.4	92.7
	72	2	3.4	4.9	97.6
	80	1	1.7	2.4	100.0
	Total	41	70.7	100.0	
Missing	999	17	29.3		
Total		58	100.0		

# homicide report queue

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	9	15.5	29.0	29.0
	1	4	6.9	12.9	41.9
	2	5	8.6	16.1	58.1
	3	2	3.4	6.5	64.5
	5	2	3.4	6.5	71.0
	7	1	1.7	3.2	74.2
	9	1	1.7	3.2	77.4
	13	2	3.4	6.5	83.9
	14	2	3.4	6.5	90.3
	15	1	1.7	3.2	93.5
	32	1	1.7	3.2	96.8
	60	1	1.7	3.2	100.0
	Total	31	53.4	100.0	
Missing	999	27	46.6		
Total		58	100.0		

# total homicide unnamed suspect

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	6	10.3	14.3	14.3
	1	2	3.4	4.8	19.0
	2	4	6.9	9.5	28.6
	3	2	3.4	4.8	33.3
	4	2	3.4	4.8	38.1
	5	2	3.4	4.8	42.9
	6	1	1.7	2.4	45.2
	8	3	5.2	7.1	52.4
	10	1	1.7	2.4	54.8
	11	1	1.7	2.4	57.1
	16	2	3.4	4.8	61.9
	18	2	3.4	4.8	66.7
	25	1	1.7	2.4	69.0
	29	1	1.7	2.4	71.4
	32	1	1.7	2.4	73.8
	41	1	1.7	2.4	76.2
	43	1	1.7	2.4	78.6
	60	2	3.4	4.8	83.3
	70	1	1.7	2.4	85.7
	119	1	1.7	2.4	88.1
	150	1	1.7	2.4	90.5
	178	1	1.7	2.4	92.9
	200	2	3.4	4.8	97.6
	600	1	1.7	2.4	100.0
	Total	42	72.4	100.0	
Missing	999	16	27.6		
Total		58	100.0		

# homicide unnamed suspect screen queue

					Cumulative
		Frequency	Percent	Valid Percent	Percent
Valid	0	7	12.1	21.2	21.2
	1	3	5.2	9.1	30.3
	2	2	3.4	6.1	36.4
	3	2	3.4	6.1	42.4
	4	1	1.7	3.0	45.5
	5	3	5.2	9.1	54.5
	8	1	1.7	3.0	57.6
	10	1	1.7	3.0	60.6
	12	1	1.7	3.0	63.6
	13	2	3.4	6.1	69.7
	14	1	1.7	3.0	72.7
	19	1	1.7	3.0	75.8
	26	1	1.7	3.0	78.8
	35	1	1.7	3.0	81.8
	60	1	1.7	3.0	84.8
	85	1	1.7	3.0	87.9
	100	1	1.7	3.0	90.9
	127	1	1.7	3.0	93.9
	178	1	1.7	3.0	97.0
	180	1	1.7	3.0	100.0
	Total	33	56.9	100.0	
Missing	999	25	43.1		
Total		58	100.0		

# homicide unnamed suspect analysis queue

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	8	13.8	25.8	25.8
	1	2	3.4	6.5	32.3
	2	2	3.4	6.5	38.7
	3	4	6.9	12.9	51.6
	4	1	1.7	3.2	54.8
	6	1	1.7	3.2	58.1
	7	1	1.7	3.2	61.3
	8	2	3.4	6.5	67.7
	9	1	1.7	3.2	71.0
	11	2	3.4	6.5	77.4
	15	1	1.7	3.2	80.6
	16	1	1.7	3.2	83.9
	20	1	1.7	3.2	87.1
	26	1	1.7	3.2	90.3
	30	1	1.7	3.2	93.5
	34	1	1.7	3.2	96.8
	60	1	1.7	3.2	100.0
	Total	31	53.4	100.0	
Missing	999	27	46.6		
Total		58	100.0		

# homicide unnamed suspect report queue

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	15	25.9	62.5	62.5
	1	1	1.7	4.2	66.7
	2	2	3.4	8.3	75.0
	4	1	1.7	4.2	79.2
	5	1	1.7	4.2	83.3
	6	1	1.7	4.2	87.5
	11	1	1.7	4.2	91.7
	14	1	1.7	4.2	95.8
	30	1	1.7	4.2	100.0
	Total	24	41.4	100.0	
Missing	999	34	58.6		
Total		58	100.0		

total property crime

					Cumulative
		Frequency	Percent	Valid Percent	Percent
Valid	0	5	8.6	12.5	12.5
	1	2	3.4	5.0	17.5
	3	2	3.4	5.0	22.5
	4	1	1.7	2.5	25.0
	5	1	1.7	2.5	27.5
	6	1	1.7	2.5	30.0
	7	1	1.7	2.5	32.5
	8	1	1.7	2.5	35.0
	10	2	3.4	5.0	40.0
	15	2	3.4	5.0	45.0
	17	1	1.7	2.5	47.5
	19	1	1.7	2.5	50.0
	20	2	3.4	5.0	55.0
	22	1	1.7	2.5	57.5
	23	1	1.7	2.5	60.0
	30	3	5.2	7.5	67.5
	31	1	1.7	2.5	70.0
	33	1	1.7	2.5	72.5
	35	1	1.7	2.5	75.0
	42	1	1.7	2.5	77.5
	46	1	1.7	2.5	80.0
	50	1	1.7	2.5	82.5
	61	1	1.7	2.5	85.0
	99	1	1.7	2.5	87.5
	100	1	1.7	2.5	90.0
	109	1	1.7	2.5	92.5
	122	1	1.7	2.5	95.0
	418	1	1.7	2.5	97.5
	576	1	1.7	2.5	100.0
	Total	40	69.0	100.0	
Missing	999	18	31.0		
Total		58	100.0		

# property crime screen queue

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	10	17.2	31.3	31.3
	1	1	1.7	3.1	34.4
	2	3	5.2	9.4	43.8
	3	2	3.4	6.3	50.0
	4	2	3.4	6.3	56.3
	5	2	3.4	6.3	62.5
	6	1	1.7	3.1	65.6
	9	1	1.7	3.1	68.8
	10	2	3.4	6.3	75.0
	17	1	1.7	3.1	78.1
	18	1	1.7	3.1	81.3
	19	1	1.7	3.1	84.4
	35	1	1.7	3.1	87.5
	41	1	1.7	3.1	90.6
	66	1	1.7	3.1	93.8
	102	1	1.7	3.1	96.9
	103	1	1.7	3.1	100.0
	Total	32	55.2	100.0	
Missing	999	26	44.8		
Total		58	100.0		

# property crime analysis queue

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	8	13.8	24.2	24.2
Valla	1	3			
	2		5.2	9.1	33.3
		1	1.7	3.0	36.4
	3	2	3.4	6.1	42.4
	4	2	3.4	6.1	48.5
	5	3	5.2	9.1	57.6
	7	1	1.7	3.0	60.6
	10	1	1.7	3.0	63.6
	11	1	1.7	3.0	66.7
	19	1	1.7	3.0	69.7
	20	1	1.7	3.0	72.7
	21	1	1.7	3.0	75.8
	25	3	5.2	9.1	84.8
	29	1	1.7	3.0	87.9
	30	1	1.7	3.0	90.9
	32	1	1.7	3.0	93.9
	33	1	1.7	3.0	97.0
	45	1	1.7	3.0	100.0
	Total	33	56.9	100.0	
Missing	999	25	43.1	100.0	
Total		58	100.0		

### property crime report queue

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	15	25.9	60.0	60.0
	1	3	5.2	12.0	72.0
	2	1	1.7	4.0	76.0
	4	3	5.2	12.0	88.0
	14	1	1.7	4.0	92.0
	15	1	1.7	4.0	96.0
	65	1	1.7	4.0	100.0
	Total	25	43.1	100.0	
Missing	999	33	56.9		
Total		58	100.0		

# total property crime unnamed suspect

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	6	10.3	19.4	19.4
	2	2	3.4	6.5	25.8
	4	1	1.7	3.2	29.0
	5	1	1.7	3.2	32.3
	7	1	1.7	3.2	35.5
	9	1	1.7	3.2	38.7
	10	1	1.7	3.2	41.9
	13	3	5.2	9.7	51.6
	14	1	1.7	3.2	54.8
	17	2	3.4	6.5	61.3
	19	1	1.7	3.2	64.5
	20	1	1.7	3.2	67.7
	25	1	1.7	3.2	71.0
	30	1	1.7	3.2	74.2
	35	1	1.7	3.2	77.4
	48	1	1.7	3.2	80.6
	50	1	1.7	3.2	83.9
	57	1	1.7	3.2	87.1
	78	1	1.7	3.2	90.3
	418	1	1.7	3.2	93.5
	479	1	1.7	3.2	96.8
	637	1	1.7	3.2	100.0
	Total	31	53.4	100.0	
Missing	999	27	46.6		
Total		58	100.0		

# property crime unnamed suspect screen queue

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	9	15.5	36.0	36.0
	1	2	3.4	8.0	44.0
	2	2	3.4	8.0	52.0
	4	2	3.4	8.0	60.0
	5	2	3.4	8.0	68.0
	7	1	1.7	4.0	72.0
	11	1	1.7	4.0	76.0
	15	1	1.7	4.0	80.0
	18	1	1.7	4.0	84.0
	22	1	1.7	4.0	88.0
	48	1	1.7	4.0	92.0
	61	1	1.7	4.0	96.0
	99	1	1.7	4.0	100.0
	Total	25	43.1	100.0	
Missing	999	33	56.9		
Total		58	100.0		

# property crime unnamed suspect analysis queue

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	7	12.1	28.0	28.0
	1	3	5.2	12.0	40.0
	2	4	6.9	16.0	56.0
	7	1	1.7	4.0	60.0
	8	1	1.7	4.0	64.0
	9	2	3.4	8.0	72.0
	12	1	1.7	4.0	76.0
	15	1	1.7	4.0	80.0
	17	1	1.7	4.0	84.0
	21	1	1.7	4.0	88.0
	30	1	1.7	4.0	92.0
	34	1	1.7	4.0	96.0
	430	1	1.7	4.0	100.0
	Total	25	43.1	100.0	
Missing	999	33	56.9		
Total		58	100.0		

# property crime unnamed suspect report queue

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	14	24.1	63.6	63.6
	1	1	1.7	4.5	68.2
	2	3	5.2	13.6	81.8
	6	1	1.7	4.5	86.4
	12	1	1.7	4.5	90.9
	63	1	1.7	4.5	95.5
	108	1	1.7	4.5	100.0
	Total	22	37.9	100.0	
Missing	999	36	62.1		
Total		58	100.0		

### total other offense

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	5	8.6	13.5	13.5
	1	3	5.2	8.1	21.6
	3	3	5.2	8.1	29.7
	5	4	6.9	10.8	40.5
	6	1	1.7	2.7	43.2
	7	2	3.4	5.4	48.6
	8	1	1.7	2.7	51.4
	10	2	3.4	5.4	56.8
	13	1	1.7	2.7	59.5
	14	1	1.7	2.7	62.2
	17	1	1.7	2.7	64.9
	20	1	1.7	2.7	67.6
	27	1	1.7	2.7	70.3
	29	1	1.7	2.7	73.0
	31	1	1.7	2.7	75.7
	44	2	3.4	5.4	81.1
	45	1	1.7	2.7	83.8
	50	1	1.7	2.7	86.5
	66	1	1.7	2.7	89.2
	85	1	1.7	2.7	91.9
	210	1	1.7	2.7	94.6
	396	1	1.7	2.7	97.3
	641	1	1.7	2.7	100.0
	Total	37	63.8	100.0	
Missing	999	21	36.2		
Total		58	100.0		

# other offense screen queue

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	7	12.1	24.1	24.1
	1	2	3.4	6.9	31.0
	2	1	1.7	3.4	34.5
	3	2	3.4	6.9	41.4
	5	3	5.2	10.3	51.7
	6	1	1.7	3.4	55.2
	7	1	1.7	3.4	58.6
	8	3	5.2	10.3	69.0
	10	1	1.7	3.4	72.4
	12	1	1.7	3.4	75.9
	15	1	1.7	3.4	79.3
	21	1	1.7	3.4	82.8
	23	1	1.7	3.4	86.2
	28	1	1.7	3.4	89.7
	44	1	1.7	3.4	93.1
	185	1	1.7	3.4	96.6
	586	1	1.7	3.4	100.0
	Total	29	50.0	100.0	
Missing	999	29	50.0		
Total		58	100.0		

# other offense analysis queue

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	10	17.2	32.3	32.3
	1	2	3.4	6.5	38.7
	2	1	1.7	3.2	41.9
	3	5	8.6	16.1	58.1
	4	2	3.4	6.5	64.5
	5	1	1.7	3.2	67.7
	7	1	1.7	3.2	71.0
	10	1	1.7	3.2	74.2
	13	1	1.7	3.2	77.4
	14	1	1.7	3.2	80.6
	23	1	1.7	3.2	83.9
	25	2	3.4	6.5	90.3
	46	1	1.7	3.2	93.5
	55	1	1.7	3.2	96.8
	66	1	1.7	3.2	100.0
	Total	31	53.4	100.0	
Missing	999	27	46.6		
Total		58	100.0		

# other offense report queue

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	15	25.9	60.0	60.0
	1	4	6.9	16.0	76.0
	3	1	1.7	4.0	80.0
	5	2	3.4	8.0	88.0
	8	1	1.7	4.0	92.0
	11	1	1.7	4.0	96.0
	23	1	1.7	4.0	100.0
	Total	25	43.1	100.0	
Missing	999	33	56.9		
Total		58	100.0		

### total other offense unnamed suspect

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	9	15.5	32.1	32.1
	1	2	3.4	7.1	39.3
	3	3	5.2	10.7	50.0
	7	1	1.7	3.6	53.6
	8	1	1.7	3.6	57.1
	9	1	1.7	3.6	60.7
	10	2	3.4	7.1	67.9
	13	2	3.4	7.1	75.0
	15	1	1.7	3.6	78.6
	29	1	1.7	3.6	82.1
	34	1	1.7	3.6	85.7
	156	1	1.7	3.6	89.3
	296	1	1.7	3.6	92.9
	365	1	1.7	3.6	96.4
	438	1	1.7	3.6	100.0
	Total	28	48.3	100.0	
Missing	999	30	51.7		
Total		58	100.0		

# other offense unnamed suspect screen queue

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	11	19.0	45.8	45.8
	1	1	1.7	4.2	50.0
	2	2	3.4	8.3	58.3
	3	1	1.7	4.2	62.5
	4	1	1.7	4.2	66.7
	5	1	1.7	4.2	70.8
	13	3	5.2	12.5	83.3
	15	1	1.7	4.2	87.5
	35	1	1.7	4.2	91.7
	270	1	1.7	4.2	95.8
	438	1	1.7	4.2	100.0
	Total	24	41.4	100.0	
Missing	999	34	58.6		
Total		58	100.0		

### other offense unnamed suspect analysis queue

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	9	15.5	40.9	40.9
	1	2	3.4	9.1	50.0
	2	1	1.7	4.5	54.5
	3	3	5.2	13.6	68.2
	4	2	3.4	9.1	77.3
	8	1	1.7	4.5	81.8
	14	1	1.7	4.5	86.4
	19	1	1.7	4.5	90.9
	26	1	1.7	4.5	95.5
	66	1	1.7	4.5	100.0
	Total	22	37.9	100.0	
Missing	999	36	62.1		
Total		58	100.0		

### other offense unnamed suspect report queue

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	16	27.6	80.0	80.0
	2	1	1.7	5.0	85.0
	4	1	1.7	5.0	90.0
	5	1	1.7	5.0	95.0
	55	1	1.7	5.0	100.0
	Total	20	34.5	100.0	
Missing	999	38	65.5		
Total		58	100.0		

### percent of rapes with possible DNA collected

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	11-25%	2	3.4	3.6	3.6
	25-50%	9	15.5	16.4	20.0
	50-75%	13	22.4	23.6	43.6
	75-100%	31	53.4	56.4	100.0
	Total	55	94.8	100.0	
Missing	999	3	5.2		
Total		58	100.0		

### percent of homicides with possible DNA collected

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	11-25%	3	5.2	5.5	5.5
	25-50%	10	17.2	18.2	23.6
	50-75%	12	20.7	21.8	45.5
	75-100%	30	51.7	54.5	100.0
	Total	55	94.8	100.0	
Missing	999	3	5.2		
Total		58	100.0		

### educated guess not possible

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	5	8.6	8.6	8.6
	No	53	91.4	91.4	100.0
	Total	58	100.0	100.0	

### percent of rape kits with DNA

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	11-25%	4	6.9	7.3	7.3
	25-50%	21	36.2	38.2	45.5
	50-75%	25	43.1	45.5	90.9
	75-100%	5	8.6	9.1	100.0
	Total	55	94.8	100.0	
Missing	999	3	5.2		
Total		58	100.0		

### percent all cases with DNA

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	11-25%	1	1.7	2.0	2.0
	25-50%	15	25.9	30.0	32.0
	50-75%	21	36.2	42.0	74.0
	75-100%	13	22.4	26.0	100.0
	Total	50	86.2	100.0	
Missing	999	8	13.8		
Total		58	100.0		

### educated guess not possible

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	5	8.6	8.6	8.6
	No	53	91.4	91.4	100.0
	Total	58	100.0	100.0	

### % backlogged rape cases with rape kits only

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	5	8.6	10.0	10.0
	11-25%	15	25.9	30.0	40.0
	25-50%	7	12.1	14.0	54.0
	50-75%	6	10.3	12.0	66.0
	75-100%	17	29.3	34.0	100.0
	Total	50	86.2	100.0	
Missing	999	8	13.8		
Total		58	100.0		

### % backlogged rape cases with rape kits and other DNA evidence

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	7	12.1	13.7	13.7
	11-25%	7	12.1	13.7	27.5
	25-50%	10	17.2	19.6	47.1
	50-75%	18	31.0	35.3	82.4
	75-100%	9	15.5	17.6	100.0
	Total	51	87.9	100.0	
Missing	999	7	12.1		
Total		58	100.0		

# % backlogged rape cases with other DNA evidence only

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	29	50.0	56.9	56.9
	11-25%	16	27.6	31.4	88.2
	25-50%	3	5.2	5.9	94.1
	50-75%	2	3.4	3.9	98.0
	75-100%	1	1.7	2.0	100.0
	Total	51	87.9	100.0	
Missing	999	7	12.1		
Total		58	100.0		

# educated guess not possible

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	6	10.3	10.3	10.3
	No	52	89.7	89.7	100.0
	Total	58	100.0	100.0	

# of rape cases within statue of limitations

					0
		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	1	1.7	3.1	3.1
	0	1	1.7	3.1	6.3
	2	1	1.7	3.1	9.4
	4	1	1.7	3.1	12.5
	8	1	1.7	3.1	15.6
	12	1	1.7	3.1	18.8
	32	1	1.7	3.1	21.9
	35	1	1.7	3.1	25.0
	39	1	1.7	3.1	28.1
	40	1	1.7	3.1	31.3
	54	1	1.7	3.1	34.4
	59	1	1.7	3.1	37.5
	60	1	1.7	3.1	40.6
	69	1	1.7	3.1	43.8
	70	1	1.7	3.1	46.9
	75	3	5.2	9.4	56.3
	88	1	1.7	3.1	59.4
	106	2	3.4	6.3	65.6
	153	1	1.7	3.1	68.8
	175	1	1.7	3.1	71.9
	200	1	1.7	3.1	75.0
	280	1	1.7	3.1	78.1
	350	1	1.7	3.1	81.3
	530	1	1.7	3.1	84.4
	600	1	1.7	3.1	87.5
	612	1	1.7	3.1	90.6
	781	1	1.7	3.1	93.8
	1350	1	1.7	3.1	96.9
	1400	1	1.7	3.1	100.0
	Total	32	55.2	100.0	
Missing	999	26	44.8		
Total		58	100.0		

# of unnamed suspect rape cases within statute of limitations

					Cumulative
		Frequency	Percent	Valid Percent	Percent
Valid	0	2	3.4	7.1	7.1
	1	1	1.7	3.6	10.7
	3	2	3.4	7.1	17.9
	10	1	1.7	3.6	21.4
	11	1	1.7	3.6	25.0
	12	1	1.7	3.6	28.6
	23	1	1.7	3.6	32.1
	29	1	1.7	3.6	35.7
	30	1	1.7	3.6	39.3
	33	1	1.7	3.6	42.9
	35	1	1.7	3.6	46.4
	43	1	1.7	3.6	50.0
	46	1	1.7	3.6	53.6
	50	1	1.7	3.6	57.1
	75	1	1.7	3.6	60.7
	88	1	1.7	3.6	64.3
	100	2	3.4	7.1	71.4
	130	1	1.7	3.6	75.0
	153	1	1.7	3.6	78.6
	200	1	1.7	3.6	82.1
	280	1	1.7	3.6	85.7
	560	1	1.7	3.6	89.3
	612	1	1.7	3.6	92.9
	781	1	1.7	3.6	96.4
	950	1	1.7	3.6	100.0
	Total	28	48.3	100.0	
Missing	999	30	51.7		
Total		58	100.0		

# # of rape cases exceeding statute of limitations by 6/30/03

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	25	43.1	80.6	80.6
	6	1	1.7	3.2	83.9
	50	1	1.7	3.2	87.1
	100	1	1.7	3.2	90.3
	106	1	1.7	3.2	93.5
	160	1	1.7	3.2	96.8
	1100	1	1.7	3.2	100.0
	Total	31	53.4	100.0	
Missing	999	27	46.6		
Total		58	100.0		

## # of unnamed suspect rape cases exceeding statute of limitations by 6/30/03

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	21	36.2	77.8	77.8
	2	1	1.7	3.7	81.5
	3	1	1.7	3.7	85.2
	5	1	1.7	3.7	88.9
	36	1	1.7	3.7	92.6
	100	1	1.7	3.7	96.3
	160	1	1.7	3.7	100.0
	Total	27	46.6	100.0	
Missing	999	31	53.4		
Total		58	100.0		

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	9	15.5	15.5	15.5
	No	49	84.5	84.5	100.0
	Total	58	100.0	100.0	

## anticipated rape case backlog

					Cumulative
		Frequency	Percent	Valid Percent	Percent
Valid	0	3	5.2	6.4	6.4
	5	2	3.4	4.3	10.6
	10	4	6.9	8.5	19.1
	20	1	1.7	2.1	21.3
	30	2	3.4	4.3	25.5
	40	2	3.4	4.3	29.8
	45	2	3.4	4.3	34.0
	50	1	1.7	2.1	36.2
	60	2	3.4	4.3	40.4
	65	2	3.4	4.3	44.7
	70	1	1.7	2.1	46.8
	75	1	1.7	2.1	48.9
	80	2	3.4	4.3	53.2
	95	1	1.7	2.1	55.3
	96	1	1.7	2.1	57.4
	100	2	3.4	4.3	61.7
	150	2	3.4	4.3	66.0
	200	1	1.7	2.1	68.1
	250	1	1.7	2.1	70.2
	260	1	1.7	2.1	72.3
	275	1	1.7	2.1	74.5
	365	1	1.7	2.1	76.6
	400	3	5.2	6.4	83.0
	478	1	1.7	2.1	85.1
	500	2	3.4	4.3	89.4
	644	1	1.7	2.1	91.5
	721	1	1.7	2.1	93.6
	1000	1	1.7	2.1	95.7
	1400	2	3.4	4.3	100.0
	Total	47	81.0	100.0	
Missing	999	11	19.0		
Total		58	100.0		

## anticipated unnamed suspect rape case backlog

					Cumulative
		Frequency	Percent	Valid Percent	Percent
Valid	0	4	6.9	9.8	9.8
	1	2	3.4	4.9	14.6
	2	2	3.4	4.9	19.5
	5	2	3.4	4.9	24.4
	7	1	1.7	2.4	26.8
	10	2	3.4	4.9	31.7
	13	1	1.7	2.4	34.1
	15	1	1.7	2.4	36.6
	20	2	3.4	4.9	41.5
	24	1	1.7	2.4	43.9
	25	2	3.4	4.9	48.8
	30	1	1.7	2.4	51.2
	40	2	3.4	4.9	56.1
	47	1	1.7	2.4	58.5
	50	3	5.2	7.3	65.9
	60	1	1.7	2.4	68.3
	75	1	1.7	2.4	70.7
	85	1	1.7	2.4	73.2
	130	1	1.7	2.4	75.6
	150	2	3.4	4.9	80.5
	180	1	1.7	2.4	82.9
	188	1	1.7	2.4	85.4
	200	1	1.7	2.4	87.8
	400	1	1.7	2.4	90.2
	451	1	1.7	2.4	92.7
	460	1	1.7	2.4	95.1
	656	1	1.7	2.4	97.6
	1200	1	1.7	2.4	100.0
	Total	41	70.7	100.0	
Missing	999	17	29.3		
Total		58	100.0		

## anticipated backlog of all cases

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	1 requericy	1.7	2.3	2.3
	10	2	3.4	4.5	6.8
	20	1	1.7	2.3	9.1
	25	1	1.7	2.3	11.4
	27	1	1.7	2.3	13.6
	40	1	1.7	2.3	15.9
	50	2	3.4	4.5	20.5
	80	1	1.7	2.3	22.7
	90	2	3.4	4.5	27.3
	95	1	1.7	2.3	29.5
	100	2	3.4	4.5	34.1
	125	1	1.7	2.3	36.4
	150	2	3.4	4.5	40.9
	160	1	1.7	2.3	43.2
	180	4	6.9	9.1	52.3
	195	1	1.7	2.3	54.5
	200	1	1.7	2.3	56.8
	225	1	1.7	2.3	59.1
	250	1	1.7	2.3	61.4
	320	1	1.7	2.3	63.6
	375	1	1.7	2.3	65.9
	400	1	1.7	2.3	68.2
	450	1	1.7	2.3	70.5
	600	2	3.4	4.5	75.0
	700	1	1.7	2.3	77.3
	794	1	1.7	2.3	79.5
	807	1	1.7	2.3	81.8
	830	1	1.7	2.3	84.1
	920	1	1.7	2.3	86.4
	1000	1	1.7	2.3	88.6
	1100	1	1.7	2.3	90.9
	1500	2	3.4	4.5	95.5
	2100	1	1.7	2.3	97.7
	3000	1	1.7	2.3	100.0
	Total	44	75.9	100.0	
Missing	999	14	24.1		
Total		58	100.0		

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	10	17.2	17.2	17.2
	No	48	82.8	82.8	100.0
	Total	58	100.0	100.0	

typical anlaysis time for unnamed suspect rape kit - In Weeks

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	.15	1	1.7	2.2	2.2
	1.00	2	3.4	4.4	6.7
	2.00	3	5.2	6.7	13.3
	3.00	1	1.7	2.2	15.6
	4.00	2	3.4	4.4	20.0
	6.00	3	5.2	6.7	26.7
	7.00	2	3.4	4.4	31.1
	8.00	8	13.8	17.8	48.9
	9.00	1	1.7	2.2	51.1
	12.00	1	1.7	2.2	53.3
	13.00	2	3.4	4.4	57.8
	16.00	1	1.7	2.2	60.0
	18.00	2	3.4	4.4	64.4
	21.00	2	3.4	4.4	68.9
	26.00	3	5.2	6.7	75.6
	32.00	1	1.7	2.2	77.8
	39.00	2	3.4	4.4	82.2
	52.00	1	1.7	2.2	84.4
	53.00	1	1.7	2.2	86.7
	75.00	1	1.7	2.2	88.9
	104.00	3	5.2	6.7	95.6
	208.00	2	3.4	4.4	100.0
	Total	45	77.6	100.0	
Missing	999.00	13	22.4		
Total		58	100.0		

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	11	19.0	19.0	19.0
	No	47	81.0	81.0	100.0
	Total	58	100.0	100.0	

## output capacity

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	6	1 requericy	1.7	2.0	2.0
1	7	1	1.7	2.0	3.9
	12	1	1.7	2.0	5.9
	17	1	1.7	2.0	7.8
	20	2	3.4	3.9	11.8
	25	1	1.7	2.0	13.7
	36	2	3.4	3.9	17.6
	39	1	1.7	2.0	17.6
	43	1	1.7	2.0	21.6
	50	1	1.7	2.0	23.5
	60	1	1.7	2.0	25.5
	69	1	1.7	2.0	27.5
	72	1	1.7	2.0	27.5
	80	2	3.4	3.9	33.3
	90	1	3. <del>4</del> 1.7	2.0	35.3 35.3
	100	2	3.4	3.9	39.2
	120	1		2.0	
	125		1.7		41.2
	130	1	1.7	2.0	43.1
	140	1	1.7	2.0	45.1
	180	1	1.7	2.0	47.1
	200	2	3.4	3.9	51.0
	200	2	3.4	3.9	54.9
	255 255	1	1.7	2.0	56.9
	300	1	1.7	2.0	58.8
	320	1	1.7	2.0	60.8
	400	1	1.7	2.0	62.7
	434	1	1.7	2.0	64.7
	454 454	1	1.7	2.0	66.7
	500	1	1.7	2.0	68.6
	600	3	5.2	5.9	74.5
	700	1	1.7	2.0	76.5
	750 750	1	1.7	2.0	78.4
	880	1	1.7	2.0	80.4
	1000	1	1.7	2.0	82.4
	1250	1	1.7	2.0	84.3
		1	1.7	2.0	86.3
	2000	1	1.7	2.0	88.2
	2500	1	1.7	2.0	90.2
	2936	1	1.7	2.0	92.2
	3000	2	3.4	3.9	96.1
	3300	1	1.7	2.0	98.0
	11300 Tatal	1	1.7	2.0	100.0
Minging	Total	51	87.9	100.0	
Missing	999	7	12.1		
Total		58	100.0		

#### month or year

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	month	21	36.2	53.8	53.8
	year	18	31.0	46.2	100.0
	Total	39	67.2	100.0	
Missing	999	19	32.8		
Total		58	100.0		

#### output capacity--samples per case

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	2	3	5.2	6.1	6.1
	3	1	1.7	2.0	8.2
	4	5	8.6	10.2	18.4
	5	20	34.5	40.8	59.2
	5	1	1.7	2.0	61.2
	6	3	5.2	6.1	67.3
	7	1	1.7	2.0	69.4
	8	5	8.6	10.2	79.6
	10	6	10.3	12.2	91.8
	11	2	3.4	4.1	95.9
	20	2	3.4	4.1	100.0
	Total	49	84.5	100.0	
Missing	999	9	15.5		
Total		58	100.0		

## likelihood that agency has rape kits unaccounted for by crime lab?

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	A Certainty	9	15.5	15.8	15.8
	Not Possible	8	13.8	14.0	29.8
	Very Likely	8	13.8	14.0	43.9
	Very Unlikely	13	22.4	22.8	66.7
	Somewhat Likely	11	19.0	19.3	86.0
	Somewhat Unlikely	7	12.1	12.3	98.2
	Unknown	1	1.7	1.8	100.0
	Total	57	98.3	100.0	
Missing	999	1	1.7		
Total		58	100.0		

## major barrier applies--not applicable

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	12	20.7	20.7	20.7
	No	46	79.3	79.3	100.0
	Total	58	100.0	100.0	

#### major barrier rank--not applicable

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	5	8.6	83.3	83.3
	Second Most Significant	1	1.7	16.7	100.0
	Total	6	10.3	100.0	
Missing	999	52	89.7		
Total		58	100.0		

#### major barrier applies--no suspect identified

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	18	31.0	31.0	31.0
	No	40	69.0	69.0	100.0
	Total	58	100.0	100.0	

#### major barrier rank--no suspect identified

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	5	8.6	35.7	35.7
	Second Most Significant	4	6.9	28.6	64.3
	Third Most Significant	5	8.6	35.7	100.0
	Total	14	24.1	100.0	
Missing	999	44	75.9		
Total		58	100.0		

#### major barrier applies--suspect not yet charged

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	13	22.4	22.4	22.4
	No	45	77.6	77.6	100.0
	Total	58	100.0	100.0	

## major barrier rank--suspect not yet charged

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Second Most Significant	4	6.9	50.0	50.0
	Third Most Significant	4	6.9	50.0	100.0
	Total	8	13.8	100.0	
Missing	999	50	86.2		
Total		58	100.0		

## major barrier applies--guilty plea anticipated

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	16	27.6	27.6	27.6
	No	42	72.4	72.4	100.0
	Total	58	100.0	100.0	

#### major barrier rank--guilty plea anticipated

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	2	3.4	22.2	22.2
	Second Most Significant	2	3.4	22.2	44.4
	Third Most Significant	5	8.6	55.6	100.0
	Total	9	15.5	100.0	
Missing	999	49	84.5		
Total		58	100.0		

#### major barrier applies--uncertain DNA use

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	14	24.1	24.1	24.1
	No	44	75.9	75.9	100.0
	Total	58	100.0	100.0	

#### major barrier rank--uncertain DNA use

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	1	1.7	12.5	12.5
	Second Most Significant	2	3.4	25.0	37.5
	Third Most Significant	5	8.6	62.5	100.0
	Total	8	13.8	100.0	
Missing	999	50	86.2		
Total		58	100.0		

#### major barrier applies--lack of DNA funding

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	14	24.1	24.1	24.1
	No	44	75.9	75.9	100.0
	Total	58	100.0	100.0	

#### major barrier rank--lack of DNA funding

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	5	8.6	50.0	50.0
	Second Most Significant	2	3.4	20.0	70.0
	Third Most Significant	3	5.2	30.0	100.0
	Total	10	17.2	100.0	
Missing	999	48	82.8		
Total		58	100.0		

#### major barrier applies--lab backlog

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	22	37.9	37.9	37.9
	No	36	62.1	62.1	100.0
	Total	58	100.0	100.0	

#### major barrier rank--lab backlog

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	7	12.1	36.8	36.8
	Second Most Significant	11	19.0	57.9	94.7
	Third Most Significant	1	1.7	5.3	100.0
	Total	19	32.8	100.0	
Missing	999	39	67.2		
Total		58	100.0		

#### major barrier applies--lab is not processing requests

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	1	1.7	1.7	1.7
	No	57	98.3	98.3	100.0
	Total	58	100.0	100.0	

#### major barrier rank--lab is not processing requests

		Frequency	Percent
Missing	999	58	100.0

#### major barrier applies--uncertain where to send for analysis

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	7	12.1	12.1	12.1
	No	51	87.9	87.9	100.0
	Total	58	100.0	100.0	

#### major barrier rank--uncertain where to send for analysis

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	2	3.4	66.7	66.7
	Third Most Significant	1	1.7	33.3	100.0
	Total	3	5.2	100.0	
Missing	999	55	94.8		
Total		58	100.0		

## major barrier applies--analysis not requested by prosecution

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	17	29.3	29.3	29.3
	No	41	70.7	70.7	100.0
	Total	58	100.0	100.0	

#### major barrier rank--analysis not requested by prosecution

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	5	8.6	35.7	35.7
	Second Most Significant	4	6.9	28.6	64.3
	Third Most Significant	5	8.6	35.7	100.0
	Total	14	24.1	100.0	
Missing	999	44	75.9		
Total		58	100.0		

#### major barrier applies--other

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	14	24.1	24.1	24.1
	No	44	75.9	75.9	100.0
	Total	58	100.0	100.0	

#### major barrier rank--analysis not requested by prosecution

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	12	20.7	92.3	92.3
	Second Most Significant	1	1.7	7.7	100.0
	Total	13	22.4	100.0	
Missing	999	45	77.6		
Total		58	100.0		

#### major barrier applies--other (explanation)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	1	9	15.5	15.5	15.5
	2	1	1.7	1.7	17.2
	999	48	82.8	82.8	100.0
	Total	58	100.0	100.0	

## educated guess not possible

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	7	12.1	12.1	12.1
	No	49	84.5	84.5	96.6
	999	2	3.4	3.4	100.0
	Total	58	100.0	100.0	

## named suspect cases--# of included suspects

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	.60	2	3.4	5.3	5.3
	.70	3	5.2	7.9	13.2
	.75	4	6.9	10.5	23.7
	.78	2	3.4	5.3	28.9
	.80	4	6.9	10.5	39.5
	.90	5	8.6	13.2	52.6
	3.00	1	1.7	2.6	55.3
	4.00	1	1.7	2.6	57.9
	12.00	1	1.7	2.6	60.5
	25.00	1	1.7	2.6	63.2
	29.00	1	1.7	2.6	65.8
	35.00	1	1.7	2.6	68.4
	55.00	2	3.4	5.3	73.7
	65.00	1	1.7	2.6	76.3
	70.00	1	1.7	2.6	78.9
	74.00	1	1.7	2.6	81.6
	75.00	1	1.7	2.6	84.2
	80.00	1	1.7	2.6	86.8
	100.00	1	1.7	2.6	89.5
	122.00	1	1.7	2.6	92.1
	140.00	1	1.7	2.6	94.7
	152.00	1	1.7	2.6	97.4
	375.00	1	1.7	2.6	100.0
	Total	38	65.5	100.0	
Missing	999.00	20	34.5		
Total	_	58	100.0		

## named suspect cases--# of excluded suspects

				ı	
					Cumulative
		Frequency	Percent	Valid Percent	Percent
Valid	.10	6	10.3	15.8	15.8
	.15	1	1.7	2.6	18.4
	.20	3	5.2	7.9	26.3
	.22	2	3.4	5.3	31.6
	.25	3	5.2	7.9	39.5
	.30	2	3.4	5.3	44.7
	.40	2	3.4	5.3	50.0
	2.00	2	3.4	5.3	55.3
	5.00	1	1.7	2.6	57.9
	7.00	1	1.7	2.6	60.5
	10.00	1	1.7	2.6	63.2
	12.00	1	1.7	2.6	65.8
	14.00	1	1.7	2.6	68.4
	16.00	1	1.7	2.6	71.1
	20.00	2	3.4	5.3	76.3
	25.00	1	1.7	2.6	78.9
	29.00	1	1.7	2.6	81.6
	41.00	1	1.7	2.6	84.2
	45.00	1	1.7	2.6	86.8
	60.00	1	1.7	2.6	89.5
	75.00	1	1.7	2.6	92.1
	87.00	1	1.7	2.6	94.7
	107.00	1	1.7	2.6	97.4
	748.00	1	1.7	2.6	100.0
	Total	38	65.5	100.0	
Missing	999.00	20	34.5		
Total		58	100.0		

#### educated guess not possible

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	16	27.6	27.6	27.6
	No	42	72.4	72.4	100.0
	Total	58	100.0	100.0	

## law enforcement routinely collect DNA from property crimes?

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	40	69.0	69.0	69.0
	No	18	31.0	31.0	100.0
	Total	58	100.0	100.0	

#### lab accepts and processes DNA from unnamed suspect property crimes?

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	40	69.0	70.2	70.2
	No	17	29.3	29.8	100.0
	Total	57	98.3	100.0	
Missing	999	1	1.7		
Total		58	100.0		

#### lab accepts but does NOT process DNA from unnamed suspect property crimes?

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	18	31.0	36.0	36.0
	No	32	55.2	64.0	100.0
	Total	50	86.2	100.0	
Missing	999	8	13.8		
Total		58	100.0		

#### prop. crime barrier applies--lack of storage space

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	3	5.2	5.2	5.2
	No	55	94.8	94.8	100.0
	Total	58	100.0	100.0	

#### prop. crime barrier rank--lack of storage space

		Frequency	Percent
Missing	999	58	100.0

#### prop. crime barier applies--more personnel needed

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	20	34.5	34.5	34.5
	No	38	65.5	65.5	100.0
	Total	58	100.0	100.0	

#### prop. crime barrier rank--more personnel needed

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	7	12.1	41.2	41.2
	Second Most Significant	8	13.8	47.1	88.2
Third	Third Most Significant	2	3.4	11.8	100.0
	Total	17	29.3	100.0	
Missing	999	41	70.7		
Total		58	100.0		

#### prop. crime barrier applies--backlog/other priority cases

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	23	39.7	39.7	39.7
	No	35	60.3	60.3	100.0
	Total	58	100.0	100.0	

#### prop. crime barrier rank--backlog/other priority cases

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	10	17.2	52.6	52.6
	Second Most Significant	5	8.6	26.3	78.9
Third Most Significant		4	6.9	21.1	100.0
	Total	19	32.8	100.0	
Missing	999	39	67.2		
Total		58	100.0		

#### prop. crime barrier applies--equipment needed

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	7	12.1	12.1	12.1
	No	51	87.9	87.9	100.0
	Total	58	100.0	100.0	

prop. crime barrier rank--equipment needed

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Second Most Significant	1	1.7	33.3	33.3
Third Mos	Third Most Significant	2	3.4	66.7	100.0
	Total	3	5.2	100.0	
Missing	999	55	94.8		
Total		58	100.0		

#### prop. crime barrier applies--lack of operational space

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	6	10.3	10.3	10.3
	No	52	89.7	89.7	100.0
	Total	58	100.0	100.0	

#### prop. crime barrier rank--lack of operational space

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	2	3.4	40.0	40.0
	Second Most Significant	2	3.4	40.0	80.0
	Third Most Significant	1	1.7	20.0	100.0
	Total	5	8.6	100.0	
Missing	999	53	91.4		
Total		58	100.0		

#### prop. crime barrier applies--prosecution must request

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	5	8.6	8.6	8.6
	No	53	91.4	91.4	100.0
	Total	58	100.0	100.0	

#### prop. crime barrier rank--prosecution must request

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Second Most Significant	2	3.4	66.7	66.7
	Third Most Significant	1	1.7	33.3	100.0
	Total	3	5.2	100.0	
Missing	999	55	94.8		
Total		58	100.0		

#### prop. crime barrier applies--investigator must request

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	10	17.2	17.2	17.2
	No	48	82.8	82.8	100.0
	Total	58	100.0	100.0	

#### prop. crime barrier rank--investigator must request

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	1	1.7	11.1	11.1
	Second Most Significant	2	3.4	22.2	33.3
	Third Most Significant	6	10.3	66.7	100.0
	Total	9	15.5	100.0	
Missing	999	49	84.5		
Total		58	100.0		

#### prop. crime barrier applies--other

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	4	6.9	6.9	6.9
	No	54	93.1	93.1	100.0
	Total	58	100.0	100.0	

#### prop. crime barrier rank--investigator must request

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	1	1.7	50.0	50.0
	Third Most Significant	1	1.7	50.0	100.0
	Total	2	3.4	100.0	
Missing	999	56	96.6		
Total		58	100.0		

prop. crime barrier applies--other (explanation)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	1	2	3.4	3.4	3.4
	999	56	96.6	96.6	100.0
	Total	58	100.0	100.0	

## total cases benefiting from mitochondrial DNA testing (but not STR)

		1			
					Cumulative
		Frequency	Percent	Valid Percent	Percent
Valid	0	2	3.4	5.1	5.1
	1	2	3.4	5.1	10.3
	2	4	6.9	10.3	20.5
	3	1	1.7	2.6	23.1
	5	9	15.5	23.1	46.2
	6	2	3.4	5.1	51.3
	7	1	1.7	2.6	53.8
	8	1	1.7	2.6	56.4
	10	8	13.8	20.5	76.9
	12	3	5.2	7.7	84.6
	15	1	1.7	2.6	87.2
	20	1	1.7	2.6	89.7
	25	4	6.9	10.3	100.0
	Total	39	67.2	100.0	
Missing	999	19	32.8		
Total		58	100.0		

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	12	20.7	20.7	20.7
	No	46	79.3	79.3	100.0
	Total	58	100.0	100.0	

## hair analysis cases that could benefit from mitochondrial DNA testing

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	2	3.4	5.6	5.6
	1	2	3.4	5.6	11.1
	2	3	5.2	8.3	19.4
	4	1	1.7	2.8	22.2
	5	10	17.2	27.8	50.0
	10	6	10.3	16.7	66.7
	12	2	3.4	5.6	72.2
	14	1	1.7	2.8	75.0
	15	4	6.9	11.1	86.1
	20	1	1.7	2.8	88.9
	25	1	1.7	2.8	91.7
	30	1	1.7	2.8	94.4
	125	1	1.7	2.8	97.2
	200	1	1.7	2.8	100.0
	Total	36	62.1	100.0	
Missing	999	22	37.9		
Total		58	100.0		

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	13	22.4	22.4	22.4
	No	45	77.6	77.6	100.0
	Total	58	100.0	100.0	

## average cost of rape kit analysis

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	30	1	1.7	2.5	2.5
	35	1	1.7	2.5	5.0
	60	1	1.7	2.5	7.5
	150	4	6.9	10.0	17.5
	300	2	3.4	5.0	22.5
	400	1	1.7	2.5	25.0
	460	1	1.7	2.5	27.5
	500	2	3.4	5.0	32.5
	552	1	1.7	2.5	35.0
	600	3	5.2	7.5	42.5
	700	1	1.7	2.5	45.0
	760	1	1.7	2.5	47.5
	780	1	1.7	2.5	50.0
	810	1	1.7	2.5	52.5
	850	1	1.7	2.5	55.0
	930	1	1.7	2.5	57.5
	975	1	1.7	2.5	60.0
	1000	1	1.7	2.5	62.5
	1200	2	3.4	5.0	67.5
	1300	1	1.7	2.5	70.0
	1400	1	1.7	2.5	72.5
	1500	3	5.2	7.5	80.0
	2000	3	5.2	7.5	87.5
	2700	1	1.7	2.5	90.0
	3100	1	1.7	2.5	92.5
	3116	1	1.7	2.5	95.0
	4300	1	1.7	2.5	97.5
	5000	1	1.7	2.5	100.0
	Total	40	69.0	100.0	
Missing	999	18	31.0		
Total		58	100.0		

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	20	34.5	34.5	34.5
	No	38	65.5	65.5	100.0
	Total	58	100.0	100.0	

#### cost factor--reagents

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	44	75.9	75.9	75.9
	No	14	24.1	24.1	100.0
	Total	58	100.0	100.0	

#### cost factor--equipment

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	20	34.5	34.5	34.5
	No	38	65.5	65.5	100.0
	Total	58	100.0	100.0	

#### cost factor--salaries

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	38	65.5	65.5	65.5
	No	20	34.5	34.5	100.0
	Total	58	100.0	100.0	

#### cost factor--overhead

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	23	39.7	39.7	39.7
	No	35	60.3	60.3	100.0
	Total	58	100.0	100.0	

#### cost factor--other

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	9	15.5	15.5	15.5
	No	49	84.5	84.5	100.0
	Total	58	100.0	100.0	

#### cost factor--other description

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	1	8	13.8	13.8	13.8
	2	37	63.8	63.8	77.6
	999	13	22.4	22.4	100.0
	Total	58	100.0	100.0	

% of budget from local funding?

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	2	3.4	3.9	3.9
	11-25%	2	3.4	3.9	7.8
	25-50%	6	10.3	11.8	19.6
	50-75%	12	20.7	23.5	43.1
	75-100%	29	50.0	56.9	100.0
	Total	51	87.9	100.0	
Missing	999	7	12.1		
Total		58	100.0		

## % of budget from state funding?

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	23	39.7	51.1	51.1
	11-25%	9	15.5	20.0	71.1
	25-50%	8	13.8	17.8	88.9
	50-75%	5	8.6	11.1	100.0
	Total	45	77.6	100.0	
Missing	999	13	22.4		
Total		58	100.0		

## % of budget from federal funding?

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	21	36.2	47.7	47.7
	11-25%	14	24.1	31.8	79.5
	25-50%	7	12.1	15.9	95.5
	50-75%	2	3.4	4.5	100.0
	Total	44	75.9	100.0	
Missing	999	14	24.1		
Total		58	100.0		

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	8	13.8	13.8	13.8
	No	50	86.2	86.2	100.0
	Total	58	100.0	100.0	

## supplemental funding exists?

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	yes	53	91.4	93.0	93.0
	no	4	6.9	7.0	100.0
	Total	57	98.3	100.0	
Missing	999	1	1.7		
Total		58	100.0		

#### use \$ applies--salaries

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	37	63.8	63.8	63.8
	No	21	36.2	36.2	100.0
	Total	58	100.0	100.0	

#### use \$ rank--salaries

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	20	34.5	76.9	76.9
	Second Most Significant	2	3.4	7.7	84.6
	Third Most Significant	4	6.9	15.4	100.0
	Total	26	44.8	100.0	
Missing	999	32	55.2		
Total		58	100.0		

#### use \$ applies--overtime

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	27	46.6	46.6	46.6
	No	31	53.4	53.4	100.0
	Total	58	100.0	100.0	

#### use \$ rank--overtime

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	3	5.2	21.4	21.4
	Second Most Significant	5	8.6	35.7	57.1
	Third Most Significant	6	10.3	42.9	100.0
	Total	14	24.1	100.0	
Missing	999	44	75.9		
Total		58	100.0		

#### use \$ applies--training

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	34	58.6	58.6	58.6
	No	24	41.4	41.4	100.0
	Total	58	100.0	100.0	

#### use \$ rank--training

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	2	3.4	12.5	12.5
	Second Most Significant	9	15.5	56.3	68.8
	Third Most Significant	5	8.6	31.3	100.0
	Total	16	27.6	100.0	
Missing	999	42	72.4		
Total		58	100.0		

#### use \$ applies--equipment

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	35	60.3	60.3	60.3
	No	23	39.7	39.7	100.0
	Total	58	100.0	100.0	

use \$ rank--equipment

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	4	6.9	20.0	20.0
	Second Most Significant	4	6.9	20.0	40.0
	Third Most Significant	12	20.7	60.0	100.0
	Total	20	34.5	100.0	
Missing	999	38	65.5		
Total		58	100.0		

#### use \$ applies--robotics

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	18	31.0	31.0	31.0
	No	40	69.0	69.0	100.0
	Total	58	100.0	100.0	

#### use \$ rank--robotics

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	2	3.4	18.2	18.2
	Second Most Significant	8	13.8	72.7	90.9
	Third Most Significant	1	1.7	9.1	100.0
	Total	11	19.0	100.0	
Missing	999	47	81.0		
Total		58	100.0		

## use \$ applies--reagents

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	39	67.2	67.2	67.2
	No	19	32.8	32.8	100.0
	Total	58	100.0	100.0	

#### use \$ rank--reagents

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	8	13.8	29.6	29.6
	Second Most Significant	11	19.0	40.7	70.4
٦	Third Most Significant	8	13.8	29.6	100.0
	Total	27	46.6	100.0	
Missing	999	31	53.4		
Total		58	100.0		

#### use \$ applies--construction/lease

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	17	29.3	29.3	29.3
	No	41	70.7	70.7	100.0
	Total	58	100.0	100.0	

#### use \$ rank--construction/lease

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	4	6.9	40.0	40.0
	Second Most Significant	2	3.4	20.0	60.0
	Third Most Significant	4	6.9	40.0	100.0
	Total	10	17.2	100.0	
Missing	999	48	82.8		
Total		58	100.0		

#### use \$ applies--offender collection

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	6	10.3	10.3	10.3
	No	52	89.7	89.7	100.0
	Total	58	100.0	100.0	

#### use \$ rank--offender collection

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	3	5.2	100.0	100.0
Missing	999	55	94.8		
Total		58	100.0		

use \$ applies--suspect case backlog

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	30	51.7	51.7	51.7
	No	28	48.3	48.3	100.0
	Total	58	100.0	100.0	

#### use \$ rank--suspect case backlog

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	2	3.4	20.0	20.0
	Second Most Significant	7	12.1	70.0	90.0
	Third Most Significant	1	1.7	10.0	100.0
	Total	10	17.2	100.0	
Missing	999	48	82.8		
Total		58	100.0		

#### use \$ applies--no suspect case backlog

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	32	55.2	55.2	55.2
	No	26	44.8	44.8	100.0
	Total	58	100.0	100.0	

#### use \$ rank--no suspect case backlog

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	5	8.6	33.3	33.3
	Second Most Significant	3	5.2	20.0	53.3
	Third Most Significant	7	12.1	46.7	100.0
	Total	15	25.9	100.0	
Missing	999	43	74.1		
Total		58	100.0		

#### use \$ applies--outsourcing

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	17	29.3	29.3	29.3
	No	41	70.7	70.7	100.0
	Total	58	100.0	100.0	

#### use \$ rank--outsourcing

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Second Most Significant	3	5.2	60.0	60.0
	Third Most Significant	2	3.4	40.0	100.0
	Total	5	8.6	100.0	
Missing	999	53	91.4		
Total		58	100.0		

## use \$ applies--other

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	4	6.9	6.9	6.9
	No	54	93.1	93.1	100.0
	Total	58	100.0	100.0	

#### use \$ rank--outsourcing

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	3	5.2	75.0	75.0
	Third Most Significant	1	1.7	25.0	100.0
	Total	4	6.9	100.0	
Missing	999	54	93.1		
Total		58	100.0		

## use \$ applies--other (description)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	1	1	1.7	1.7	1.7
	2	41	70.7	70.7	72.4
	999	16	27.6	27.6	100.0
	Total	58	100.0	100.0	

#### Comments?

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	13	22.4	22.4	22.4
	No	45	77.6	77.6	100.0
	Total	58	100.0	100.0	

# **APPENDIX 3d**

State Laboratory Response Frequencies by Question

# Appendix 3D – State Laboratory Response Frequencies by Question STATE

	Frequency	Percent	Valid Percent	Cumulative Percent
Valid AK	1	2.3	2.3	2.3
AL	1	2.3	2.3	4.5
AR	1	2.3	2.3	6.8
AZ	1	2.3	2.3	9.1
CA	1	2.3	2.3	11.4
CO	1	2.3	2.3	13.6
CT	1	2.3	2.3	15.9
DE	1	2.3	2.3	18.2
FL	1	2.3	2.3	20.5
IA	1	2.3	2.3	22.7
ID	1	2.3	2.3	25.0
IL	1	2.3	2.3	27.3
IN	1	2.3	2.3	29.5
KS	1	2.3	2.3	31.8
KY	2	4.5	4.5	36.4
LA	1	2.3	2.3	38.6
MA	1	2.3	2.3	40.9
ME	1	2.3	2.3	43.2
MI	1	2.3	2.3	45.5
MN	1	2.3	2.3	47.7
MO	1	2.3	2.3	50.0
MT	1	2.3	2.3	52.3
NC	1	2.3	2.3	54.5
ND	1	2.3	2.3	56.8
NE	1	2.3	2.3	59.1
NH	1	2.3	2.3	61.4
NJ	1	2.3	2.3	63.6
NM	1	2.3	2.3	65.9
NY	1	2.3	2.3	68.2
ОН	1	2.3	2.3	70.5
OK	1	2.3	2.3	72.7
OR	1	2.3	2.3	75.0
PA	1	2.3	2.3	77.3
SC	1	2.3	2.3	79.5
SD	1	2.3	2.3	81.8
TN	1	2.3	2.3	84.1
TX	1	2.3	2.3	86.4
UT	2	4.5	4.5	90.9
VA	1	2.3	2.3	93.2
WA	1	2.3	2.3	95.5
WV	1	2.3	2.3	97.7
WY	1	2.3	2.3	100.0
Total	44	100.0	100.0	

**Total Number of Forensic Cases** 

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	1	2.3	2.4	2.4
	3	1	2.3	2.4	4.9
	4	1	2.3	2.4	7.3
	5	1	2.3	2.4	9.8
	22	1	2.3	2.4	12.2
	34	1	2.3	2.4	14.6
	49	1	2.3	2.4	17.1
	84	1	2.3	2.4	19.5
	100	1	2.3	2.4	22.0
	144	1	2.3	2.4	24.4
	168	1	2.3	2.4	26.8
	182	1	2.3	2.4	29.3
	200	1	2.3	2.4	31.7
	208	1	2.3	2.4	34.1
	213	1	2.3	2.4	36.6
	310	2	4.5	4.9	41.5
	312	1	2.3	2.4	43.9
	405	1	2.3	2.4	46.3
	469	1	2.3	2.4	48.8
	550	1	2.3	2.4	51.2
	560	1	2.3	2.4	53.7
	683	1	2.3	2.4	56.1
	780	1	2.3	2.4	58.5
	900	2	4.5	4.9	63.4
	936	1	2.3	2.4	65.9
	956	1	2.3	2.4	68.3
	1003	1	2.3	2.4	70.7
	1286	1	2.3	2.4	73.2
	1300	1	2.3	2.4	75.6
	1644	1	2.3	2.4	78.0
	1898	1	2.3	2.4	80.5
	2394	1	2.3	2.4	82.9
	2950	1	2.3	2.4	85.4
	3000	1	2.3	2.4	87.8
	4154	1	2.3	2.4	90.2
	4800	1	2.3	2.4	92.7
	5279	1	2.3	2.4	95.1
	7014	1	2.3	2.4	97.6
	10685	1	2.3	2.4	100.0
	Total	41	93.2	100.0	
Missing	999	3	6.8		
Total		44	100.0		

## Number of Homicides

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	3	3	6.8	21.4	21.4
	5	1	2.3	7.1	28.6
	6	1	2.3	7.1	35.7
	10	1	2.3	7.1	42.9
	78	1	2.3	7.1	50.0
	97	1	2.3	7.1	57.1
	193	1	2.3	7.1	64.3
	608	1	2.3	7.1	71.4
	750	1	2.3	7.1	78.6
	900	1	2.3	7.1	85.7
	1250	1	2.3	7.1	92.9
	2729	1	2.3	7.1	100.0
	Total	14	31.8	100.0	
Missing	999	30	68.2		
Total		44	100.0		

## **Number of Rapes**

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	1	1	2.3	7.7	7.7
	2	1	2.3	7.7	15.4
	8	1	2.3	7.7	23.1
	142	1	2.3	7.7	30.8
	186	1	2.3	7.7	38.5
	371	1	2.3	7.7	46.2
	385	1	2.3	7.7	53.8
	900	1	2.3	7.7	61.5
	1271	1	2.3	7.7	69.2
	1950	1	2.3	7.7	76.9
	2000	1	2.3	7.7	84.6
	2649	1	2.3	7.7	92.3
	5583	1	2.3	7.7	100.0
	Total	13	29.5	100.0	
Missing	999	31	70.5		
Total		44	100.0		

## **Number of Property Crimes**

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	2	4.5	16.7	16.7
	3	1	2.3	8.3	25.0
	6	1	2.3	8.3	33.3
	7	1	2.3	8.3	41.7
	10	1	2.3	8.3	50.0
	19	1	2.3	8.3	58.3
	64	1	2.3	8.3	66.7
	150	1	2.3	8.3	75.0
	238	1	2.3	8.3	83.3
	340	1	2.3	8.3	91.7
	1725	1	2.3	8.3	100.0
	Total	12	27.3	100.0	
Missing	999	32	72.7		
Total		44	100.0		

## Other Forensic Cases

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	1	2.3	9.1	9.1
	1	1	2.3	9.1	18.2
	5	1	2.3	9.1	27.3
	6	1	2.3	9.1	36.4
	43	1	2.3	9.1	45.5
	50	1	2.3	9.1	54.5
	118	1	2.3	9.1	63.6
	129	1	2.3	9.1	72.7
	150	1	2.3	9.1	81.8
	648	1	2.3	9.1	90.9
	2877	1	2.3	9.1	100.0
	Total	11	25.0	100.0	
Missing	999	33	75.0		
Total		44	100.0		

total rape

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	1	2.3	2.7	2.7
	5	1	2.3	2.7	5.4
	10	1	2.3	2.7	8.1
	15	1	2.3	2.7	10.8
	20	1	2.3	2.7	13.5
	26	1	2.3	2.7	16.2
	29	1	2.3	2.7	18.9
	32	1	2.3	2.7	21.6
	37	1	2.3	2.7	24.3
	57	1	2.3	2.7	27.0
	65	1	2.3	2.7	29.7
	75	1	2.3	2.7	32.4
	101	1	2.3	2.7	35.1
	105	1	2.3	2.7	37.8
	119	1	2.3	2.7	40.5
	143	1	2.3	2.7	43.2
	165	1	2.3	2.7	45.9
	187	1	2.3	2.7	48.6
	202	1	2.3	2.7	51.4
	203	1	2.3	2.7	54.1
	205	1	2.3	2.7	56.8
	220	1	2.3	2.7	59.5
	241	1	2.3	2.7	62.2
	317	1	2.3	2.7	64.9
	336	1	2.3	2.7	67.6
	369	1	2.3	2.7	70.3
	382	1	2.3	2.7	73.0
	385	1	2.3	2.7	75.7
	436	1	2.3	2.7	78.4
	510	1	2.3	2.7	81.1
	713	1	2.3	2.7	83.8
	747	1	2.3	2.7	86.5
	872	1	2.3	2.7	89.2
	1309	1	2.3	2.7	91.9
	2200	1	2.3	2.7	94.6
	4449	1	2.3	2.7	97.3
	4665	1	2.3	2.7	100.0
	Total	37	84.1	100.0	
Missing	999	7	15.9		
Total		44	100.0		

rape screen queue

		Глажизанан	Downount	Valid Dansant	Cumulative
Valid	8	Frequency	Percent	Valid Percent	Percent
Vallu	10	1	2.3	3.6	3.6
		1	2.3	3.6	7.1
	15	1	2.3	3.6	10.7
	16	1	2.3	3.6	14.3
	22	1	2.3	3.6	17.9
	25	3	6.8	10.7	28.6
	36	1	2.3	3.6	32.1
	51	1	2.3	3.6	35.7
	52	1	2.3	3.6	39.3
	55	1	2.3	3.6	42.9
	94	1	2.3	3.6	46.4
	110	1	2.3	3.6	50.0
	147	1	2.3	3.6	53.6
	150	2	4.5	7.1	60.7
	164	1	2.3	3.6	64.3
	182	1	2.3	3.6	67.9
	200	1	2.3	3.6	71.4
	237	1	2.3	3.6	75.0
	241	1	2.3	3.6	78.6
	479	1	2.3	3.6	82.1
	523	1	2.3	3.6	85.7
	547	1	2.3	3.6	89.3
	600	1		3.6	92.9
	683	1	2.3	3.6	96.4
	3681	1			100.0
	Total				
Missing	999				
Total		44			
Missing Total	600 683 3681	1 1 1 28 16	2.3 2.3 2.3 2.3 63.6 36.4 100.0		92.9 96.4

rape analysis queue

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	1	2.3	3.7	3.7
	1	1	2.3	3.7	7.4
	5	1	2.3	3.7	11.1
	10	1	2.3	3.7	14.8
	12	1	2.3	3.7	18.5
	14	1	2.3	3.7	22.2
	25	1	2.3	3.7	25.9
	29	1	2.3	3.7	29.6
	39	1	2.3	3.7	33.3
	52	1	2.3	3.7	37.0
	55	1	2.3	3.7	40.7
	71	1	2.3	3.7	44.4
	77	1	2.3	3.7	48.1
	101	1	2.3	3.7	51.9
	113	1	2.3	3.7	55.6
	119	1	2.3	3.7	59.3
	121	1	2.3	3.7	63.0
	131	2	4.5	7.4	70.4
	152	1	2.3	3.7	74.1
	199	1	2.3	3.7	77.8
	200	1	2.3	3.7	81.5
	262	1	2.3	3.7	85.2
	300	1	2.3	3.7	88.9
	400	1	2.3	3.7	92.6
	626	1	2.3	3.7	96.3
	768	1	2.3	3.7	100.0
	Total	27	61.4	100.0	
Missing	999	17	38.6		
Total		44	100.0		

### rape report queue

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	7	15.9	46.7	46.7
	7	1	2.3	6.7	53.3
	10	2	4.5	13.3	66.7
	16	2	4.5	13.3	80.0
	25	1	2.3	6.7	86.7
	55	1	2.3	6.7	93.3
	87	1	2.3	6.7	100.0
	Total	15	34.1	100.0	
Missing	999	29	65.9		
Total		44	100.0		

# total rape unnamed suspect

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	1 requericy	2.3	4.2	4.2
	5	2	4.5	8.3	12.5
	11	1	2.3	4.2	16.7
	15	2	4.5	8.3	25.0
	30	1	2.3	4.2	29.2
	40	1	2.3	4.2	33.3
	42	1	2.3	4.2	37.5
	50	1	2.3	4.2	41.7
	60	2	4.5	8.3	50.0
	61	1	2.3	4.2	54.2
	74	1	2.3	4.2	58.3
	122	1	2.3	4.2	62.5
	124	1	2.3	4.2	66.7
	148	1	2.3	4.2	70.8
	262	1	2.3	4.2	75.0
	300	1	2.3	4.2	79.2
	330	1	2.3	4.2	83.3
	350	1	2.3	4.2	87.5
	382	1	2.3	4.2	91.7
	889	1	2.3	4.2	95.8
	1518	1	2.3	4.2	100.0
	Total	24	54.5	100.0	
Missing	999	20	45.5		
Total		44	100.0		

### rape unnamed suspect queue

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	3	6.8	16.7	16.7
	5	2	4.5	11.1	27.8
	7	1	2.3	5.6	33.3
	10	1	2.3	5.6	38.9
	11	1	2.3	5.6	44.4
	12	1	2.3	5.6	50.0
	39	1	2.3	5.6	55.6
	40	2	4.5	11.1	66.7
	42	1	2.3	5.6	72.2
	45	1	2.3	5.6	77.8
	82	1	2.3	5.6	83.3
	200	1	2.3	5.6	88.9
	300	1	2.3	5.6	94.4
	736	1	2.3	5.6	100.0
	Total	18	40.9	100.0	
Missing	999	26	59.1		
Total		44	100.0		

rape unnamed suspect analysis queue

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	5	11.4	31.3	31.3
	5	2	4.5	12.5	43.8
	30	1	2.3	6.3	50.0
	34	1	2.3	6.3	56.3
	42	1	2.3	6.3	62.5
	45	1	2.3	6.3	68.8
	50	1	2.3	6.3	75.0
	71	1	2.3	6.3	81.3
	115	1	2.3	6.3	87.5
	153	1	2.3	6.3	93.8
	300	1	2.3	6.3	100.0
	Total	16	36.4	100.0	
Missing	999	28	63.6		
Total		44	100.0		

### rape unnamed suspect report queue

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	7	15.9	63.6	63.6
	5	1	2.3	9.1	72.7
	6	1	2.3	9.1	81.8
	20	1	2.3	9.1	90.9
	30	1	2.3	9.1	100.0
	Total	11	25.0	100.0	
Missing	999	33	75.0		
Total		44	100.0		

# total homicide

					Cumulative
		Frequency	Percent	Valid Percent	Percent
Valid	0	3	6.8	8.8	8.8
	3	2	4.5	5.9	14.7
	5	1	2.3	2.9	17.6
	7	1	2.3	2.9	20.6
	10	2	4.5	5.9	26.5
	11	2	4.5	5.9	32.4
	24	1	2.3	2.9	35.3
	25	1	2.3	2.9	38.2
	30	1	2.3	2.9	41.2
	32	1	2.3	2.9	44.1
	34	1	2.3	2.9	47.1
	36	1	2.3	2.9	50.0
	38	1	2.3	2.9	52.9
	55	1	2.3	2.9	55.9
	60	1	2.3	2.9	58.8
	61	1	2.3	2.9	61.8
	72	1	2.3	2.9	64.7
	74	1	2.3	2.9	67.6
	84	2	4.5	5.9	73.5
	95	1	2.3	2.9	76.5
	100	1	2.3	2.9	79.4
	127	1	2.3	2.9	82.4
	142	1	2.3	2.9	85.3
	150	1	2.3	2.9	88.2
	192	1	2.3	2.9	91.2
	234	1	2.3	2.9	94.1
	525	1	2.3	2.9	97.1
	2416	1	2.3	2.9	100.0
	Total	34	77.3	100.0	
Missing	999	10	22.7		
Total		44	100.0		

# homicide screen queue

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	1	2.3	4.0	4.0
	3	4	9.1	16.0	20.0
	4	1	2.3	4.0	24.0
	8	1	2.3	4.0	28.0
	9	2	4.5	8.0	36.0
	18	1	2.3	4.0	40.0
	21	1	2.3	4.0	44.0
	26	2	4.5	8.0	52.0
	27	2	4.5	8.0	60.0
	44	2	4.5	8.0	68.0
	50	1	2.3	4.0	72.0
	70	2	4.5	8.0	80.0
	78	1	2.3	4.0	84.0
	91	1	2.3	4.0	88.0
	100	1	2.3	4.0	92.0
	108	1	2.3	4.0	96.0
	330	1	2.3	4.0	100.0
	Total	25	56.8	100.0	
Missing	999	19	43.2		
Total		44	100.0		

# homicide analysis queue

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	2	4.5	7.7	7.7
	1	1	2.3	3.8	11.5
	2	1	2.3	3.8	15.4
	3	3	6.8	11.5	26.9
	7	1	2.3	3.8	30.8
	9	1	2.3	3.8	34.6
	11	1	2.3	3.8	38.5
	14	2	4.5	7.7	46.2
	15	1	2.3	3.8	50.0
	16	1	2.3	3.8	53.8
	18	1	2.3	3.8	57.7
	24	1	2.3	3.8	61.5
	26	1	2.3	3.8	65.4
	28	1	2.3	3.8	69.2
	36	1	2.3	3.8	73.1
	42	1	2.3	3.8	76.9
	45	1	2.3	3.8	80.8
	48	1	2.3	3.8	84.6
	60	2	4.5	7.7	92.3
	126	1	2.3	3.8	96.2
	195	1	2.3	3.8	100.0
	Total	26	59.1	100.0	
Missing	999	18	40.9		
Total		44	100.0		

### homicide report queue

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	9	20.5	64.3	64.3
	2	1	2.3	7.1	71.4
	3	1	2.3	7.1	78.6
	6	1	2.3	7.1	85.7
	9	1	2.3	7.1	92.9
	23	1	2.3	7.1	100.0
	Total	14	31.8	100.0	
Missing	999	30	68.2		
Total		44	100.0		

# total homicide unnamed suspect

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	2	4.5	8.3	8.3
	1	1	2.3	4.2	12.5
	2	2	4.5	8.3	20.8
	3	1	2.3	4.2	25.0
	4	1	2.3	4.2	29.2
	5	1	2.3	4.2	33.3
	6	3	6.8	12.5	45.8
	10	2	4.5	8.3	54.2
	16	1	2.3	4.2	58.3
	17	1	2.3	4.2	62.5
	18	1	2.3	4.2	66.7
	26	1	2.3	4.2	70.8
	27	1	2.3	4.2	75.0
	38	1	2.3	4.2	79.2
	60	1	2.3	4.2	83.3
	61	1	2.3	4.2	87.5
	105	1	2.3	4.2	91.7
	115	1	2.3	4.2	95.8
	1019	1	2.3	4.2	100.0
	Total	24	54.5	100.0	
Missing	999	20	45.5		
Total		44	100.0		

### homicide unnamed suspect screen queue

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	6	13.6	40.0	40.0
	1	1	2.3	6.7	46.7
	2	1	2.3	6.7	53.3
	4	2	4.5	13.3	66.7
	7	1	2.3	6.7	73.3
	11	1	2.3	6.7	80.0
	37	1	2.3	6.7	86.7
	40	1	2.3	6.7	93.3
	66	1	2.3	6.7	100.0
	Total	15	34.1	100.0	
Missing	999	29	65.9		
Total		44	100.0		

# homicide unnamed suspect analysis queue

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	6	13.6	37.5	37.5
	2	1	2.3	6.3	43.8
	5	2	4.5	12.5	56.3
	8	1	2.3	6.3	62.5
	10	1	2.3	6.3	68.8
	14	1	2.3	6.3	75.0
	17	1	2.3	6.3	81.3
	18	1	2.3	6.3	87.5
	20	1	2.3	6.3	93.8
	39	1	2.3	6.3	100.0
	Total	16	36.4	100.0	
Missing	999	28	63.6		
Total		44	100.0		

### homicide unnamed suspect report queue

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	9	20.5	81.8	81.8
	6	1	2.3	9.1	90.9
	8	1	2.3	9.1	100.0
	Total	11	25.0	100.0	
Missing	999	33	75.0		
Total		44	100.0		

total property crime

		Fraguanay	Doroont	Valid Darsont	Cumulative
Valid	0	Frequency	Percent	Valid Percent	Percent
valiu	1	3	6.8	9.1	9.1
		1	2.3	3.0	12.1
	5	1	2.3	3.0	15.2
	7	2	4.5	6.1	21.2
	10	2	4.5	6.1	27.3
	13	1	2.3	3.0	30.3
	14	1	2.3	3.0	33.3
	23	1	2.3	3.0	36.4
	25	3	6.8	9.1	45.5
	29	1	2.3	3.0	48.5
	30	1	2.3	3.0	51.5
	31	1	2.3	3.0	54.5
	33	1	2.3	3.0	57.6
	40	1	2.3	3.0	60.6
	50	1	2.3	3.0	63.6
	58	1	2.3	3.0	66.7
	79	1	2.3	3.0	69.7
	99	1	2.3	3.0	72.7
	143	1	2.3	3.0	75.8
	144	1	2.3	3.0	78.8
	205	1	2.3	3.0	81.8
	212	1	2.3	3.0	84.8
	231	1	2.3	3.0	87.9
	398	1	2.3	3.0	90.9
	411	1	2.3	3.0	93.9
	730	1	2.3	3.0	97.0
	1356	1	2.3	3.0	100.0
	Total	33	75.0	100.0	.55.6
Missing	999	11	25.0		
Total		44	100.0		

# property crime screen queue

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	2	4.5	8.3	8.3
	1	2	4.5	8.3	16.7
	5	1	2.3	4.2	20.8
	9	2	4.5	8.3	29.2
	10	2	4.5	8.3	37.5
	11	1	2.3	4.2	41.7
	15	1	2.3	4.2	45.8
	16	1	2.3	4.2	50.0
	20	2	4.5	8.3	58.3
	27	1	2.3	4.2	62.5
	28	1	2.3	4.2	66.7
	40	1	2.3	4.2	70.8
	58	2	4.5	8.3	79.2
	86	1	2.3	4.2	83.3
	100	1	2.3	4.2	87.5
	128	1	2.3	4.2	91.7
	133	1	2.3	4.2	95.8
	336	1	2.3	4.2	100.0
	Total	24	54.5	100.0	
Missing	999	20	45.5		
Total		44	100.0		

### property crime analysis queue

					0 1 "
		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	2	4.5	9.1	9.1
	1	1	2.3	4.5	13.6
	4	1	2.3	4.5	18.2
	5	2	4.5	9.1	27.3
	8	1		4.5	31.8
	15	-	2.3		
	-	1	2.3	4.5	36.4
	16	4	9.1	18.2	54.5
	17	1	2.3	4.5	59.1
	21	2	4.5	9.1	68.2
	23	1	2.3	4.5	72.7
	25	1	2.3	4.5	77.3
	29	1	2.3	4.5	81.8
	62	1	2.3	4.5	86.4
	100	1	2.3	4.5	90.9
	124	1	2.3	4.5	95.5
	205	1	2.3	4.5	100.0
	Total	22	50.0	100.0	
Missing	999	22	50.0		
Total		44	100.0		

### property crime report queue

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	7	15.9	50.0	50.0
	1	2	4.5	14.3	64.3
	7	1	2.3	7.1	71.4
	10	1	2.3	7.1	78.6
	12	1	2.3	7.1	85.7
	21	1	2.3	7.1	92.9
	100	1	2.3	7.1	100.0
	Total	14	31.8	100.0	
Missing	999	30	68.2		
Total		44	100.0		

# total property crime unnamed suspect

		<b>-</b>	<b>D</b> 1	Valid Daniel	Cumulative
		Frequency	Percent	Valid Percent	Percent
Valid	0	1	2.3	4.2	4.2
	2	3	6.8	12.5	16.7
	4	1	2.3	4.2	20.8
	5	1	2.3	4.2	25.0
	7	2	4.5	8.3	33.3
	8	2	4.5	8.3	41.7
	13	1	2.3	4.2	45.8
	20	2	4.5	8.3	54.2
	23	1	2.3	4.2	58.3
	29	1	2.3	4.2	62.5
	57	1	2.3	4.2	66.7
	137	1	2.3	4.2	70.8
	144	1	2.3	4.2	75.0
	150	1	2.3	4.2	79.2
	190	1	2.3	4.2	83.3
	300	1	2.3	4.2	87.5
	342	1	2.3	4.2	91.7
	548	1	2.3	4.2	95.8
	555	1	2.3	4.2	100.0
	Total	24	54.5	100.0	
Missing	999	20	45.5		
Total		44	100.0		

### property crime unnamed suspect screen queue

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	4	9.1	28.6	28.6
	2	2	4.5	14.3	42.9
	8	1	2.3	7.1	50.0
	10	1	2.3	7.1	57.1
	12	1	2.3	7.1	64.3
	16	1	2.3	7.1	71.4
	25	2	4.5	14.3	85.7
	49	1	2.3	7.1	92.9
	205	1	2.3	7.1	100.0
	Total	14	31.8	100.0	
Missing	999	30	68.2		
Total		44	100.0		

### property crime unnamed suspect analysis queue

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	5	11.4	33.3	33.3
	2	2	4.5	13.3	46.7
	4	1	2.3	6.7	53.3
	7	1	2.3	6.7	60.0
	8	1	2.3	6.7	66.7
	18	1	2.3	6.7	73.3
	25	1	2.3	6.7	80.0
	71	1	2.3	6.7	86.7
	103	1	2.3	6.7	93.3
	180	1	2.3	6.7	100.0
	Total	15	34.1	100.0	
Missing	999	29	65.9		
Total		44	100.0		

### property crime unnamed suspect report queue

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	6	13.6	50.0	50.0
	2	2	4.5	16.7	66.7
	17	1	2.3	8.3	75.0
	25	1	2.3	8.3	83.3
	34	1	2.3	8.3	91.7
	35	1	2.3	8.3	100.0
	Total	12	27.3	100.0	
Missing	999	32	72.7		
Total		44	100.0		

total other offense

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	2	4.5	6.3	6.3
	2	3	6.8	9.4	15.6
	5	1	2.3	3.1	18.8
	10	1	2.3	3.1	21.9
	12	1	2.3	3.1	25.0
	14	1	2.3	3.1	28.1
	29	2	4.5	6.3	34.4
	30	1	2.3	3.1	37.5
	35	1	2.3	3.1	40.6
	40	2	4.5	6.3	46.9
	50	1	2.3	3.1	50.0
	51	1	2.3	3.1	53.1
	52	1	2.3	3.1	56.3
	69	1	2.3	3.1	59.4
	100	2	4.5	6.3	65.6
	101	1	2.3	3.1	68.8
	114	1	2.3	3.1	71.9
	130	1	2.3	3.1	75.0
	145	1	2.3	3.1	78.1
	172	1	2.3	3.1	81.3
	174	1	2.3	3.1	84.4
	243	1	2.3	3.1	87.5
	263	1	2.3	3.1	90.6
	382	1	2.3	3.1	93.8
	1096	1	2.3	3.1	96.9
	1285	1	2.3	3.1	100.0
	Total	32	72.7	100.0	
Missing	999	12	27.3		
Total		44	100.0		

# other offense screen queue

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	1	2.3	4.3	4.3
	1	1	2.3	4.3	8.7
	2	3	6.8	13.0	21.7
	5	2	4.5	8.7	30.4
	9	1	2.3	4.3	34.8
	10	1	2.3	4.3	39.1
	11	2	4.5	8.7	47.8
	20	1	2.3	4.3	52.2
	30	1	2.3	4.3	56.5
	41	1	2.3	4.3	60.9
	45	1	2.3	4.3	65.2
	50	1	2.3	4.3	69.6
	55	1	2.3	4.3	73.9
	82	1	2.3	4.3	78.3
	128	1	2.3	4.3	82.6
	133	1	2.3	4.3	87.0
	140	1	2.3	4.3	91.3
	178	1	2.3	4.3	95.7
	182	1	2.3	4.3	100.0
	Total	23	52.3	100.0	
Missing	999	21	47.7		
Total		44	100.0		

# other offense analysis queue

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	2	4.5	10.0	10.0
	1	1	2.3	5.0	15.0
	5	1	2.3	5.0	20.0
	8	1	2.3	5.0	25.0
	9	1	2.3	5.0	30.0
	10	1	2.3	5.0	35.0
	12	2	4.5	10.0	45.0
	18	1	2.3	5.0	50.0
	20	1	2.3	5.0	55.0
	27	1	2.3	5.0	60.0
	29	1	2.3	5.0	65.0
	40	1	2.3	5.0	70.0
	41	1	2.3	5.0	75.0
	44	1	2.3	5.0	80.0
	45	1	2.3	5.0	85.0
	55	1	2.3	5.0	90.0
	61	1	2.3	5.0	95.0
	242	1	2.3	5.0	100.0
	Total	20	45.5	100.0	
Missing	999	24	54.5		
Total		44	100.0		

### other offense report queue

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	5	11.4	41.7	41.7
	2	2	4.5	16.7	58.3
	5	2	4.5	16.7	75.0
	11	1	2.3	8.3	83.3
	20	1	2.3	8.3	91.7
	45	1	2.3	8.3	100.0
	Total	12	27.3	100.0	
Missing	999	32	72.7		
Total		44	100.0		

### total other offense unnamed suspect

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	2	4.5	11.8	11.8
	1	1	2.3	5.9	17.6
	2	1	2.3	5.9	23.5
	5	3	6.8	17.6	41.2
	6	1	2.3	5.9	47.1
	7	1	2.3	5.9	52.9
	8	1	2.3	5.9	58.8
	28	1	2.3	5.9	64.7
	34	1	2.3	5.9	70.6
	35	1	2.3	5.9	76.5
	40	1	2.3	5.9	82.4
	66	1	2.3	5.9	88.2
	180	1	2.3	5.9	94.1
	259	1	2.3	5.9	100.0
	Total	17	38.6	100.0	
Missing	999	27	61.4		
Total		44	100.0		

### other offense unnamed suspect screen queue

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	6	13.6	50.0	50.0
	1	2	4.5	16.7	66.7
	2	1	2.3	8.3	75.0
	10	1	2.3	8.3	83.3
	25	1	2.3	8.3	91.7
	38	1	2.3	8.3	100.0
	Total	12	27.3	100.0	
Missing	999	32	72.7		
Total		44	100.0		

#### other offense unnamed suspect analysis queue

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	5	11.4	38.5	38.5
	1	1	2.3	7.7	46.2
	2	2	4.5	15.4	61.5
	5	2	4.5	15.4	76.9
	8	1	2.3	7.7	84.6
	9	1	2.3	7.7	92.3
	14	1	2.3	7.7	100.0
	Total	13	29.5	100.0	
Missing	999	31	70.5		
Total		44	100.0		

other offense unnamed suspect report queue	other	offense	unnamed	susi	oect i	report	aueue
--	-------	---------	---------	------	--------	--------	-------

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	8	18.2	80.0	80.0
	4	2	4.5	20.0	100.0
	Total	10	22.7	100.0	
Missing	999	34	77.3		
Total		44	100.0		

### percent of rapes with possible DNA collected

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	11-25%	2	4.5	6.3	6.3
	25-50%	5	11.4	15.6	21.9
	50-75%	11	25.0	34.4	56.3
	75-100%	14	31.8	43.8	100.0
	Total	32	72.7	100.0	
Missing	999	12	27.3		
Total		44	100.0		

### percent of homicides with possible DNA collected

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	11-25%	3	6.8	9.4	9.4
	25-50%	3	6.8	9.4	18.8
	50-75%	9	20.5	28.1	46.9
	75-100%	17	38.6	53.1	100.0
	Total	32	72.7	100.0	
Missing	999	12	27.3		
Total		44	100.0		

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	13	29.5	30.2	30.2
	No	30	68.2	69.8	100.0
	Total	43	97.7	100.0	
Missing	999	1	2.3		
Total		44	100.0		

### percent of rape kits with DNA

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	11-25%	1	2.3	2.7	2.7
	25-50%	9	20.5	24.3	27.0
	50-75%	24	54.5	64.9	91.9
	75-100%	3	6.8	8.1	100.0
	Total	37	84.1	100.0	
Missing	999	7	15.9		
Total		44	100.0		

### percent all cases with DNA

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	25-50%	8	18.2	21.6	21.6
	50-75%	17	38.6	45.9	67.6
	75-100%	12	27.3	32.4	100.0
	Total	37	84.1	100.0	
Missing	999	7	15.9		
Total		44	100.0		

### educated guess not possible

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	9	20.5	20.9	20.9
	No	34	77.3	79.1	100.0
	Total	43	97.7	100.0	
Missing	999	1	2.3		
Total		44	100.0		

### % backlogged rape cases with rape kits only

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	1	2.3	3.0	3.0
	11-25%	6	13.6	18.2	21.2
	25-50%	7	15.9	21.2	42.4
	50-75%	8	18.2	24.2	66.7
	75-100%	11	25.0	33.3	100.0
	Total	33	75.0	100.0	
Missing	999	11	25.0		
Total		44	100.0		

### % backlogged rape cases with rape kits and other DNA evidence

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	1	2.3	2.9	2.9
	11-25%	5	11.4	14.7	17.6
	25-50%	14	31.8	41.2	58.8
	50-75%	7	15.9	20.6	79.4
	75-100%	7	15.9	20.6	100.0
	Total	34	77.3	100.0	
Missing	999	10	22.7		
Total		44	100.0		

### % backlogged rape cases with other DNA evidence only

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	22	50.0	64.7	64.7
	11-25%	9	20.5	26.5	91.2
	25-50%	2	4.5	5.9	97.1
	50-75%	1	2.3	2.9	100.0
	Total	34	77.3	100.0	
Missing	999	10	22.7		
Total		44	100.0		

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	9	20.5	20.9	20.9
	No	34	77.3	79.1	100.0
	Total	43	97.7	100.0	
Missing	999	1	2.3		
Total		44	100.0		

# of rape cases within statue of limitations

					Cumulative
		Frequency	Percent	Valid Percent	Percent
Valid	1	1	2.3	5.3	5.3
	10	1	2.3	5.3	10.5
	35	1	2.3	5.3	15.8
	37	1	2.3	5.3	21.1
	57	1	2.3	5.3	26.3
	65	1	2.3	5.3	31.6
	75	1	2.3	5.3	36.8
	100	2	4.5	10.5	47.4
	143	1	2.3	5.3	52.6
	187	1	2.3	5.3	57.9
	199	1	2.3	5.3	63.2
	202	1	2.3	5.3	68.4
	203	1	2.3	5.3	73.7
	241	1	2.3	5.3	78.9
	382	1	2.3	5.3	84.2
	479	1	2.3	5.3	89.5
	872	1	2.3	5.3	94.7
	3681	1	2.3	5.3	100.0
	Total	19	43.2	100.0	
Missing	999	25	56.8		
Total		44	100.0		

### # of unnamed suspect rape cases within statute of limitations

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	1	2.3	5.9	5.9
	1	2	4.5	11.8	17.6
	14	1	2.3	5.9	23.5
	15	2	4.5	11.8	35.3
	42	1	2.3	5.9	41.2
	50	1	2.3	5.9	47.1
	56	1	2.3	5.9	52.9
	100	1	2.3	5.9	58.8
	122	1	2.3	5.9	64.7
	153	1	2.3	5.9	70.6
	200	1	2.3	5.9	76.5
	262	1	2.3	5.9	82.4
	325	1	2.3	5.9	88.2
	382	1	2.3	5.9	94.1
	736	1	2.3	5.9	100.0
	Total	17	38.6	100.0	
Missing	999	27	61.4		
Total		44	100.0		

### # of rape cases exceeding statute of limitations by 6/30/03

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	20	45.5	95.2	95.2
	115	1	2.3	4.8	100.0
	Total	21	47.7	100.0	
Missing	999	23	52.3		
Total		44	100.0		

### # of unnamed suspect rape cases exceeding statute of limitations by 6/30/03

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	12	27.3	85.7	85.7
	24	1	2.3	7.1	92.9
	25	1	2.3	7.1	100.0
	Total	14	31.8	100.0	
Missing	999	30	68.2		
Total		44	100.0		

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	6	13.6	14.0	14.0
	No	37	84.1	86.0	100.0
	Total	43	97.7	100.0	
Missing	999	1	2.3		
Total		44	100.0		

# anticipated rape case backlog

					Cumulative
		Frequency	Percent	Valid Percent	Percent
Valid	0	1	2.3	2.7	2.7
	10	1	2.3	2.7	5.4
	15	1	2.3	2.7	8.1
	30	1	2.3	2.7	10.8
	35	1	2.3	2.7	13.5
	50	3	6.8	8.1	21.6
	80	1	2.3	2.7	24.3
	87	1	2.3	2.7	27.0
	90	1	2.3	2.7	29.7
	100	3	6.8	8.1	37.8
	115	1	2.3	2.7	40.5
	120	2	4.5	5.4	45.9
	150	3	6.8	8.1	54.1
	180	2	4.5	5.4	59.5
	207	1	2.3	2.7	62.2
	250	2	4.5	5.4	67.6
	256	1	2.3	2.7	70.3
	300	1	2.3	2.7	73.0
	400	1	2.3	2.7	75.7
	450	1	2.3	2.7	78.4
	500	1	2.3	2.7	81.1
	720	1	2.3	2.7	83.8
	750	1	2.3	2.7	86.5
	1053	1	2.3	2.7	89.2
	1234	1	2.3	2.7	91.9
	1608	1	2.3	2.7	94.6
	1800	1	2.3	2.7	97.3
	2232	1	2.3	2.7	100.0
	Total	37	84.1	100.0	
Missing	999	7	15.9		
Total		44	100.0		

# anticipated unnamed suspect rape case backlog

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	4	9.1	15.4	15.4
	10	1	2.3	3.8	19.2
	15	1	2.3	3.8	23.1
	17	1	2.3	3.8	26.9
	25	1	2.3	3.8	30.8
	30	3	6.8	11.5	42.3
	45	1	2.3	3.8	46.2
	50	2	4.5	7.7	53.8
	75	2	4.5	7.7	61.5
	85	1	2.3	3.8	65.4
	150	1	2.3	3.8	69.2
	200	1	2.3	3.8	73.1
	210	1	2.3	3.8	76.9
	300	1	2.3	3.8	80.8
	320	1	2.3	3.8	84.6
	450	1	2.3	3.8	88.5
	493	1	2.3	3.8	92.3
	500	1	2.3	3.8	96.2
	1000	1	2.3	3.8	100.0
	Total	26	59.1	100.0	
Missing	999	18	40.9		
Total		44	100.0		

# anticipated backlog of all cases

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	1 requericy	2.3	2.9	2.9
Valla	20	2	2.3 4.5	5.7	8.6
	50	1	2.3	2.9	11.4
	70	1	2.3	2.9	14.3
	75 75	1	2.3	2.9	17.1
	85	1	2.3	2.9	20.0
	100	1	2.3	2.9	22.9
	107	1	2.3	2.9	25.7
	125	1	2.3	2.9	28.6
	150	2	2.3 4.5	5.7	34.3
	170	1	2.3	2.9	3 <del>4</del> .3 37.1
	200	2	2.3 4.5	5.7	42.9
	220	1	2.3	2.9	45.7
	250	1	2.3	2.9	48.6
	300	4	2.3 9.1	11.4	60.0
	350	1	2.3	2.9	62.9
	400	1	2.3	2.9	65.7
	500	1	2.3	2.9	68.6
	700	1	2.3	2.9	71.4
	750	2	4.5	5.7	71.4
	800	1	2.3	2.9	80.0
	935	1	2.3	2.9	82.9
	1100	1	2.3	2.9	85.7
	1573	1	2.3	2.9	88.6
	2056	1	2.3	2.9	91.4
	2200	1	2.3	2.9	94.3
	2400	1	2.3	2.9	97.1
	2665	1	2.3	2.9	100.0
	Total	35	79.5	100.0	100.0
Missing	999	9	20.5	100.0	
Total		44	100.0		

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	12	27.3	27.9	27.9
	No	31	70.5	72.1	100.0
	Total	43	97.7	100.0	
Missing	999	1	2.3		
Total		44	100.0		

typical analysis time for unnamed suspect rape kit

					0 1
		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	.50	1	2.3	2.8	2.8
	2.00	1	2.3	2.8	5.6
	4.00	1	2.3	2.8	8.3
	5.00	1	2.3	2.8	11.1
	6.00	2	4.5	5.6	16.7
	8.00	1	2.3	2.8	19.4
	10.00	4	9.1	11.1	30.6
	11.00	2	4.5	5.6	36.1
	12.00	1	2.3	2.8	38.9
	14.00	1	2.3	2.8	41.7
	16.00	3	6.8	8.3	50.0
	18.00	2	4.5	5.6	55.6
	20.00	1	2.3	2.8	58.3
	22.00	1	2.3	2.8	61.1
	26.00	3	6.8	8.3	69.4
	30.00	1	2.3	2.8	72.2
	32.00	2	4.5	5.6	77.8
	40.00	1	2.3	2.8	80.6
	42.00	1	2.3	2.8	83.3
	52.00	4	9.1	11.1	94.4
	78.00	2	4.5	5.6	100.0
	Total	36	81.8	100.0	
Missing	999.00	8	18.2		
Total		44	100.0		

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	7	15.9	16.3	16.3
	No	36	81.8	83.7	100.0
	Total	43	97.7	100.0	
Missing	999	1	2.3		
Total		44	100.0		

# output capacity

					Cumulative
		Frequency	Percent	Valid Percent	Percent
Valid	24	2	4.5	5.6	5.6
	30	1	2.3	2.8	8.3
	32	1	2.3	2.8	11.1
	40	3	6.8	8.3	19.4
	70	3	6.8	8.3	27.8
	80	1	2.3	2.8	30.6
	90	1	2.3	2.8	33.3
	100	2	4.5	5.6	38.9
	135	1	2.3	2.8	41.7
	144	1	2.3	2.8	44.4
	250	2	4.5	5.6	50.0
	266	1	2.3	2.8	52.8
	336	1	2.3	2.8	55.6
	350	1	2.3	2.8	58.3
	410	1	2.3	2.8	61.1
	430	1	2.3	2.8	63.9
	512	1	2.3	2.8	66.7
	800	2	4.5	5.6	72.2
	952	1	2.3	2.8	75.0
	1000	1	2.3	2.8	77.8
	1200	1	2.3	2.8	80.6
	1500	1	2.3	2.8	83.3
	2141	1	2.3	2.8	86.1
	2250	1	2.3	2.8	88.9
	3375	1	2.3	2.8	91.7
	4800	1	2.3	2.8	94.4
	8172	1	2.3	2.8	97.2
	15360	1	2.3	2.8	100.0
	Total	36	81.8	100.0	
Missing	999	8	18.2		
Total		44	100.0		

### month or year

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	month	15	34.1	55.6	55.6
	year	12	27.3	44.4	100.0
	Total	27	61.4	100.0	
Missing	999	17	38.6		
Total		44	100.0		

### output capacity--samples per case

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	1	1	2.3	2.9	2.9
	2	1	2.3	2.9	5.9
	3	1	2.3	2.9	8.8
	4	7	15.9	20.6	29.4
	5	7	15.9	20.6	50.0
	6	4	9.1	11.8	61.8
	7	2	4.5	5.9	67.6
	8	5	11.4	14.7	82.4
	10	4	9.1	11.8	94.1
	20	1	2.3	2.9	97.1
	520	1	2.3	2.9	100.0
	Total	34	77.3	100.0	
Missing	999	10	22.7		
Total		44	100.0		

#### explanation of aberration or indifference at laboratories

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	1	1	2.3	9.1	9.1
	2	9	20.5	81.8	90.9
	4	1	2.3	9.1	100.0
	Total	11	25.0	100.0	
Missing	999	33	75.0		
Total		44	100.0		

### likelihood that agency has rape kits unaccounted for by crime lab?

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	A Certainty	22	50.0	51.2	51.2
	Not Possible	5	11.4	11.6	62.8
	Very Likely	5	11.4	11.6	74.4
	Very Unlikely	1	2.3	2.3	76.7
	Somewhat Likely	4	9.1	9.3	86.0
	Somewhat Unlikely	3	6.8	7.0	93.0
	Unknown	3	6.8	7.0	100.0
	Total	43	97.7	100.0	
Missing	999	1	2.3		
Total		44	100.0		

#### law enforcement routinely collect DNA from property crimes?

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	33	75.0	80.5	80.5
	No	8	18.2	19.5	100.0
	Total	41	93.2	100.0	
Missing	999	3	6.8		
Total		44	100.0		

#### lab accepts and processes DNA from unnamed suspect property crimes?

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	32	72.7	76.2	76.2
	No	10	22.7	23.8	100.0
	Total	42	95.5	100.0	
Missing	999	2	4.5		
Total		44	100.0		

#### lab accepts but does NOT process DNA from unnamed suspect property crimes?

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	8	18.2	26.7	26.7
	No	22	50.0	73.3	100.0
	Total	30	68.2	100.0	
Missing	999	14	31.8		
Total		44	100.0		

#### explanation of aberration or indifference at laboratories

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	2	8	18.2	100.0	100.0
Missing	999	36	81.8		
Total		44	100.0		

#### prop. crime barrier applies--lack of storage space

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	4	9.1	9.3	9.3
	No	39	88.6	90.7	100.0
	Total	43	97.7	100.0	
Missing	999	1	2.3		
Total		44	100.0		

### prop. crime barrier rank--lack of storage space

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	1	2.3	25.0	25.0
	Second Most Significant	1	2.3	25.0	50.0
	5	1	2.3	25.0	75.0
	99	1	2.3	25.0	100.0
	Total	4	9.1	100.0	
Missing	999	40	90.9		
Total		44	100.0		

#### prop. crime barrier applies--more personnel needed

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	13	29.5	30.2	30.2
	No	30	68.2	69.8	100.0
	Total	43	97.7	100.0	
Missing	999	1	2.3		
Total		44	100.0		

### prop. crime barrier rank--more personnel needed

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	9	20.5	75.0	75.0
	Second Most Significant	1	2.3	8.3	83.3
	Third Most Significant	2	4.5	16.7	100.0
	Total	12	27.3	100.0	
Missing	999	32	72.7		
Total		44	100.0		

### prop. crime barrier applies--backlog/other priority cases

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	13	29.5	30.2	30.2
	No	30	68.2	69.8	100.0
	Total	43	97.7	100.0	
Missing	999	1	2.3		
Total		44	100.0		

### prop. crime barrier rank--backlog/other priority cases

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	5	11.4	45.5	45.5
	Second Most Significant	3	6.8	27.3	72.7
	Third Most Significant	3	6.8	27.3	100.0
	Total	11	25.0	100.0	
Missing	999	33	75.0		
Total		44	100.0		

#### prop. crime barrier applies--equipment needed

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	8	18.2	18.6	18.6
	No	35	79.5	81.4	100.0
	Total	43	97.7	100.0	
Missing	999	1	2.3		
Total		44	100.0		

#### prop. crime barrier rank--equipment needed

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	3	6.8	42.9	42.9
	Second Most Significant	1	2.3	14.3	57.1
	Third Most Significant	2	4.5	28.6	85.7
	99	1	2.3	14.3	100.0
	Total	7	15.9	100.0	
Missing	999	37	84.1		
Total		44	100.0		

#### prop. crime barrier applies--lack of operational space

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	7	15.9	16.3	16.3
	No	36	81.8	83.7	100.0
	Total	43	97.7	100.0	
Missing	999	1	2.3		
Total		44	100.0		

#### prop. crime barrier rank--lack of operational space

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	1	2.3	16.7	16.7
	Second Most Significant	2	4.5	33.3	50.0
	Third Most Significant		4.5	33.3	83.3
	4	1	2.3	16.7	100.0
	Total	6	13.6	100.0	
Missing	999	38	86.4		
Total		44	100.0		

### prop. crime barrier applies--prosecution must request

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	2	4.5	4.7	4.7
	No	41	93.2	95.3	100.0
	Total	43	97.7	100.0	
Missing	999	1	2.3		
Total		44	100.0		

#### prop. crime barrier rank--prosecution must request

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	1	2.3	100.0	100.0
Missing	999	43	97.7		
Total		44	100.0		

#### prop. crime barrier applies--investigator must request

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	2	4.5	4.7	4.7
	No	41	93.2	95.3	100.0
	Total	43	97.7	100.0	
Missing	999	1	2.3		
Total		44	100.0		

#### prop. crime barrier rank--investigator must request

		Frequency	Percent
Missing	999	44	100.0

### prop. crime barrier applies--other

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	2	4.5	4.7	4.7
	No	41	93.2	95.3	100.0
	Total	43	97.7	100.0	
Missing	999	1	2.3		
Total		44	100.0		

#### prop. crime barrier rank--other

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Second Most Significant	1	2.3	100.0	100.0
Missing	999	43	97.7		
Total		44	100.0		

### prop. crime barrier applies--other (explanation)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	2	1	2.3	2.3	2.3
	999	43	97.7	97.7	100.0
	Total	44	100.0	100.0	

# named suspect cases--# of included suspects

			Downant	Valid Dansont	Cumulative
Valid	.46	Frequency 2	Percent 4.5	Valid Percent 6.5	Percent 6.5
Valid	.65	_	_		
	.67	1	2.3	3.2	9.7
	.75	1	2.3	3.2	12.9
	.80	5	11.4	16.1	29.0
	.85	2	4.5	6.5	35.5
		2	4.5	6.5	41.9
	.90	1	2.3	3.2	45.2
	.95	1	2.3	3.2	48.4
	11.00	1	2.3	3.2	51.6
	12.00	1	2.3	3.2	54.8
	47.00	1	2.3	3.2	58.1
	80.00	2	4.5	6.5	64.5
	150.00	3	6.8	9.7	74.2
	171.00	1	2.3	3.2	77.4
	213.00	1	2.3	3.2	80.6
	290.00	1	2.3	3.2	83.9
	308.00	1	2.3	3.2	87.1
	312.00	1	2.3	3.2	90.3
	365.00	1	2.3	3.2	93.5
	451.00	1	2.3	3.2	96.8
	720.00	1	2.3	3.2	100.0
	Total	31	70.5	100.0	
Missing	999.00	13	29.5		
Total		44	100.0		

# named suspect cases--# of excluded suspects

					Cumulative
		Frequency	Percent	Valid Percent	Percent
Valid	.05	1	2.3	3.2	3.2
	.10	1	2.3	3.2	6.5
	.15	2	4.5	6.5	12.9
	.20	4	9.1	12.9	25.8
	.25	6	13.6	19.4	45.2
	.33	1	2.3	3.2	48.4
	.35	1	2.3	3.2	51.6
	3.00	1	2.3	3.2	54.8
	4.00	1	2.3	3.2	58.1
	10.00	1	2.3	3.2	61.3
	20.00	3	6.8	9.7	71.0
	30.00	1	2.3	3.2	74.2
	42.00	1	2.3	3.2	77.4
	70.00	1	2.3	3.2	80.6
	90.00	1	2.3	3.2	83.9
	98.00	1	2.3	3.2	87.1
	110.00	1	2.3	3.2	90.3
	133.00	1	2.3	3.2	93.5
	150.00	1	2.3	3.2	96.8
	180.00	1	2.3	3.2	100.0
	Total	31	70.5	100.0	
Missing	999.00	13	29.5		
Total		44	100.0		

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	16	36.4	37.2	37.2
	No	27	61.4	62.8	100.0
	Total	43	97.7	100.0	
Missing	999	1	2.3		
Total		44	100.0		

# total cases benefiting from mitochondrial DNA testing (but not STR)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	1	2.3	3.2	3.2
	2	1	2.3	3.2	6.5
	3	1	2.3	3.2	9.7
	4	1	2.3	3.2	12.9
	5	4	9.1	12.9	25.8
	6	1	2.3	3.2	29.0
	8	1	2.3	3.2	32.3
	10	5	11.4	16.1	48.4
	15	4	9.1	12.9	61.3
	19	1	2.3	3.2	64.5
	20	3	6.8	9.7	74.2
	35	1	2.3	3.2	77.4
	40	2	4.5	6.5	83.9
	45	1	2.3	3.2	87.1
	75	1	2.3	3.2	90.3
	80	1	2.3	3.2	93.5
	100	1	2.3	3.2	96.8
	250	1	2.3	3.2	100.0
	Total	31	70.5	100.0	
Missing	999	13	29.5		
Total		44	100.0		

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	12	27.3	27.9	27.9
	No	31	70.5	72.1	100.0
	Total	43	97.7	100.0	
Missing	999	1	2.3		
Total		44	100.0		

# hair analysis cases that could benefit from mitochondrial DNA testing

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	1	1	2.3	4.2	4.2
	2	1	2.3	4.2	8.3
	3	2	4.5	8.3	16.7
	4	1	2.3	4.2	20.8
	5	2	4.5	8.3	29.2
	8	1	2.3	4.2	33.3
	10	5	11.4	20.8	54.2
	20	1	2.3	4.2	58.3
	25	1	2.3	4.2	62.5
	35	2	4.5	8.3	70.8
	40	1	2.3	4.2	75.0
	50	2	4.5	8.3	83.3
	80	1	2.3	4.2	87.5
	150	1	2.3	4.2	91.7
	200	1	2.3	4.2	95.8
	380	1	2.3	4.2	100.0
	Total	24	54.5	100.0	
Missing	999	20	45.5		
Total		44	100.0		

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	19	43.2	44.2	44.2
	No	24	54.5	55.8	100.0
	Total	43	97.7	100.0	
Missing	999	1	2.3		
Total		44	100.0		

# number of current offender profiles in DNA database as of November 1, 2002

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	458	1	2.3	2.3	2.3
	650	1	2.3	2.3	4.7
	1000	1	2.3	2.3	7.0
	1443	1	2.3	2.3	9.3
	1804	1	2.3	2.3	11.6
	2000	2	4.5	4.7	16.3
	2038	1	2.3	2.3	18.6
	2612	1	2.3	2.3	20.9
	3185	1	2.3	2.3	23.3
	3200	1	2.3	2.3	25.6
	3300	1	2.3	2.3	27.9
	3752	2	4.5	4.7	32.6
	3910	1	2.3	2.3	34.9
	4363	1	2.3	2.3	37.2
	5500	1	2.3	2.3	39.5
	7000	1	2.3	2.3	41.9
	7500	1	2.3	2.3	44.2
	9186	1	2.3	2.3	46.5
	9634	1	2.3	2.3	48.8
	12000	1	2.3	2.3	51.2
	13000	1	2.3	2.3	53.5
	14000	1	2.3	2.3	55.8
	15082	1	2.3	2.3	58.1
	16387	1	2.3	2.3	60.5
	20804	1	2.3	2.3	62.8
	20968	1	2.3	2.3	65.1
	22000	1	2.3	2.3	67.4
	22448	1	2.3	2.3	69.8
	24129	1	2.3	2.3	72.1
	26390	1	2.3	2.3	74.4
	29000	1	2.3	2.3	76.7
	31926	1	2.3	2.3	79.1
	34541	1	2.3	2.3	81.4
	35388	1	2.3	2.3	83.7
	46078	1	2.3	2.3	86.0
	51655	1	2.3	2.3	88.4
	99499	1	2.3	2.3	90.7
	137247	1	2.3	2.3	93.0
	143787	1	2.3	2.3	95.3
	187846	1	2.3	2.3	97.7
	205047	1	2.3	2.3	100.0
	Total	43	97.7	100.0	
Missing	999	1	2.3		
Total		44	100.0		

number of expected offender profiles in DNA database by June 30, 2003

					Cumulative
		Frequency	Percent	Valid Percent	Percent
Valid	0	1	2.3	2.4	2.4
	200	1	2.3	2.4	4.8
	300	1	2.3	2.4	7.1
	400	2	4.5	4.8	11.9
	500	3	6.8	7.1	19.0
	800	1	2.3	2.4	21.4
	1000	3	6.8	7.1	28.6
	1110	1	2.3	2.4	31.0
	1200	1	2.3	2.4	33.3
	2000	4	9.1	9.5	42.9
	2100	1	2.3	2.4	45.2
	2800	1	2.3	2.4	47.6
	4000	1	2.3	2.4	50.0
	5000	2	4.5	4.8	54.8
	5500	1	2.3	2.4	57.1
	6000	1	2.3	2.4	59.5
	7000	2	4.5	4.8	64.3
	7200	1	2.3	2.4	66.7
	8500	1	2.3	2.4	69.0
	10000	3	6.8	7.1	76.2
	12000	1	2.3	2.4	78.6
	13759	1	2.3	2.4	81.0
	15000	1	2.3	2.4	83.3
	18000	1	2.3	2.4	85.7
	20000	1	2.3	2.4	88.1
	25000	1	2.3	2.4	90.5
	30000	1	2.3	2.4	92.9
	32082	1	2.3	2.4	95.2
	33390	1	2.3	2.4	97.6
	84765	1	2.3	2.4	100.0
	Total	42	95.5	100.0	
Missing	999	2	4.5		
Total		44	100.0		

### educated guess not possible

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	6	13.6	14.0	14.0
	No	37	84.1	86.0	100.0
	Total	43	97.7	100.0	
Missing	999	1	2.3		
Total		44	100.0		

size of current offender backlog as of November 1, 2002

					Cumulative
		Frequency	Percent	Valid Percent	Percent
Valid	0	4	9.1	9.5	9.5
	150	2	4.5	4.8	14.3
	200	2	4.5	4.8	19.0
	237	1	2.3	2.4	21.4
	400	1	2.3	2.4	23.8
	450	1	2.3	2.4	26.2
	500	1	2.3	2.4	28.6
	554	1	2.3	2.4	31.0
	581	1	2.3	2.4	33.3
	698	2	4.5	4.8	38.1
	800	1	2.3	2.4	40.5
	1000	2	4.5	4.8	45.2
	1613	1	2.3	2.4	47.6
	1800	1	2.3	2.4	50.0
	1838	1	2.3	2.4	52.4
	2000	3	6.8	7.1	59.5
	2305	1	2.3	2.4	61.9
	3000	1	2.3	2.4	64.3
	3230	1	2.3	2.4	66.7
	3700	1	2.3	2.4	69.0
	4000	1	2.3	2.4	71.4
	4942	1	2.3	2.4	73.8
	5000	1	2.3	2.4	76.2
	6000	1	2.3	2.4	78.6
	9090	1	2.3	2.4	81.0
	10875	1	2.3	2.4	83.3
	12000	1	2.3	2.4	85.7
	12500	1	2.3	2.4	88.1
	13020	1	2.3	2.4	90.5
	17000	1	2.3	2.4	92.9
	36448	1	2.3	2.4	95.2
	60000	1	2.3	2.4	97.6
	61156	1	2.3	2.4	100.0
	Total	42	95.5	100.0	
Missing	999	2	4.5		
Total		44	100.0		

# size of projected offender backlog by June 30, 2003

					Cumulative
		Frequency	Percent	Valid Percent	Percent
Valid	0	8	18.2	19.5	19.5
	100	1	2.3	2.4	22.0
	150	1	2.3	2.4	24.4
	200	2	4.5	4.9	29.3
	600	1	2.3	2.4	31.7
	900	1	2.3	2.4	34.1
	1000	3	6.8	7.3	41.5
	1500	1	2.3	2.4	43.9
	2113	1	2.3	2.4	46.3
	2500	1	2.3	2.4	48.8
	3000	1	2.3	2.4	51.2
	3400	2	4.5	4.9	56.1
	4224	1	2.3	2.4	58.5
	5000	1	2.3	2.4	61.0
	6000	1	2.3	2.4	63.4
	6100	1	2.3	2.4	65.9
	7860	1	2.3	2.4	68.3
	8000	1	2.3	2.4	70.7
	9000	1	2.3	2.4	73.2
	11000	1	2.3	2.4	75.6
	16500	1	2.3	2.4	78.0
	20000	1	2.3	2.4	80.5
	23000	1	2.3	2.4	82.9
	25000	2	4.5	4.9	87.8
	29000	1	2.3	2.4	90.2
	30000	1	2.3	2.4	92.7
	60000	1	2.3	2.4	95.1
	66462	1	2.3	2.4	97.6
	90000	1	2.3	2.4	100.0
	Total	41	93.2	100.0	
Missing	999	3	6.8		
Total		44	100.0		

# educated guess not possible

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	4	9.1	9.3	9.3
	No	39	88.6	90.7	100.0
	Total	43	97.7	100.0	
Missing	999	1	2.3		
Total		44	100.0		

new samples 1st year

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	2000	1	2.3	4.8	4.8
	3000	1	2.3	4.8	9.5
	4000	1	2.3	4.8	14.3
	4700	1	2.3	4.8	19.0
	5500	1	2.3	4.8	23.8
	11800	2	4.5	9.5	33.3
	16000	1	2.3	4.8	38.1
	18000	1	2.3	4.8	42.9
	20000	2	4.5	9.5	52.4
	24000	1	2.3	4.8	57.1
	27400	1	2.3	4.8	61.9
	35000	1	2.3	4.8	66.7
	50000	1	2.3	4.8	71.4
	59000	2	4.5	9.5	81.0
	110000	1	2.3	4.8	85.7
	138000	1	2.3	4.8	90.5
	161800	1	2.3	4.8	95.2
	1500000	1	2.3	4.8	100.0
	Total	21	47.7	100.0	
Missing	999	23	52.3		
Total		44	100.0		

# annual samples in future years

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	1000	1	2.3	4.5	4.5
	1300	1	2.3	4.5	9.1
	1500	1	2.3	4.5	13.6
	1718	1	2.3	4.5	18.2
	2000	1	2.3	4.5	22.7
	2500	1	2.3	4.5	27.3
	6000	2	4.5	9.1	36.4
	11800	2	4.5	9.1	45.5
	12000	1	2.3	4.5	50.0
	12166	1	2.3	4.5	54.5
	17000	1	2.3	4.5	59.1
	17500	1	2.3	4.5	63.6
	18000	1	2.3	4.5	68.2
	20000	1	2.3	4.5	72.7
	22000	1	2.3	4.5	77.3
	25000	1	2.3	4.5	81.8
	42700	1	2.3	4.5	86.4
	60500	1	2.3	4.5	90.9
	72000	1	2.3	4.5	95.5
	140000	1	2.3	4.5	100.0
	Total	22	50.0	100.0	
Missing	999	22	50.0		
Total		44	100.0		

### educated guess not possible

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	9	20.5	20.9	20.9
	No	34	77.3	79.1	100.0
	Total	43	97.7	100.0	
Missing	999	1	2.3		
Total		44	100.0		

### offender DNA samples taken at sentencing or intake

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	1	32	72.7	76.2	76.2
	2	10	22.7	23.8	100.0
	Total	42	95.5	100.0	
Missing	999	2	4.5		
Total		44	100.0		

# offender DNA samples taken prior to release

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	1	12	27.3	28.6	28.6
	2	30	68.2	71.4	100.0
	Total	42	95.5	100.0	
Missing	999	2	4.5		
Total		44	100.0		

# offender DNA sample processing time

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	2.00	3	6.8	8.1	8.1
	4.00	3	6.8	8.1	16.2
	5.00	1	2.3	2.7	18.9
	6.00	3	6.8	8.1	27.0
	8.00	3	6.8	8.1	35.1
	12.00	2	4.5	5.4	40.5
	14.00	3	6.8	8.1	48.6
	16.00	3	6.8	8.1	56.8
	18.00	1	2.3	2.7	59.5
	26.00	3	6.8	8.1	67.6
	39.00	5	11.4	13.5	81.1
	52.00	5	11.4	13.5	94.6
	78.00	2	4.5	5.4	100.0
	Total	37	84.1	100.0	
Missing	999.00	7	15.9		
Total		44	100.0		

# educated guess not possible

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	1	2.3	2.3	2.3
	Yes	6	13.6	14.0	16.3
	No	36	81.8	83.7	100.0
	Total	43	97.7	100.0	
Missing	999	1	2.3		
Total		44	100.0		

# average cost of rape kit analysis

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	50	2	4.5	7.1	7.1
	90	1	2.3	3.6	10.7
	130	1	2.3	3.6	14.3
	250	1	2.3	3.6	17.9
	300	3	6.8	10.7	28.6
	313	1	2.3	3.6	32.1
	385	1	2.3	3.6	35.7
	400	1	2.3	3.6	39.3
	436	1	2.3	3.6	42.9
	500	1	2.3	3.6	46.4
	520	1	2.3	3.6	50.0
	640	1	2.3	3.6	53.6
	740	1	2.3	3.6	57.1
	750	1	2.3	3.6	60.7
	1000	2	4.5	7.1	67.9
	1002	1	2.3	3.6	71.4
	1200	1	2.3	3.6	75.0
	1400	1	2.3	3.6	78.6
	1500	1	2.3	3.6	82.1
	1600	2	4.5	7.1	89.3
	1700	1	2.3	3.6	92.9
	2000	1	2.3	3.6	96.4
	9000	1	2.3	3.6	100.0
	Total	28	63.6	100.0	
Missing	999	16	36.4		
Total		44	100.0		

### educated guess not possible

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	11	25.0	25.6	25.6
	No	32	72.7	74.4	100.0
	Total	43	97.7	100.0	
Missing	999	1	2.3		
Total		44	100.0		

# cost factor--reagents

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	29	65.9	67.4	67.4
	No	14	31.8	32.6	100.0
	Total	43	97.7	100.0	
Missing	999	1	2.3		
Total		44	100.0		

### cost factor--equipment

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	12	27.3	27.9	27.9
	No	31	70.5	72.1	100.0
	Total	43	97.7	100.0	
Missing	999	1	2.3		
Total		44	100.0		

### cost factor--salaries

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	21	47.7	48.8	48.8
	No	22	50.0	51.2	100.0
	Total	43	97.7	100.0	
Missing	999	1	2.3		
Total		44	100.0		

#### cost factor--overhead

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	5	11.4	11.6	11.6
	No	38	86.4	88.4	100.0
	Total	43	97.7	100.0	
Missing	999	1	2.3		
Total		44	100.0		

#### cost factor--other

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	4	9.1	9.3	9.3
	No	39	88.6	90.7	100.0
	Total	43	97.7	100.0	
Missing	999	1	2.3		
Total		44	100.0		

### other-explanation

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	1.00	7	15.9	15.9	15.9
	2.00	5	11.4	11.4	27.3
	999.00	32	72.7	72.7	100.0
	Total	44	100.0	100.0	

### type of DNA sample collected

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	buccal	12	27.3	28.6	28.6
	blood	22	50.0	52.4	81.0
	both	8	18.2	19.0	100.0
	Total	42	95.5	100.0	
Missing	999	2	4.5		
Total		44	100.0		

# per sample estimate of collection

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	2	3	6.8	10.7	10.7
	3	1	2.3	3.6	14.3
	3	1	2.3	3.6	17.9
	3	6	13.6	21.4	39.3
	4	1	2.3	3.6	42.9
	4	3	6.8	10.7	53.6
	5	1	2.3	3.6	57.1
	5	6	13.6	21.4	78.6
	6	1	2.3	3.6	82.1
	11	1	2.3	3.6	85.7
	12	1	2.3	3.6	89.3
	25	1	2.3	3.6	92.9
	30	1	2.3	3.6	96.4
	35	1	2.3	3.6	100.0
	Total	28	63.6	100.0	
Missing	999	16	36.4		
Total		44	100.0		

# per sample estimate of equipment

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	1	2.3	10.0	10.0
	1	2	4.5	20.0	30.0
	2	1	2.3	10.0	40.0
	6	1	2.3	10.0	50.0
	10	2	4.5	20.0	70.0
	13	1	2.3	10.0	80.0
	13	1	2.3	10.0	90.0
	50	1	2.3	10.0	100.0
	Total	10	22.7	100.0	
Missing	999	34	77.3		
Total		44	100.0		

### per sample estimate of reagents

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	1	2.3	5.0	5.0
	2	1	2.3	5.0	10.0
	5	1	2.3	5.0	15.0
	9	1	2.3	5.0	20.0
	11	1	2.3	5.0	25.0
	13	1	2.3	5.0	30.0
	22	2	4.5	10.0	40.0
	24	1	2.3	5.0	45.0
	25	1	2.3	5.0	50.0
	30	3	6.8	15.0	65.0
	32	1	2.3	5.0	70.0
	35	1	2.3	5.0	75.0
	39	1	2.3	5.0	80.0
	40	1	2.3	5.0	85.0
	42	1	2.3	5.0	90.0
	50	1	2.3	5.0	95.0
	60	1	2.3	5.0	100.0
	Total	20	45.5	100.0	
Missing	999	24	54.5		
Total		44	100.0		

# per sample estimate of salaries

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	5	2	4.5	13.3	13.3
	8	2	4.5	13.3	26.7
	14	1	2.3	6.7	33.3
	15	1	2.3	6.7	40.0
	18	1	2.3	6.7	46.7
	19	1	2.3	6.7	53.3
	20	1	2.3	6.7	60.0
	27	1	2.3	6.7	66.7
	30	1	2.3	6.7	73.3
	34	1	2.3	6.7	80.0
	50	1	2.3	6.7	86.7
	140	1	2.3	6.7	93.3
	400	1	2.3	6.7	100.0
	Total	15	34.1	100.0	
Missing	999	29	65.9		
Total		44	100.0		

### per sample estimate of storage

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	1	2.3	12.5	12.5
	1	3	6.8	37.5	50.0
	2	1	2.3	12.5	62.5
	7	1	2.3	12.5	75.0
	10	1	2.3	12.5	87.5
	37	1	2.3	12.5	100.0
	Total	8	18.2	100.0	
Missing	999	36	81.8		
Total		44	100.0		

# per sample estimate of other

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	1	1	2.3	7.1	7.1
	2	1	2.3	7.1	14.3
	3	1	2.3	7.1	21.4
	14	1	2.3	7.1	28.6
	20	1	2.3	7.1	35.7
	29	1	2.3	7.1	42.9
	35	1	2.3	7.1	50.0
	39	1	2.3	7.1	57.1
	40	2	4.5	14.3	71.4
	43	1	2.3	7.1	78.6
	45	1	2.3	7.1	85.7
	49	1	2.3	7.1	92.9
	100	1	2.3	7.1	100.0
	Total	14	31.8	100.0	
Missing	999	30	68.2		
Total		44	100.0		

### educated guess not possible

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	11	25.0	25.6	25.6
	No	32	72.7	74.4	100.0
	Total	43	97.7	100.0	
Missing	999	1	2.3		
Total		44	100.0		

# funding mechanism for analysis?

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	14	31.8	32.6	32.6
	No	29	65.9	67.4	100.0
	Total	43	97.7	100.0	
Missing	999	1	2.3		
Total		44	100.0		

# mechanism description?

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	13	29.5	31.0	31.0
	No	29	65.9	69.0	100.0
	Total	42	95.5	100.0	
Missing	999	2	4.5		
Total		44	100.0		

# explanation of mechanism description

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	\$0.07 per court fee with amount capped 350 K	1	2.3	2.3	2.3
	\$0.07 per court fee with amount capped at 350K	1	2.3	2.3	4.5
	\$100 fee from each offender	1	2.3	2.3	6.8
	\$1200 Cost per case upon conviction	1	2.3	2.3	9.1
	\$150.00 Fee	1	2.3	2.3	11.4
	\$2 fee for DNA analysis court costs	1	2.3	2.3	13.6
	250 mandatory fine	1	2.3	2.3	15.9
	3% surcharge on fines	1	2.3	2.3	18.2
	999	31	70.5	70.5	88.6
	dedicated appropriated funds	1	2.3	2.3	90.9
	fee	1	2.3	2.3	93.2
	fee collected from each offender	1	2.3	2.3	95.5
	forensic fee assessment	1	2.3	2.3	97.7
	Missing persons program	1	2.3	2.3	100.0
	Total	44	100.0	100.0	

# amount collected?

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	12	27.3	27.9	27.9
	No	31	70.5	72.1	100.0
	Total	43	97.7	100.0	
Missing	999	1	2.3		
Total		44	100.0		

#### amount collected

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	1	1	2.3	2.3	2.3
	999	12	27.3	27.3	29.5
	\$14,700	1	2.3	2.3	31.8
	\$315000	1	2.3	2.3	34.1
	\$4000	1	2.3	2.3	36.4
	\$48095	1	2.3	2.3	38.6
	0	1	2.3	2.3	40.9
	1.1 million total	1	2.3	2.3	43.2
	2 million/ 100,00 to DNA	1	2.3	2.3	45.5
	3657000	1	2.3	2.3	47.7
	80,000 per year on average	1	2.3	2.3	50.0
	999	20	45.5	45.5	95.5
	Expected to collect 200K this year	1	2.3	2.3	97.7
	expected to collect less than 200K this year	1	2.3	2.3	100.0
	Total	44	100.0	100.0	

# % caseload funded?

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	11	25.0	25.6	25.6
	No	32	72.7	74.4	100.0
	Total	43	97.7	100.0	
Missing	999	1	2.3		
Total		44	100.0		

# explanation of % caseload funded

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0	1	2.3	2.3	2.3
	0 %	1	2.3	2.3	4.5
	0%	1	2.3	2.3	6.8
	0.6 %	1	2.3	2.3	9.1
	2%	1	2.3	2.3	11.4
	30%	1	2.3	2.3	13.6
	40%	1	2.3	2.3	15.9
	999	33	75.0	75.0	90.9
	As of now 0	1	2.3	2.3	93.2
	As of now, 0	1	2.3	2.3	95.5
	less than 1%	1	2.3	2.3	97.7
	Not all allocat	1	2.3	2.3	100.0
	Total	44	100.0	100.0	

### educated guess not possible

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	No	43	97.7	100.0	100.0
Missing	999	1	2.3		
Total		44	100.0		

### funding mechanism of collection?

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	8	18.2	20.0	20.0
	No	32	72.7	80.0	100.0
	Total	40	90.9	100.0	
Missing	999	4	9.1		
Total		44	100.0		

### mechanism description?

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	9	20.5	21.4	21.4
	No	33	75.0	78.6	100.0
	Total	42	95.5	100.0	
Missing	999	2	4.5		
Total		44	100.0		

# explanation of mechanism description

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	\$25.00 collection fee assessed as part of criminal costs	1	2.3	2.3	2.3
	150.00 Fee	1	2.3	2.3	4.5
	999	36	81.8	81.8	86.4
	dedicated appropriated funds	1	2.3	2.3	88.6
	discretionary \$60 assessment	1	2.3	2.3	90.9
	Funded through inmate fees and fiscal notes	1	2.3	2.3	93.2
	kit purchase and description	1	2.3	2.3	95.5
	Mandatory \$250 cost per individual convicted of crimes	1	2.3	2.3	97.7
	See item 24	1	2.3	2.3	100.0
	Total	44	100.0	100.0	

# amount collected?

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	6	13.6	14.3	14.3
	No	36	81.8	85.7	100.0
	Total	42	95.5	100.0	
Missing	999	2	4.5		
Total		44	100.0		

# explanation of amount collected

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	999	11	25.0	25.0	25.0
	\$169,000	1	2.3	2.3	27.3
	\$196,000 fees; \$269,000 fiscal note	1	2.3	2.3	29.5
	\$3	1	2.3	2.3	31.8
	\$614,000	1	2.3	2.3	34.1
	999	27	61.4	61.4	95.5
	crimes \$109,209	1	2.3	2.3	97.7
	Included in item 24	1	2.3	2.3	100.0
	Total	44	100.0	100.0	

### % caseload funded?

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	5	11.4	11.6	11.6
	No	38	86.4	88.4	100.0
	Total	43	97.7	100.0	
Missing	999	1	2.3		
Total		44	100.0		

### explanation of % caseload funded

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	1.00	1	2.3	2.3	2.3
ξ	999.00	10	22.7	22.7	25.0
\$	3100	1	2.3	2.3	27.3
0	,	1	2.3	2.3	29.5
20 ui	00% for 20002-03; Insure after 2003	1	2.3	2.3	31.8
1/	5%	1	2.3	2.3	34.1
5/	50	1	2.3	2.3	36.4
9:	99	28	63.6	63.6	100.0
Т	otal	44	100.0	100.0	

# educated guess not possible

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	1	2.3	2.4	2.4
	No	41	93.2	97.6	100.0
	Total	42	95.5	100.0	
Missing	999	2	4.5		
Total		44	100.0		

# % of budget from local funding?

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	21	47.7	87.5	87.5
	25-50%	1	2.3	4.2	91.7
	50-75%	1	2.3	4.2	95.8
	75-100%	1	2.3	4.2	100.0
	Total	24	54.5	100.0	
Missing	999	20	45.5		
Total		44	100.0		

### % of budget from state funding?

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	1	2.3	2.6	2.6
	11-25%	4	9.1	10.3	12.8
	25-50%	6	13.6	15.4	28.2
	50-75%	10	22.7	25.6	53.8
	75-100%	18	40.9	46.2	100.0
	Total	39	88.6	100.0	
Missing	999	5	11.4		
Total		44	100.0		

# % of budget from federal funding?

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	0-10%	9	20.5	23.1	23.1
	11-25%	12	27.3	30.8	53.8
	25-50%	10	22.7	25.6	79.5
	50-75%	7	15.9	17.9	97.4
	75-100%	1	2.3	2.6	100.0
	Total	39	88.6	100.0	
Missing	999	5	11.4		
Total		44	100.0		

# educated guess not possible

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	6	13.6	14.0	14.0
	No	37	84.1	86.0	100.0
	Total	43	97.7	100.0	
Missing	999	1	2.3		
Total		44	100.0		

# supplemental funding exists?

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	yes	39	88.6	90.7	90.7
	no	4	9.1	9.3	100.0
	Total	43	97.7	100.0	
Missing	999	1	2.3		
Total		44	100.0		

### use \$ applies--salaries

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	25	56.8	59.5	59.5
	No	17	38.6	40.5	100.0
	Total	42	95.5	100.0	
Missing	999	2	4.5		
Total		44	100.0		

#### use \$ rank--salaries

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	15	34.1	75.0	75.0
	Second Most Significant	3	6.8	15.0	90.0
	Third Most Significant	2	4.5	10.0	100.0
	Total	20	45.5	100.0	
Missing	999	24	54.5		
Total		44	100.0		

# use \$ applies--overtime

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	21	47.7	48.8	48.8
	No	22	50.0	51.2	100.0
	Total	43	97.7	100.0	
Missing	999	1	2.3		
Total		44	100.0		

#### use \$ rank--overtime

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	3	6.8	42.9	42.9
	Second Most Significant	1	2.3	14.3	57.1
	Third Most Significant	3	6.8	42.9	100.0
	Total	7	15.9	100.0	
Missing	999	37	84.1		
Total		44	100.0		

# use \$ applies--training

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	27	61.4	62.8	62.8
	No	16	36.4	37.2	100.0
	Total	43	97.7	100.0	
Missing	999	1	2.3		
Total		44	100.0		

### use \$ rank--training

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	2	4.5	16.7	16.7
	Second Most Significant	3	6.8	25.0	41.7
	Third Most Significant	6	13.6	50.0	91.7
	5	1	2.3	8.3	100.0
	Total	12	27.3	100.0	
Missing	999	32	72.7		
Total		44	100.0		

# use \$ applies--equipment

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	29	65.9	67.4	67.4
	No	14	31.8	32.6	100.0
	Total	43	97.7	100.0	
Missing	999	1	2.3		
Total		44	100.0		

### use \$ rank--equipment

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	3	6.8	16.7	16.7
	Second Most Significant	8	18.2	44.4	61.1
	Third Most Significant	7	15.9	38.9	100.0
	Total	18	40.9	100.0	
Missing	999	26	59.1		
Total		44	100.0		

### use \$ applies--robotics

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	25	56.8	58.1	58.1
	No	18	40.9	41.9	100.0
	Total	43	97.7	100.0	
Missing	999	1	2.3		
Total		44	100.0		

#### use \$ rank--robotics

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	2	4.5	18.2	18.2
	Second Most Significant	4	9.1	36.4	54.5
	Third Most Significant	4	9.1	36.4	90.9
	4	1	2.3	9.1	100.0
	Total	11	25.0	100.0	
Missing	999	33	75.0		
Total		44	100.0		

# use \$ applies--reagents

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	33	75.0	76.7	76.7
	No	10	22.7	23.3	100.0
	Total	43	97.7	100.0	
Missing	999	1	2.3		
Total		44	100.0		

### use \$ rank--reagents

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	2	4.5	11.1	11.1
	Second Most Significant	10	22.7	55.6	66.7
	Third Most Significant	6	13.6	33.3	100.0
	Total	18	40.9	100.0	
Missing	999	26	59.1		
Total		44	100.0		

### use \$ applies--construction/lease

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	21	47.7	48.8	48.8
	No	22	50.0	51.2	100.0
	Total	43	97.7	100.0	
Missing	999	1	2.3		
Total		44	100.0		

#### use \$ rank--construction/lease

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	7	15.9	70.0	70.0
	Second Most Significant	2	4.5	20.0	90.0
	Third Most Significant	1	2.3	10.0	100.0
	Total	10	22.7	100.0	
Missing	999	34	77.3		
Total		44	100.0		

# use \$ applies--offender collection

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	18	40.9	41.9	41.9
	No	25	56.8	58.1	100.0
	Total	43	97.7	100.0	
Missing	999	1	2.3		
Total		44	100.0		

#### use \$ rank--offender collection

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	2	4.5	40.0	40.0
	Second Most Significant	1	2.3	20.0	60.0
	Third Most Significant	2	4.5	40.0	100.0
	Total	5	11.4	100.0	
Missing	999	39	88.6		
Total		44	100.0		

### use \$ applies--suspect case backlog

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	18	40.9	41.9	41.9
	No	25	56.8	58.1	100.0
	Total	43	97.7	100.0	
Missing	999	1	2.3		
Total		44	100.0		

### use \$ rank--suspect case backlog

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	1	2.3	14.3	14.3
	Second Most Significant	3	6.8	42.9	57.1
	Third Most Significant	3	6.8	42.9	100.0
	Total	7	15.9	100.0	
Missing	999	37	84.1		
Total		44	100.0		

# use \$ applies--no suspect case backlog

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	22	50.0	51.2	51.2
	No	21	47.7	48.8	100.0
	Total	43	97.7	100.0	
Missing	999	1	2.3		
Total		44	100.0		

### use \$ rank--no suspect case backlog

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	1	2.3	25.0	25.0
	Second Most Significant	2	4.5	50.0	75.0
	Third Most Significant	1	2.3	25.0	100.0
	Total	4	9.1	100.0	
Missing	999	40	90.9		
Total		44	100.0		

### use \$ applies--outsourcing

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	15	34.1	34.9	34.9
	No	28	63.6	65.1	100.0
	Total	43	97.7	100.0	
Missing	999	1	2.3		
Total		44	100.0		

### use \$ rank--outsourcing

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	3	6.8	30.0	30.0
	Second Most Significant	2	4.5	20.0	50.0
	Third Most Significant	4	9.1	40.0	90.0
	6	1	2.3	10.0	100.0
	Total	10	22.7	100.0	
Missing	999	34	77.3		
Total		44	100.0		

# use \$ applies--other

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	12	27.3	27.9	27.9
	No	31	70.5	72.1	100.0
	Total	43	97.7	100.0	
Missing	999	1	2.3		
Total		44	100.0		

#### use \$ rank--outsourcing

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Most Significant	3	6.8	33.3	33.3
	Second Most Significant	5	11.4	55.6	88.9
	Third Most Significant	1	2.3	11.1	100.0
	Total	9	20.5	100.0	
Missing	999	35	79.5		
Total		44	100.0		

### use \$ applies--other (description)

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	1	5	11.4	11.4	11.4
	999	38	86.4	86.4	97.7
	personn el	1	2.3	2.3	100.0
	Total	44	100.0	100.0	

#### DNA database match information included?

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	20	45.5	46.5	46.5
	No	23	52.3	53.5	100.0
	Total	43	97.7	100.0	
Missing	999	1	2.3		
Total		44	100.0		

#### Comments?

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Yes	6	13.6	14.0	14.0
	No	37	84.1	86.0	100.0
	Total	43	97.7	100.0	
Missing	999	1	2.3		
Total		44	100.0		

# **APPENDIX 4**

"Salaries" Clarification

ID	Salaries Clarification
1	Both
2	Augment salaries
44	Both
3	New Hires
4	Augment salaries
	3
46 5	New Hires New Hires
	New Hire
42	
45	New Hires
6	Both
7	New Hires
8	New Hires
43	Augment salaries
9	New Hires
10	Both
11	Augment salaries
47	New Hires
12	Both
13	Both
14	New Hires
15	New Hires
16	New Hires
17	New Hires
18	Augment salaries
19	New Hires
20	New Hires
21	New Hires
22	New Hires
23	Both
24	New Hires
41	Both
25	New Hires
26	Augment salaries
27	New Hires
28	New Hires
29	New Hires
30	New Hires
31	Augment salaries
32	New Hires
33	New Hires
34	New Hires
35	Augment salaries
36	New Hires
37	New Hires
38	Both
39	Both
40	Both

# **APPENDIX 5a**

SDIS Hits

# **Appendix 5 – CODIS Hit Information**

# type of lab

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	local	250	8.5	8.5	8.5
	state	2704	91.5	91.5	100.0
	Total	2954	100.0	100.0	

State recode to numeric

					Cumulative
\	4.7	Frequency	Percent	Valid Percent	Percent
Valid	AZ	2	.1	.1	.1
	CA	19	.6	.6	.7
	GA	214	7.2	7.2	8.0
	MA	42	1.4	1.4	9.4
	MD	46	1.6	1.6	10.9
	NM	38	1.3	1.3	12.2
	NY	182	6.2	6.2	18.4
	ОН	2	.1	.1	18.5
	VA	948	32.1	32.1	50.6
	WA	90	3.0	3.0	53.6
	Kansas	10	.3	.3	53.9
	Alaska	14	.5	.5	54.4
	MO	62	2.1	2.1	56.5
	TN	40	1.4	1.4	57.9
	SC	13	.4	.4	58.3
	IL	424	14.4	14.4	72.7
	TX	104	3.5	3.5	76.2
	KY	33	1.1	1.1	77.3
	NC	70	2.4	2.4	79.7
	NV	18	.6	.6	80.3
	ID	2	.1	.1	80.4
	MT	3	.1	.1	80.5
	AR	29	1.0	1.0	81.4
	MI	16	.5	.5	82.0
	CO	38	1.3	1.3	83.3
	NE	7	.2	.2	83.5
	WY	6	.2	.2	83.7
	NJ	67	2.3	2.3	86.0
	ME	99	3.4	3.4	89.3
	UT	9	.3	.3	89.6
	PA	97	3.3	3.3	92.9
	IN	76	2.6	2.6	95.5
	OR	133	4.5	4.5	100.0
	Total	2953	100.0	100.0	
Missing	System	1	.0		
Total		2954	100.0		

# Qualifying offense

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	property (theft, forg., b&e)	123	4.2	4.2	4.2
	murder/homicide (w/rape-hom.)	68	2.3	2.3	6.5
	rape & sex offenses	421	14.3	14.3	20.7
	assaults	121	4.1	4.1	24.8
	drug offenses	228	7.7	7.7	32.5
	robbery	208	7.0	7.0	39.6
	burglary	431	14.6	14.6	54.2
	supervision violations	9	.3	.3	54.5
	misc.	211	7.1	7.1	61.6
	unknown	1134	38.4	38.4	100.0
	Total	2954	100.0	100.0	

# Type of crime solved

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	property offenses (theft, forg., b&e)	87	2.9	2.9	2.9
	murder/homicide (w/rape-hom.)	290	9.8	9.8	12.8
	rape & sex offenses	1282	43.4	43.4	56.2
	assaults	39	1.3	1.3	57.5
	drug offenses	18	.6	.6	58.1
	robbery	176	6.0	6.0	64.0
	burglary	552	18.7	18.7	82.7
	supervision violations	7	.2	.2	83.0
	misc.	40	1.4	1.4	84.3
	unknown	463	15.7	15.7	100.0
	Total	2954	100.0	100.0	

# Type of hit

					Cumulative
		Frequency	Percent	Valid Percent	Percent
Valid	offender	1891	64.0	71.1	71.1
	forensic	767	26.0	28.9	100.0
	Total	2658	90.0	100.0	
Missing	999.00	296	10.0		
Total		2954	100.0		

### inter/tra state hits

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	intra	2706	91.6	91.6	91.6
	inter	248	8.4	8.4	100.0
	Total	2954	100.0	100.0	

# **APPENDIX 5b**

**NDIS Hits** 

ID	Reporting S	Hit ST	Forensic Case	Case 2	Offender
2	AK	CA	х		х
3	IL	AR	Burglary	Burglary	
4	IL	AR	Rape		х
5	KY	AR	X	x	
6	MO	AR	Homicide	Homicide	
7	TX	AR	Rape		х
8	TX	Army	Rape		х
9	MN	AZ	Rape		Rape
10	MN	AZ	Burglary	Burglary	
11	NM	AZ	Rape	Rape	
12	NM	AZ	Rape	Rape	
13	TX	AZ	Rape		x
14	WA	AZ	Rape	x	
15	WA	AZ	x		Assault
16	WA	AZ	Rape		Х
17	WA	AZ	х		Rape
18	WY	AZ	Burglary		Robbery
	CO	CA	Homicide		Rape
20	CO	CA	Homicide		Rape
21	CO	CA	Rape		Rape
22	CO	CA	Rape		Rape
23	IL	CA	Rape		x
24	IN	CA	Rape	Homicide	
25	MN	CA	Rape		Rape
26	NC	CA	Rape		х
27	NV	CA	Rape		Rape
28	NV	CA	Rape		Homicide
	TX	CA	RApe		х
			X	X	
31	WA	CA	X		Rape
32			X		х
		CO	Burglary		Burglary
34	MO	CO	Rape	Rape	
		CO	Homicide		Receiving Stolen Property
		CO	Rape	Homicide	
			X	x	
		DC	Rape		GL
			X		Х
			X	x	
		FBI	Rape		Rape
42		FBI	X		X
			X		X
			X		X
		FBI	Homicide		Assault
46		FL	Homicide		Х
47			X		Х
48		FL	X		Х
49	KY	FL	Assault		Х

ID	Reporting S	Hit ST	Forensic Case	Case 2	Offender
50	ME	FL	Rape		Rape
51	ME	FL	Burglary		x
52	NC	FL	Homicide		Robbery
53	NC	FL	B&E		x
54	NC	FL	Rape		Х
55	NJ	FL	Burglary		x
56	NM	FL	Rape		Burglary
57	NY	FL	X	x	
58	NY	FL	X		X
59	NY	FL	X		X
60	NY	FL	X		X
			X		X
	NY		X		X
	NY		х		X
	NY		х		X
				x	
			Homicide		Rape
	SC		Vehicle Theft		Burglary
	TN		Homicide		Rape
	TN		X		X
			Burglary		X
			Assault		x
	WA	FL	Rape		X
73			Rape	Rape	
74		GA	Rape		X
75			X	X	
	KY	GA	Rape	Rape, 2 Homi	
	MO	GA	Rape	Rape	
	NC	GA	Rape		Rape
	SC	GA	Rape		Homicide
	SC		Homicide		Theft
	TN		Burglary		Homicide
	TN		X		Drugs
	TN		Burglary		X
	TN		X		X
	NC		X Divinale min		Robbery
	CO		Burglary		Burglary
88		IN	Rape		Χ
89			Battery		X
	MD		X		X
				X	
	MD		X	X	Llamiaida
	MN	IL	Rape	Dono	Homicide
		IL II	Rape	Rape	
100		IL II	Rape	rape	Accoult
101		IL II	Rape	Done	Assault
104		IL	Homicide	Rape	
105	IVIO	IL	Homicide	Homicide	

ID	Reporting S	Hit ST	Forensic Case	Case 2	Offender
106	MO	IL	Rape	Rape	
107	MO	IL	Rape		Felonious Restraint
108	NY	IL	x		x
110	NY	IL	x		x
116	TN	IL	Theft		x
121	ОН	KY	Rape		x
122	TN	KY	Rape		x
123	TN	KY	Rape		Rape
124	ME	MA	2 Burglary		Burglary
125	ME	MA	Burglary		Burglary
126	ME	MA	Burglary		Burglary
127	ME	MA	4 Rapes	Rape	
128	ME	MA	Homicide		Rape
131	IL	MI	Burglary	x	
132	IL	MI	x		Rape
133	IN	MI	X		X
134	IL	MN	Homicide	Rape	x
135	CO	MO	Rape		Rape
136	KS	MO	Rape		Rape
137	KS	MO	Burglary	Burglary	
139	NM	MO	Rape		Assault
140	NY	NC	х	x	
142	MD	NJ	х	x	
152	MT	NM	3 Rapes	Rape	
153	IN	NV	Rape	Homicide	
154	NY	NV	X		x
155	NY	NV	x		x
156	MD	NY	X		x
157	MD	NY	X	x	
158	MN	NY	X		Rape
159	MN	NY	x		Rape
160	NC	NY	Rape		Assault
161	NC	NY	Rape		X
162	NJ	NY	Homicide	x	
163	SC	NY	Rape		Rape
164	IN	OH	X		X
165		OH	X	x	
166		OH	Rape		Kidnap
167		OH	Rape	Homicide	
168	NY	OH	X	x	
170	WA	OH	Rape	Rape	
171		OH	Rape		Rape
173		OK	Rape		Rape
174	NV	OR	Burglary		Burglary
176	IL	PA	2 Burglaries	x	
177	NY	PA	x		Х
178	NY	PA	x	x	
179	NY	PA	Х		Х

ID	Reporting S	Hit ST	Forensic Case	Case 2	Offender
180	NY	PA	х		х
181	NY	PA	Х	х	
182	IL	TN	Homicide		Rape
183	IL	TN	Homicide		x
184	IN	TN	Rape		x
185	IN	TN	Rape		Homicide
186	MO	TN	Rape	Rape	
187	OH	TN	Rape		Assault
188	ОН	TN	Homicide		Kidnap
189	TX	TN	Robbery		Burglary
190	ME	TX	Rape		Weapons
191	MN	TX	Rape		Rape
192	MO	TX	Homicide	Homicide	
193	NM	TX	Rape		Weapons
194	NM	TX	Burglary		Burglary
195	NM	TX	Rape		Assault
196	NM	TX	Homicide		Tampering w/ Evid.
197	TN	TX	Rape		x
198	WA	TX	х		Burglary
199	WY	TX	Burglary		Forgery
200	WY	UT	Rape	Rape	
202	MD	VA	Rape		Drugs
203	MD	VA	х		х
204	ME	VA	Rape		х
205	MN	VA	Robbery		Robbery
206	NC	VA	B&E		Robbery
207	NC	VA	B&E		Robbery
208	NC	VA	B&E		х
209	NC	VA	Rape		Kidnap
210	NM	VA	Rape		х
211		VA	Rape		Assault
212	NY	VA	Burglary		GL
213	NY	VA	Rape		Drugs
214	NY	VA	Rape		Juvenile
215			X	x	
216			X	x	
217			X	x	
218		VA	B&E		Forgery
219		WA	Rape		Rape
220		WA	Assault	Assault	
221		WA	Homicide		Burglary
222		WA	Rape		Robbery
223		WA	Burglary		Rape
224		WA	Rape		Assault
225		WI	Rape		Х
226		WI	Burglary		Rape
227		WI	Rape		Х
228	IL	WI	Rape		x

ID Reporting S	Hit ST	Forensic Case	Case 2	Offender
229 IL	WI	Homicide		х
230 IL	WI	Rape		X
231 IL	WI	Rape		X
232 IL	WI	Rape		X
233 IL	WI	Rape		x
234 IL	WI	Rape		x
235 IL	WI	Burglary		x
236 IL	WI	Burglary		x
237 IL	WI	Burglary		х
238 IL	WI	Rape		х
239 ME	WI	Criminal Trespass		Criminal Trespass
240 MN	WI	Homicide		Burglary
241 MN	WI	Homicide		Rape
242 MN	WI	Rape		Rape
243 MN	WI	Rape		Rape
244 MN	WI	Rape		Rape
245 MN	WI	X		Rape
246 TN	WI	X		Drugs
247 CT	x	Homicide		x
248 CT	x	Homicide		х
249 CT	X	Rape		x
250 CT	x	Rape		x
251 CT	X	Rape		x
252 NC	X	B&E		x
253 NC	X	Rape		x
254 VA	FL	Burglary		Rape
255 VA	FL	Rape		GL
256 VA	DC/FBI	Homicide		GL
257 VA	FBI	Rape		Maiming
258 VA	GA	Rape		GL
259 VA	PA	B&E		Drugs
260 VA	PA	Rape		Larceny
261 VA	PA	Homicide		Larceny
262 VA	NY	Rape		Bad Checks
263 VA	NY	Rape		Rape
264 VA	TX	Assault		Larceny
265 VA	NY	Rape		Weapons
266 VA	PA	Rape		Rape
267 VA	AZ	Rape		Forgery
268 VA	PA	DUI/Vehicle Acc.		B&E
269 VA	NY	Rape		Rape
270 VA	DC	Rape		Drugs
271 VA	NY	Rape		Drugs
272 AR	CO	Assault		Assault
273 CO	OK	Rape	Pana	Rape
274 CO	PA	Rape	Rape	
275 CO	TX	3 Rapes	Rape	Dono
276 IL	MN	Rape		Rape

ID	Reporting S	Hit ST	Forensic Case	Case 2	Offender
277			Rape		Rape
278	IL		Rape		x
279	IL		Homicide		Burglary
280	IL	SC	Rape		Burglary
281	IL	WI	Rape		x
282	IL	MN	Homicide	x	
283	IL	MO	Homicide	Homicide	
284	IL	MO	Rape	2 Rapes	
285	IL	FBI	Rape	Homicide	
286	IL	NY	Rape	Rape	
287			Burglary	x	
288			Rape	X	
289		MN	Rape	Rape	
290			Rape	rape	
291			Homicide	Rape	
292			Burglary	2 Burglaries	
293			Rape	2 Rape, 1 Ho	
294		KS	Rape	Rape	
295			X		X
296			Homicide		Battery
298			Х		X
299			X		X
300				X	
301			X Llamaiaida	X	Dana
304			Homicide		Rape
305 306			Rape		Rape
307			Rape		X
308			Robbery Rape		x x
309			Robbery		X
310			Rape		X
311			4 Rapes	X	^
312			2 Burglaries	X	
313			Homicide	X	
314				X	
315				x	
316			Rape		Rape
317			Burglary		Weapon
318			Burglary		x
319			Homicide		х
320	TN	GA	Rape		х
321	VA	MD	Rape	2 Rapes, 1 R	
323	MI		X		Rape
324	MI	GA	Х		Rape
325	MI	TX	Homicide		B&E
326	NY	Canada	X		X